# The Jordanian Corporate Governance Code: a Study of Institutional Investors' Perception of the Reliability of the Audit Report and the Audit Expectation Gap

Submitted by

Mohammad Khaled Shbeilat

BSc Accounting (Mutah), MA Accounting (AABFS)

A thesis submitted in total fulfilment of the requirements of the degree of

Doctor of Philosophy

School of Business, North Sydney Campus (MacKillop)

Faculty of Business

Australian Catholic University

Research Services | Graduate Research

P.O. Box 968 North Sydney NSW 2059

AUSTRALIA

Date 22/May/2013

## **Statement of Sources**

This thesis contains no material published elsewhere or extracted in whole or in part from a thesis by which I have qualified for or been awarded another degree or diploma.

No other person's work has been used without due acknowledgment in the main text of the thesis.

This thesis has not been submitted for the award of any degree or diploma in any other tertiary institution.

All research procedures reported in the thesis received the approval of the relevant Ethics/Safety Committees (where applicable)

na na

Mohammad Khaled Shbeilat

## **Acknowledgments**

First and foremost, I thank the Almighty Allah for enabling me to successfully complete this thesis.

This thesis would not have been realized without the assistance and encouragement of my supervisors, participants and friends who have made invaluable contributions and smoothed the path towards completing this thesis.

I owe deep gratitude to my principal supervisor, Dr. Waleed Abdel-Qader, for his ongoing support, guidance and patience throughout the stages of my Ph.D. I acknowledge the contribution made by him as well as our friendship since I started the journey towards earning my Ph.D.

I would like to express my sincere thanks to my Co-supervisor, Professor Donald Ross. I am forever grateful for his guidance throughout the thesis and for his assistance in capturing the methodology which has enriched the contents of my thesis.

I would also like to thank my Associate supervisor, Professor Garry Tibbits from the University of Western Sydney, whose invaluable constructive comments and advice were essential to the completion of my thesis.

I wish to express my appreciation to Tafila Technical University for granting me a scholarship to pursue my Ph.D study, and I would like to thank the Australian Catholic University for giving me the opportunity to teach accounting courses for three years, further enriching my academic experience. I also wish to acknowledge the assistance and encouragement of my colleagues at the faculty of business, of all campuses, who showed kind and overwhelming support during the course of my study. I must express my thanks to Dr. Mike Hopkins who devoted considerable time helping me analyze data via SPSS. I would like to thank Professor Lynne Bennington for organizing many training series and for her support of HDR students. I am also greatly indebted to all of my doctoral colleagues and staff at the faculty of business for their ongoing support, with special thanks to Sebastian Krook, Dr. Boonseng and Linda Gao.

This thesis has had the benefit of professional editing from Clarity Editors, and I extend my thanks for their invaluable efforts. The services rendered included advice and editing regarding thesis language, completeness and consistency, and did not involve the substance or structure of the thesis. A certificate of services rendered by Clarity Editors during the preparation of this thesis is available in Appendix 9.

The fieldwork study went smoothly in Jordan due, in no small part, to the assistance of the Certified Financial Analysts Society in Jordan (CFA-Jordan) in approaching participants for this study. Many thanks also go to all the kind individuals who responded to my survey and also to those who generously gave me of their time in order to conduct face-to-face meetings.

Finally, there are a number of people whom I appreciate for their companionship in Australia. To Salma Jibreel Al-Musadien, Birgit, Ashraf Al-sukker and his family, Raed Qatamin, Hakam Ajarmeh, Ahmad Karaja, Amine Tahboub, Asa'ad & Hannah Rawda, I say you made my life in Australia joyful and meaningful.

## **Dedication**

This thesis is dedicated with all my love and respect to my family. It is to my inspiring parents, supportive brothers and my compassionate sisters that I owe all my accomplishments and it is to them I dedicate this thesis.

## Abstract

This thesis examines, from the perspective of Jordanian financial analysts, the impact of selected components of the Jordanian Corporate Governance Code (JCGC) on the reliability of audit reports and the extent to which the JCGC contributes in narrowing the audit expectations gap. Thus this study focuses on those elements of the JCGC that relate to auditing and seeks to fill a gap in the existing literature regarding the determinants of changes in the audit expectation.

Previous research in this area either considered a smaller range of variables or considered codes other than the JCGC. Moreover, Jordanian research has focused on studying the relative main effects of the independent variables on the dependent variables, and thus less attention has been placed on identifying interactive or configural relationships. Also the research methods used were incapable of identifying cause and effect relationships. Furthermore, given that the JCGC was only relatively recently implemented, on a "comply or explain" basis, the full impact of the code has, perhaps, not been fully reflected in prior research.

This thesis investigates, experimentally, the main and interactive effects of four independent variables on two dependent variables. The independent variables are: (1) the external auditor's independence, (2) the internal auditor's effectiveness, (3) disclosure and transparency, and (4) corporate accountability. The dependent variables are (1) the perceived reliability of the audit report and (2) the perceived level of the audit expectation gap. In addition, this thesis investigates the extent to which Jordanian financial analysts possess self-insight as to the weight they place on the decision cues when evaluating the reliability of audit reports and when assessing the size of the audit expectation gap. By comparing financial analyst stated beliefs about the weights they placed on various factors, on one hand, to the weights revealed by the experiment, on the other, it is possible to gauge the accuracy of self-insight of the group.

The study design is based on the agency theory and uses a mixed method approach. The study uses semi-structured interviews, concentrating on 'how' and 'why' questions, aiming to investigate how institutional investors assess the selected elements of the JCGC in terms of

their effect on the perceived reliability of the audit report and the level of the audit expectation gap. The study also uses a fully crossed 2 level factorial experiment based on the repeated measures technique, thus forming 16 possible combinations of the four independent variables (i.e. 16 scenarios). In the survey, each subject was requested to provide their assessment of the two dependent variables in each of the 16 scenarios. The interviews deliver an in-depth understanding of the investors' perceptions regarding the selected elements of the JCGC and facilitate the interpretation of the study's experimental results.

It was predicted that the sophisticated investors may make decisions regarding the assessment of audits and auditors differently from unsophisticated investors. Therefore this study focused on financial analysts as an important type of sophisticated investor from the following types of institutional investors: banks, insurance companies, brokerage houses, investment companies and funds.

An analysis of 47 completed surveys and ten interviews suggested that external auditors' independence had the greatest effect on the perceived reliability of the audit report while, contrary to the existing literature, corporate accountability had the greatest influence on the perceived level of the audit expectation gap. Several other significant interactions have been identified between the study variables. One-third of the overall effect size in relation to the perceived reliability of the audit report is attributed to interaction between the independent variables, whilst approximately one-quarter of the overall effect size in relation to the level of the audit expectation gap is attributed to interaction between the independent variables. This indicates that institutional investors configurally process decision making information when considering the influence of the selected elements of the Jordanian corporate governance code. This has policy implications for the Jordanian regulating bodies.

# **Table of Contents**

### Contents

Statement of Sources	2
Acknowledgments	3
Dedication	5
Abstract	
Table of Contents	8
Appendices	13
List of Figures	13
List of Tables	14

Chapter 1: Thesis Introduction	15
1.1 Introduction	15
1.2 Background and Overview of the Thesis	15
1.3 Reviewing the Jordanian Literature and the Contribution of the Thesis	20
1.3.1 Highlighting the Problems before the Establishment of the Jordanian Corpora Code.	
1.3.2 Reviewing the Related Empirical Jordanian Studies.	27
1.3.3 Contribution of the Thesis and to the Jordanian Studies	32
1.4 Research Statement	35
1.4.1 Research Questions	35
1.4.2 Objectives of the Study	41
1.5 Research Methodology	42
1.5.1 The mixed method approach	43
1.5.2 In-depth Interviews	43
1.5.3 Experimental Treatments	44
1.6 Study Population	45

1.7 Definitions	47
1.8 Summary of the Study Limitations and Prospective Future Studies	50
1.9 Structure and Overview of the Thesis	51

Chapter 2: An Overview of Corporate Governance in Jordan53
2.1 Introduction53
2.2 The Theoretical Framework53
2.2.1 Corporate Governance & Agency Theory54
2.2.2 Agency Theory & the Auditing Profession56
2.3 Definitions of Corporate Governance
2.3.1 Introduction
2.3.2 Translation Dilemma
2.3.3 Corporate Governance Definition59
2.4 Importance of Corporate Governance61
2.5 OECD Principles of Corporate Governance63
2.6 Corporate Governance Assessment in Jordan64
2.7 Background of the Auditing Profession in Jordan
2.7.1 The Companies' Law
2.7.2 Auditing Profession Law67
2.8 The Development of Audit Committees71
2.9 The Development of the Audit Committees in Jordan71
2.9.1 Audit Committees in Jordan before the establishment of the Jordanian Securities Commission (JSC) in 199772
2.9.2 The 1998 JSC Instructions of Disclosure, Accounting and Auditing Standards73
2.9.3 The 2004 JSC Instructions of Disclosure, Accounting and Auditing Standards for Listed companies74
2.9.4 The Jordanian Corporate Governance Code for listed Companies
2.10 Chapter Summary81

Chapter 3: The Literature Review	82
3.1 Introduction	82
3.2 An Overview of the Audit Expectation Gap	82
3.3 The Audit Expectations Gap in Jordan	86
3.4 Investor Reliance on the Audit Report	
3.5 Types of Audit Reports and Opinions	92
3.6 The Role and the Potential Support by an Audit Committee	93
3.7 External Auditors' Independence	
3.7.1 External Auditors' Selection, Retention and Remuneration	
3.7.2 External auditor's Independence and the Audit Expectation Gap	
3.7.3 Non-Audit Services	
3.7.4 Audit Rotation	
3.8 The JCGC and the Internal Audit	
3.9 Disclosure & Transparency	118
3.10 Corporate Accountability	122
3.11 Chapter Summary	128

130
130
135
138
143
149
149
151
153

4.4 Institutional and individual investors in the context of Jordan.	154
4.5 Research Implementation Strategy	156
4.5.1 Experimental Treatments	. 157
4.5.2 In-depth Interviews	. 163
4.5.3 Pretesting: Survey Instrument & Interview Protocol	164
4.5.4 Data Collection process	165
4.6 Methodological Limitations	.166
4.7 Chapter Summary	. 167

hapter 5: Results & Analysis	. 168
5.1 Introduction	. 168
5.2 Experiment Results	. 168
5.2.1 Description of Response Rates and Respondents	. 168
5.2.2 The Validity of the Survey Experiment	. 171
5.2.3 Factor Weightings	. 174
5.3 Interview Data	. 187
5.3.1 Interview Summaries	. 188
5.3.2 Cross Case Comparison	. 197
5.4 Integration of the Mixed Methods Data	. 229
5.4.1 Context	. 230
5.4.2 Self insight	. 230
5.4.3 External Auditor's Independence	. 231
5.4.4 Internal Auditor's Effectiveness	. 232
5.4.5 Disclosure and Transparency	. 232
5.4.6 Corporate Accountability	. 233
5.4.7 Significant Interactions	. 234

Chapter 6: Discussion and Conclusions	235
6.1 Introduction	235
6.2 Discussion of the Independent Variables	235
6.2.1 External Auditors' Independence	236
6.2.2 Internal Auditors' Effectiveness	239
6.2.3 Disclosure & Transparency	240
6.2.4 Corporate Accountability	241
6.3 Discussion of Analysts' Decision-Making	243
6.3.1 Configurality	243
6.3.2 Self Insight	244
6.4 Perceptions of the Elements of the Corporate Governance Code in Jordan	245
6.5 Theoretical Implications	246
6.6 Conclusions from the Analysis and Discussion of Findings	248
6.6.1 The relative main and interactive weights of the JCGC Selected Dimensions	248
6.6.2 The Constituents of the JCGC and Cue Processing	253
6.6.3 A Note on Self-Insight	254
6.6.4 In-depth interviews	254
6.7 Practical Implications	255
6.8 The limitations of the Study	256
6.8.1 Limitations of Design	256
6.8.2 Limitations of Scope	258
6.8.3 Limitations in Applications	259
6.9 Suggestions for Future Research	259

References
------------

# Appendices

Appendix 1: Ethics Approval	293
Appendix 2: Information letter to Participants	295
Appendix 3: Letter to the Professional Association of Financial Analysts	299
Appendix 4: Reply From CFA-Jordan	
Appendix 5: Consent Form	
Appendix 6: Data Collection - Experimental Survey	
Appendix 7: Data Collection – Interview	
Appendix 8: List of Abbreviations	313
Appendix 9: Proofreading Certificate	315

# **List of Figures**

Figure 1: The Study Model: The key Constituents of the JCGC & the Auditing Profession17
Figure 2: The Mixed Method Approach Adopted in this Thesis46
Figure 3: Effective Elements of Corporate Governance60
Figure 4: ROSC Assessment of the Compliance to OECD Principles of Corporate Governance in Jordan65
Figure 5: Relationships between the Major Constituents of Corporate Governance as they Relate to
Auditing
Auditing 96   Figure 6: Research Implementation Strategy 157
Figure 6: Research Implementation Strategy157

gure 10: Level of Confidence that the Selected Variables of the JCGC Represent the Major Variables in	
terms of their Impact on the Dependent Variables	. 171
Figure 11: Weights of Factors' Influence on the Reliability of the Audit Report	. 176
Figure 12: Weights of Factors' Influence on the Level of the Audit Expectation Gap	. 183
Figure 13: External auditor Independence and Reliability of the Audit Report	. 237

# **List of Tables**

Table 1: Jordanian Corporate Governance and Auditing Profession Studies, Highlighting the	
Methodology Adopted	
Table 2: Respondents' Distribution by Sectors	169
Table 3: Respondents' Distribution by Positions	
Table 4: Respondents' Distribution by Experience	170
Table 5: Confidence Level of the Factors	172
Table 6: Frequency of Mention of other Factors Seen as Having an Important Impact on the Dep	pendent
Variables	172
Table 7: Factor Weightings for the Reliability of the Audit Report	175
Table 8: Effect Sizes for the Reliability of the Audit Report	178
Table 9: Factor Weightings for the Audit Expectation Gap	182
Table 10: Effect Sizes for the Audit Expectation Gap	185
Table 11: Effect Sizes as a Percentage of Total Effect Sizes- Reliability of the Audit Report	250
Table 12: Effect Sizes as a Percentage of Total Effect Sizes- Audit Expectation Gap	

## **Chapter 1: Thesis Introduction**

#### **1.1 Introduction**

This chapter introduces the research agenda for this thesis. It provides an overview of the study, including the study's historical context, research problem and research questions, research objectives, research conclusions and contribution to the literature, research methodology and data collection methods, as well as structure of the remaining chapters of the thesis.

## 1.2 Background and Overview of the Thesis

The Jordanian corporate governance code draws on a number of local and international laws and codes, such as Jordan's Securities Act and Companies' Law of 1997 and the Organization for Economic Co-operation and Development's (OECD) Principles of Corporate Governance (Corporate Governance Code in Jordan, 2009). The OECD principles of corporate governance were established by OECD Ministers in 1999, then revised and replaced in 2004. The OECD code became an "international benchmark" for developing countries worldwide (OECD, 2004, p. 3). The OECD defined corporate governance as: "Procedures and processes according to which an organization is directed and controlled. The corporate governance structure specifies the distribution of rights and responsibilities among the different participants in the organization – such as the board, managers, shareholders and other stakeholders – and lays down the rules and procedures for decision-making" (OECD Glossary, 2007, P. 151).

The Jordanian corporate governance code for shareholding companies listed on the Amman stock Exchange (ASE) was approved by the board of the commissioners of the Jordanian Securities Commission (JSC) on the 29th of July 2008 and became operational by the beginning of 2009<sup>1</sup> (Corporate Governance Code in Jordan, 2009). This Jordanian Corporate Governance Code (hereafter referred to as the **JCGC**), which regulates the activities of all Jordanian listed companies, is the subject of this study's investigation.

Audit committees are a cornerstone of corporate governance as they play a significant role in improving the quality and reliability of annual reporting. One way in which they accomplish this is by protecting external and internal auditors, starting from their nomination and continuously thereafter. Audit committees have the responsibility to discuss and resolve any conflict between auditors and management. The audit committee must ensure external auditor independence and internal audit effectiveness (Colbert, 2002). Colbert (2002) also stated that "under the International Standards on Auditing (ISAs) [ISA No. 260 Communications of audit matters with those charged with governance], external auditors have a responsibility to communicate certain matters, discovered during the financial statement audit, with those persons charged with governance of the entity. Similarly, the authoritative Practice Advisories (PA), issued by the Institute of Internal Auditors (IIA), require internal auditors to share observations and recommendations with those charged with oversight responsibilities" (p.147), typically the governance body of the entity.

This study investigates four primary independent variables derived from the JCGC, namely (a) external auditor independence, (b) internal auditor effectiveness, (c) disclosure and transparency levels and (d) corporate accountability (See Figure 1). These four variables are posited to influence the expert judgments of institutional investors as to (1) the perceived

<sup>&</sup>lt;sup>1</sup> See also Circular Number: 12/1/4659. Dated December 15, 2010. Issued by the Jordanian Securities Commission. Available on: http://www.jsc.gov.jo/Public/english.aspx?site\_id=1&Lang=3&Page\_Id=2550&Menu\_ID2=257

reliability of the audit report and (2) the level of the audit expectation gap. This thesis reviews and analyzes the variables of the following model, whose variables are discussed in more detail in the literature review in Chapter Three:

## Figure 1: The Study Model: The key Constituents of the JCGC & the Auditing Profession



Understanding the influence of these variables could be especially valuable for Jordan. There have been criticisms that low levels of disclosure, transparency, corporate accountability, external auditor independence and internal auditor effectiveness, have undermined confidence in the Jordanian capital market, compromised the integrity of financial reporting, and contributed to a widening of the audit expectation gap (Swaiti, 2006; Malkawi, 2008; Abdullatif, 2006; Ameera, 2004; Al-Saudi, 2007).

The first independent variable (external auditor independence) is measured by assessing institutional investors' perceptions of how and why independence is crucial, and reviewing related prior studies of (1) the role of well-structured audit committees in supporting and communicating with the external auditors, based on the JCGC and the ISA No. 260, (2) the impact of the processes of nominating, dismissing and setting audit remuneration by the audit committee, (3) audit partner rotation and (4) prohibiting the supply of Non-Audit Services (NAS). These four inputs are expected to enhance external auditor independence, both in fact and appearance, leading to reliable independent audit opinions and reporting about the company's ability to continue as a going concern.

The second independent variable (internal auditor effectiveness) is measured by assessing institutional investors' perceptions of how and why some internal audit functions are more effective than others. This measurement was adopted after a review of prior related studies focusing on (1) the role of audit committees in nominating internal auditors, (2) the impact of providing internal audit reports to audit committees and executive managers, (3) the role of audit committees in reviewing internal audit reports, especially those related to violations detected by the internal auditor and (4) the role of audit committees in reviewing the external auditor's assessment of the internal audit function. These measures, exercised by a well-established audit committee, are expected to lead to enhanced internal auditor independence and effectiveness in detecting frauds and errors and preventing malpractices, thereby protecting the assets of the company. This should also lead to greater perceived reliability of the audit report and a reduction in the audit expectation gap.

The reliability and the credibility of financial reports provided by a company, whether in the annual report or at any point during the year, depend on the quality, and the level, of disclosure and transparency. In regard to the third independent variable (disclosure and transparency), the JCGC has devoted a chapter to the topic of disclosure and transparency. This thesis explores institutional investors' perceptions of how and why JCGC disclosure and transparency

requirements for listed companies improve the quality of financial reporting and reduce the audit expectation gap. Prior research is examined to help guide this investigation.

Accountability is defined as "The quality of being accountable; liability to give account of, and answer for, discharge of duties or conduct", according to The Oxford English Dictionary. The fourth independent variable of this study involves corporate accountability. It refers to the ability of shareholders and stakeholders to hold the governing body of the company accountable to the laws, regulations and company's bylaws, based on their power and responsibilities. This factor is measured by interviewing and exploring institutional investors' perceptions, and by reviewing the related previous studies of how and why the appropriately constituted responsibilities, duties and authorities stipulated in the JCGC align the actions of governance professionals with the interests of the company and shareholders. The relevant governance professionals here are the external auditors and other governance bodies involved with auditing, such as the executive management, members of the board of directors, the audit committees and the internal auditors, all of whom are accountable for their decisions and responsibilities towards the company and the shareholders. In addition to the literature review and investor interviews, the study surveys financial analyst perceptions of how different levels of corporate accountability influence the reliability of the audit report and reduce the audit expectation gap.

Auditing has been defined as "being concerned with the verification of accounting data, with determining the accuracy and reliability accounting statements and reports" (Mautz, 1964, P.1). External auditors use their reports to communicate with the users of financial statements, such as investors, lenders and stakeholders. These parties assess investment and lending options based on the assurance provided by external auditors. Failure, by the external auditors, to give warning about the entity's viability, along with occurrences of corporate collapses and scandals, have raised the issue of the audit expectations gap; this gap partly exists because of the difference in perceptions and beliefs, between the users of the audit reports and the auditors, in regard to the auditors' duties and responsibilities (Koh and Woo, 1998).

## **1.3 Reviewing the Jordanian Literature and the Contribution of the** Thesis

As this study is focused on corporate governance in Jordan and its impact on the audit expectation gap and reliability of audit reports, as perceived by the financial analysts, a brief overview of corporate governance in Jordan, and relevant auditing research, is necessary both as background and for the design of the study. The overview also highlights the audit problems existing prior to the establishment of the Jordanian corporate governance code, in order to assist in evaluating the perceptions of institutional investors, in the post JCGC adoption period, of the reliability of the audit report and the size of the audit expectations gap.

# **1.3.1** Highlighting the Problems before the Establishment of the Jordanian Corporate Governance Code.

Over the years, there have been criticisms of the level of disclosure released by listed companies in Jordan, the use of unqualified internal auditors coupled with weak internal control procedures, and the appointment of negligent and incompetent members of the board of directors. In some instances, there have been claims of malpractices conducted by the directors and the auditors (Malkawi, 2008).

While the world had its headline-grabbing corporate collapses, Jordan also experienced its share of corporate scandals, creating the need for enhanced corporate governance. Following the Petra Bank bankruptcy in 1989, one of the most notorious scandals in Jordan's history, Jordan faced a severe financial crisis, resulting in further corporate collapses and a decrease in the Jordanian Dinar (JD) exchange rate from USD 3.35 to USD 1.41 (Al-Awaqleh, 2008). Other examples of scandals in large companies include the case of the Phosphate Company and the case of the Magnesia Company, referred to the Attorney General after a loss JD130 million (Al-Awaqleh, 2008) and involving legal actions, rescheduling and a compensation plan to protect

company employees (Awartani, 2011; Alrai, 2011; Zayadeen, 2011). Poor corporate governance practices were also associated with another scandal, involving four Jordanian banks, in which facilities amounting to millions of Jordanian Dinars were extended to individuals without getting the necessary approvals (Malkawi, 2008). In another corporate scandal, one billion US Dollars in loans, provided to an IT company, went astray (The Economist, 2003).

Insufficient accountability impairs entities' compliance with laws and regulations, which consequently leads to inadequate disclosure and a lack of transparent information, and weakens Jordanian entities' compliance with the best practices of corporate governance (Malkawi, 2008). Jordanian entities' compliance with the best practices of corporate governance is questionable. In July 2009, the governor of the Jordanian Central Bank o (CBJ) dissolved the board of directors of the Jordanian Capital Bank. The CBJ's officers discovered some administrative mismanagement and irregularities contradicted corporate governance best practices (Reuters, 2009).

The auditor is considered the most capable party to assess whether the company is managed according to good corporate governance practices. A working paper, presented by Al-Basheer (2003) to the 5th Scientific Conference of the Jordanian Association of Certified Public Accountants (JACPA), about the role of the auditors in the light of the corporate governance environment in Jordan, pointed out that audit committees play a vital role in maintaining the soundness of the financial reporting system since their primary functions are examining the reports of auditors and their conformity with ISAs, and reviewing the comprehensiveness of the scope of audits. Audit committees should also ensure top management responsiveness to the observations of both external and internal auditors, facilitate the work of auditors and remove any obstacles facing auditors.

Audit committees in Jordan are mostly formed to meet the requirements and conditions of the laws rather than for their essential role in maintaining the soundness of the financial reporting system (Swaiti, 2006; Abdullatif, 2006; Al-Farah, 2001; Al-Khadash and Al-Sartawi, 2010). The literature showed that audit committees did not possess the basic requirements to carry out

their responsibilities effectively, did not have sufficient influence to reinforce the independence of external auditors, and did not effectively play their role in appointing, dismissing, and determining the fees of external auditors. These findings are supported by the results of the World Bank's assessment of the corporate governance environment in Jordan (World Bank: Corporate Governance Assessment, 2004). The literature attributed these problems to (1) insufficient legislation, (2) a lack of requirements for audit committee members to be independent, and (3) a lack of requirements specifying the minimum financial/accounting qualifications and experience that an audit committee member should hold.

Al-Basheer (2003) pointed out that greater fairness and balance in setting audit fees, taking into account the heavy responsibilities required by shareholders and legislators, will facilitate greater audit effectiveness due to motivated auditors, ensure the company's compliance with the corporate governance code, and support external auditors in expressing their independent opinions about the entity's long-term viability.

Auditing plays a key role in corporate governance. The Jordanian Association of Certified Public Accountants (JACPA) adopted the International Standards on Auditing (ISA) in 1990. Under ISA No. 570, the external auditor is responsible for evaluating whether the company is a going concern and giving early warning, to the public and users of the audit report, of any impending corporate collapse. Failure to give such early warning has raised the issue of the audit expectations gap in Jordan, where there is a discrepancy between what auditors are perceived to have done and what a group of financial statement users believes they should have done.

Al-Awaqleh (2008) investigated the evaluation of the going concern assumption by the Jordanian external auditors, revealing that Jordan also has experienced the problem of unqualified audit opinions in the auditor's annual reports followed by unexpected corporate failures and collapses. The study reached these findings by conducting a pilot study on 160 audit reports related to 32 failed shareholding entities. Examples of listed companies in Jordan which had clean audit reports in the six months prior to their receivership or liquidation include The Jordanian Candy and Chocolate Factories, in 2001; The Jordanian Clothing Industrial

Company, in 2003; The Developed Medication Industrial Company, in 2003; The Mineral Investment Company, in 2003; The Ala Aldeen Industrial Company, in 2001; The Al-Kawthar Company for Investment, in 2001.

The external auditor's reports are the medium of communication between auditors and investors. Users of financial statements, such as investors, lenders and stakeholders, look to the assurance provided by the auditors as an input to their investment decisions. Therefore, an external auditors' forthright judgment regarding client viability is valued by investors (Jones, 1996; Epstein and Geiger, 1994; O'Reilly, 2009; Citron, Taffler, & Jinn-Yang 2008).

In Jordan, investors analyze the contents of the audit report as they evaluate the alternative investment options based, in part, on the assurance provided by the external auditors. Market share prices in Jordan are also influenced by the released audited financial statements (Saaydah, 1998; Zureigat, 2006; Zureigat, 2010; Al-Awaqleh, 2008). The importance of the auditor's opinion in Jordan means that Jordanian laws and the regulatory bodies of the Jordanian capital market and the Jordanian Association of Certified Public Accountants (JACPA) should reinforce the external auditor's independence. This independence should be in both fact and appearance, in order to ensure that auditors can express, clearly and unambiguously, their independent opinion.

The failures to provide adequate warning were attributed, by Jordanian literature, to the external auditor's lack of independence, which has been compromised by (1) Inadequate audit fees, (2) Providing significant non-audit services, (3) Unlimited audit partner tenure, (4) Lack of support from audit committees to both external and internal auditors, (5) Insufficient laws and regulations, (6) Social and personal relationship between the auditors and the board of directors and the management and (7) Management pressure on the auditors and weak corporate accountability (Swaiti, 2006; Malkawi, 2008; Abdullatif, 2006; Dahmash, 1989; Matar, 1995; Matar, 2000; Momany, 1994; Al-Awaqleh, 2008; Hamdan, 1996; Abu-Tapanjeh, 2006; Al-Saudi, 2007; World Bank: Accounting and Auditing, 2004; Bani-Ahmad, 2000; Al-Basheer 2003; Malhas, 1992).

An audit committee should also ensure the soundness of the internal control procedures in detecting and preventing errors, frauds and malpractices (Cadbury Report, 1992; Braiotta, 2004). Audit committees are responsible for supporting the internal auditors and reinforcing their effectiveness (Allison, 1994). Audit committee support to the internal audit could be achieved by the nomination of the internal auditors, evaluating the requirements and the budget of the internal audit department, ensuring that the internal auditors have a direct reporting relationship and communication with the audit committee as an independent body (in addition to any reporting to company management), hold meetings with the internal auditors without the presence of the company management and by reviewing the external auditor's assessment of the internal audit function (Pickett, 2010; Goodwin, 2003; Qin, 2007; Goodwin & Yeo, 2001).

Thnaibat and Shunnaq (2010) investigated the role of the external auditors, in Jordan, in evaluating the internal audit function according to the International Standard on Auditing (ISA) No. 610. The study revealed that there were statistically significant differences between the internal and the external auditors' perceptions in regard to the implementation of the ISA No. 610. The study explained this, pointing out that internal auditors in Jordan are still doing their job in a traditional manner, focusing mainly on the accuracy of the accounts. The study also revealed that the internal auditors' reliance on and evaluation of the internal audit function in Jordan. The study recommended that internal auditors should report to the audit committee to maintain their independence and effectiveness.

This variance of perceptions between external and internal auditors in the evaluation of the internal audit function in Jordan highlights the audit committee's vital role as a liaison between the external and the internal auditors, the company management, and the board of directors. This thesis, through in-depth interviews, aims to evaluate how and why the communication between the audit committees and the external and the internal auditors, as stated by the JCGC, lead to strengthening auditor independence and effectiveness.

The Jordanian Company Law (1997) specifies that shareholders appoint the external auditors. Prior to the JCGC, it was management, normally with the collaboration of the board of directors, who appointed external auditors and determined their remuneration. Malkawi (2008) criticized that situation, arguing that it undermined the independence of the external auditors.

The selection of the external auditors in Jordan prior to the release of the JCGC was influenced by the auditors' personal relationships with the board of directors, the company management and the audit committee members (Al-Saudi, 2007; Khsharmh, 2003; Ameera, 2004; Abdullatif, 2003). Malkawi (2008) suggests that comprehensive legislation to ensure audit partner rotation, disciplinary systems, an ethical code of conduct for external auditors and prohibit nonaudit services for audit clients, could mitigate problems in auditor independence.

Jordanian companies are often described as family-owned businesses (MENA Corporate Governance Workshop, 2003; Zureigat, 2011b; Al-Adeli, 1993; Abed et al, 2012; Matar, 1994). Abu-Tapanjeh (2006) revealed that a number of shareholding companies in Jordan are owned by members of a family, and prefer to elect their family and relatives to be members of both of the board of directors and executives in the company. This ownership characteristic in Jordanian shareholding companies has created obstacles for implementing corporate governance best practices. With family relationships determining appointments to positions in both the board of directors and high in the executive management, decision makers often lack the flexibility and objectivity to fulfill their jobs and oversee the company's activities effectively (Sharar 2006). Moreover, family-owned businesses in Jordan sometimes do not use competence as a criterion for appointing external auditors; rather, personal relationships might be the most influential criterion (Zureigat, 2011b).

On the 29th of July 2008, the board of the commissioners of the Jordanian Securities Commission (JSC) issued and approved the Jordanian Corporate Governance Code (JCGC) for the publicly traded companies listed on the Amman Stock Exchange (ASE) which has come into effect as of the first of January 2009. The JCGC was established to develop the investment and economic climate, including the national capital market and its regulatory and organizational

framework. Corporate governance best practices became a tool to attract foreign and local investment alike. They enhance confidence in the capital market since they reflect the level of fairness, disclosure, transparency and accountability of the listed corporations and identify the measures that should be taken to limit corruption (Corporate Governance Code in Jordan, 2009).

The Jordanian corporate governance code was established to help build a clear framework to regulate the roles of the shareholders and safeguard the rights of all stakeholders. The code states that the duty of the board of directors and the company management is to establish adequate, and effective, internal control procedures to detect and prevent errors and frauds. It also specifies the role of shareholders and the audit committees in nominating the external auditors of the company and ensuring their independence. It also states the roles of the audit committee in nominating and communicating with the internal and the external auditors, evaluating their performance and removing unnecessary barriers in the way of external auditors.

According to the JCGC, Jordanian listed companies should disclose accurate, clear and timely information to shareholders and investors; this includes dealings with company insiders and their relatives as well as related party transactions. The JCGC has prohibited any person from simultaneously holding the position of chairperson of the board of directors and any executive position in the same entity, and it has also prohibited simultaneous membership by any person in more than five boards of directors of different listed companies. This measure aimed at limiting the opportunistic behavior of the managers of some closely-held corporations and ensuring accountability in corporate governance of the companies listed at Amman Stock Exchange (ASE).

The main research objective of this thesis is to investigate (1) the impact of the Jordanian corporate governance code (JCGC), in the post implementation environment, on external auditor independence and internal auditor effectiveness, and (2) the extent of the JCGC

disclosure, transparency and the accountability requirements on institutional investor perceptions regarding the reliability of the audit report and reducing the audit expectation gap.

#### 1.3.2 Reviewing the Related Empirical Jordanian Studies.

This section presents relevant Jordanian literature about Jordanian corporate governance and the existence and effect of the audit expectation gap on the auditing profession in Jordan. It also focuses on the roles of audit committees in relation to external and internal auditors. This section aims to explore prior empirical work in relation to corporate governance environment in Jordan. The results of this exploration will be used, in the next section, to show how the current study contributes to the knowledge of Jordanian corporate governance.

A study aimed at developing a system for corporate governance in Jordanian shareholding companies and reinforcing external auditor independence was conducted by Al-Hanini and Dahmash (2008). Al-Hanini and Dahmash designed and distributed a questionnaire to 120 external auditors and conducted an unspecified number of interviews. The study pointed out that the auditors thought that both effective corporate governance systems and sub-committees (e.g. Audit committee) derived from the board of directors positively enhance external auditor independence. The study recommended that auditors should express their opinion about corporate governance through their audit report.

Swaiti (2006) argued that, according to Jordanian regulations and laws in force, all public shareholding companies are required to form audit committees. However, Swaiti argues that the audit committee functions and responsibilities stipulated in those regulations and laws were insufficient. For example, they did not require any qualifications for audit committee members, such as having financial management experience or holding a degree in accounting or any related fields.

Swaiti (2006) proposed a model for the role of audit committees in Jordan. Swaiti designed a survey questionnaire which was given to financial managers, general managers, external audit firms, and internal audit managers, and then conducted personal interviews with some of those who were involved in his study<sup>2</sup>. The study concluded that audit committees at that time did not have the basic requirements to fulfil their roles effectively, and they did not have a substantial impact on external auditor independence. The study recommended that all audit committee members should be independent and suggested they have an industry expert and a legal expert among their members.

The researcher would argue that all audit committee experience and qualifications must be in the fields of auditing, accounting and finance, not just because the committee's name is 'the audit committee' and its main function is to ensure the integrity and the accuracy of the financial reporting, it is also because the audit committee has the authority to consult external legal consultants or experts for any issue related to fulfilling their roles properly. The JCGC stipulated that each company "shall put at the disposal of [its Audit] Committee all facilities that it needs to perform its duties, including the authority to seek expert assistance whenever needed" (Corporate Governance Code in Jordan, 2009, Chapter 5, section 1-3).

The effectiveness of audit committees in Jordan and the factors affecting the effectiveness of audit committees have been explored by Abdullatif (2006). In his study, a questionnaire was distributed to external auditors. The study concluded that audit committees in Jordan were somewhat effective, and they marginally fulfilled their jobs as stipulated in the related Jordanian regulations. The study revealed that those regulations were insufficient and recommended reinforcing audit committee independence from company management in order to strengthen their role in overseeing the financial reporting processes, and their role in nominating, dismissing and setting external auditor's remuneration.

Abdullatif (2007) conducted another study to investigate the relevance of audit committee responsibilities to corporate governance. The study utilized the same methodology as the previous

<sup>&</sup>lt;sup>2</sup> The study interviewed executives, board members, auditors, and staff at Jordanian regulating bodies.

one and the same sample category. The study concluded that external auditors in Jordan perceived audit committee responsibilities as being relevant in general, and they suggested some additional responsibilities such as providing the external auditor with any necessary information and coordinating between external and internal auditors. The study also revealed that external auditors in Jordan were against giving audit committees the authority of nominating, dismissing, and setting external auditors' fees.

External auditor opinions against the role of the audit committees in nominating external auditors were also observed by Swaiti (2006). Abdullatif (2007) attributed this partially to the external auditors' criticisms of the weak composition of audit committees at that time, such as there being no requirements of being independent or having any financial experts. He also partially attributed this reluctance to the external auditors' unwillingness to undertake additional tasks or to change their personal relationship with their clients. Swaiti's study (2006) revealed that there was consensus between auditors and company executives in rejecting the role of audit committees in selecting the external auditors. Swaiti (2006), however, pointed out that the auditors' rejection was based on the audit committee members being unqualified to evaluate auditor work<sup>3</sup>. On the other hand, the study also attributed this rejection to the possibility that external auditors desire to maintain a personal relationship with their company (management) clients.

The causes and structure of the audit expectations gap have been investigated by Al-Thuneibat (2003). A quantitative method approach was adopted in the study. The questionnaire explored expectations about auditor responsibilities and duties, including whether they were reasonable or unreasonable<sup>4</sup>, and was distributed to the study sample which consisted of the external

<sup>&</sup>lt;sup>3</sup> This could be a suggestion that audit committee members are incompetent, or it could be a mistaken belief that only experts can evaluate experts when in fact their output is for non-auditors and ultimately it must be the shareholders who decide whether they have performed a useful role.

<sup>&</sup>lt;sup>4</sup> The study identified the unreasonable auditors' responsibilities, based on the previous literature especially the study of Porter (1993). The study defined the unreasonable responsibilities as those duties and responsibilities which were not stipulated in the international standards on auditing (the adopted standards in Jordan) or in the legislation that regulates the auditing profession in Jordan.

auditors, and executive and financial managers of Jordanian listed companies. Based on the analysis of the data, the study concluded that the audit expectations gap existed in Jordan, and it consisted of the following three basic components: deficient standards gap, ignorance gap and deficient performance gap.

To narrow these gaps, the study recommended that the regulatory bodies of the auditing profession in Jordan should play a more effective role in educating society as to the auditor's duties and responsibilities, developing the profession and overseeing the auditors' job, ensuring they continuously participate in seminars related to the profession, and imposing quality control on the auditors.

The performance gap could also be narrowed by strengthening the external auditor's performance. This thesis aims to investigate the role of the JCGC and support from the audit committees in reinforcing the external auditor's independence, which is expected to encourage external auditors to express an independent opinion about the company's financial position.

The gap between auditors and investors' perceptions about the qualitative characteristics of accounting information has been investigated by Obaidat (2007). These qualitative characteristics include the reliability and relevance of information in relation to helping investors in decision making (Kieso, Waygandt, & Warfield 2010). The Obaidat study adopted quantitative methods, with questionnaires distributed to a sample which consisted of 30 investors and 40 auditors. The study revealed that the major focus of the investors was on timeliness while the auditors gave neutrality the first rank in importance.

Obaidat (2007) explained the difference in priorities between investors and auditors, explaining that individual investors in Jordan are not interested in long-term investment, being more interested in gaining profits as soon as possible through speculation in the stock exchange. The auditors focused on neutrality, which means providing information that has not been prepared to favor one party of interested users over another (Kieso et al, 2010), which is the basis of the auditor's independence (Arens, Best, Shailer, Fiedler, Elder, & Beasley 2010). To ensure

neutrality, audit committees should strengthen the external auditor's independence, even in the face of management pressure. In Jordan, management pressure to issue clean opinions has been a significant problem for external auditors (Omari, 2003; Dahmash, 1989; Bani-Ahmad, 2000; Matar, 1995; Al-Adeli, 1993; Malhas, 1992). This pressure impairs the auditor's independence and highlights the necessity of adopting effective corporate governance systems.

The audit expectations gap and auditors' responsibilities, from investors' perspective, have been investigated by Hajir (2001). The researcher designed and distributed 72 questionnaires to individual investors, present at the Amman Stock Exchange at the time of data collection, and 28 questionnaires to institutional investors (the brokerage companies). The study found that perceptions of inadequate disclosure<sup>5</sup>, inadequate independence of auditors and inadequate performance by auditors, all contributed to a widening of the audit expectation gap.

The study also revealed that the only difference in perceptions between individual investors and institutional investors, regarding the factors affecting the existence of the expectations gap, related to the level of disclosure. The study found that the individual investors believed, at that time, that the levels of disclosure in the audit and financial reports, alongside the accompanying notes in the annual report, were adequate and provided relevant information for decision making. The institutional investors thought the level of disclosure inadequate. A possible explanation for this issue is that the institutional investors employ qualified and sophisticated financial analysts who can devote more time and skill to the analysis of the reports and notes.

The study of Omari (2003) investigated the expectation gap in relation to the credibility of the accounting figures in the financial reports, the extent of the auditor's responsibility in expressing an opinion about the entity's ability to continue as a going concern, the auditor's

<sup>&</sup>lt;sup>5</sup> The study defined the level of disclosure by stating that financial reports and the notes accompanying the financial statements should include all necessary information that reflect the real financial position of the company to make the investors informed and in helping them to make their investing decisions.

independence and the scope of the auditor's responsibility in finding the misleading financial data during the auditing process. In addition, that study set out to identify which of the investor groups (individual investors, financial brokers, employees of banks' investment units) have the greatest expectations gap. A quantitative approach was adopted in the study, with questionnaires distributed to investors and external auditors. The study concluded that the audit expectations gap in Jordan was mainly focused on the independence of the external auditors, which was affected by non-audit services, unlimited audit tenures, and perceptions of management pressure.

Al-Khadash and Al-Sartawi (2010) investigated the usefulness of legislation along the lines of the Sarbanes-Oxley Act (SOX) in enhancing the independence of external auditors in Jordan and narrowing the audit expectation gap. The study, as did most of the previous Jordanian studies, adopted a quantitative approach in which questionnaires were used to collect data from Jordanian Institutional Investors and external auditors. The study concluded that such legislation was expected to reinforce the external auditor's independence and narrow the audit expectation gap. Based on this, the study recommended that the Jordanian regulatory supervisory bodies of the auditing profession should define a list of non-audit services prohibited to their audit clients, and those prohibitions should be enforced by relevant legislation (using the SOX legislation as a guide). In addition, the legislation should reinforce the auditor's independence, limit the audit partner's tenure, and activate the role of well-structured audit committees.

#### 1.3.3 Contribution of the Thesis and to the Jordanian Studies

The objective of this thesis is to gain insights into the perceptions of institutional investors in the period subsequent to the adoption of the JCGC, in so far as those perceptions relate to the reliability of the audit reports and the extent to which they contribute in narrowing the audit expectations gap. This thesis investigates, experimentally, the impact of the external auditor's independence, the internal auditor's effectiveness, disclosure and transparency, and corporate accountability, on the level of the audit expectation gap and the perceived reliability of the audit report. Thus, the contribution of this thesis is to complement the existing research by investigating the relative main and interactive weights of the posited four variables on the reliability of the audit report and the reduction of the audit expectation gap, following the relatively recent introduction of the JCGC in Jordan which was intended to increase the actual and perceived independence of auditors. In addition, the thesis explores cause and effect relationships, including interactions, in the Jordanian context, in contrast to prior research focusing on associations.

The results of this thesis fill a gap in the extant body of the literature and are more comprehensive than the prior research in Jordan. For instance, the study of Al-Hanini and Dahmash (2008) focused on corporate governance's role in reinforcing the external auditor's independence. Another study, Al-Khadash and Al-Sartawi (2010), investigated the capability of legislation like the American Sarbanes-Oxley Act (SOX), rather than the JCGC, to reinforce the independence of external auditors in Jordan and narrow the audit expectation gap<sup>6</sup>. The current study considers a wider range of variables and focuses on investigating the relationships in the Jordanian environment.

In addition, this thesis investigates the extent to which Jordanian financial analysts possess selfinsight as to the weight they place on the decision cues when evaluating the reliability of audit reports and when assessing the size of the audit expectation gap. By comparing financial analyst beliefs about the weights they placed on various factors, on one hand, to the weights revealed by an experiment, on the other, it is possible to gauge the accuracy of self-insight of the group. This comparison provides clues as to the quality of the decision making and the likelihood of consistency in judgments (Libby, 1981; Hooper & Trotman, 1996).

<sup>&</sup>lt;sup>6</sup> The study of Al-Khadash and Al-Sartawi (2010) had been conducted before the establishment of the JCGC, the study was sent for publishing in September 2008 as mentioned in the footnotes of the study.

This thesis contributes to the Jordanian literature by using experimentation. This thesis presents a new methodology to the area of understanding corporate governance in Jordan. The use of the experiment, along with in-depth interviews, aims to enhance understanding of how and why Jordanian listed companies are affected by the key components of the newly adopted Jordanian corporate governance code, from the point of view of institutional investors. The use of an experiment is intended to overcome methodological limitations inherent in prior Jordanian studies which mostly adopted quantitative approaches based on questionnaires. The use of the experiment, employing the repeated measures design, enabled cause and effect relations to be investigated and some insights to be developed, including insights into interactions between the independent variables in their impact on the dependent variables. The study results will provide valuable information to external users such as the Jordanian Securities Commission (JSC), the Certified Financial Analysts Society in Jordan (CFA-Jordan), the Jordanian Association for Certified Public Accountants (JACPA), other Jordanian regulatory bodies in addition to shareholders, as well as to the Internal users such as members of boards of directors, audit committee members, executive managers and the internal auditors.

The research's importance also stems from the lack of previous empirical Jordanian studies in this regard. Most local and international studies investigated the effect of corporate governance on market share prices, **company performance**, earnings management and financial reporting quality, and dealt primarily with exceptionally large firms in the U.S.A and Europe, countries with highly developed and active markets for corporate control in comparison to those of emerging markets like Jordan (Tomar & Bino, 2012; Ishhadat & Abed Al-Jaleel, 2012; Al-Manzou, 2009; Exi et al, 2001; Al-Najjar, 2007; Al-Khouri, 2005; Gompers, Ishii, & Metrick 2001; Al-khabash & Al-Thuneibat, 2009; Klein, 2006). Nevertheless, corporate governance best practices and the roles of the board of directors and the audit committees in communicating and supporting the internal and the external auditors may also influence the auditing profession and the demand for audit quality (Ballesta & García-Meca, 2005)

Finally, this study has the potential to contribute further to the knowledge of the impact of corporate governance on the auditing profession literature through the database that it has

developed for Jordanian listed companies. This database could be used to study other corporate governance issues.

## **1.4 Research Statement**

This study enhances understanding of the newly adopted JCGC and audit reports in Jordan. The purpose of this thesis is to corroborate primary and interactive effects of factors and variables related to the constituents of the JCGC on the reliability of the audit report and the extent they contribute in narrowing the audit expectation gap as perceived by the institutional investors. This research employs a combination of qualitative and quantitative methods to provide a grounded investigation of the elements of the JCGC in so far they relate to the auditing profession, specifically the external auditor's independence, the internal auditor's effectiveness, disclosure, transparency and accountability. The study also includes consideration of the relative weights of effects, casual relationships, how the factors influence and interact and how they are evaluated.

#### **1.4.1 Research Questions**

The main question addressed in this thesis is:

How and why do the selected elements of the JCGC affect the institutional investor's judgment of (a) the reliability of the audit report and (b) a narrowing of the audit expectation gap?<sup>7</sup>

<sup>&</sup>lt;sup>7</sup> The audit expectations gap exists in Jordan (Al-Thuneibat, 2003; Obaidat, 2007; Abdullatif, 2003; Omari, 2003; Hajir, 2001) as discussed earlier. This thesis endeavors to contribute to narrowing the audit expectation gap by investigating the posited independent variables of the JCGC experimentally.

In answering this fundamental research question, this thesis both explores institutional investors' perceptions and reviews previous studies, investigating how and why corporate governance affects the auditing profession, in particular, how and why external auditor independence, internal auditor effectiveness, disclosure, transparency and corporate accountability affect institutional investors' perceptions of the reliability of the audit report and the size of the audit expectations gap. In addition, this thesis utilized the mixed method approach to answer the research questions by interviewing institutional investors to facilitate additional insight into their decision-making processes. Moreover, this thesis employs a factorial experiment, based on a questionnaire, to identify the causal relationship between the hypothesized elements of the JCGC and auditing. Also, by using the experimental technique, this study aims to identify the relative weights of the independent variables and their interactions with the dependent variables, in other words the perceived reliability of the audit report and the level of the audit expectation gap.

The first research question relates to the impact of the selected elements of JCGC, in so far as they relate to auditing, on the reliability of the audit report.


The second research question relates to the perceived impact of the selected elements of JCGC, in so far they relate to auditing, on the level of the audit expectation gap.



# 1.4.1.1 Configurality

Slovic, Fleissner, & Bauman (1972) argued "Analysts believe that factors relevant to investment decisions should often be interpreted configurally" (p. 286). However, they defined configurality as meaning "that the analyst's interpretation of an item of information varies depending on the nature of other available information" (Slovic et al, 1972, p. 286). Previous literature that investigated judgmental decisions for capital market investment (Wood & Ross, 2006; Hopkins, 2009; Nguyen & Ross, 2006) and judgmental decision making in auditing (Ashton, 1974; Hofstedt & Hughes, 1977; Brown & Solomon, 1990, 1991; Hooper & Trotman, 1996) have revealed that influential factors interact and intermingle in their influence in judgmental decision making. "An interaction means that the effect of one independent variable depends on the level of the other independent variable being considered" (Trotman, 1996, p. 21).

The four independent variables hypothesized to influence the dependent variables in this study might interact and result in different impacts depending on the combination of values of the other independent variables. For instance, the impact of evidence of uniformly high or low performance on all the variables might have a greater impact than would have been expected from examining the impact of each variable in the absence of the other factors. Thus, high levels of performance on all four factors may convince the investors that the company and its auditors take governance seriously because of the strong and consistent message. In such circumstances, it is possible to envisage investors having a perception of reliable reporting and robust auditing.

A further example of the interaction of the elements of corporate governance would be that high or strong levels of external auditor independence and performance, coupled with high levels of internal auditor effectiveness, might not ensure the perceived maximum influence and outcomes of corporate governance if they were associated with inadequate and untimely disclosure and transparency and a poor accountability system.

It has been evidenced that interactions between variables in experimental studies are particularly important and deliver great value to the decision makers because they lend the findings additional useful dimensions; for instance, Hopkins (2009) found statistically significant interaction between intensity of private equity involvement and duration of involvement on the attractiveness of the Initial Public Offering (IPO). Hopkins pointed out that the result was noteworthy and interesting because the intensity of private equity involvement failed to satisfy standard statistical tests of significance, but it was a relevant factor when it interacted with the duration of involvement. Teoh and Lim (1996) found two statistically significant interactions in their experiment investigating the factors affecting the level of confidence in the auditor's independence. They also found that the existence of audit committees mitigates perceptions of impairment of auditor independence when that auditor gained large audit fees (more that 15 %) from one client. Moreover, the study also found that the perceived external auditor independence was enhanced by the absence of both non-audit fees and a fee dependency situation.

Evidently, manipulation of the four independent variables may provide combined effects that are relevant information for investors in their decision making. Therefore, it is hypothesized that institutional investors will recognize the potential interactions and interrelations between the four factors and evaluate and process them configurally when considering the influence of the Jordanian corporate governance code. Given the potential significance of interactions between the four independent variables, this thesis establishes the following hypothesis:

# **H1**:

Institutional investors process decision making information configurally when considering the influence of the selected components of the Jordanian corporate governance code.

#### 1.4.1.2 Self Insight

The unit of analysis in this thesis consisted of institutional investors. Institutional investors are perceived as experienced and sophisticated analysts whose judgments and opinions are trusted as sources of information related to investment assessment (Bouwman, Frishkoff, & Frishkoff, 1987; Schipper, 1991; O'Brien & Bhushan, 1990). This thesis also aims to investigate the degree of self-insight demonstrated by Jordanian financial analysts. Self-insight is the extent of insight that individuals have into their judgment process (Trotman, 1996). Typically, the degree of self-insight has "been assessed as the correlation, over all cues, between a subject's cue usage and the importance rating subjectively attached to each cue" (Solomon & Shields, 1995, p. 151). This technique has been employed in studies about judgment decision making in finance, including Hopkins (2009), Wood & Ross (2006), Mear & Firth (1987) and Nguyen & Ross (2006). This thesis builds on these studies and considers the level of self-insight among Jordanian financial analysts regarding their assessments of the elements of the Jordanian corporate

governance code, in so far they relate to auditing among, by addressing the following questions:

# RQ.3

What degree of self-insight do institutional investors demonstrate in their assessments of the selected key constituents of the Jordanian corporate governance code on the reliability of the audit report?

# RQ.4

What degree of self-insight do institutional investors demonstrate in their assessments of the selected key constituents of the Jordanian corporate governance code on the audit expectation gap?

This thesis also aims to provide an in-depth understanding of how institutional investors assess the selected constituents of the JCGC, and to provide a justification for their conclusions. This could be achieved by conducting in-depth semi-structured interviews with a concentration on 'how' and 'why' questions (Symon & Cassel, 1998; Yin, 2009). The fifth research question of this study is:

# RQ.5

How and why do institutional investors' assessments of the selected constituents of the JCGC affect their perception of:

- The reliability of the audit report, and
- The level of the audit expectation gap?

The study's research questions and hypothesis are the basis for the research objectives.

#### 1.4.2 Objectives of the Study

The primary objective of this thesis is to improve understanding of how implementation of the Jordanian corporate governance code, in so far as it relates to auditing, has affected the perceived reliability of an independent audit opinion and the extent to which the code might help in narrowing the audit expectation gap. The experiment conducted during this study aims to test the inter-relationships between the elements of the corporate governance code (namely the external auditor's independence, the internal auditor's effectiveness, disclosure and transparency, and accountability) as regards to their influence on the auditing profession in Jordan, from the institutional investors' perspective. In particular the study sets out to:

- 1- Investigate the impact of the external auditor's independence, in the post JCGC implementation period, on the reliability of the audit report and the size of the audit expectation gap. This includes reviewing the related literature of the impact of the audit partner rotation, preventing non-audit services, external auditor's selection, remuneration, and dismissal by the audit committees, and the role of a well-established audit committee in supporting and communicating with the external auditors.
- 2- Investigate the impact of the internal auditor's effectiveness, in the post JCGC implementation period, on the reliability of the audit report and the size of the audit expectation gap. This includes reviewing the previous studies and highlighting the role of appropriately constituted audit committees in appointing the internal auditors of the company, setting their budget and communicating with the internal auditors, in addition to their role as liaisons between internal and external auditors.
- 3- Investigate the impact of the disclosure and the transparency, in the post JCGC implementation period, on the reliability of the audit report and the size of the audit expectation gap. This also includes reviewing the literature of how adequate, timely, and transparent information influence the institutional investor's perception.

- 4- Investigate the impact of accountability, in the post JCGC implementation period, on the reliability of the audit report and the size of the audit expectation gap. This includes reviewing the literature of how and why the appropriately constituted responsibilities and authorities of external auditors and governance bodies involved with auditor accountability affect institutional investor's perception of the reliability of the audit report and the size of their expectation gap.
- 5- Determine whether Jordanian institutional investors consider and evaluate the impact of the elements of the JCGC individually or in combination (configurally).

# **1.5 Research Methodology**

Before embarking on any research project, the researcher should develop a plan, outlining a research approach and the techniques that will be used to collect the necessary data. The importance of obtaining high quality data cannot be overestimated as, according to Gill & Johnson (2010), the quality of the findings of the research project are determined by the quality of the data on which they are based. Corporate governance studies tend to use an agency theoretical frame, with research objectives expressed in testable hypotheses and using publicly available information, and in some instances a survey method is used to collect data (Saunders, Lewis, & Thornhill, 2012). The research design can be either qualitative or quantitative or a mixture of both. This research will use a strategic combination of approaches including semi-structured interviews and a factorial experimental design.

#### 1.5.1 The mixed method approach

Although business research predominantly uses quantitative methods, the use of mixed methodology has become common in some fields of business research (Collis & Hussey, 2003). The mixed method approach uses both qualitative and quantitative methods to achieve the study objectives. Mixed methods can provide both quantitative and qualitative research strengths and offset their different weaknesses, and can also increase the generalizability of the results (Bryman, 2006; Johnson and Christensen, 2010). Conducting an experimental research design alongside a qualitative method may deliver better understanding of the research problem (Burns, 2000).

#### 1.5.2 In-depth Interviews

Semi-structured interviews are used in this thesis because there have been few studies dealing with Jordanian corporate governance in the context of the role of audit committees as liaisons between corporate governance constituents. In Jordan, as in many developing countries, there is limited availability of financial information beyond the financial statements (Afifi, 2009). A possible explanation could be that Jordan has only recently applied a formal code of corporate governance, or it could be due to the availability of other mechanisms of acquiring information, or it could be that there is greater reliance on substantial investor-directors to look after all shareholders in a close knit commercial community, or it could be due to yet unrecognized factors.

Given the current incompleteness of knowledge, interviews offer various useful advantages, such as getting higher response rates, greater depth of information and the availability of the interviewer to clarify any misunderstood questions. The use of semi-structured interviews provides flexibility that allows new questions and topics to arise as a result of what the interviewee says (Lindlof and Taylor, 2002; Gendron and Bedard, 2006; Al-Lehaidan, 2006). In

this way, the insights gained are less constrained by the researcher's preconceptions of the likely relationships.

#### **1.5.3 Experimental Treatments**

Factorial experimentation enables researchers to evaluate interaction effects between two or more independent variables, something that cannot be obtained from single-factor experimentation (Winer et al, 1991). "Interaction effects" are effects that cannot be predicted from the sum of effects of individual variables but, rather, from the interaction of these variables together (Winer et al, 1991, p.284).

The factorial experiment method is a systematic treatment of the study variables. It allows the researcher to manipulate and control the independent variables and observe the resulting variation in the dependent variables under each scenario (Kerlinger, 1986; Collis & Hussey, 2003).

The research objectives of this study are addressed by conducting a qualitative in-depth investigation (interview) alongside the experimentation. A completely crossed (2^4 = 16 cases) factorial experiment is employed to investigate the influence (including relative and interactive weights) of the selected elements of the Jordanian corporate governance code on the perceptions of their impact on the value of the audit opinion as measured by the reliability of the audit report and the size of the audit expectation gap. "Another advantage to this design is that we can use each participant as his or her own control 'group,' thereby reducing the amount of variance within each treatment condition in the ANOVA" (Weathington, Cunningham & Pittenger, 2010, p. 446).

Experimental treatment provides a strong platform to ensure the evidencing of the causality of the effect of the independent variables on the dependent variables, and also helps identify any

interaction between the four independent variables in influencing the dependent variables (Keppel & Wickens, 2004; Creswell, 2009). The experiment method is recommended and well established in auditing, accounting and financial research (Libby & Lewis, 1982; Black, 1986; Chewning & Harrell, 1990; Baker and Nofsinger, 2002; Trotman, 1996; Wang & Tuttle, 2009; Krishnamoorthy, 2002; Brown, 1983; Ashton, 1974; Hofstedt & Hughes, 1977; Stocks & Harrell, 1995; Ashton & Ashton, 1995; Hooper & Trotman, 1996), and particularly in those studies that have investigated informed and knowledgeable participants such as financial analysts, professional auditors and accountants (Arnold, Bedard, Phillips, & Sutton 2011; Teoh & Lim, 1996; O'Reilly, 2009; Ganzach, 1994).

# **1.6 Study Population**

The main purpose of sampling is collecting the data needed to answer the research questions. This thesis aims at evaluating the current JCGC as regards to its impact on the reliability of the audit report and the level of the audit expectation gap in Jordan. This expectations gap arises from the differences in perceptions between the auditors and users of audit reports. The main users of audit reports are the investors; therefore, this thesis investigates the perceptions of investors in Jordan.

Investors could be either individuals or institutions. In order to provide more reliable results based on well-informed respondents' feedback, in addition to the justifications mentioned in chapter four<sup>8</sup>, particularly the differences in perceptions of individual and institutional investors as disclosed in previous Jordanian research, the study focuses on institutional investors. In addition, the suggestion that developing markets are more reliant on institutional investors to ensure the integrity of the market further justifies the focus on institutional investors (Afifi, 2009; Zureigat, 2010; O'Reilly, 2009). Therefore, the study sample of this thesis is derived from institutional investors in Jordan rather than individual investors; they are the financial analysts

<sup>&</sup>lt;sup>8</sup> See sections: 4.3 & 4.4

who work in institutional investment companies including insurance companies, investment and portfolio departments in banks, investment funds and companies, retirement funds, mutual funds and brokerage companies. The following figure summarizes the mixed method approach of this study.

## Figure 2: The Mixed Method Approach Adopted in this Thesis

# Research Methodology & Data Collection



## **1.7 Definitions**

This section provides definitions of key terms used in this thesis. An operational definition is "specification of how you will define and measure the variable in your study" (Creswell, 2008, p. 160). As the definitions presented in this section indicate the meaning of the terms used in this thesis, where applicable, this study adopts definitions from the Jordanian corporate governance code, The International Standards on Auditing (ISA) (which are the valid standards in Jordan), The Organization for Economic Cooperation and Development (OECD) and other resources that provide comprehensive definitions that help the researcher in achieving the goals of this thesis and/or in constructing the study's own definitions.

The Jordanian Corporate Governance Code (JCGC): Refers to the JCGC approved and issued by the board of the commissioners of the Jordanian Securities Commission (JSC) on the 29th of July 2008 for the publicly traded companies listed on the Amman Stock Exchange (ASE) and which came into effect as of the first of January 2009. The JCGC contains five chapters as follows: Definitions, The Board of Directors of the Shareholding Company, General Assembly Meetings, Shareholders Rights, and Disclosure and Transparency.

**Auditors' Independence:** "Freedom from those pressures and other factors that compromise, or can reasonably be expected to compromise, an auditor's ability to make unbiased audit decisions" (ISB, 2000, para. 4).

Audit committees: Audit committees are sub-committees of the board of directors of at least 3 members established by and amongst the members of the board of directors of a company whose main responsibilities are reviewing the annual financial statements and reporting before submission to the board of directors, safeguarding the internal audit function, nominating and reviewing the findings of the external auditors, in addition to their role as liaisons between internal auditors, external auditors, executive management and the board of directors.

**Internal Audit Effectiveness**: Internal Audit Effectiveness refers to the extent to which the designated internal audit objectives and functions are achieved properly, are unbiased, and are free from management pressure that may compromise the internal auditor's performance. Examples of those designated internal audit functions are safeguarding assets against loss and theft, providing reasonable assurances that the financial and operating information are accurate and reliable, and ensuring the entity's compliance with laws and regulations.

**Disclosure and Transparency**: Refers to the quality and reliability of information, whether financial or non-financial, provided by the company to the shareholders and users of the financial reporting (International Chamber of Commerce, 2011). For the purpose of this study, it refers to the disclosure and transparency items as mentioned in chapter five of the Jordanian Corporate Governance Code.

**Corporate Accountability:** Refers to the ability of the shareholders and the stakeholders to hold the governing body of the company, such as the executive management, board of directors and external auditors, accountable and answerable to the laws, the regulations and the company's bylaw, based on their power and responsibilities (Porter, 2009; Gay & Simnett, 2010; Harris & Spannier, 1976).

**The going concern assumption:** The going concern assumption states that the financial statements of an entity should reflect that the value of its assets and liabilities are recorded on the basis that the entity will be able to realize its assets and discharge its debts and liabilities when they fall due in the normal course of operation for the foreseeable future, unless there is evidence to the contrary.

**The audit expectation gap:** The audit expectation gap refers to the different perceptions between the auditors' actual performance and to what report users believe the auditors should be performing and achieving as auditors.

**Institutional Investors:** An institutional investor is an individual who works in the investment section of an organization that trades securities for investment purposes on behalf of the organization's members. Common examples are an insurance company, a bank, a hedge fund, a retirement fund, a mutual fund, brokerage, or another such group that has a large amount of cash or assets to invest.

**Reliability of the audit report:** refers to the extent to which the audit report clearly conveys significant instances of poor internal controls, accounting errors, biases, or omissions; and concerns regarding the application of the going concern assumption. Such reports can be relied upon by institutional investors, enabling them to make informed investment decisions.

**Error**: "The errors refer to the unintentional misstatement in the financial report, including the omission of an amount or a disclosure" (IFAC, 2010b, p. 18).

**Fraud**: "Intentional act by one or more individuals among management, those charged with governance, employees, or third parties, involving the use of deception to obtain an unjust or illegal advantage" (IFAC, 2010b, p. 19).

**Independent Member of the Board of directors:** "A member of the board of directors who is not tied to the company or any of its upper executive management, affiliate companies, or its external auditors by any financial interests or relationships other than their shareholding in the company that may be suspected to bring that member benefit, whether financial or incorporeal, or that may affect his/ her decisions or lead to exploitation of his/ her position with the company" (Corporate Governance Code in Jordan, 2009, P. 5). Nor should the member be a recent employee of the company or its suppliers as those relationships could cause conflicts of interests.

**Non-Executive Board Member:** The board member who is not staff or employed by the entity, and accordingly does not get a regular salary or remuneration from the entity and is not

involved in the day to day management of the company. Such members may be remunerated for their role as a board member only.

#### 1.8 Summary of the Study Limitations and Prospective Future Studies

As mentioned earlier, there are three sets of corporate governance codes in Jordan; one for each of the following groups: the banking sector, the insurance sector and publicly traded companies. The code for publicly traded companies is the corporate governance code investigated in this thesis.

This thesis aims to identify the impact of the elements of the Jordanian corporate governance code in so far as they relate to the perceived reliability of the audit report and the extent to which they narrow the level of the audit expectations gap, in the perception of institutional investors in Jordan. In doing so, this thesis endeavours to contribute to narrowing the expectation gap. This thesis focuses on the external auditor's independence, the internal auditor's effectiveness, disclosure and transparency, and corporate accountability. Other studies (Ojo, 2009; Siddiqui, Nasreen, & Lema 2009; Dickins & Higgs, 2009; Rehana, 2010; Sikka, Puxty, Willmott, & Cooper, 1992; Sidani & Olayan, 2007; Haniffa & Hudaib, 2007; Fadzly & Ahmad, 2004; Pierce & Kilcommins, 1997; Wolf, Tackett, & Claypool 1999) have investigated and pointed out the importance of other factors on the reliability of the audit report, such as the role of accounting professional associations, the code of professional ethical conduct in relation to auditing, the role of audit education on narrowing the audit expectations gap, and the role of the auditor's experience, integrity and the size of the non-audit services provided by the auditors. The scope of this thesis and the unit of analysis is limited to institutional investors in Jordan and excludes other users of financial statements such as the managers, members of the board of directors, creditors and the credit staff at banks, as well as the preparers of the financial statements, such as the company financial managers and accountants.

These limitations might open the door for further studies. Accordingly, there are opportunities for further research to be undertaken in Jordan in order to identify a comprehensive approach, to be adopted alongside implementation of and education about the JCGC, to minimize the audit expectation gap.

#### 1.9 Structure and Overview of the Thesis

This thesis - including this introductory chapter - is organized into six chapters, which are briefly described below:

Chapter One provided a background for and an overview of the thesis; the justification for the study, a statement of the study research questions and objectives, and the organization of the remaining chapters of the thesis. Chapter One also looked at selected prior Jordanian research, with a view to identifying the gaps in the research which can be addressed by this thesis. Chapter Two provides a theoretical and historical perspective on corporate governance and the auditing profession in Jordan, and the definition, establishment and need for the corporate governance code. Chapter Two also discusses, in the context of authoritative guidance, the development of audit committees, internationally and locally, and communication between both external and internal auditors with the audit committees. Chapter Three reviews the literature relevant to the main themes of the thesis. Among the themes discussed in Chapter Three is the audit expectation gap phenomena and its effect on the auditing profession and investors' confidence, including the nature and the structure of the audit expectation gap in Jordan. The review presented in Chapter Three covers and critically analyzes the study of independent and dependent variables mentioned in the model of this thesis.

Chapter Four reviews the research methodology employed in the fieldwork of the thesis and based on the study model. Chapter Four provides a justification for the chosen research strategy and provides a rationale for combining the qualitative and the quantitative methods.

Sampling, data collection techniques and the limitations of this methodological approach are also presented and discussed.

Chapter Five reports an analysis and interpretation of the results of both research methods. The chapter describes and discusses the interviews and analyzes the interviewees' themes and feedback. Chapter Five also presents a description of the respondents to the survey instrument and discusses the validity of the experiment. The chapter concludes with a presentation of the integrated data from the two methods.

Finally, Chapter Six summarizes and discusses the main findings of the study, in line with the extant literature. Recommendations, potential areas for future research and recognition of study limitations on the scope and design are presented.

# Chapter 2: An Overview of Corporate Governance in Jordan

#### 2.1 Introduction

This chapter addresses the application of corporate governance and provides an overview of the auditing profession in Jordan. The chapter explores the theoretical framework of the corporate governance, and the importance of adopting the corporate governance codes to reform the capital markets and mandate that listed companies provide the shareholders and other stakeholders with adequate disclosure, and transparent information. In addition, this chapter provides a review of the definitions of corporate governance; as well it focuses on the development of the roles of the audit committees in order to enhance the independence, and the effectiveness of the external and the internal auditors. In conclusion, this chapter discusses the agency theory, and delivers an understanding of the nature of corporate governance and auditing profession in Jordan.

#### 2.2 The Theoretical Framework

Theoretical frameworks are tools for explaining and analyzing complex concepts. Several such frameworks have been developed for corporate governance, each examining the concept from the perspective of a different discipline, profession or school of thought (Solomon, 2010). These different theoretical frameworks often varied only slightly, and some applied different terms to

describe the same concepts (Solomon, 2010). Leung, Paul & Cooper (2011) reviewed three explanatory theories that have justified the use of audits such as the information hypothesis, an insurance hypothesis, and agency theory, while Adelopo (2010), in his study about the impact of corporate governance on auditor independence, referred to the stakeholder theory, and pointed out that the essence of corporate activity is not only for the benefit of the shareholders, but also for the benefit of all relevant stakeholders.

#### 2.2.1 Corporate Governance & Agency Theory

Theoretically, the agency theory framework has been employed in corporate governance studies (Saunders et al, 2012). Dalton, Daily, & Roengpitya (2003) argued that most research on corporate governance is based on agency theory. The critical concern of corporate governance emerges from the separation of ownership of the entities, and the control of those entities (Shleifer and Vishny, 1997b). This separation gives rise to agency problems<sup>9</sup> (Jensen and Meckling, 1976).

When the era of limited liability shareholding companies first began, it heralded a dramatic expansion in investment, and ownership in companies, by the general public, but it also resulted in a fundamental change in the way that companies were controlled (Solomon, 2010). Under the new system of ownership, shareholders generally delegate the day to day control of the company to managers, essentially separating company ownership and control and resulting in the now infamous phenomenon known as the 'agency problem' (Solomon, 2010).

<sup>&</sup>lt;sup>9</sup> There is a double agency problem. Individuals invest in superannuation funds and in financial institutions. These organisations then invest in listed companies on behalf of their contributors. The executives and directors of the superannuation funds and financial institutions are often influential in appointing or removing directors of listed companies who in turn appoint senior managers and structure their reward systems. Thus primary investors are subject to two agency problems namely their initial investment being managed by others which is in turn placed with agents in the form of directors of the listed companies. The audit committees of the listed companies address the second agency problem. Also there are potential conflicts between the interests of major shareholders and minority shareholders.

Jensen and Meckling (1976) argued that conflicts of interest inevitably exist between the management and the owners of businesses in cases where owners are not managers. This is because the theory assumes a model of a man (manager) who is self-serving, individualistic and opportunistic in nature, prefers to maximize his own utility functions at the expense of the owners<sup>10</sup>. As a result, the theory is built on the assumption that there is almost always a divergence of objectives between the goals of the management, and those of the shareholders. The Effective corporate governance is seen as an active tool in reducing agency problems because it "promotes open dialogue and communication while ineffective corporate governance gives rise to conflicts and lack of transparency" (Leung, Paul, & Cooper 2011, p. 814).

The relationship amongst corporate governance constituents has been described based on the agency theory (Jensen and Meckling, 1976; Adams, 1994; Christopher, Sarens & Leung 2009) as follow: The shareholders elect a board of directors which meets a few times each year; the board of directors is responsible for appointing the company's management, including appointing a chief executive officer to lead the managers; as the shareholders' agents, the board of directors and the company management are charged with the responsibility of controlling the company; Independent external auditors are appointed to provide the shareholders with assurance on the quality of the financial statements prepared by company management but, in practice, the value of this assurance depends on the external auditors perceiving that their responsibility is towards the company owners rather than to the company management. In order to protect the interests of the owners, an audit committee is appointed to ensure that an appropriate relationship is maintained between external auditors and company management. An independent internal audit function that is adequately funded can further monitor internal controls and adherence to policies and risk profiles, and in that way provides further assurance to the board, via the audit committee.

<sup>&</sup>lt;sup>10</sup> Some theories such as stewardship theory which sees managers as stewards carrying out the wishes of the shareholders. The reality is probably a mixture of good stewards and managers who are gaming the system. Agency systems are designed to curb the latter.

As shown by this description of the relationship between the various constituents of corporate governance systems, a separation between ownership and management is a key characteristic of public shareholding companies. The problems created by this separation have been prompted significant debate among scholars, investors, finance and accounting professionals, regulators and law makers, regarding how to best ensure that management works in the best interests of company owners (Demsetz & Lehn, 1985; Allen & Gale, 2001; Adelopo, 2010). In the face of the ineffectiveness of market mechanisms alone to control management behavior, it would seem, as suggested by Solomon (2010), that there is a need to reform corporate governance.

#### 2.2.2 Agency Theory & the Auditing Profession

The need for independent auditors, according to Imhoff (2003), was a direct result of the separation of company ownership and management that arose following the formation of capital markets that grew to feed the industrial revolution in the 18<sup>th</sup> century. The separation of ownership and management created opportunities for management to use their authority in ways that did not serve the owners' best interests, thus creating a need for an independent party to provide owners with assurance regarding management's reports (Imhoff, 2003; Leung, Paul, & Cooper 2011). In the framework of the agency theory, auditors are responsible for monitoring and controlling the quality of financial reporting and, consequently, are part of the corporate governance system (Beasley & Salterio, 2001). Auditors play a significant corporate governance role in helping shareholders monitor and controlling the quality of financial statements and reports fairly present all materially significant facts, thus lending credibility to the disclosures in management's financial reports and, consequently, increasing investor confidence in the transparency of the company (Solomon, 2010). Given the importance of the role of independent auditors, procedures for

enhancing the independence, objectivity and professionalism of auditors are included in most governance codes and guidelines (UNCTAD, 2006).

#### 2.3 Definitions of Corporate Governance

The following three sections review the development of the term corporate governance and explore the different translations of the term in the Arab world in general, and in Jordan in particular, and the most popular and generally accepted definitions of the corporate governance are discussed in the last section.

#### 2.3.1 Introduction

Corporate governance is a "fashionable" terminology, albeit an ambiguous concept (Farrar 2005, p. 141). In order to understand the concept of corporate governance, it is important to highlight its emergence and definition. Corporate governance has not identified with a unique accepted definition (Ching, Tan, & Ching, 2006). Abu-Tapanjeh (2009) traced the emergence of the term governance and pointed out that this term was originally derived from the Greek word "kyberman" which referred to guide, direct, and govern, then that term was transferred to the Latin and had been known as "gubernare", while the French version of the term is "governor".

Corporate governance is not a phenomenon solely for shareholding companies, it is also applicable for all sorts of entities and its definitions could be expanded to cover all of the economic and non-economic firms and government organizations. Abdullah and Valentine (2009) pointed out that corporate governance literature contributed to ambiguity around the terms regulate, control, manage, govern and governance.

#### 2.3.2 Translation Dilemma

The concept of corporate governance in the Arab world has not been understood properly since its first emergence and adoption, in fact, there is no definite acceptable translation of the term 'governance' in Arabic (Boutros-Ghali, 2001). This confusion of terminology has impacted the discussion seminars<sup>11</sup> and the implementation of corporate governance in Jordan. Moreover it has affected academics and researchers in designing their studies. As an example, Abdullatif (2007) avoided the use of the Arabic translation of the term corporate governance in his questionnaire in order to reduce terminological issues.

The Academy of Arabic Language has endorsed the word 'Hawkama' as an Arabic translation after discussions of the other alternatives (Saeed, 2003), while the Jordanian Academy of Arabic language adopted the word 'Hakemeyya'<sup>12</sup>. In spite of the adoption of the word 'Hakemeyya' by the Jordanian Academy of Arabic, there is no consensus on a single term in the Jordanian literature. Most of Jordanian researches used the term 'Hakemeyya' (Al-Hanini and Dahmash 2008; Matar, & Nour, 2007; Dahmash & Abu-Zerr; Zureigat, 2011a; Al-Barghouthi, 2009; Ishhadat & Abed Al-Jaleel, 2012). Some other Jordanian literature used the term 'Hawkama' (Al-Manzou, 2009; Ahmad, 2003; Ashmawi, 2005; Shehadeh, & Barghouthy, 2009; Al-Aqdeh, Juda, & Saadah, 2011; Al-Waked, 2007). Others have used the term 'Tahakkom' as suggested by Al-Basheer (2003), the previous chairperson of the Jordanian Association of Certified Public Accountants, (Abdullatif, 2007; Bashtawi and Suleiman, 2003; Matar, 2003).

Interestingly, it is noteworthy to mention that the inconsistencies of the translation can also be seen in the Jordanian governmental regulatory bodies themselves, they have not adopted a specific term consistently; in the Jordanian corporate governance code for the publicly shareholding companies, issued by the JSC, the term 'Hawkama' was used, while the corporate

<sup>&</sup>lt;sup>11</sup> The Policy Debate on Economic Issue has referred to the confusion of these multiple translations on its periodic. See (Jordanian Forum for Economic Development, 2004) page 5.

<sup>&</sup>lt;sup>12</sup> The Decision No. 200406 issued by The Jordanian Academy of Arabic language on 8 June 2004.

governance code for the bank sector, issued by the CBJ, and the corporate governance code of the insurance sector, issued by IRC have used the term 'Hakemeyya'.

It is important to reach an acceptable Arabic version that does not compromise the theme of the term to be used consistently in the literature. This thesis is written in English, since an Arabic version of the questionnaire that should be provided to the subjects and the interviews should be conducted in Arabic, this thesis used the term "Hawkama" because this is the term that has been employed in the JCGC for the Jordanian listed companies. However, the ambiguity of the new term caused initial confusion, currently academicians and institutional investors are aware of these different translations.

#### 2.3.3 Corporate Governance Definition

Discussions of the corporate governance issues came into the spotlight due to big corporate scandals all over the world. Issues such as the agency theory problems and how to ensure the rights of the shareholders in the entities, joint corporate-audit firm collusion, poor internal control, weak accountability system and insufficient evaluation of the risks are examples of some catalysts that have contributed to corporate collapses and raised the call to reform the capital market and to establish and adopt best practices guidelines of corporate governance. Plessis, McConvill, & Bagaric (2005) pointed out that there is no consensus on a specific definition for corporate governance although many academicians and scholars have attempted to establish a comprehensive definition.

According to Solomon (2010) and Millstein (1998) the concept of corporate governance has many definitions. Most of the definitions have a common denominator which are: the accountability, the disclosure and transparency, protecting the rights of the shareholders and the stakeholders alike, organizing the roles and the responsibilities of the board of directors and the company management, and to reinforce the effectiveness and the performance of the

external and the internal auditors in order to ensure the soundness and the accuracy of the financial reporting system (Archambeault, DeZoort, & Holt 2008; Bushman, Piotroski, & Smith 2004; Dalton et al, 2003; Monks and Minow, 2011; Farrar, 2005; Levitt, 1999; Cadbury Report, 1992; Zingales, 1994; Shleifer and Vishny, 1997a; IIA, 2011; Parkinson, 1995).

The most widely used definition of corporate governance is the definition of the Cadbury Report (Applied Corporate Governance, 2012). It defined corporate governance as "the system by which companies are directed and controlled" (The Cadbury Report, 1992, section 2.5). The codes of best practice of the Cadbury Report focused on the principles of accountability, disclosure and transparency, and probity. In addition to the equity, these four concepts have become the benchmark and the foundation for forming effective corporate governance (IIA, 2012). The four concepts are explained in figure 3.



Figure 3: Effective Elements of Corporate Governance

This figure is quoted from the Institute of Internal Auditors website (IIA, 2012)

Finally, as Jordan has mainly referred to the Organization for Economic Cooperation and Development (OECD) principles of corporate governance, this thesis adopted the definition of the OECD. The OECD defined corporate governance as "Procedures and processes according to which an organization is directed and controlled. The corporate governance structure specifies the distribution of rights and responsibilities among the different participants in the organization – such as the board, managers, shareholders and other stakeholders – and lays down the rules and procedures for decision-making" (OECD Glossary, 2007. p. 151).

#### 2.4 Importance of Corporate Governance

Corporate collapses have negative consequences for the society because the companies are the livelihood of many communities. Corporations' governance has affected economies, and a wide range of aspect of the social environment. Corporations are banks, insurers, factories besides many other sorts working under the private sector system. Corporations generate jobs, income, incorporate capital with labor, raw materials, and management expertise to produce goods and services. Corporations also supply the state treasury with revenue and taxes (Prentice and Holland, 1993). Therefore, countries and states around the world, in relation to their public social security responsibilities, should take decisions, and implement measures to maintain the soundness of the private sector. Hence, the regulatory and the supervisory bodies of the capital market issues their own corporate governance codes of best practices and they also monitor the shareholding companies' compliance with it financially, politically, and even ethically (Cohen & Boyd, 2000).

Shareholder's confidence in the capital market and affects the market value of securities and the availability of investment funds. With the emergence of globalization, there is a threat to the ability of any government to control large multinational companies. That state of affairs places increased importance on the role of the economic regulatory bodies and also a need for establishing sufficient accountability (Crane and Matten, 2010). The corporate governance

codes specify guiding principles for all participants involved with the entities corporate affairs in regards to their rights, duties, obligations and accountability, in addition to that it identifies and evaluate any potential challenges that corporations may encounter (Healy, 2003).

Internationally, big corporate scandals in the early 21st century such as the demise of Enron-Arthur Andersen, HIH and WorldCom have undermined the stakeholder's confidence in the world's capital markets and in audit reports<sup>13</sup>, and also have motivated economic policymakers, researchers and academics to develop and find solutions of the weak points of the corporate governance structure. In 2008 the world has witnessed the global financial crisis which was triggered by the collapse of Lehman Brother Bank in August 2008. This financial crisis has raised and directed critical questions to the auditing profession such as the age-old question of where were the auditors? (Tricker, 2009). Cooper & Grose (2010) reviewed audit firm scandals and raised the question of can auditors be trusted? Cheng and Abdel-Qader (2010) raised the argument of whether auditors were guilty for surprise corporate collapses.

Building and adopting a strong corporate governance structure helps achieve an entity's objectives such as implementing their strategic long-term goals by maximizing the owners' wealth, securing the rights of the employees by ensuring the company is providing them fair health coverage, training, and retirement security, it also aims to maintain good relationships with outsider parties by providing good quality services to the entity's clients and suppliers, and finally a strong corporate governance system enforces the company's compliance with the laws and the regulations (Sheridan & Kendall, 1992; Cadbury, 2002; Millstein, 1998; Gregory & Simms 1999).

A survey of the Institutional investors' perception of good corporate governance in the U.S has been conducted by McKinsey and Company in 1996. The survey findings revealed that investors believe that good corporate governance is important in their decision making to the same

<sup>&</sup>lt;sup>13</sup> The audit expectation gap is at least partly due to the failure of audit firms to stand for clear disclosure to shareholders rather than assisting management to avoid their accountability to shareholders e.g. Enron. Hence the need for an audit committee to ensure they have the right priorities in spite of the financial incentives for partners to align their interests with management.

degree of the importance as the financial performance. The study found that two-thirds of the participants revealed that they would readily pay on average over ten percentage points for well-governed<sup>14</sup> entities. In June 2000, McKinsey and Company repeated this survey again and expanded the countries from which participants were drawn; the study surveyed Europe, Asia, and Latin America. Over 200 institutional investors in Asia, Latin America, Europe in addition to USA, representing US \$3.25 trillion in assets were surveyed and similar results were obtained (For more details about the two surveys see: McKinsey and Company, 2000; Felton, Hudnut, & Heeckren 1996; Hopkins, 2007; Block & Epstein, 1995).

#### 2.5 OECD Principles of Corporate Governance

As mentioned earlier, the Jordanian corporate governance code for listed companies is based on the non-binding OECD principles of corporate governance which were endorsed by the OECD Ministers in 1999. These codes were revised and replaced in 2004, albeit remain nonbinding, OECD principles proved its efficiency and effectiveness to the extent that even nonmember countries adopted and implemented these principles (Morck, 2006; Jesover & Kirkpatrick, 2005)

The OECD principles of corporate governance aim to provide countries, both OECD and non-OECD countries, with guidelines and good practices of corporate governance provided in order to assist in achieving these goals. "The Principles are a living instrument offering non-binding standards and good practices as well as guidance on implementation, which can be adapted to the specific circumstances of individual countries and regions" (OECD, 2004, p. 4). The OECD cooperated with the International Monetary Fund (IMF) and the World Bank in promoting

<sup>&</sup>lt;sup>14</sup> The study defined well-governed company by referring to those companies with majority of outside directors, responsive to investors' requests for any data they ask for, and had a board of directors independent from the company management to hold the executive management accountable to shareholders.

corporate governance best practices, in fact, the IMF used the OECD corporate governance as a benchmark for observing its members (Low, 2002; OECD, 2004; Kirkpatrick, 2005).

Low (2002) argued that the main four themes of OECD corporate governance principles of 1999 lies on the basis of transparency, accountability, fairness and responsibilities, Abu-Tapanjeh (2009) argued that the revised OECD principles of 2004 were built on four core concepts that are: business ethics, decision making methods, disclosure and transparency, and the integrity of the financial reporting system.

The six OECD corporate governance principles are outlined below (OECD, 2004):

Principle One: "Ensuring the basis for an effective corporate governance framework".

Principle Two: "The rights of shareholders and key ownership functions".

Principle Three: "The equitable treatment of shareholders".

Principle Four: "The role of stakeholders in corporate governance".

Principle Five: "Disclosure and transparency".

Principle Six: "The responsibilities of the board".

#### 2.6 Corporate Governance Assessment in Jordan

An important assessment of corporate governance was conducted in Jordan in 2004 by the World Bank and the International Monetary Fund joint committee. This joint initiative was referred to as (Reports on the Observance of Standards and Codes "ROSC"). It benchmarked developing country's assessment of corporate governance code against those developed by

OECD (World Bank: Corporate Governance Assessment, 2004). The conducted program provides a diagnostic tool for evaluating the strengths and deficiencies of the Jordanian corporate governance framework. The ROSC evaluation of the Jordanian corporate governance was released for publication by the Jordan Securities Commission in 2005. The main ROSC results were:

- Disclosure standards (Largely observed).
- Standards of Accounting and Audit (Observed).
- Independent audit oversight body (Partially observed).
- Fair & timely dissemination (Largely observed).

# Figure 4: ROSC Assessment of the Compliance to OECD Principles of Corporate Governance in Jordan



The figure quoted from (World Bank ROSC experience in MENA and the World, 2005, p. 5)

#### 2.7 Background of the Auditing Profession in Jordan

This section discusses the historical development of the auditing profession in Jordan simultaneously along with the relevant laws and regulations especially the companies' law and the auditing profession law in Jordan.

#### 2.7.1 The Companies' Law

The establishment of the auditing profession regulations in the Middle Eastern countries first appeared in the early twentieth century, and that exactly was in 1909 in Egypt and in 1919 in Iraq. By the passage of the time, Middle Eastern countries developed their auditing profession laws, and then established their own auditing profession associations to regulate and control the auditing profession. In Jordan, the Palestinian companies' law which was primarily based on the British Acts was applied in both countries from 1929 since they were under the British colonialism and those acts remained operational until the issuance of the provisional law in 1962 which became permanent in 1964. The companies' law of 1964 regulated the formation of the companies, liquidation, management, the financial reporting system and added some general responsibilities and rights to the external auditors (Abdullah, 1982). In 1989 another companies' law enacted and added some details to the previous one in addition to regulate the consolidations of the foreign companies working in Jordan (Abdullah, 2007).

In the middle of the last decade of the last century, Jordan was concerned of openness and being involved with the world, setting government plant reform programs, legislating and regulating the privatization to cope and deal with the globalization. In January 1994, Jordan submitted an application for the General Agreement on Tariffs and Trade (GATT) in order to gain a membership in the World Trade Organization (WTO). Jordan conducted several meetings of negotiations to answer and respond to the queries of the WTO Members on Jordan's foreign trade system and economic policies and facilities. In an endeavor to achieve this objective,

Jordan has amended, revised and enacted major legislative reforms to promote the Jordanian economic climate and to bring the Jordanian foreign trade system into conformity with WTO conditions. The majority of these amended or enacted legislations were to those related to the intellectual property rights, the free movement of capital, and taxation; examples: Trademarks law, Patents law, Copyrights laws, Trade Secrets and Unfair Competition, Customs law, General sales tax law, law on unifying fees and taxes. On the 17th of December 1999 The WTO General Council officially accepted the membership of Jordan which made Jordan the 136 WTO member (Jordan's Membership to WTO, 2011).

In 1997 Jordan issued the companies' law No. 22, this current law has been established to work in line with other commercial and capital market laws and regulations, such as the securities law of 1997<sup>15</sup>, to facilitate Jordan accession to the WTO and to accelerate the government economic reform plans. The companies' law of 1997 also added more details concerning the issuance of the shares, the face value and types of companies. The companies' law of 1997 law has been amended several times to enforce and encourage the Jordanian companies to adopt the corporate governance codes best practices. In fact, the Jordanian capital market regulatory bodies started drafting for another revised and developed companies' law (Companies Law 1997; Company Law Draft 2011; Securities law of 2002).

#### 2.7.2 Auditing Profession Law

The first specific law dedicated to regulating the auditing profession in Jordan was established in 1961. Prior to the establishment of the Practicing of the Auditing Profession in 1961, there were no regulations or conditions to control the profession including the absence of a minimum qualification requirement that the practitioners should meet to be eligible for practicing auditing. The law of the auditing profession of 1961 set license's conditions to be met by any person willing to practice auditing and required specific academic and professional

<sup>&</sup>lt;sup>15</sup> Currently, The Securities law of 1997 has been replaced with the Securities law of 2002.

qualifications and experience, the law also established committees to be responsible for overseeing the profession which includes licensing and penalizing of auditors, this committee was mainly composed of governmental members (Abdullatif, 2003; Abdullah, 1982).

In 1985, the new law of the auditing profession was established to keep pace with the developments of the economic environment, the capital market, the auditing profession and to add more details to the 1961 law especially to the external auditor's rights and responsibilities and to provide a more precise framework as a guidance to the profession. In this law, the external auditors were given the right not to be rejected by the companies during the financial period of engagement. Non audit services were not prohibited in this law, the law identified and stipulated ten prohibited acts that the auditors should avoid such as the auditors must not have another job beside the auditing, unethical competition to gain audit job or unethical advertising and disclosing their client's secrets as examples. Under this law, the Auditing Profession Council was established to oversee and control the profession and its member practices. Similar to the previous committee, the majority of the council's 12 members were from the government. Based on this law, the regulation of classifying the external auditors (No. 30 for 1986) was enacted to classify the auditors on three main categories which are A, B and C where the A category represents the highest level or the highest qualified and experienced auditors.

Another important regulation enacted by the same law is the establishment of the Jordanian Association of Certified Public Accountants (JACPA) in 1987 and became legally effective in 1988. The JACPA aims at considering the members' interest; setting codes of professional ethics conducts, encouraging scientific research, protecting the rights of the auditors, conducting seminars and training workshops to follow up and explain any new additions or amendments to the International Standards on Auditing (ISAs) in order to enhance the capacity of the auditors. The membership of the JACPA is mandatory, and all of the licensed auditors represent the JACPA's general assembly and they have the right to elect the JACPA board of directors. Contrary to the companies' law of 1989 that required companies to prepare the financial statements in accordance with the generally accepted accounting and auditing standards,

taking in consideration that this expression was ambiguous since there were no specific generally accepted accounting and auditing standards in Jordan, the JACPA adopted the ISAs and International Accounting Standards (IASs)<sup>16</sup> after about two months of the operation of the companies' law of 1989 (Khouri, 1994; Abdullah, 2007; Abdullatif, 2003).

The current auditing law that regulates the auditing profession in Jordan is Law No. 73 of regulating the Auditing Profession for the year of 2003. The article No. 3 stated that this law aims at regulating and developing the profession, ensuring the implementation of the ISAs and the International Financial Reporting Standards, develop the scientific and the professional level of JACPA members, and reinforce the certified public accountants independence and neutrality (Al-Tamimi, 2006; Al-Matarneh, 2009).

Based on Articles 29 and 45 of 2003 law, the bylaw of practicing the auditing profession No. 7 for the year of 2006 was issued to set a framework of the roles, the responsibilities, the conditions and fields for practicing the profession. The bylaw stressed the importance of following up the updates of the auditing profession and standards by the auditors. The article No. 8 identified the continuous training as following up all updates, and issues related to the profession by attending seminars and workshops or participating for training in specific institutions or by writing related articles and essays in topics related to the profession. The article No. 8 also added that the auditors should provide the JACPA with a certificate of attending at least 20 training hours or 10 hours with either participating of a research in a conference or two published articles in topics related to the auditing profession. The same article stressed that the license of practicing auditing must not be renewed unless the auditor satisfy the above requirement.

<sup>&</sup>lt;sup>16</sup> Replaced with the International Financial Reporting Standards (IFRS) after 2005.

Article No. 10 regulated the qualifications and the training experience <sup>17</sup>conditions for any individual who wants to apply for a license to practice auditing as follows (Law of Regulating the auditing profession No. 73, 2003; Bylaw of Practicing the Auditing Profession, 2006):

A. Individuals who hold an accounting degree:

1- One year of training for those who hold a PhD degree.

2- Two years of training for those who hold a Master degree.

3- Three years training for those who hold Bachelor degree.

4- Five years of training for those who hold a Diploma degree<sup>18</sup>.

B. Individuals who hold the bachelor certificate majoring in fields related to the auditing profession such as finance, banking and management as an examples with at least 24 credit hours in accounting, have the right to apply for the training as follows:

1- Two year of training for those who hold a PhD degree.

2- Three years of training for those who hold MA degree.

3- Four years training for those who hold Bachelor degree.

<sup>&</sup>lt;sup>17</sup> Item (b) from article 10 stated that the training should be conducted under the supervision of licensed auditors.

<sup>&</sup>lt;sup>18</sup> In general, the Diploma Certificate in Jordan issued by the community colleges and it is equivalent to two years after secondary school,

#### 2.8 The Development of Audit Committees

In response to recent concerns about the incidence of fraudulent financial reporting, expressed by various groups including accountants, users of financial reports, law makers and regulators, companies have, among other things, established audit committees (McMullen, 1996). These concerns have also brought about debates about the so called audit expectation gap. Narrowing the audit expectations gap continues to be a key issue of interest to the accounting profession (Dewing & Russell, 2002).

Many recall that the success of corporate governance in the organization depends significantly on the success of audit committees and the failure of the composition, responsibilities and the effectiveness of the audit committee can then cause a gap in the corporate governance system of the organizations (Abdel-Al, 2008).

A historical review of the development of audit committees in the US, Canada, UK, Australia, and Saudi Arabia will highlight the importance of such committees to ensure reliable, high quality financial reporting and thus by implication the need for an effective audit committee to enhance the integrity of companies' financial reports. In fact, the main theme of establishing audit committees emerged from the necessity of having another tool to protect the stakeholders, in addition to those protections provided by the external and internal auditors. The first introduction of the audit committees in the U.S. came in response to the McKesson & Robbins scandal where the management exaggerated the value of the company's assets and the auditors had relied on the management assertions of the receivables and the inventory (Teed, 2010; Mautz & Neary, 1979).

#### 2.9 The Development of the Audit Committees in Jordan

This section reviews, and discusses the development of the audit committees in Jordan in three stages; that are the period before the establishment of Jordanian Securities Commission (JSC) in

1997, followed by the first instructions issued by the Jordanian Securities Commission in 1998 about the disclosure requirements, accounting standards, and auditing standards in which all listed companies were mandated to establish audit committees for the first time in Jordan and then followed by the revised instructions in 2004, and finally the issuance of the Jordanian corporate governance codes which became operational in 2009.

# 2.9.1 Audit Committees in Jordan before the establishment of the Jordanian Securities Commission (JSC) in 1997

The beginning of the appearance of the shareholding companies in Jordan started in the early thirties of the last century, that time witnessed the establishment of the Arab Bank in Amman followed by the establishment of Jordan Tobacco & Cigarette Company and the Jordanian Electricity Company. By 1978 the number of shareholding companies in Jordan reached 66 companies, and their shares used to be traded in unlicensed offices. The Amman Capital Market was established in 1978 to control and organize the issuance of the shares and to regulate the trading process. In 1997, the corporation law, and the securities law were issued and the Amman Capital Market was replaced by three financial regulating bodies; that are the Jordanian Securities Commission (JSC), Amman Stock Exchange (ASE), and Securities Depository Center (SDC). The divisions of these financial institutions came in response to the calls for separating the executives and the supervisory roles of the regulating bodies and to bridge the gap between Jordan and other developed countries (Swaiti, 2008). Following to these calls for economic reforms, in 1998 the JSC issued instructions relating to the disclosure and accounting and auditing standards in which all of the listed companies were mandated to establish audit committees.
#### 2.9.2 The 1998 JSC Instructions of Disclosure, Accounting and Auditing Standards

Forming the audit committees in Jordan became mandatory for the listed companies in 1998. Based on the securities law of 1997 the Jordanian Securities Commission (JSC) issued the Instructions on Disclosure and Accounting Standards and Auditing Standards (1998) in which the JSC mandated all listed companies to establish audit committees (Jordan Securities Commission, 1998).

Jumaa (1999) criticized these instructions and regulations and argued that selecting audit committee members by and amongst the board of directors especially from those who lack accounting and financial experience compromises their effectiveness and leads the external auditors to waste time explaining their point of views. Jumaa also criticized lack of written charter and that there were no requirements regarding the qualifications of the audit committee members. Al-Basheer (2003) and Dahmash, Hajjier & Al-Farah (2003) pointed out that audit committees' primary functions are examining the reports of auditors, and its conformity with ISAs and the comprehensiveness of the scope of audits. They should also ensure that top management responds to the observations of both external and internal auditors, and make sure that the external auditors are doing their job without any obstacles facing them in effectively conducting their audit. The study also recommended a greater role and more supervision to be taken by the Jordanian capital market regulatory bodies over the shareholding companies and the auditing profession. Al-Wardat (2003) also criticized that selecting audit committee's members without proper accounting and financial qualification increases the bureaucracy of the job of the audit committees as a liaison between the external and the internal auditors.

# 2.9.3 The 2004 JSC Instructions of Disclosure, Accounting and Auditing Standards for Listed companies.

The instructions on Disclosure and Accounting Standards and Auditing Standards (1998) have been revised and replaced with the Instructions for Issuing Companies Disclosure, Accounting and Auditing Standards (2004). Abdullatif (2006) pointed out that the new revision of the Instructions in 2004 added some extended details, but did not include any new responsibilities to audit committees. The new roles and responsibilities of the audit committees have also been criticized in the Jordanian literature (Malkawi, 2008; Swaity, 2006; Abdullatif, 2006; Abdullatif, 2007; Al-Farah, 2001; Al-Khadash and Al-Sartawi, 2010; Al-Saudi, 2007; Khsharmh, 2003; Ameera, 2004) as follows:

In spite of fact that the instructions on disclosure and accounting standards and auditing standards (1998), the revised instructions in 2004, and the 'Directives for Listing Securities on the Amman Stock Exchange' (2004) mandated all Jordanian listed companies to form audit committees, the Jordanian listed companies' compliance with this rule was not fully implemented, although the latter one has clearly stated in article No.1 that any security can be listed in the ASE once the issuing company verifies that it has established an audit committee.

Swaiti (2006) and Abdullatif (2006) pointed out that at that time only about 75% of listed companies had audit committees. The study also revealed that both the banking and insurance sectors responded 100% to the requirement of establishing an audit committee and hence it was the other categories of the listed companies which had substantial non-compliance. The explanation for the high response by banking and insurance sector was due to the strict legislation and regulations set by other supervisory bodies that regulate the work of banks and insurance companies in Jordan, i.e. there are additional specific regulatory bodies whose main job is to follow up their activities. Banks in Jordan work under the supervision of the Central Bank of Jordan (CBJ), while the Insurance Regulatory Commission (IRC) supervises the work of insurance Companies.

The Jordanian literature - mentioned above of this section - also showed that audit committees were ineffective due to the insufficient power and functions stipulated in the laws and regulations, and they were not required to be independent or required to have a minimum experience in the accountancy and finance. Abdullatif (2006; 2007) pointed out that most of listed companies in Jordan formed audit committees in order to meet the revised Instructions issued by the Jordanian securities commission in 2004. Abdullatif (2006) and Malkawi (2008) also added that listed companies established audit committees without adequate power and even their role in relation to the auditor-related decision was undertaken by the company management. The audit committees did not get fair remunerations in line with their responsibilities. Swaity (2006) revealed that the audit committees in Jordan used to get similar remuneration to the other members of the board of directors, and attributed that to the assumption that audit committee members did not have extra burdens, tasks or fulfil a role that justified extra remuneration.

The literature has also criticized the lack of a written audit committee charters which affected the efficiency and the objectivity of their work especially in the decision of nominating the external auditor's where the personal relationships between audit committee members and the auditors affected the selection process (Al-Saudi, 2007; Khsharmh, 2003; Ameera, 2004).

#### 2.9.4 The Jordanian Corporate Governance Code for listed Companies

On the 29th of July 2008, the board of the commissioners of the Jordanian Securities Commission (JSC) has issued, and approved the Jordanian Corporate Governance Code (JCGC) for the publicly traded companies listed on the Amman Stock Exchange (ASE) which has come into effect as of the first of January 2009. The JCGC consist of five chapters as follows: Definitions, The Board of Directors of the Shareholding Company, General Assembly Meetings, Shareholders Rights, and the Disclosure and Transparency respectively. Chapter five of the JCGC contains four sections, in addition to the disclosure and the transparency, the chapter has also

addresses audit committees, duties of the audit committee, powers of the audit committee, and the external auditor respectively.

#### 2.9.4.1 Audit Committees Formation

The JCGC stipulated the conditions of the two committees that should be formed by the board of directors, these two committees are: the audit committee, and the nominations and compensations committee. The committees shall be composed of three or more non-executive members of the board of directors. The JCGC also stated that at least two of the members must be independent, and one of the two independent members must be the chairperson of the committee. Members of the audit committees in Jordan should have knowledge in accounting and finance, and at least one of them must have previous work experience in the accounting or financial fields, as well it must possess scientific qualifications such as a professional or an academic certificate in accounting or finance or related fields.

# 2.9.4.2 The Role of Audit Committees in Nomination and Communication with the External Auditors

The audit committees must oversee, and monitor the accounting and the internal control procedures and the auditing to ensure the integrity of the financial reporting system. The right of electing the external auditor is in the hands of the shareholders according to the Jordanian laws and the corporate governance code (corporate governance code in Jordan, 2009, chapter five, section, 2). The role of the audit committees in the selection process is to discuss matters associated with the external auditor to check whether they meet all the conditions stipulated in legislations in force, and also to ensure that their independence were not violated. The next step after the nomination is to submit recommendations to the board of directors in order to

recommend the election of the nominated external auditor by the shareholder at the general assembly meeting.

The role of the audit committees is not confined to the nomination of the external auditors in the JCGC, it goes further; once the auditors started fulfilling their job and setting their audit plan, the audit committees have to review their plan of work, and ensure that the entity provides them with all facilities needed to fulfill their job properly. In order to strengthen the attitude and the independence of the external auditors, the audit committee has to conduct meetings with external auditor of the company at least once a year without the presence of the company management or any of their representatives. The audit committee also has the power to request the presence of the external auditors to discuss and ask for clarifications of any issues related to their job and to seek their opinion in writing if necessary. Moreover, one of the audit committees' tasks is to discuss the external auditors' observations, reservations and suggestions, following up the extent of the company management's responsiveness to them, and submit recommendations to the board of directors in this regard.

All of the correspondences between the company management and the external auditors should be reviewed and evaluated by the audit committees and they should comment and submit recommendations to the board of directors. The audit committees also must review the periodic reports before they submit them to the board of directors.

# 2.9.4.3 The Role of Audit Committees in the Nomination and Communication with the Internal Auditors

The internal auditors play an important role of protecting the company's assets and the company's financial and accounting records from errors, frauds and misappropriation. The JCGC stated that the board of directors has the responsibility to take the necessary measures to ensure the integrity of the internal control procedures by establishing an internal audit unit

with functions such as: ensuring compliance with the legislations in force, the regulations of regulatory bodies, policies, procedures, and plans set by the board of directors (corporate governance code in Jordan, 2009, chapter two, section one, item 12).

The audit committees also have a role to reinforce the effectiveness of the internal auditors, and they coordinate the work of the internal and external auditors to ensure the maximum benefits of these two controlling arms. According to the JCGC, the audit committees have the power to nominate the company's internal auditors. Moreover, the audit committees have to evaluate the internal control and audit procedures, examining external auditor's assessment of the internal audit function, reviewing the internal audit reports especially those related to any breaches detected by the internal auditor, finally, the audit committee must provide the board of directors of recommendations regarding internal audit function and suggestions of any further necessary requirements for internal auditors.

#### 2.9.4.4 Criticisms to the Current Corporate Governance Codes

Suleiman (2011) criticized the power of the audit committees section as stipulated in the JCGC and pointed out that those powers are vague, and limited, in other words they should be expanded to facilitate the audit committee members achieving their objectives. Suleiman argued that they should have the right and the power to investigate those matters that affect the performance of the external auditors and the accuracy of the financial reports issued by them and recommended that all companies must improve the skills and the experience of the audit committee members.

The JCGC stipulated that "the company shall take appropriate actions to ensure that the external auditors perform their duties impartially without interference from the board of directors or the executive management" (corporate governance code in Jordan, 2009 ch.5, Section 4, D). The JCGC did not explain how the external auditors could avoid their interference

or any possible pressure from the company management. The literature<sup>19</sup> revealed that a strong and independent audit committee supports the external auditors and mitigates the management pressure to issue a clean or favorable audit opinion and also protecting them from dismissal after issuing qualified or a going concern modification. Therefore, the researcher would argue that one of the most important deficiencies of the current composition of the audit committees as stipulated in the JCGC is that it does not require all the audit committee members be independent.

Al-Jazy<sup>20</sup> (2005) criticized the selection process of the audit committee members in Jordan, and argued that it is not efficient and lacks of transparency. Al-Jazy explained and pointed out that the board of director's decision of selecting or changing any director does not require the general assembly approval. In this regard, the researcher would argue that the selection process of the audit committee members should be conducted directly by the shareholders at the general assembly meeting whereas only eligible candidates who satisfy the conditions and possess the appropriate qualifications can be nominated for the election.

Finally, the researcher found that the role of audit committee in the selection process of the external auditors is not identified clearly as it stated that the audit committees should discuss matters related to the nomination of the external auditors, and the role of the audit committees in dismissing, retaining and compensating the external auditors has not been clearly defined. Similarly, the JCGC has not mentioned any role for the audit committee to set the budget of the internal audit. Moreover, it has not required the audit committee to conduct meetings with internal auditors without the presence of the company management, and also it has not given the internal auditors the authority of accessing and reporting to the audit committee directly, although the JCGC stated that the audit committee should review the internal audit reports.

<sup>&</sup>lt;sup>19</sup> These studies have been discussed in more details in chapter three. Section 3.6

<sup>&</sup>lt;sup>20</sup> Managing Partner for the Jordanian Centre for Dispute Resolution (JCDR)

Corporate governance code for Jordanian banks is more precise than the JCGC for all listed companies in regard to the roles of the audit committees. For instance it is clearly mentioned that the role of the audit committee to suggest to the board of directors "the appointment, the removal, the remuneration, and other contractual terms of the external auditors, in addition to assessing the objectivity of the external auditors, including the consideration of any other non-audit work performed by the external auditors" (Corporate Governance Code for Banks in Jordan, Chapter 6, Item b-vi), where the JCGC stated the right of the audit committee in discussing all matters related to the nomination of the external auditors without referencing to the external auditor's dismissal or fees.

Additionally, corporate governance code of Jordanian banks stressed the necessity of that audit committees must work under a charter approved by the board of directors in which their powers and responsibilities are clearly documented. Furthermore, it states that "The Audit Committee has, by a specific provision in the written charter of its functions and responsibilities, the ability to obtain any piece of information from executive management, and the ability to call any executive or director to attend its meetings" (Corporate Governance Code for Banks in Jordan, Chapter 3, Item b-v)

Corporate governance code of Jordanian banks since its issuance in 2007 requested banks to state their compliance with corporate governance code every year within their annual reports, as part of its commitment to the disclosure and transparency requirements, (Corporate Governance Code for Banks in Jordan, Chapter 6, Item vi-1), while the JCGC requested the same in 2010, i.e. in the subsequent year of the issuance of the JCGC in a separate circulation<sup>21</sup>.

<sup>&</sup>lt;sup>21</sup> See Circular Number: 12/1/4659. dated on December 15, 2010. Issued by Jordanian Securities Commission. Available on: http://www.jsc.gov.jo/Public/english.aspx?site\_id=1&Lang=3&Page\_Id=2550&Menu\_ID2=257

#### 2.10 Chapter Summary

This chapter aimed to shed the light on the environment of corporate governance, and the auditing profession in Jordan along with the theoretical framework of these two topics. Starting by the agency theory, the context then extended by a detailed discussion, and analysis of its impact on corporate governance and the auditing profession. The chapter highlighted and discussed the needs, and benefits of adopting corporate governance in Jordan added to its emergence, assessment, the different Arabic translations of the term, and concluded with criticisms directed to the current Jordanian corporate governance code. The chapter also reviewed the development of the auditing profession in Jordan and the legislations that regulate the auditor's work and the conditions of licensing the Jordanian certified public accountant. Finally, the chapter reviewed the development of the auditories and authorities as stipulated on the JCGC in addition to their relation with both of external and internal auditors.

The next chapter (chapter 3) reviews the literature relevant to the main themes of the thesis. That chapter starts by discussing the extent of investor's reliability of the audit report followed by discussing the audit expectation gap phenomena, their effect on the auditing profession, and investors' confidence, and reviews literature of this issue in Jordan to explain the nature, and the structure of the audit expectation gap. Additionally, the extent of investor's reliability of the audit report is discussed. The presented review in this chapter has covered, and critically analyzed the study's both independent and dependent variables which were mentioned in the model of this thesis.

# **Chapter 3: The Literature Review**

#### **3.1 Introduction**

This chapter presents the literature review of this thesis. Starting with the audit expectation gap, this chapter sheds light on the meaning, nature, and causes of the audit expectation gap in Jordan. The chapter also discusses audit opinions and the reliability of the audit report as perceived by the Jordanian financial analysts. In line with the study model presented in Chapter One, this chapter reviews the Jordanian corporate governance codes and critically analyzes the influence of the four selected factors in regards to bridging the audit expectation gap in Jordan, namely; the external auditor's independence, the effectiveness of internal auditors, disclosure & transparency, and corporate governance mechanisms such as external and internal auditors, enabling them to produce reliable, accurate and independent financial reports.

#### 3.2 An Overview of the Audit Expectation Gap

Over the last decade, calls for reforming corporate governance structures have become a significant global issue. The Asian financial crisis in 1997 and corporate scandals and collapses of large entities such as Barings, HIH, Parmalat, Enron (Demirag & Solomon, 2003) and Lehman Brothers have raised fears of a global economic meltdown and raised the importance of reforming corporate governance systems. The auditing profession has also come under the

spotlight, especially after the dramatic collapse of famous companies without any prior warning signals in audit reports. These events have highlighted the phenomena of the audit expectation gap (Koh & Woo, 1998).

Ojo (2009) defined the audit expectation gap as "the difference between what users of financial statements, the general public perceive an audit to be and what the audit profession claim is expected of them in conducting an audit" (p. 3). The audit expectation gap might have a substantial effect on the continuing existence of the auditing profession, especially given recent corporate collapses. These collapses have been costly to the audit profession in many ways such as compromising the profession's reputation, incurring high costs of litigation and settling these cases in the courts, and the fear that they may increase the burdens (more responsibilities) placed on the auditors (Wolf et al, 1999). Moreover, Hassan (2004) pointed out that the audit expectation gap is considered one of the greatest dilemmas affecting the auditing profession around the world and it has affected the public's confidence in the profession's role in economic life. These potential negative impacts of the audit expectation gap have caused it to be of considerable interest to researchers of the auditing profession and to practitioners, especially in the last two decades of the twentieth century.

The expectations gap in auditing is an old phenomenon and goes back to the start of the compulsory audit in the late eighteenth century (Al-Duneibat, 1998). The use of the term 'audit expectation gap', however, likely appeared for the first time in the 1970s. Porter (1991) traced and attributed the first appearance of the term to Liggio (1974) who is considered the first academic to use the term in accounting and auditing literature. Criticisms and attacks directed at the auditing profession since the late 1960s motivated Liggio to identify the two dimensions defining the gap; the expected performance of auditors as envisioned by external auditors and as envisioned by the public who rely on the audit report's opinion of the financial statements.

In regards to Jordan, the results of the study of Al-Thuneibat (2003), which aimed to investigate the structure of the expectations gap in Jordan, are consistent with those found by Porter (1991

and 1993) and Humphrey (1997) in regard to an ignorance gap, a deficient standards gap, and a deficient performance gap.

Porter (1991, P. 4) reviewed and analyzed the components of the audit expectationperformance gap and provided the following breakdown of key components:

- 1- "The reasonableness gap: defined as the difference between what the public expects auditors to achieve and what they can reasonably be expected to accomplish".
- 2- "The performance gap: defined as the difference between what the public can reasonably expect auditors to accomplish and what they are perceived to achieve".

The latter gap i.e. "performance gap" is further subdivided into:

A- "The deficient standards gap: defined as the difference between what can reasonably be expected from auditors and auditors' existing duties as defined by the law and professional promulgations".

B- "The deficient performance gap: defined as the difference between the expected standard of performance of auditors' existing duties and auditors' perceived performance, as perceived by the public".

An audit expectation gap can, according to Humphrey (1997), be defined narrowly as a "roleperception gap" between what could reasonably be expected from auditors and what is actually expected by the users of financial reports. This definition of the audit expectation gap suggests that some portion the gap (perhaps the entire gap) is, in fact, an "ignorance gap" and that it can thus be narrowed or closed by educating the users of financial reports. Knutson (1994), on the other hand, argued that the audit expectation gap was a difference between the expectations of users of the audit reports and the insufficient performance of external auditors, thus suggesting that the "ignorance gap" was not the problem. As mentioned in Chapter One, in the definitions and limitations sections, this thesis deals with the auditors' performance gap. For the purposes of this thesis, the audit expectation gap refers to the difference between the auditors' actual performance and to what report users believe the auditors should be performing and achieving as auditors.

In response to Porter's analysis of the gap, Gray and Manson (2006) argued that lack of external auditor independence and competence are the main causes of deficient performance. The performance gap could also be narrowed by strengthening the external auditor's performance. This thesis aims to investigate the role of the JCGC and support from the audit committees in reinforcing perceived external auditor's independence and performance. External auditor independence is the cornerstone of the auditing profession (Leung, Paul, & Cooper 2011), and reinforcing external auditors' independence leads them to express an independent opinion about the company's financial position unaffected by any management pressure.

According to Sidani and Olayan (2007), resolving the audit expectation gap does not necessarily require an increase in the auditors' responsibilities. The gap can be bridged by adopting an effective corporate governance mechanism which ensures the sharing of the organization's responsibilities and accountability amongst corporate governance constituents, including top management, the board of directors, external auditors, internal auditors and the capital market regulatory bodies.

Corporate governance bodies with effective constituents and co-operation between the auditors and audit committees can help establish effective internal control procedures, enhancing directors' accountability and transparency, which are crucial factors in narrowing the expectations gap. Hakim (2002) pointed out that agency problems could be reduced by adopting corporate governance best practices.

### 3.3 The Audit Expectations Gap in Jordan

The audit expectation gap partly exists because of the difference in perceptions and beliefs between the users of the audit reports and the auditors in regard to the auditors' duties and responsibilities and the messages implied by the external auditor's reports (Koh & Woo, 1998).

The audit expectations gap has re-emerged in Jordan during the last two decades due to the modernizing of audit legislation, the privatization of some large public institutions which has led to an increase in the size, number, and power of shareholding companies in the market (Abdullatif, 2003), and due to the "increasing expectations from auditors" in Jordan (Al-Thuneibat et al, 2007, p. 85). Asfoor<sup>22</sup> (2003), in the 5th Scientific Conference of JACPA, pointed out that the practical reality of the auditing profession in Jordan reveals a gap between auditors and non-auditors regarding external auditor's performance. In addition to that, corporate collapses and scandals in Jordanian entities, such as the failures of Petra Bank, the scandal of Magnesia Company and the scandal involving credit facilities to an IT company (Al-Awaqla, 2008; Malkawi, 2008; The Economist, 2003).

As mentioned in Chapter One, the audit expectations gap does exist in Jordan (Al-Thuneibat, 2003; Obaidat, 2007; Abdullatif, 2003; Omari, 2003; Hajir, 2001). The nature and the structure of the audit expectation gap are briefly presented below:

Al-Thuneibat's study (2003) investigated the causes and the structure of the audit expectations gap in the perception of external auditors and company managers, and concluded that the audit expectations gap did exist in Jordan, and consisted of three main components: a deficient standards gap, an ignorance gap, and a deficient performance gap.

A study by Obaidat (2007) aimed at investigating the gap between external auditors and investors in regard to the qualitative characteristics of accounting information. The study

<sup>&</sup>lt;sup>22</sup> Member of the founding committee of the Jordanian Association of Certified Public Accountants.

revealed that the primary focus of investors lay on timeliness while auditors primarily focused on neutrality.

The audit expectations gap and the auditors' responsibilities in investors' perceptions have been investigated by Hajir (2001). The study found that the low levels of disclosure of financial information, auditor independence and auditor performance and efficacy contributed to widening the expectation gap in Jordan. The study defined the appropriate level of disclosure, stating that financial reports and the notes accompanying financial statements should include all the information required to reflect the financial position of the company and enable investors to make informed investment decisions.

A study by Omari (2003) investigated the expectation gap in regard to auditors' responsibility to express their opinion about the entity's ability to continue as a going concern, the auditor's independence, and the scope of the auditor's responsibility in identifying misleading and incomplete financial data. The study concluded that the audit expectations gap in Jordan was mainly focused on the independence of the external auditors, which was negatively impacted by the potential loss of sales in lucrative non-audit services, unlimited audit tenures and company management pressure on the external auditors.

In summary, previous studies that investigated the audit expectation gap in Jordan revealed that the gap did exist in Jordan and that it was partially affected by external auditors' performance and independence (Al-Thuneibat, 2003; Asfoor, 2003 Omari, 2003; Al-Khadash & Al-Sartawi, 2010) and the level of disclosure & transparency (Hajir, 2001). The gap was also affected by differences between the perceptions of auditors and investors regarding the importance of different qualitative characteristics of accounting information (Obaidat, 2007). External auditors' independence, neutrality and objectivity and the level of disclosure & transparency could be improved by adopting effective corporate governance codes.

This thesis aims to investigate the role of the JCGC and audit committees in reinforcing external auditor independence. Such reinforcement is expected to lead external auditors to express

independent opinions about companies' financial positions, in the face of any management pressure, thus enhancing investor confidence in audit reports. The study also evaluates the extent of the JCGC requirements mandating Jordanian shareholding companies to provide adequate and timely disclosure and transparency and adopt sufficient accountability systems. The study also investigates the JCGC's impact on the perceived reliability of the audit report and the narrowing of the audit expectation gap. Moreover, this thesis investigates the role of the JCGC in reinforcing internal auditor's effectiveness, enabling them to detecting fraud and errors effectively, which should lead to greater reliability of audit reports and a reduction of the audit expectation gap.

Through a factorial experiment, this study investigates the degree of each variable's influence on institutional investors' perceptions of the reliability of the audit report and the level of the audit expectation gap. The factorial experimentation method identifies if there are any causal relationships rather than just correlations between variables.

## 3.4 Investor Reliance on the Audit Report

Financial reports published by corporations every year contain very useful information used in the decision-making processes of many parties, such as shareholders, investors, lenders and the government. The growing interest about financial reports and the extent to which they could be trusted by investors and stakeholders suggests that they should be prepared objectively and audited by an independent and neutral body, this body normally being the independent external auditors. It is a mandatory condition that public companies appoint an external auditor during the company general assembly. The Jordanian companies' law specified that listed companies, private shareholding companies, limited partnerships in shares, and limited liability companies must appoint licensed external auditors (Companies Law, 1997, Article 192). The auditing profession is vital, playing a crucial role in reinforcing the "quality and reliability of financial statements" (Zureigat, 2011b, p. 38). The audit report adds value to the credibility of financial reports. Auditing has been defined as "being concerned with the verification of accounting data, with determining the accuracy and reliability of accounting statements and reports" (Mautz, 1964, p.1). Audited financial statements enhance investors' judgment decision-making in the capital market (Sudsomboon & Ussahawanitchakit, 2009), and different sorts of audit reports have a significant impact on investor reactions toward the capital market (Herboh, Ragunathan & Garsden 2007).

Reliability of audit reports refers to a condition in which investors, and all those interested in a company's business affairs, consistently find the audit reports and opinions about a company's financial statements and position to be dependable and credible. Reliable audit reports clearly identify any significant weaknesses in internal controls, as such weaknesses limit the auditor's ability to confirm that the accounting systems produce accounts which reflect all the company's transactions in accordance with relevant accounting standards. Reliable audit reports also reveal whether accounting reports are reasonably free from error and bias and whether the accountants are justified in making a 'going concern' assumption. In other words, for the audit report to be reliable it must give institutional investors adequate information about the quality and comprehensiveness of the accounting reports so that they can decide to what extent they can rely on the accounting reports in making investment decisions.

The extent of the reliability and credibility of the audit report is considered the starting point for decisions by stakeholders and potential investors about the viability of an entity (Jones, 1996; Epstein & Geiger, 1994; Gul, 1990; O'Reilly, 2009; Firth, 1980; Citron et al, 2008; Saaydah, 1998; Guillamon, 2003; Zureigat, 2006; Zureigat, 2010; Al-Awaqleh, 2008). The audit report has been seen as the primary tool or medium of communication between auditors and the rest of the financial society (Hatherly & Skuse, 1991).

Auditors are assumed to be belong to one of two categories: rigid external auditors who do not compromise on their probity and reliability and always report their judgments truthfully, and

other auditors who compromise their professional reporting decisions based on a cost-benefit analysis (Beyer & Sridhar, 2006; Antle, 1982). Beyer & Sridhar (2006) argued that the latter type misreport their professional assessment due to the absence of the auditing regulatory bodies while others attribute that to management pressure and auditor competition on fees (Knapp, 1985; Malhas, 1992; Omari, 2003).

This has caused Taylor, DeZoort, Munn, & Thomas (2003) to argue that the reliability of the audit report is the backbone of the auditing profession and "our view is that auditor reliability rather than auditor independence is, or should be, the profession's cornerstone for protecting the public interest" (p. 258). They argued that the external auditor's independence was an input into stakeholders' reliance on the audit report, as were external auditor experience and integrity. Taylor et al (2003) defined the reliability of the audit report as the condition in which the users of the audit report perceive the audit report as dependable and credible.

Zureigat (2010) conducted an event study to investigate the impact of the modified auditor's opinion on the share prices of Jordanian listed companies during the period 2002 to 2007. Based on the analysis of the abnormal returns of 53 listed companies during the test period (i.e. 10 days before the assembly meeting<sup>23</sup> date and 10 days after), the study concluded that modified audit opinions significantly influenced the listed companies' share prices. The study attributed this to the Jordanian investor's capability to evaluate the contents of the audit report and assess the audited financial statements.

Zureigat's (2010) study confirmed the results (i.e. the issuance of qualified audit opinions influenced the shares prices of the Jordanian companies) of two prior studies utilizing that same methodology (Zureigat, 2006; Saaydah, 1998). Zureigat's study (2006) covered the period 1993 to 2005, and Saaydah's study (1998) investigated 19 listed companies that received qualified opinions during the period from 1991 to 1994.

<sup>&</sup>lt;sup>23</sup> The study adopted the general assembly meeting date rather than the release of the audited annual report date, because that is the date in which the financial statements disclosed for the first time to shareholders and usually this meeting includes the ratification of the auditor's report.

The effect of the international auditing standard ISA No. 700 and its amendments<sup>24</sup> on the audit expectation gap and the level of disclosure in the audit report has been investigated by Al-Khadash, et al, (2012). The study surveyed external auditors, preparers of the financial statements, investors and academicians. The results of the study showed that ISA No. 700 and its amendments contributed in bridging the audit expectation gap and enhancing the disclosure level of the audit report. Based on the participants' opinions, the study recommended an expansion to the audit report whereby another separate paragraph would be added at the end of the audit report – after the opinion paragraph - in which the auditors should state their opinion about the effectiveness of the internal control system, the extent they relied on it, compliance with the corporate governance codes, and earnings management practices.

O'Reilly (2009) pointed out that the general assumption is that expressing an explicit statement about the likelihood of the firm being a going concern should be useful to users of the audit report. The study experimentally investigated the issuance of (uncertain) going-concern opinions, including financial analysts' expectations about how share prices in the USA will react to such opinions. The results of the experiment provided strong evidence that the goingconcern opinion influenced perceptions of share values and that financial analysts perceived the auditor's opinion about the financial viability of the entities as useful information for valuing companies' shares. The results of the experiment also found that the importance of the auditor's opinion was higher when it provided information different from the market expectations.

Gómez-Guillamón (2003) investigated the extent that institutional investors rely on the audit report in investment and financing decisions in Spain. The study surveyed a sample of head officers of credit institutions who actively make investment and credit decisions and managers of investment and analysis companies. The study showed that the types of audit opinion (clean, qualified, disclaimer or adverse) provided by the external auditor affected the institutional investors in their investment and lending decisions.

<sup>&</sup>lt;sup>24</sup> The study referred to the amendments of The ISA 700 "Forming an Opinion and Reporting on Financial Statements" that became operational in the first of January 2006.

## 3.5 Types of Audit Reports and Opinions

The previous section indicated that assurance provided by the external auditors is relied upon by users of financial reports such as shareholders, lenders and investors to help evaluate and make investment decisions. Moreover, the literature suggested that the external auditor's opinion about the ability of companies to continue as going concerns is appreciated by the users of the audit report, and they reconsider their investment decisions accordingly (Jones, 1996; Gul, 1990; Epstein & Geiger, 1994; Firth, 1980; O'Reilly, 2009; Guillamon, 2003; Citron et al, 2008). This section summarizes the types of audit report as categorized in International Standards.

As stated in International Standard on Auditing (ISA) No. 700, auditors are required to provide an opinion on whether, in all material respects, financial statements are prepared in accordance with all applicable financial reporting frameworks. That opinion, according to ISA No. 700, is to be reached by obtaining reasonable assurance that the financial statements are free from all material misstatement, whether from error or fraud, and thus fairly reflect the true financial condition of the organization.

Auditing standard (ISA 700) requires that the external auditors must clearly express their opinion as one of the following types:

1- An unqualified<sup>25</sup> audit report: is the most common type of audit report and is issued when the financial statements are presented fairly, in all material respects, according with applicable standards and the relevant statutory such as the corporations act (Leung, Paul, & Cooper 2011).

2- A qualified audit report: "issued when there is a disagreement with management concerning appropriate accounting policies, a conflict between applicable financial reporting frameworks, or a limitation on the scope of the audit" (Arens et al, 2010, p. 48). This opinion is issued when material misstatements exist, but their effects are not pervasive to the overall financial report.

<sup>&</sup>lt;sup>25</sup> This type is also referred to as un unmodified opinion

3- A disclaimer of opinion is issued when the auditors are unable to express an opinion about the fairness of the financial reports due to a substantial restriction or a severe limitation on the scope of the audit such as the client concealing substantial evidence.

4- An adverse opinion: issued when the external auditors believe that the overall financial reports are not presented fairly nor prepared according to the applicable accounting standards.

The executive management and the board of directors of companies always prefer the first type of the audit report in order to send a message to the shareholders that they are diligent, competent, and deserve to keep their positions and be elected for another period. Therefore, executive managements put pressure on the external auditors to ensure the issuance of the unmodified audit report. In the context of Jordan, the literature pointed out that poor corporate governance mechanisms, including the lack of well-established and independent audit committees, encourages top management to undertake the role of selecting, retaining and setting the external auditor's fees (World Bank: Corporate Governance Assessment, 2004; Swaiti, 2006; Abdullatif, 2006; Al-Farah, 2001; Al-Khadash & Al-Sartawi, 2010) which consequently compromises the objectivity of the external auditors and leads them to issue audit reports in favor of company management.

# 3.6 The Role and the Potential Support by an Audit Committee

The International Standards on Auditing (ISAs) also require auditors to communicate significant findings, arising from or discovered during the financial statement audit, with the appropriate persons charged with the governance of the organization, these persons usually being the audit committee (Colbert, 2002). ISA No. 260 (Communications of audit matters with those charged with governance) and ISA No. 265 (Communicating Deficiencies in Internal Control to Those Charged with Governance and Management) provide guidance on communicating matters of interest to the governance body of an entity (IFAC 2010b). Jubb, Rittenberg, Johnstone, &

Gramling (2012) argued that, under these standards, the external auditors have a responsibility to discuss with the audit committee all significant audit matters, deficiencies in internal control and any pressure, restriction and disagreements with company management in relation to the accounting. Jubb et al (2012) also added that the audit committee must support auditors and ensure that they are free of management pressure during the course of the audit.

Audit committees play a role in bridging the audit expectation gap. Sharaby (2010) investigated the role of audit committees in narrowing the audit expectations gap in commercial banks in Yemen. The study found that audit committees play a significant role in reinforcing external auditors' independence and performance, and the quality of financial reporting. The study also pointed out the importance of effective communications with external auditors in enhancing the quality of the audit process which would, consequently, increase the auditor's capability in reporting about the entity's ability to continue as a going concern. Another study from the region, this time conducted on Egyptian pharmaceutical companies during the period from 2005 to 2008, investigated the role of the audit committee as a key pillar of corporate governance for improving the quality of financial reports (Ahmad, 2011). The study's regression analysis of pooled data suggested that audit committee independence, the size of the audit committee in terms of number of members, and the financial experience of audit committee members are positively related to the issuance of clean opinions in the external auditor's report.

Effective audit committee is considered a cornerstone and crucial pillar of effective corporate governance (Gramling, Maleta, Schmeider, & Church 2004). Independent audit committees are an essential part of the corporate governance mosaic (NYSE & NASD, 1999; The UK Corporate Governance Codes, 2010).

The audit committee has been considered a safeguard of the external auditor's independence (albeit among some other factors and bodies) due to its role in overseeing the financial reporting system and monitoring compliance with the regulatory instructions (ISB, 2000, para. 14). Furthermore, the audit committees have been seen as a safeguard for reinforcing the internal auditor's effectiveness as well; The Malaysian Institute of Accountants (MIA) recommended that all publicly-traded company establish audit committees in order to prevent frauds and increase the integrity of the financial reports. The MIA added "audit committees provide additional safeguards against fraud and malpractice by monitoring the effectiveness of record keeping, internal controls and internal audit. They also provide an assurance to regulatory authorities, investors and depositors of the objectivity, credibility and integrity of a company" (Malaysia Institute of Accountants, 1990, p. 27, cited by Teoh & Lim, 1996). Libby, Libby & Shor (2001) went further, affirming the pivotal role of the audit committee among other safeguards. They pointed out that internal and external auditors provide a safeguard for maintaining the soundness, accuracy and reliability of financial reporting, and added that companies establish an audit committee "whose job is to review these other two safeguards" (p. 24).

The figure below shows the relationships between the major constituents of the auditcorporate governance system, emphasizing the crucial role of the audit committees in the corporate financial reporting system. The audit committee is an important body in the corporate structure because it is involved with overseeing and communicating with all other controlling and auditing bodies.

# Figure 5: Relationships between the Major Constituents of Corporate Governance as they Relate to Auditing



**Internal Auditors** 

#### **External Auditors**

Whittington and Pany (2012) pointed out that one of the main duties of the audit committee is to oversee and ensure accountability and responsibility over the company's financial reporting process to the external users, monitoring the company's risk and related control processes, in addition to coordinating the work of the internal and the external auditors. Rittenberg and Schweiger (2007) pointed out that proper authority and powers of audit committees enable them to oversee the flow of the financial reporting and prevent fraud by company management.

The board of directors must ensure that audit committees have the proper authority and power to do their job properly. Audit committees with inadequate authority cannot ensure corporate accountability and cannot enhance disclosure and transparency, thus negatively impacting investor confidence and creating incorrect impressions amongst them. Audit committee effectiveness depends on the efficacy of its members. Megat (2000) pointed out that audit committees cannot help in reinforcing corporate governance unless they are composed of truly independent members, and they should have experience and analytical skills and the capability to confront management pressure. Megat (2000) also added that if audit committees do not play their vital role and are nothing more than just "window dressing", then the audit expectations gap from investors' perception will be widened.

Goodwin and Seow (2000) argued that the investors, auditors and managers all believe that a powerful and effective audit committee support external auditors to perform their job effectively and independently.

Corporate failure in Malaysia has raised questions regarding the professionalism of external auditors and affected public confidence negatively as the public blamed auditors for those scandals. Teoh and Lim (1996) conducted an empirical study of the factors affecting the perception of the external auditor's independence in Malaysia. The study employed the repeated measures experimental design. An experimental instrument consisting of 32 scenarios was distributed to a sample of 100 accountants, 50 of which worked in auditing firms and the other 50 worked in non-auditing firms. The study concluded that the establishment of audit committees has a strong positive influence in reinforcing the perceived independence of external auditors. Moreover, the study found that providing non-auditing services and the large size of audit fees gained from a single client impair the perception of auditor independence. However, the negative effect of the latter could be mitigated by the formation of audit committees. The study revealed a statistically significant interaction between the existence of an audit committee and large audit fees obtained from a single client.

Abdel-Qader (2002) highlighted the importance of establishing audit committees to bridge the audit expectations gap. The study pointed out that the AICPA, the Treadway Commission, and the SEC have affirmed that the main objective of audit committees is to foster auditors' independence. Abdel-Qader (2002) argued that an external auditor's independence is

enhanced when the auditor has direct communications with and a path to an independent party. Therefore, the existence of an independent audit committee is seen as enhancing and reinforcing the quality of the auditor's work and thus mitigating control of management over the financial reporting process. The committee should review the scope of the audit, significant accounting estimates and provisions<sup>26</sup>, and policies on external audit services, thus helping the auditors reach an independent opinion in the audit report.

Prior to the JCGC, audit committees in Jordan were ineffective and had insufficient power. That lack of power could be attributed to the fact that there were no requirements regarding the qualifications that the audit committee members should hold, and no requirement for them to be independent (Swaiti, 2006; Abdullatif, 2006; Malkawi, 2008).

The JCGC has clearly defined the requirements for the independent and the non-executive (i.e. non-management) members of boards of directors, and defined the conditions that impair the independence of the board of directors. The JCGC (Chapter Five, Section One, Item 1) required that each and every member of the audit committee have knowledge and experience in the field of accounting and finance. Moreover, the JCGC requires that at least one member of the audit committee must have actually worked in accounting, finance or a related field and must also hold an academic degree or professional certificate in one of those fields.

Additionally, the JCGC (Chapter Two, Section two, Item 2) required audit committees be comprised of at least three non-executive (i.e. non-management) members selected from the board of directors, of which at least two must be independent (i.e. not be an employee of the company). One of these independent members is required to chair the audit committee.

Requiring only two independent members means that one of the audit committee members might have a beneficial interest other than ownership of shares with the company or its external auditor or the company upper executive management. Therefore, the presence of

<sup>&</sup>lt;sup>26</sup> Provisions refer to the "liabilities for which the amount of the future sacrifice is uncertain" (Carlon, Mladenovic-McAlpine, Loftus, Palm, Kimmel, Kieso, & Weygandt 2008).

even one member who is not independent means that all discussions could leak back to the management and the non-independent member may exercise a heavy influence particularly if they are a person of high prestige. In the case of Jordan, some literature showed that upper executive management had undertaken the roles of the board and the shareholders, especially in decisions related to the selection and remuneration of external auditors (World Bank: Corporate Governance Assessment, 2004; Swaiti, 2006; Abdullatif, 2006; Al-Farah, 2001; Al-Khadash & Al-Sartawi, 2010).

Green (2004) argued that "a more independent audit committee means a more independent auditor" (p. 12). "Qualified, independent and tough minded audit committees represent the most reliable guardians of the public interest" (Levitt, 1998)<sup>27</sup>. Abbott, Park, & Parker (2000) found that companies with audit committees with independent members, and which meet at least twice a year, are associated with a less likelihood of both fraud and malpractice.

In the U.S.A., the National Association of Securities Dealers Automated Quotations (NASDAQ), the New York Stock Exchange (NYSE), and the Securities and Exchange Commission (SEC) have taken many actions to enhance the quality of companies' disclosures and transparency, and have forced the corporations to alter the composition of their audit committees.

Felo and Soliere (2009) pointed out that the SOX legislation requires all audit committee members be independent of the company management. They argued that independent audit committee members are stronger, unbiased and provide better oversight over the company management because they are less likely to be influenced by management pressure. They also added that insider audit committee members are more likely to be influenced by management pressure and that CEOs could override their powers.

The SOX initiative in the U.S.A. has influenced other countries to establish reforms to strengthen the independence requirements of their capital markets. Tafara (2006) pointed out

<sup>&</sup>lt;sup>27</sup> The former US Securities and Exchange Commission chairman

that, since the issuance of the SOX in 2002, the UK, Australia, Canada, Hong Kong and Mexico have introduced reforms.

The difficulty and the vagueness associated with reporting about the going-concern principle may put the auditor under pressure of the executive management. Independent and effective audit committee members could help mitigate such pressure by supporting the auditor in any conflicts with management (Parker, 2000). It is expected that a non-executive audit committee independent of the company management is more likely to support the external auditors when they face any management pressure to issue an unqualified opinion when the assumption of the entity's ability to continue as a going-concern is questionable (McMullen, 1996).

Carcello and Neal (2000) have examined the relation between the characteristics of the audit committee's composition and the likelihood of issuing going-concern opinions for USA corporations experiencing financial distress during 1994. The study developed a logistic regression model and found that the higher the percentage of affiliated<sup>28</sup> directors amongst the audit committee members, the lower the likelihood that the companies receive a going-concern opinion by the auditor. The study supported the call of some capital market regulators for more independent audit committee members.

Carcello and Neal (2003) investigated the relationship between audit committee characteristics and auditor dismissals following the issuance of a going concern opinion between 1988 to 1999. The study utilized the previous study's methodology to, this time, investigate the relation between the likelihood that the company dismisses its external auditor and the qualities of the audit committee of that company. The study concluded that audit committees with a higher

<sup>&</sup>lt;sup>28</sup> Affiliated directors are directors who lack independence; which includes current or past officers, employees and the staff of the company or of a related entity, relatives of executives, professional advisors to the company such as bank officers and legal consultants, officers of significant customers and/or suppliers of the entity, and interlocking directors.

percentage of independent members, governance and financial expertise, work effectively, and support and shield the auditors from dismissal especially after expressing a going-concern opinion.

Carcello (2000) and Neal (2003) found a relationship between the variables, using the logistic regression model rather than causation relationship. This thesis includes an experiment aimed at identifying the causal relationships, rather than just correlations, between the independent variables and both the reliability of the audit report and the audit expectation gap.

# 3.7 External Auditors' Independence

Auditing or assurance services can be defined as an analysis of an auditee's financial statements conducted to improve the quality and reliability of the presented information for the users of the financial statements, enabling them to make informed investment decisions. This includes checking accounts to prove their truth and fairness by an independent third party and the issue of a credible and independent opinion in a special report (Gray & Manson, 2006; Arens et al, 2010; Strawser & Strawser, 2001; Whittington & Pany, 2012). The ISB considered external auditors' independence to provide significant benefits to investors and other users of financial information. The ISB stated "auditor independence helps ensure quality audits and the reliability of the financial reporting process, which also may lead to increased confidence in that reliability" (ISB, 2000, para. 28-a). Moreover, in paragraph 28-c, the ISB added that external auditors' independence is also beneficial to the company management, board of directors and to the audit committee, ensuring the credibility of the financial reporting proteendence is also beneficial to the financial reporting proteons and to the audit committee, ensuring the credibility of the financial reporting prepared by lower-level management.

The concept of Independence is one of the most important foundations on which the auditing profession is built. Knapp (1985) defines auditor independence as the auditor's ability to resist pressure from the client (i.e. the company being audited).

Potential ethical dilemmas exist in any situation where auditors face outside pressure to compromise their professional integrity or objectivity (e.g. management pressure to decide in favor of the appropriateness of the accounting policies chosen by management) (Leung, Paul, & Cooper, 2011). Furthermore, potential problems arising from a close relationship between the auditors and company management may compromise the external auditor's independence (Cadbury Report, 1992).

Auditors' independence and the quality of the audit can be enhanced by support from effective audit committees, the maintaining of audit working papers and auditor rotation (Arens et al, 2010).

The auditing profession and the business society are interested in ensuring that the users of the audit report perceive auditors as being independent (independence in appearance) and that the auditors actually maintain an unbiased attitude throughout the audit (independence in fact) (Arens et al, 2010). According to Leung, Paul, & Cooper (2011) "independence in mind relates to the state of mind that permits the expression of a conclusion without being affected by influences that compromise professional judgment. It requires the professional accountant to exercise skepticism and act with integrity and objectivity. Independence in appearance means avoiding situations and facts that are so significant that a reasonable person, knowing all relevant facts and having considered the safeguards in place, would reasonably conclude that a firm's or a professional accountant's integrity and objectivity may have been impaired" (p. 85).

#### 3.7.1 External Auditors' Selection, Retention and Remuneration

Audit committees are expected to choose the external auditors, presenting their choice of audit to the board of directors who usually have the final say (although, legally, the shareholders have the final say). The board of directors normally accepts the auditor proposed by the audit committee. It is also expected that the external auditor should report any problems to the audit committee. As such, the audit committee is, in effect, the guarantor of the auditor's independence, an independence that is enhanced if the audit committee, not management, is truly responsible for selecting the audit. By selecting truly independent auditors and enforcing the integrity of the auditing process in the face of possible opportunistic behavior by management, audit committees serve the best interests of company shareholders (Alleyne, Howard & Greenidge, 2006).

In response to Enron's scandal in 2002, Abdel-khalik (2002) argued that the auditor's independence is critically affected by the body which has the responsibility of nominating, dismissing, and setting the external auditors' fees, and suggested establishing another board, called the Shareholders' Board of Trustees (SBT), whose sole function is to select, retain and compensate the external auditors.

Nagy (2005) pointed out that by making a voluntary auditor change, the company can:

1- Seek an accounting firm whose views on accounting and reporting matters are more consonant with those of the company.

2- Seek an accounting firm that the company views as more pliable.

3- Signal that the company is willing to change auditors in the future.

In the case of Jordan, personal relationships play a pivotal role in the external auditor selection process. Malkawi (2008) pointed out that, in practice, the board of directors and the company management<sup>29</sup> select the external auditors and, moreover, set the auditors' remuneration, negatively impacting the independence of the external auditors. Zureigat (2011b) found that ownership concentration in Jordan negatively impacted auditing quality in Jordan. He attributed this ownership concentration to the nature of family owned companies in Jordan. Zureigat (2011b) argued that family owned companies tend to hire external auditors based on

<sup>&</sup>lt;sup>29</sup> Prior to the establishment of the Jordanian Corporate Governance Codes there were no requirement to separate the positions of the chairperson of the board of directors and any executive position.

personal relationships; therefore, the study recommended moving the authority of nominating the external auditors to the audit committees.

Means of reinforcing external auditor independence have been investigated in Jordan by Matar (1994). The researcher designed and distributed questionnaires to supervisory bodies of the auditing profession, the users of the financial statements, and external auditors. The study found that the role of the directors in appointing, dismissing and setting audit fees was the most influential factor affecting external auditors' independence. However, the study found significant negative impact of other factors on the external auditor's independence, such as audit tenure, non-audit services, and audit fees.

A lack of written audit committee charters in Jordan has possibly affected the auditor selection process. The lack of such systematic guidelines affects the work of audit committees and may contribute to them compromising their own objectivity and failing to operate in the company stakeholders' best interests, especially in decisions to nominate an external auditor (Al-Saudi, 2007; Khsharmh, 2003; Ameera, 2004). For instance, Al-Saudi (2007), whose study investigated audit committee members, pointed out that personal relationship between the audit committee and the auditors affected the selection process of the external auditors, in the perception of the audit committee members. He attributed this to the absence of systematic procedures regulating external auditors' nomination by those charged with governance, especially the audit committee.

External auditor's selection and retention dilemmas have been experienced in other countries. In the USA, for example, Abdel-khalik (2002) pointed out that, in practice, auditor-related decisions are taken by the company management, and even non-executive members of the board of directors are often chosen by the executives. In order to resolve this dilemma, Abdelkhalik proposed establishing another board, independent of the company board of directors and the company management, whose sole function will be to select, retain and compensate the external auditors. He called this board the Shareholders' Board of Trustees (SBT). Abdel-

khalik also suggested that the stock exchange administer the process of selecting the SBT, in exchange for a fee.

From the researcher's point of view, the potential weaknesses of the model proposed by Abdel-khalik (2002) are as follows:

1- The model stated "the BoD will not have any role in the choice of SBT other than acting as shareholders voting their shares" (Abdel-khalik, 2002, page 101). The researcher argues that the shareholders who elect the board of directors (BoD) are themselves the same people who elect the Shareholders' Board of Trustees (SBT). Through this lacuna, the probability of the Board of Directors having control over the Board of Trustees is high because they would elect representatives who somehow "belong" to them and work as they desire, thus compromising the independence of the SBT when selecting external auditors. This proposal might not work efficiently in Jordan due to the presence of a large portion of family businesses, thus impacting the election process. Abu-Tapanjeh (2006) pointed out that many Jordanian shareholders elect members of their families and relatives, during general assembly meetings, to become members of the board of directors and then to occupy the key executive positions.

The researcher would argue that this lacuna could be narrowed by excluding major shareholders, directors, executives, and associates (relatives and companies, trusts, etc controlled by them) from voting in the election of the Shareholders' Board of Trustees.

2- The cost-benefit of paying fees to the stock exchange to administer the election process of the SBT, in addition to the costs of establishing and maintaining a new board whose sole function is to select or retain an external auditor, should be taken in to consideration.

In Jordan, factors affecting decisions to select and change the external auditors have been investigated in several studies (Al-Qam, 1997; Khsharmh, 2003; Al-Saudi, 2007). These three studies were quantitative, and utilized questionnaires distributed to internal auditors, company

management, external auditors and financial managers, and audit committees. Khsharmh's study (2003) suggested that, in order to mitigate the influence of the board of directors in the external auditor's selection, the selection should be made by the ministry of trade and industry through the companies' controller. Al-Qam (1997) recommended establishing non-executive qualified audit committee to approve the appointment of an external auditor. Al-Saudi (2007) concluded that audit committees in Jordan had, at that time, no standards or systematic guidelines for the process of selecting auditors and that personal relationships between audit committee members and auditors affects the selection process. The study also revealed that both of audit fees and audit committees' powers<sup>30</sup> affect the external auditor's nomination in Jordan.

#### 3.7.2 External auditor's Independence and the Audit Expectation Gap

Jarbou (2004) pointed out that the lack of external auditor independence is the primary factor for the audit expectation gap. He argued that the financial community and the stakeholders should be convinced that external auditors are independent because the survival of the audit profession depends on the independence of auditors. He argued that if the independence of auditors in general came to be doubted, then their reports would have limited value and, consequently, the auditing profession would become worthless. The study concluded that the perceived independence of external auditors is a crucial factor in narrowing the expectations gap, and emphasized the importance of the role of the audit committees in nominating, dismissing, and maintaining the independence, of the external auditor.

As an aside, it is worth noting that early warnings in an audit report can reduce the likelihood of unsustainable firms growing and causing losses in the wider community. Thus, while this study

<sup>&</sup>lt;sup>30</sup> The study referred the powers of the audit committees in (1) the committee's ability in accessing and communicating with external auditors freely, and (2) the committee's ability to hold private meeting with the external auditors without the presence of the company management.

focuses on the relevance to the audit profession, it also has considerable significance for the achievement of satisfactory allocations of investments in the Jordanian economy.

Similarly, Jarbou (2007) concluded that the lack, and the suspicion of the lack, of the external auditor's independence was the principal factor widening the expectations gap. In that study, a questionnaire was developed and distributed to external auditors, bank managers, and income tax department employees in the Gaza province in Palestine. The study recommended encouraging the independence and the neutrality of the external auditor in practicing the auditing profession, and encouraging the role of the professional associations and their control over the accounting and auditing profession.

The audit expectation gap in Britain has been empirically investigated by Humphrey, Moizer & Turley (1993). The study surveyed auditors, financial analysts, directors of corporate finance, credit officers at banks, in addition to financial journalists. The study identified auditor independence as a key element of the audit expectations gap. The study confirmed that the critical components of the expectations gap included expectations regarding auditors' role in fraud detection, auditors' responsibility to third parties, auditors' valuation of companies' financial positions, and threats to auditors' independence. Humphrey et al (1993) also concluded that auditor independence was, perhaps, the core around which the debate about audit expectation gap is waged.

Much of the literature related to the audit expectations gap referred to the perceived lack of external auditors' independence as being the main factor in widening the gap. Those studies include Jordanian (Abdel-Qader, 2002; Hajir, 2001; Al-Thuneibat, 2003; Omari, 2003; Al-Khadash & Al-Sartawi, 2010), regional<sup>31</sup> (Jarbou, 2004; Jarbou, 2007; Sidani & Olayan, 2007; Al-Qarni, 2004; Elmeligy, 2006; Al-Husaini, 2000) and international studies (Humphrey et al, 1993; Koh & Woo, 1998; Dewing & Russell, 2002; Salehi & Rostami, 2009; Gloeck & Jager, 1993; Gray

<sup>&</sup>lt;sup>31</sup> Regional studies referred to those studies that have been conducted in the Middle East and North Africa's area and normally it is referred to these countries as MENA.

& Manson, 2008; Lin & Chen, 2004). This lineup of studies reflects the importance of external auditors' independence and performance in bridging the expectation gap.

The issue of independence is vital because auditing, as a function, was established to provide independent and expert opinions on financial statements to assist users in making decisions regarding financial statements. These vital functions of the external auditors can be seen in the JCGC. The JCGC (Chapter Five, Section 4, Item 6- D) stated the duties of the external auditors. One of these duties is "examining the financial, administrative, and internal auditing systems of the company and submitting their opinion on their effectiveness and ensuring their suitability for the company's business and safeguarding of its assets". Therefore, the Jordanian regulatory bodies and professional association should ensure that external auditors meet the standards of independence.

There have been many debates and criticisms surrounding the external auditor's independence in Jordan (Swaity, 2006; Malkawi, 2008; Al-Khadash & Al-Sartawi, 2010; Abdullatif, 2006; Omari 2003; Dahmash, 1989; Matar, 1995; Momany, 1994). Dahmash (1989) pointed out that 54% of auditors in Jordan feel themselves under pressure from management. Matar (1995) also pointed out that the external auditor's independence in Jordan is impaired because of management pressure on auditors. Moreover, Omari (2003) revealed that investors in Jordan believed that auditors are responding to management's desires.

In 2004, World Bank and International Monetary Fund (IMF) joint initiatives reviewed the accounting and auditing standards and practices in Jordan. They reported several drawbacks that affected the audit profession in Jordan. In regard to the external auditor's independence, the report revealed that some auditors are dependent on one key client, have close relationships with the company management and provide non-audit services such as internal audit services to audit clients (World Bank: Accounting and Auditing, 2004).

The Jordanian corporate governance codes, in Chapter Five – Section 2 and 4, focused on the importance of the external auditor's independence, and the role of the audit committees in
reviewing issues related to the selection of the external auditor, this aims to ensure that companies satisfy all the conditions as stipulated in the legislation in force. The codes also stated that the "company shall take appropriate actions to ensure that the external auditor is independent in accordance with International Standards on Auditing". The main external auditor's duty is to fulfill their job with impartiality and independence without interference from the board of directors or the executive management. (Corporate Governance Codes in Jordan, 2009 Chapter Five: Section Two Item 1; Section Four items 3- C, D and 6- A).

#### 3.7.3 Non-Audit Services

The provision of audit and non-audit services by one audit firm to the same client, and the resulting potential impairment of the independence of the auditor when issuing the audit report, has repeatedly been a controversial issue. Non-audit services are all services provided by the incumbent external auditor that are not considered as an audit. Such services may include bookkeeping, tax services, financial information systems design and implementation, management consultation and any service other than the audit service. In the United States, the Sarbanes Oxley Act (SOX) (2002) banned audit firms from providing most non-audit services. In Jordan, there were also many criticisms directed at the non-audit services provided by auditors to their clients prior to the JCGC's establishment.

Abdel-Qader (2002) pointed out that Jordanian commercial laws did not require public companies to disclose non-audit services. The study argued that, in order to maintain integrity, objectivity, and independence, the public accounting firm should not act as internal and external auditor for the same entity. Malkawi (2008) pointed out that the Law of the Audit Profession of 2003 in Jordan does not specify whether auditors can perform non-audit services. Omari (2003) concluded that the expectation gap is concentrated on the independence of the external auditors and that Investors believed external auditors' independence to be negatively

affected when auditors provide non-audit services to their audit clients as this would create a conflict of interest and affect their performance.

Chan, Wong, & Wong (2002) pointed out that, in the last two to three decades, audit firms have expanded their services beyond auditing companies' financial statements, and the non-audit services business had grown to the point where fees generated from non-audit services were substantial. Chan et al (2002) also found that the largest New Zealand audit companies indicated that audit firms generated as much from non-audit services as they do from their audit work. The analysis suggested that audit firms have increased their provision of non-audit services to companies over the 1996 to 2000 period. Given this rebalancing of audit and non-audit services, auditors should consider how auditor independence and, consequently, investor confidence will be impacted and take action to address related problems.

These changes in the structure of services provided by audit companies must also be addressed by regulators, perhaps placing appropriate restrictions on the type or value of services that an audit company, whether directly or through partnerships or affiliations with other service providers, can provide.

There are other ways in which these potential conflicts of interest can be addressed. The auditing profession continues to establish broad guidelines for audit independence, as it has done for quite some time, but these guidelines, perhaps, give auditors too large a margin for maneuver. Alternatively, auditors could be required to disclose the types and value of services provided the client, perhaps within the audited financial reports of the client. Such an approach would be useful in reducing some, but not all, perceived conflicts of interest. A more extreme, but perhaps more effective, measure would be to totally forbid certain services on the premise that they compromise auditor independence. In applying this approach, regulators could balance the potential impact on the auditor's independence against the potential benefit of each non-audit service to the quality of the audit.

At the same time, audit committees could help investigate the types of services that auditors perform and question whether it would be in the investors' interests to have some of those services performed by another party. An effective audit committee can moderate management attempts to pressure auditors into accepting inappropriate accounting treatments.

Non-audit services are prohibited in Jordan. The JCGC states that each company must ensure that its external audit service provider not provide them with any additional services, including administrative support or technical consultation (Corporate Governance Codes in Jordan, 2009, Chapter Five, Section Four, Item 3-b).

#### 3.7.4 Audit Rotation

There are two types of audit rotation: rotation of the audit partner and rotation of the audit firm. The rotation stipulated in the JCGC is the audit partner rotation.

Mandatory rotation of audit partners plays a role in enhancing auditor independence and bridging the expectation gap. Nagy (2005) pointed out that enforcing mandatory rotation of auditors would break the auditor's expectation that the audit client is a perpetual source of income as well as bring a fresh and perhaps more skeptical point of view to the audit.

In a case study by Dewing and Russell (2002) describing how UK investment fund managers define the audit expectation gap and their perceptions regarding the degree to which it be closed, audit rotation was identified as a significant contributor towards narrowing the expectation gap. The same study also suggested that frequent rotation of audit management partners would help to improve auditor's independence.

Governing bodies and major professional accounting associations viewed management fraud as a major influence on the reliability of the audit report and the audit expectation gap. "A gap exists between the expectations of the users of financial statements and the performance of the auditing profession" (Pankaj, 2003, p. 22). Pankaj investigated the relationship between corporate governance and management fraud in the USA. The study investigated instances of management fraud and reviewed the Accounting & Auditing Enforcement Releases issued by the U.S Security Commission. The results of the study suggested the importance of the role of corporate governance mechanisms in limiting management fraud, particularly in the way these mechanisms handle the selection of external auditors<sup>32</sup>. The study also added that these mechanisms lead to a decreasing of the level of the audit expectation gap as they enhance auditors' ability to detect and prevent management fraud.

Both the revised 8th Directive of the European Countries (Council of the European Communities, 2006) and the United States' SOX require key audit partner rotation every five years. In the context of Jordan, Omari (2003) referred to the impact of audit firm rotation on the auditors' independence. The study indicated that the tenure of audit firms was almost unlimited, negatively impacting external auditors' independence. Al-Khadash and Al-Sartawi (2010) also recommended audit partner rotation to reinforce external auditors' independence in Jordan. Matar (1994) also found that long audit tenure was a factor, among several other factors, that had negatively impacted external auditors' independence in Jordan. Swaiti's (2006) proposed model for the role of audit committees in Jordan suggested mandating audit partner rotation every five years.

The newly adopted JCGC mandated audit partner rotation. It stated "the external auditor shall exercise his duties for one year renewable, provided that the renewal for the partner at the external auditor may not be for more than four consecutive years, and the re-election may not take place before a minimum of two years" (Corporate Governance Codes in Jordan, 2009, Chapter Five, Section four, Item 2).

<sup>&</sup>lt;sup>32</sup> In regards to the selection process, the study took into account the "identity" of the audit firm. The Identity variable indicated if the audit firm was (1) a member of the Big-Six audit firms or (2) not a member of the Big-Six.

# 3.8 The JCGC and the Internal Audit

Effective internal audit contributes to good corporate governance systems (Leung, Cooper & Perera, 2011). The principal duties for internal control are to prevent and detect frauds and errors. Hence, organizations should establish an adequate internal control system to protect their assets. As insider staff, and due to their subordination to company management, the internal auditors are not perceived to be as independent as the external auditors. Mandating that decisions on appointing, dismissing and compensating internal auditors is the responsibility of an independent body, such as the audit committee, may enhance the independence and the effectiveness of internal auditors.

The JCGC pointed out the importance of establishing an adequate internal control system. The JCGC (in Chapter Five, section 3, item 3) stated that nomination of a candidate to be appointed as the company's internal auditor is one of powers of the audit committee. The JCGC also pointed out the roles of the audit committee in reinforcing the effectiveness of the internal audit. In Chapter Five, Section 2, Items 6, 7, 8, 9, and 10, it stated these duties of audit committees as:

- Assessing the adopted procedures of the internal control and internal audit functions.

- Examining the external auditor's evaluation of the adopted procedures of the internal control and internal audit functions.
- Examining the internal auditor's reports, especially those related to any irregularities
- The audit committees must provide the board of directors with recommendations regarding the adopted procedures of internal control and the work of the internal auditor.

Zureigat (2011a) investigated the impact of the corporate governance codes of the banking sector, issued by Central Bank, in improving internal audit quality and effectiveness in Jordan. The researcher designed a questionnaire and surveyed the internal auditors of Jordanian banks.

The study concluded that corporate governance of the Jordanian banking sector significantly improved the quality of the internal audit. The study revealed that giving internal auditors the authority to report directly to the audit committee was the main factor enhancing the internal auditor's performance.

The current research contributes further to Zureigat's study (2011a), which focused on investigating internal audit quality and effectiveness. This thesis covers external auditors' independence, disclosure and transparency, corporate accountability, and internal auditors' effectiveness as key elements of the corporate governance codes of publicly traded companies on the Amman stock exchange.

In a study of internal auditing in Australia and New Zealand, Goodwin (2003) examined the impact of audit committee independence and financial experience on the committee's relationship with Internal Audit. While both Audit committee independence and experience were found to have an effect on the internal audit function, independence was more strongly linked to having fewer audit process-related issues. On the other hand, committee experience was more strongly linked to the degree to which Internal Audit work was reviewed by the audit committee.

Goodwin & Seow (2000) surveyed the chief internal auditors in Singapore to investigate the factors affecting internal auditor's independence. They argued that internal auditors' independence and objectivity might be compromised because of company management pressure, which could be mitigated by giving the internal auditors private and direct access to the audit committee, and by giving the audit committee the power of appointing and dismissing the internal auditors. The study concluded that the relationship between the internal auditor's independence. This is also consistent with Braiotta (2004) and Leung, Cooper, & Perera (2011) who argued that holding several meetings with the internal auditors, without the presence of the company management, enhances their independence. Leung, Cooper & Perera (2011) also added "In order to provide relevant, timely and complete information for the board and senior

management, chief audit executives should have a clear reporting role directly to the audit committee of the board . . . . A close relationship between the chief audit executive and CEO, rather than with the audit committee, can potentially result in the risk of intimidation or management pressure on the chief audit executive" (p. 813).

Qin (2007) investigated the impact of audit committee financial expertise on earnings quality in the U.S. The author developed a regression model to test his hypothesis, concluding that members of audit committees with financial expertise strengthen the internal control system of companies, and enhance the reliability of financial reporting. The results of the study, contradicting the SEC rules, suggested that only accounting-literate professional members are associated with good quality of reported earnings.

In the period after the passing of the Sarbanes-Oxley Act (SOX) in the U.S., Krishnan & Visvanathan (2007) investigated how reporting of weakness in internal control is influenced by the audit committee and external auditors. The study found that companies that reported weaknesses were characterized by a higher frequency of audit committee meetings, lower financial experience within the audit committee and more frequent change of auditor, in comparison with companies that did not report weaknesses.

Another study, in which 114 chief internal auditors participated, also found a significant relationship between the membership structure of the audit committee and the Internal Auditing Function (Raghunandan, Rama & Read, 2001). Audit committees that consisted only of independent members, of which at least one member had experience in accounting, finance or a related field, were found to be more likely allow the chief internal auditor to address the committee in private (i.e. outside the presence of company management) and tended to have longer meetings with the chief internal auditor. Such committees were also more likely to review the work of internal auditors.

Christopher et al (2009) examined the internal auditor's independence in Australia. In the study, a questionnaire was designed and sent to Australian chief audit executives to examine

the impact of the relationships between the internal auditors, company management and the audit committee on the internal auditor's independence.

The study concluded there were three threats affecting internal auditor's independence:

1- The internal auditor's dependency on the management to move to other positions or seek promotion.

2- The setting of the internal audit budget by the company's chief executive officer or chief finance officer.

3- The heavy involvement of the upper management in developing the internal audit plan.

The study attributed these threats to (1) Non-compliance with the corporate governance codes of best practices that recommended that internal auditors report about technical matters to the audit committee and administratively to the company upper management; (2) Inconsistent adherence to audit committee compositions to guarantee the independence of the internal audit function, even though "the majority of audit committees have the right composition to guarantee the independence of the internal audit function" (p. 214); (3) The management's power in selecting and dismissing the head of the internal audit. The study revealed that in onequarter of entities, company executive management, rather than the audit committee, made the decisions to select and dismiss internal auditors.

By reviewing the plans and work of internal auditors, audit committees can ensure that internal auditors are actually able to investigate all potential risks that should be disclosed in financial reports (Mohamed & Hussain, 2005). Such a review reveals potential problems that may prevent internal auditors from adequately accomplishing their roles, and thus allows the committee to correct problems impairing internal auditors' independence.

Reporting about internal audit effectiveness is one of several aspects that may help in bridging the audit expectation gap. Abdel-Qader (2002) pointed out the importance of external auditors'

reporting on management responsibilities and internal control, the auditor being the only independent authority who can evaluate and report on the internal control system to users of the audit report. The study concluded that the auditor should review and evaluate the internal control system in sufficient depth to form an opinion as to its effectiveness, and any weaknesses which present potential for significant injury to the client should be reported to the public.

Audit committee support also plays a significant role in increasing internal auditors' confidence in their ability to detect frauds and errors. In regard to audit effectiveness, Davies (2009) argued that audit committees must actively support the work of the internal audit function. This active role requires that audit committees evaluate the work of the internal audit function and remove any obstacles standing in the way of the accomplishing of the function's duties. The audit committee should also select the heads of the internal audit function, helping ensure that the appointees are willing and able to protect the independence of the internal audit function in the face of potential pressure from management. Internal auditors should be able to respect the audit committee's ability to evaluate and support the work of the internal audit function.

According to Vinten & Lee (1993), audit committees have significant roles to play in enhancing the work of internal auditors. As previously mentioned, Audit committees can ensure that internal auditors investigate all the company's high-risk areas of activity. This role can be enhanced through coordinating work with both internal and external auditors. Furthermore, by facilitating coordination between internal and external audit efforts, the committee can optimize the costs of both audits.

In 1994, a study conducted in Australia by the Australian Society of Certified Practicing Accountants and The Institute of Chartered Accountants recommended requiring company managements to disclose the effectiveness of their company's internal controls (Koh & Woo, 1998). These disclosures would be made in the financial reports and external auditors would provide their opinions regarding these disclosures.

Muneer (2010) focused on the importance of communication as an approach to bridge the gap. The study pointed to the role of International Standard of Auditing ISA No. 700 that stated that the auditor's report on financial statements improves awareness and understanding of the financial reporting users about the nature, duties and limitations of the auditors' function, and the level of assurance they provide. The results of the study indicate that (1) conveying vital information, observed during the audit process, to the Audit Committee and (2) evaluating and conveying the fundamental deficiencies in internal control to management and the audit committee, increases the external auditor's performance and the effectiveness of the internal audit in detecting frauds and errors, and increases the credibility of financial statements, consequently contributing to narrowing the audit expectation gap.

In summary, the literature revealed the crucial role of the audit committees in maintaining the independence and the effectiveness of internal auditors. Well-established and independent audit committees appoint the chief internal auditors, review the work of the internal audit function, report recommendations to the board of directors and work as liaisons between external and internal auditors, thus reinforcing the effectiveness of the internal auditors in safeguarding company assets and increasing the credibility of the financial reporting system. Independent audit committees can provide auditors with direct access to the committee, without the presence of the company management, thus preventing any potential management pressure and reinforcing the independence of auditors.

# **3.9 Disclosure & Transparency**

Without transparence of corporate governance, shareholders and other stakeholders do not have adequate confidence in the information provided by company management. According to Hakim (2002) and Leung & Cooper (2005), effective corporate governance cannot exists in the absence of acceptable levels of transparency, accountability and disclosure. Hence, it can be concluded that corporate governance is a set of mechanisms that help in confirming the

fairness and justness of company dealings with all the stakeholders, and in reinforcing the level of disclosure, transparency, and accountability. Bushman et al (2004) defined governance transparency as "availability of firm-specific to those outside publicly traded firms" (p. 207).

An audit committee also has a significant role in meeting investors' expectations and ensuring that a company has complied with approved accounting standards and disclosure rules for corporate reporting. In this respect, Audit committees are expected to adopt good practices aimed at maintaining high standards of corporate disclosure and integrity of information, with prime concern for responsibility and accountability to shareholders and other stakeholders (Megat, 2000).

Disclosure aims to provide financial report users with adequate information needed for sound financial decisions. Jordanian corporate governance codes have devoted a complete chapter (Chapter Five) to the subject of Disclosure and Transparency. The chapter mainly deals with the role and duties of audit committees, external auditors and internal auditors, and communications between corporate governance constituents, in regards to ensuring and maximizing transparency, disclosure and reporting quality.

According to the JCGC and the instructions on disclosure for listed companies and accounting and auditing standards (2004, Articles 8 and 9), all listed companies on the Amman Stock Exchange (ASE) are obliged to establish their own websites, yet many companies have not generated their websites, which is a clear contradiction of the JCGC's disclosure & transparency section. As per the code, all listed companies are required to inform the public of any new material facts, as soon as possible. Within a week of this public disclosure, a copy of the public disclosure, in addition to a detailed report on the new facts, should be submitted to the Jordanian Securities Commission (JSC). Surprisingly, companies are also required to promptly and publically confirm or deny any and all news items published in the media regarding material facts about the company, and are required to provide a copy of the official statement to the JSC.

Similarly, all listed companies are required to provide the Amman Stock Exchange (ASE) with all material financial and non-financial information, without delay, and to release that information to the public through the ASE's website under the circulars and disclosure section.

The importance of having a website for each listed company is maximized when the companies provide continuously updated information to the shareholders and potential investors. "Accounting information has relevance if it is timely; it must be available to decision makers before it loses its capacity to influence decisions" (Kieso et al, 2010, p. 491). Listed companies must use their websites, in addition to the annual and interim reports, to release all necessary information to the public. This information, in the hands of current and potential shareholders, will help them take the right decision regarding keeping, selling and buying shares. Concealing information will have a negative impact on the quality of investors' decisions, creating potential future surprises for investors and, consequently, raise the dilemma of accountability regarding the audit expectations. On the other hand, there is such a thing as having too much communication, whereby the crucial information is lost within a deluge of routine disclosures.

The JCGC (Chapter 5, item 2) required companies to publically disclose all relevant information in a timely, accurate and clear manner so as to allow investors to make informed decisions. These disclosures should include periodic reports, all emerging material facts, insider dealing, related party transactions, and board of directors and executive management's remuneration and benefits.

Before the establishment of the JCGC, Malkawi (2008) pointed out that related-party transactions, such as transactions between the company and the directors and general managers, were prohibited. For example, loans to directors are prohibited. The law required companies to provide relevant disclosure for loans made to the related parties such as the company's executives, directors, its major clients and suppliers, and parent companies. However, the definition of related-party transactions was unclear.

The newly adopted JCGC has clearly defined "Related Party Transactions" in Chapter One- The Definitions. It has been defined as "Any deal or contract, the value of which exceeds JD 50,000 (fifty thousand Jordanian Dinars JD)<sup>33</sup> made between the company and any the related parties". The codes go on to describe and identify related parties: "families and relatives of directors, partners, and companies under the control of members of the board of directors are within the definition of related parties." The JCGC stated that the company shall provide shareholders and investors with accurate, clear and timely disclosure of information about the related party transactions. The codes also stated that the audit committees must ensure that no conflict of interest may arise from the company's transactions, contracts or projects with related parties. Audit committees also must review and approve related party transactions prior to their ratification by the company (JCGC: Chapter Five, section two, items 11 & 12).

Wright (1996) pointed out that the role of the corporate governance system is to ensure the company provides adequate disclosure and transparencies on the financial reporting to the shareholders. The study found a strong positive relationship between the soundness of the governance structure and the quality of disclosure and transparency. The study also found non-independent audit committee members are associated with negative financial reporting quality, in the perception of analysts.

Cheng and Abdel-Qader (2010) have investigated whether auditors were responsible for surprise corporate failures. The paper suggested that the lack of management disclosure of how they assess going concern assumptions, along with the lack of auditors' disclosure of how they evaluate management's going concern assumptions, in addition to the passive approach of the auditing standards, were associated with surprise corporate failures and widened the audit expectation gap between investors and auditors. In regards to narrowing the expectation gap, the article recommended improvement in disclosures by both management and auditors, with adequate disclosure of material uncertainties and an explicit qualification on the audit going concern opinion. Abdel-Qader (2002) recommended that the audit report should be expanded

 $<sup>^{33}</sup>$  JD has fixed exchange rate with the USD; Approximately, 1 JD = 1.41 USD

by including a statement about the auditor's evaluation of the internal control system and the results of reviewing the entity's ability to continue in the future.

# 3.10 Corporate Accountability

"Accountability has definitely become a topic of concern throughout governance literature" (ERKKILÄ, 2007). The recent spate of big-name corporate collapses, such as Enron, WorldCom and Tyco in the U.S, Barings Bank in UK, Philipp Holzmann and Comroad in Germany, Yline in Austria, HIH in Australia, SAirGroup in Switzerland, Lernout & Hauspie in Belgium and Parmalat in Italy, have increased public pressure to reform corporate governance systems and demand greater accountability over the corporations. In response to corporate failures, governmental and non-governmental regulating and supervisory bodies worldwide issued strict legislation, regulations and directives in order to restore public trust in capital markets and the auditing profession (Pott, Mock & Watrin, 2009). For instance, the Sarbanes-Oxley Act of the United States and the revised 8th Directive of The European Union (Council of the European Communities, 2006), both included strict penalties to be implemented on those who violate the regulations. Mohamed and Hussain (2005) pointed out that dramatic pressure for greater accountability has placed an immense emphasis on the functions and the vital roles of the audit committees in reinforcing corporate governance. The external auditor profession also came under fire for their failure to develop a 'whistle blower' type mechanism.

It is perceived that increasing the awareness of accountability, in addition to having a clear delineation of powers and responsibilities, helps in limiting and /or preventing fraud and malpractice. Corporate governance can be defined, in its narrowest sense, as a system for improving accountability (Farrar, 2003; Wolfensohn, 1999; Higgs Report, 2003; Tricker, 1984; Jubb et al, 2012). Hence, corporate governance initiatives have placed a great deal of emphasis on ensuring accountability over corporate governance bodies. For instance, the British Corporate Governance Codes, issued by The Financial Reporting Council (FRC), devoted a

separate section to accountability, emphasizing the duties and responsibilities of the board of directors, the company management, the internal auditors, the audit committees and the external auditors. The section included the role of the directors in preparing the annual report, specifying their responsibility for the soundness of the financial reporting system and that the entity's ability to continue as a going concern. The codes also stipulated the role of the directors in assessing the company's risk and reviewing the internal control system, in addition to the board's responsibility to establish an independent audit committee whose main functions and responsibilities must be set in a written charter (The UK Corporate Governance Codes, 2010).

Solomon (2010) pointed out that the most common characteristic in the definitions of corporate governance is accountability because it helps protect the wealth of the shareholders and the rights of the stakeholders. Sherer & Turley (1997) pointed out that the most significant elements of corporate governance were ensuring the proper implementation of accountability and overseeing the company executive management. Millstein (1998) pointed out that the corporate governance simply aims to ensure that the managers are accountable to the board of directors and directors are accountable to the shareholders. He also argued that having the managers and the directors accountable for corporate assets leads to improved corporate performance.

Accountability, in the context of financial reporting, is defined as "the responsibility to provide information to enable users to make informed judgments about the performance, financial position, financing and investing, and compliance of the reporting entity" (Statement of Accounting Concepts, SAC 2.5, 2012). Gay & Simnett (2010) pointed out that accountability refers to the responsibilities of the governing body of the company to be held accountable for reporting, and is subject to an examination according to an accountability system that involves both internal and external parties.

Wild (1994) argued that the audit committees are the most effective pillar of corporate governance, overseeing accountability and reducing fraud, error and the risk surrounding the

financial reporting. Wild (1994) examined and reviewed data from 260 companies in the U.S.A., both before and after the establishment of an audit committee. That data covered the period from 1966 to 1980, and was analyzed to establish empirical evidence about the relationship between accounting earnings and audit committees. The results showed that the formation of audit committees has significantly made the earnings more informative to users of financial statements and reports. The study attributed that to the vital role of the audit committee in overseeing the quality of the management accountability to the shareholders.

Porter (2009) reviewed the corporate governance accountability literature, pointing out that accountability was demanded, due to giant corporate collapses, to counter any possible abuse of executives' powers and to produce reliable and informative financial statements that were free of error and fraud. Porter stressed on the importance of the tripartite or "trinity" audit function (external auditors, internal auditors and audit committees) in maintaining the integrity of corporate accountability, and argued that the audit trinity members have interlocking and mutually supportive functions. Porter also added that, in addition to the vital roles of the audit trinity in securing the integrity of the financial reporting, audit committees should have the power and responsibility to ask probing questions of executives and employees, and should possess a high degree of skepticism in order not to be easily satisfied with any reply.

Corporate governance systems characterized as having a poor accountability system impair internal and external auditors' ability to do their jobs effectively and with due professional care which, in turn, affects the integrity of the financial reporting system. In Jordan, Qadamani (2007) criticized the accountability of corporate governance systems, and pointed out that some of the large controlling investors, especially those who hold both positions on the board of directors and are executive managers, believe that they own the assets of the entity regardless of the interest of the minority shareholders, and feel that they are not accountable. The former controller of the Companies Control Directorate in Jordan, Khaza'alah (2001), pointed out that, due to the lack of accountability, external auditors did not accomplish their tasks as agents of shareholders, and were mainly tasked with preparing the financial statements from the ending balances as per the trial balance while external auditors were

supposed to assess their client's risk, the efficiency of their internal control, and perform preliminary analytical procedures. Khaza'alah also attributed this situation to the unbalanced (low) audit fees compared to the external auditors' legal responsibilities. Furthermore, weak accountability in Jordan brought about demands for implementing strict punishments against external auditors who violate the code of the auditing profession (Matar, 1994).

Abdullatif (2003) has interviewed Jordanian lawyers and revealed that the weak accountability system, insufficient legal responsibilities and the low probability of punishment falling on corporate governance constituents in Jordan, have led some external auditors to cooperate with dishonest managers at client companies to keep their jobs as auditors. He also added that in the case of Jordan, where the owner/manager was a dominant characteristic among companies (i.e there was no separation between the positions of owners and management), the likelihood of malpractice and theft of company assets was high. This duality of positions (owner/manager) in Jordan was also criticized by Al-Hasani (1999), Saadah (1993), and Al-Jazy (2005), all of whom criticized relevant laws and demanded reform of corporate accountability to separate management of listed companies from their owners. They also argued that the separation enhances the external auditor's independence and performance when the appointers of the auditor are different from the company executive management.

Abed et al (2012) investigated, over the period 2006-2009, the impact of Jordanian corporate governance characteristics on management's ability to manipulate financial reporting and income. The study revealed that Jordanian listed companies' compliance with the corporate governance code was poor due to insufficient accountability and penalties. The authors also related this to the fact that the JCGC became operational in 2009 on a comply or explain basis. The study recommended Jordanian policy makers encourage compliance with the codes in order to control the opportunistic behavior of boards of directors and management regarding manipulation of financial statements, such compliance thereby enhancing the reliability of the financial reporting system.

Compliance with the JCGC might be enhanced by mandating the external auditors to report, within the audit report, the entity's compliance with the JCGC, as suggested by Al-Hanini & Dahmash (2008) and Al-Khadash, et al, (2012). In the UK, Porter (2009) pointed out that, under the United Kingdom Listing requirements, the external auditors report about the publicly traded companies' compliance with the accountability provisions of the UK corporate governance codes.

The JCGC (Chapter 2, Item 5) has prohibited duality of the roles CEO and chairman. This segregation of duties aims clarify the responsibility and accountability of the company board and management. The JCGC stipulated the role of the audit committee in making recommendations on the selection of the internal and external auditor and securing the independence of the latter. The audit committees also have the responsibility to evaluate the internal control and auditing procedures. The JCGC stated that the external auditors have the responsibility to audit the company's accounts according to the ISAs and to answer any queries and questions raised by the shareholder during the assembly meetings. In addition, the auditor has the responsibility to inform the authorities concerning any violation of the legislation in force and regulations, or any administrative or financial matters that affect the company's financial position negatively (Corporate Governance Codes in Jordan, 2009, Chapter Five, Section 4).

In regard to corporate governance bodies' legal liabilities toward the laws, the JCGC (Chapter 2, Item 12) stated that members of the boards, the company executive management and the external auditors are legally accountable to notify the supervisory authorities if:

- The board of directors or the general manager exploits their powers for their personal benefit.

- The board of directors or the general manager performs any act that implies fraud, embezzlement, bribery or betrayal of confidence that influences the shareholders and the stakeholder's rights.

- If the entity suffers severe financial or administrative deficiencies, or if the company suffers massive losses that might impact shareholders' and/or stakeholders rights.

The JCGC (Chapter 2, Section 1, Items: 13, 14, 17) also stipulated the responsibility of the board of directors to oversee and evaluate the performance of company management, and to annually review their implementation of the codes of corporate governance. The board also has the responsibility to establish a mechanism to receive complaints and recommendations from the shareholders.

In regard to the shareholders' rights, the JCGC has devoted a chapter to the discussion of their rights. In an effort to give shareholders the power to hold company management and the board of directors accountable for their actions and decisions, the JCGC (Chapter 4, Section 1, Items 8 & 9) stated that the shareholders have the right to file a lawsuit against the company management and directors. The JCGC added that shareholders can further claim for compensation for damages or losses incurred as a result of revealing the company's secrets, confidentiality, and breaches of the legislation in force or of the company's bylaw, or any fraud or negligence in managing the company.

Interestingly, the JCGC has not identified the legal responsibilities of the external auditors, although they were stipulated in the Company's Law of 1997 (Articles 201, 202 and 203). The law stated that the auditors are legally accountable toward the company, the shareholders and the stakeholders for any violation of the legislation in force, the company bylaws and the internationally accepted (adopted) accounting and auditing standards. The auditors are also liable to compensate any realized loss incurred as a result of negligence and errors committed by the auditors, or as a result of their failure to fulfill their duties, or as a result of expressing opinions about the financial statements that significantly do not confirm with real facts, or for approving disordered financial statements. Furthermore, the law also stipulated penalties if the auditors disclose client company secrets or speculate in the company's shares.

In summary, this section, reviewed the literature of corporate accountability and their impact on the reliability of the audit report and the level of the audit expectation gap as perceived by financial analysts. The section reviewed corporate failure and the need to adopt effective corporate governance systems in order to protect the capital market. Corporate accountability was considered one of the major constituents of corporate governance, as can be seen in many cited definitions of corporate governance. This study discussed how and why the institutional investors consider, or might be affected by, the extent of corporate accountability and the responsibility of the board of directors, the company management, the external auditors, the internal auditors and the audit committees in formulating their judgment decisions.

# **3.11 Chapter Summary**

Narrowing the audit expectations gap continues to be a key issue of interest to the accounting profession (Dewing & Russell, 2002). The expectation gap is considered one of the major issues confronting the accountancy profession. The expectation gap cannot be ignored as it is a key strategic risk for the audit firms, and claims that the expectations of the community are not objective, or that criticisms directed to the auditing profession are not fair, do not reduce the risk entailed (Jarbou, 2007).

Based on reviewing and analyzing the literature, uncertainty about the external auditor's independence is considered the main reason for the expectations gap in Jordan. The literature showed that an independent and well-established audit committee enhances external auditor's independence and mitigates management pressure to issue a clean opinion, and shield the auditors from being dismissed after issuing a qualified or a going concern opinion.

Insufficient internal control procedures, coupled with weak and unqualified internal auditors, compromise internal controls' effectiveness in detecting fraud and errors and in preventing malpractice. The literature highlighted the importance of the audit committees in nominating

the internal auditors and reinforcing their effectiveness. The effectiveness of internal auditors could be enhanced by giving them direct access to the audit committees. The literature also showed that the audit committee should evaluate the work of auditors, the internal audit budget, and the observations of the internal auditors, and should work as a liaison between the external and internal auditors and between the auditors and the company management and board of directors.

Adequate and timely disclosure of transparent information, in addition to sufficient corporate accountability systems, help direct the work of executive management, board of directors and company employees towards achieving the interests of the company's stakeholders, resulting in enhanced investor confidence in financial reports and, consequently, contributes to narrowing the audit expectation gap.

The next chapter reviews the research methodology employed in the fieldwork for the thesis and outlined in the study model. The chapter also provides a justification for the chosen research strategy and provides a rationale for combining the qualitative and the quantitative methods. The chapter also identifies the population of the study and the utilized data collection techniques and justifies the chosen sample. The chapter also presents the limitations of the study's methodological approach.

# Chapter 4: Research Methodology

# 4.1 Introduction and Chapter Overview

This chapter discusses the study hypothesis and the research questions that are utilized to address the gap in knowledge regarding how the selected elements of the Jordanian corporate governance code affect the auditing profession in Jordan in terms of the reliability of the audit report and the level audit expectation gap as perceived by institutional investors. These elements of the JCGC selected for this study are external auditors' independence, internal auditors' effectiveness, disclosure & transparency, and corporate accountability.

The rationale for the adopted method (the mixed method approach) and the research design are addressed in this chapter. The research design includes a combination of two methodologies that integrate and complement each other to provide a thorough analysis of the research topic. In addition to the design of the methodology, this chapter also describes the unit of analysis, the data collection techniques and the research implementation.

The postpositive approach is adopted in this study. According to Creswell (2003, p. 7) "Postpositivism reflects a deterministic philosophy in which causes probably determine effects and outcomes. Thus, the problems studied by postpositivists reflect a need to examine causes that influence outcomes such as issues examined in experiments".

### 4.2 Research Method

This section discusses the prevailing methods used in prior studies of corporate governance and the auditing profession in Jordan, in relation to the external auditor's report and the level of the audit expectation gap. This section also justifies the rationale for the mixed method approach and discusses the strengths of the combination of the experimental and interview methods.

#### 4.2.1 The Related Research Methods Previously Adopted

Cooper & Schindler (2010) point out that good research should establish dependable data, based on professional and systematic processes, and provide reliable information for decision making. According to Veal (2005), there are many points to be taken into consideration while the researcher chooses an appropriate research methodology, such as the research questions or the study hypothesis, previous research and data availability. Veal (2005) point out that the research questions guide the researchers to the data sources, and previous research might provide the researcher with methods to be adopted, improved or replicated to achieve comparability between the new and prior research.

This thesis adopted the mixed method approach to achieve the study objectives. This approach involves integration of qualitative and quantitative data collection and analysis. The quantitative element was designed to investigate, through experiment, the relative and interactive weights of the influence of selected elements of the Jordanian corporate governance on (1) the perceived reliability of the audit report and (2) the level of the audit expectation gap as perceived by financial analysts. This thesis is the first study to adopt an experimental approach (i.e. the repeated measures design) in the field of corporate governance and auditing in Jordan. The design of the questionnaire for the factorial experiment of this thesis was prepared based on many previous research studies that adopted the same method (e.g. Hopkins, 2009; Wood & Ross, 2002).

Quantitative approaches, utilizing financial data or perceptual data, are in common use in Jordan, with quantitative research utilizing perceptual data being dominant, especially in studies related to corporate governance and the auditing profession. In fact, most non-Jordanian studies that have investigated the audit expectation gap utilized questionnaires as an instrument for achieving their study objectives (McEnroe & Martens, 2001; Porter, 1993; García Benau, Humphrey Moizer, & Turley 1993; Gloeck & Jager, 1993; Humphrey et al, 1993; Al Husaini, 2000).

The mixed method approach is rarely used in Jordan. This approach is more frequently adopted in academic PhD theses rather than other types of applied research<sup>34</sup>. The researcher noted that none of the previous Jordanian studies discussed earlier (in Chapter One and Three) adopted the experimental approach, although the use of experimental studies delivers great value by identifying causal correlations among the variables, according to Black (1986) who also added "experimental studies in economics and finance have value" (p. 537). Hence, the use of the experiment approach is common and is recommended for use in developed countries (Black, 1986; Baker & Nofsinger, 2002; Trotman, 1996). Table 1 shows a list of some of the key related Jordanian studies in the fields of corporate governance and the auditing profession<sup>35</sup>, with a brief description of the methodology adopted in each case.

# Table 1: Jordanian Corporate Governance and Auditing Profession Studies, Highlighting the Methodology Adopted.

	Research Title/ Author(s) & Year of Publication <sup>36</sup>	Brief description of the method
1	Developing a System for Corporate Governance in Jordanian Stock Corporations to Reinforce the Independence of the Auditor/ Al-Hanini & Dahmash (2008)	Mixed method: Questionnaires administered to external auditors, and interviews conducted with external auditors
2	A Proposed Model for the Role of Audit Committees in the Jordanian Public Shareholding Companies, and their Influence on the	Mixed method: Questionnaires and interviews conducted with Financial managers, external auditors, internal

 <sup>&</sup>lt;sup>34</sup> The first two studies in table 1 are Ph.D. theses.
<sup>35</sup> All of these studies have been reviewed in chapters One and Three.

<sup>&</sup>lt;sup>36</sup> Full citation is presented in the References Section

	Effectiveness and Independence of External Audit/ Swaiti (2006)	auditors, chairpersons of the board of directors and audit committees.
3	The Effect of ISA 700 (Revised) "The Independent Auditor's Report on Narrowing the Audit Expectations Gap./ (Al-Khadash, et al, 2012).	Quantitative: Questionnaires distributed to auditors, preparers of financial statements, investors and academicians.
4	The Effectiveness of Audit Committees in Jordanian Public Shareholding Companies and Potential Characteristics Affecting It: Perceptions from Auditors in Jordan/ Abdullatif (2006)	Quantitative: Questionnaires distributed to the external auditors
5	Structure and Causes of the Audit Expectations Gap: Evidence From Jordan/ Al- Thuneibat (2003)	Quantitative: Questionnaires distributed to external auditors and executive and financial managers of Jordanian listed companies
6	Expectations gap and the responsibilities of auditors: Jordanian Investors' perspective/ Hajir (2001)	Quantitative: Questionnaires distributed to individual and institutional investors.
7	The Effect of Modified Auditors Opinions on Shares Prices: Evidence from Amman Stock Exchange/ Zureigat (2010)	Event Study: based on the data available for the Jordanian listed companies at ASE during the period from 2002 to 2007
8	Accounting Information Qualitative Characteristics Gap: Evidence from Jordan/ Obaidat (2007)	Quantitative: Questionnaires administered to a sample of external auditors and investors.
9	Audit Committees' Responsibilities and Their Relevance to the Corporate Governance Process/ Abdullatif (2007)	Quantitative: Questionnaires administered to a sample of external auditors
10	Assessment of Corporate Governance in Jordan: An Empirical Study/ Shanikat & Abbadi (2011)	<i>Qualitative</i> : 20 Interviews carried out with key employees of 10 listed companies
11	The expectation gap between investors and auditors in Jordan/ Omari (2003)	Quantitative: Questionnaires distributed to investors (individual investors, financial brokers, employees of banks' investment units) and external auditors
12	The Capability of Sarbanes-Oxley Act in Enhancing the Independence of the Jordanian Certified Public Accountant and its Impact on Reducing the Audit Expectation Gap "An Empirical Investigation from the Perspectives of Auditors and Institutional Investors/ Al-Khadash & Al- Sartawi (2010)	Quantitative: Questionnaires were used to collect data from Jordanian Institutional Investors and external auditors
13	Factors that Affect Decisions on Selecting and Changing Public Accountants in the Jordanian Public Companies from the Perspective of the Audit Committees/ Al-Saudi (2007)	Quantitative : Questionnaires administered to chairpersons and members of audit committees in Jordan

14	External Auditors' Assessment of the Internal Audit Function in the Light of Implementation of the International Standard on Auditing No. (610): Analytical Study from the Point of View of the External and Internal Auditors in Jordan/ Thnaibat & Shunnaq (2010)	Quantitative: Questionnaires distributed to samples of external and internal auditors
15	The Effect of Corporate Governance Guide Issued by the Central Bank of Jordan in Improving Internal Audit Quality in Jordanian Banks/ Zureigat (2011)	Quantitative: Questionnaires distributed to internal audit staff in Jordanian banks.

One of the major advantages of qualitative studies is that they provide a high level of understanding, but, at the same time, they lack empirical robustness, which consequently affects their generalizability and makes them questionable (Bryman, 1988). Quantitative methods offer stronger statistical evidence than qualitative methods do. However, quantitative studies that aim to capture attitudes (usually via survey instruments such as self-administered questionnaires) are subject to criticism regarding their objectivity, as they tend to be self-completed, and usually suffer from low return rates (Buglear, 2012; Cohen & Manion, 2007; Mitchell & Jolley, 2012). On the other hand, quantitative research that depend on publicly available data (such as corporate annual reports and data available on stock exchange sites) enjoy an advantage in objectivity.

The studies mentioned in Table 1 provided evidence about the existence and the nature of the audit expectation gap in Jordan, including suggestions on how to narrow the gap; they also explained the situation of corporate governance in the Jordanian environment. However, there are several points, inherent in their structure, that possibly limited their added value, such as they were mostly quantitative and questionnaire-driven studies with much focus on external auditors' independence and performance. This might lead researchers to miss or ignore some other factors given their focus on testing their study hypothesis rather than producing new theories (Johnson & Christensen, 2010). Whilst such studies can identify assumed relationships, they cannot show the direction of the relationship nor their relative Importance to the variables being studied.

As mentioned earlier, the JCGC has been implemented relatively recently in Jordan, becoming operational at the beginning of 2009. However, the Jordanian Securities Commission has mandated all listed companies to report, in a separate chapter of their 2010 Annual Reports, the extent of their compliance with the code<sup>37</sup>. Due to the recentness of the implementation of the JCGC, publicly available data about the corporate governance of listed companies is still rare. Therefore, in order to answer the research questions of the study, which aimed to provide empirical evidence of the impact of the selected factors of JCGC and their potential interactive weights on both the perceived reliability of the audit report and the level of the audit expectation gap in Jordan, this thesis adopted the mixed method.

#### 4.2.2 The Rationale for the Mixed Methods Approach

The research questions and objectives, set out in sections 1.4.1 and 1.4.2 respectively, are best addressed by adopting a mixed-method approach. This approach proffers a mix of contextual interpretive relevance and generalizability, in addition to providing deeper perceptions, closer insights and greater diversity of viewpoints than any single method alone (Borkan, 2004). Johnson & Onwuegbuzie (2004) defined mixed methods research as "the class of research where the researcher mixes or combines quantitative and qualitative research techniques, methods, approaches, concepts or language into a single study" (p. 17). The Authors also added "A key feature of mixed methods research is its methodological pluralism or eclecticism, which frequently results in superior research" (p. 14). Therefore, the adoption of the mixed method has become common in business research (Collis & Hussey, 2003).

<sup>&</sup>lt;sup>37</sup> See Circular Number: 12/1/4659. Dated December 15, 2010. Issued by Jordanian Securities Commission. Available on: http://www.jsc.gov.jo/Public/english.aspx?site\_id=1&Lang=3&Page\_Id=2550&Menu\_ID2=257

To conduct a mixed method study, several combinations of quantitative and qualitative techniques might be used. Smith (2003) suggested four general types of quantitative approaches for use in mixed methods: Experimental Method, Quasi-Experimental Method, Survey Method and Archival Method. Mason (1996) pointed out that qualitative studies are of three main types: Case Studies, Ethnographic Studies and Phenomenological Studies. The mixed method produces better, higher quality findings if it is conducted appropriately; this means selecting the best combination of qualitative and quantitative techniques and choosing the appropriate sequence that best achieves study objectives (Hurmerinta-Peltomäki & Nummela, 2006; Bryman, 1992; Bryman, 2006; Bryman & Bell, 2007; Creswell, 2009; Onwuegbuzie & Leech, 2005; Collins, Onwuegbuzie, & Sutton 2006). Combining experimental research design with a qualitative method delivers better understanding of the research problem (Burns, 2000).

There have been, during the 1970s and 1980s, arguments about the usefulness of using a mix of quantitative and qualitative techniques in a single study. However, the adoption of mixed methods is now commonplace and widely accepted by academicians and professionals alike (Torrance, 2012; Creswell, 2010; Tashakkori & Teddlie, 2003; Johnson et al, 2007; Clark & Creswell, 2007; Teddlie & Tashakkori, 2009; Bryman, 2006; Sale, Lohfeld, & Brazil 2002; Johnson & Onwuegbuzie, 2004). In the health care and clinical fields, it is common to conduct research that combines both qualitative and quantitative approaches to improve generalizability, reliability, validity and accuracy (Morgan, 2007; Borkan, 2004; Johnstone, 2004; Andrew & Halcomb, 2009). It is noteworthy to mention that some scholars classified the mixed method as a third paradigm in research (Cameron, 2011; Creswell, 2009; Howe, 1988; Datta, 1994; Johnson & Onwuegbuzie, 2004). Scholars also pointed out that the adoption of the mixed method approach delivers potential complementarities for qualitative and quantitative research (Luyt, 2012; Yauch & Steudel, 2003; Bryman, 2006; Bryman, 2008; Sale et al, 2002).

Reichardt & Cook (1979, p. 225) remarked that there existed no convincing reasons not to use the mixed method, after appropriately considering how to best use it to achieve the research objectives. Moreover, they suggested that there was no essential conflict in world view or outcomes generated from the various components of the mixed method (Datta, 1994; Johnson & Onwuegbuzie, 2004).

The quantitative part of this thesis delivers objective evidence and verification of the magnitude of influences and the relative weights of selected JCGC variables on the perceived reliability of the audit report and the level of the audit expectation gap. The qualitative part of this study lends additional depth of understanding of the judgment process. In this regard, this thesis adopts the recommendations of Reichardt & Cook (1979): "comprehensive evaluations should be process-oriented as well as outcome oriented, exploratory as well as confirmatory and valid as well as reliable" (p. 225). The use of the mixed method enhances the construct validity of the study findings since it enables discovery of (1) the level of confidence in the operational definitions of variables, (2) whether the relationships between the study variables logically reflect the true theoretical meaning of a phenomenon under investigation and (3) the extent to which the variables were actually measured (Zachary, McKenny, Short, & Payne 2011; Shadish et al, 2002; Keats, 1988; Wenger, 1999; Ariño, 2002).

The adoption of the mixed method delivers many advantages, such as triangulation, complementarity, and expansion, as outlined by Greene, Caracelli, & Graham (1989). Triangulation refers to combining qualitative and quantitative research to get mutually corroborated conclusions and to demonstrate convergence (Bryman, 2006; Bryman, 2008; Greene et al, 1989; Yin, 2009). Complementarity research utilizes quantitative and qualitative methods to deliver, and support, robustness, clarification, enhancement, and interpretation of the findings of the adopted methods (DeCuir–Gunby, 2008; Brannen, 1992; Wenger, 1999; Yin, 2009). Expansion is to use various approaches in order to "extend the breadth and range of the study" (Greene et al, 1989, p. 259).

#### 4.2.3 Judgment and Decision Making

External auditors play a significant role, via their audit reports, in lending independent integrity to published financial statements, which are essential inputs in investment decisions made by financial analysts, shareholders, creditors and other stakeholders. However, investment decision-making is in the hands of individual human beings, which make them subject to their biases, partiality and prejudices (Statman, 2005). This thesis investigates the perceptions of financial analysts regarding the impact of the JCGC on the value of audit opinion, as measured by the reliability of the audit report and the size of the audit expectation gap; those financial analysts are also subject to normal biases and idiosyncratic errors in their judgments (Hunter & Coggin, 1988).

Several authors have studied the importance of human information processing research in relation to the field of accounting and auditing (Libby & Lewis, 1982; Trotman, 1996; Maletta & Kida, Cohen & Paquette 1990; Hooper & Trotman, 1996; Solomon & Trotman, 2003) and the field of behavioral finance, which adopted techniques from the psychology literature (Snowball, 1986; Kahneman, 2011; Andersen, 2008; Fama, 1998). In the context of financial decisions, human information processing and judgments are categorized as a field of study under behavioral finance as described by Pike & Neale (2006): "the study of how psychological and sociological factors influence financial decision making and financial markets" (p. 665). Al-Khafaji, Aly, Gheyara, & Metawae (1993) provided a broader definition, arguing that, under conditions of uncertainty, decision makers need to use the ability to interpret information about a given event, they may need to interpret indirect signals or cues. These information cues could arrive simultaneously or over time and may, therefore, be missed since the decision maker may not be expecting them or may be unable to separate them from other cues.

Payne (1982) argued that decision making is connected to information processing. He also pointed that individuals consider common characteristics when considering similarities, and distinguishing characteristics when considering different options of choices (See also Slovic &

MacPhillamy, 1974; Einhorn & Hogarth, 1981). Libby (1981) stated that causal data is thought to be more sufficient and has more influence on decision makers than diagnostic data does.

Financial analysts appreciate the added value that external auditors integrate into financial statements; that added value depends on the corporate governance system and is believed to be influenced by many factors, such as external auditors' independence, internal auditors' effectiveness, disclosure & transparency, and corporate accountability, as outlined in the previous chapter. Financial analysts consider these factors in formulating their judgments, based upon multiple cues. They might assess that these cues interact in their influence (configurally) or impact independently (main effects). Information Processing & Judgment Decisions are discussed in the following section, followed by a review of the literature relevant to main effects and configurality, self-insight and decision confidence.

#### 4.2.3.1 Information Processing & Judgment Decisions

When Individuals consider their judgment decisions, as human beings, they are subject to several factors that may lend potential bias into their decisions (For examples of potential influential factors see Hilbert, 2012; Merkle, 2009; Pronin, 2007; Tversky & Koehler, 1994; Kahneman & Tversky, 1972). During the 1970s, Kahneman & Tversky affirmed the importance of the following influential factors: the representativeness heuristic, the availability heuristic, and the anchoring and adjustment heuristic.

Kahneman & Tversky (1972) pointed out that the representativeness heuristic emerges when factor (A), for example, resembles another factor, (B), and so may lead to systematic biases. Kahneman (2011) added "when 'A' is highly representative of 'B', the probability that 'A' originates from 'B' is judged to be high. On the other hand, if 'A' is not similar to 'B', the probability that 'A' originates from 'B' is judged to be low" (p. 420). In other words, the

representativeness heuristic is likely to occur when incidents are described by their general characteristics.

The availability heuristic refers to the assessment, by individuals, of the probability of an event by the ease with which relevant, frequent or recent events come to their mind. Steginga, Occhipinti, Gardiner, Yaxley, & Heathcote (2002) added "when a judgment or a decision is mediated through an instance that comes easily to mind, [it] may refer to an event with a positive or a negative outcome" (p. 6). According to Kahneman (2011), "availability is affected by factors other than frequency and probability. Consequently, the reliance on availability leads to predictable biases" (p. 425).

The anchoring and adjustment heuristic emerges, for instance, when individuals underweight new information when adjusting past overweighting of another event or initial value (Libby, 1981). According to Kahneman (2011), "the initial value, or starting point may, be suggested by the formulation of the problem, or it may be the result of a partial computation . . . . different starting points yield different estimates, which are biased toward the initial values" (p. 427).

In the accounting and auditing arena, judgment decision making has been employed in experiments (Solomon & Trotman, 2003; Schultz, Bierstaker, & O'Donnell 2010; Martinov-Bennie, Cohen & Simnett 2011; Ng & Tan, 2003; Trotman, 1996), and it has been influenced by psychological research as noted by Gibbins & Swieringa (1995). The experimental design is primarily based on statistical methods to measure treatment usage, and the ANOVA technique is the most widely used statistical analysis employed in judgment decision making (Solomon & Shields, 1995).

The experimental research undertaken in this thesis involved presenting participants with a series of different treatments of independent variables (the four selected elements of the JCGC). In each treatment, participants' judgment decisions regarding the value of the dependent variables (the perceived reliability of the audit report and the level of the audit expectation gap) are analyzed in order to establish relationships with combinations of cues or

even with a particular cue. Cues must be balanced in their quality and quantity; because a mix of less and more functional cues limits their performance (Libby, 1981).

Human information processing is influenced by individual experience, age and time pressure. Individuals with less experience usually evaluate information based on how they were presented to them (the organization of the information), whereas more experienced people usually look for additional, and specific details within the, information (Bouwman et al, 1987). Taylor (1975) related age with longer decision making processes, seeking more information combined with an ability to evaluate this information. Time pressures mitigates efficiency in human information processing as individuals are more risk averse when they feel that they are under time pressure (Wright, 1974; Ben Zur & Breznitz, 1981). Finally, personality characteristics have not been seen to have a significant impact on human information processing (McGhee, Shields, & Birnberg 1978).

#### 4.2.3.2 Main Effects, Linearity and Configurality

Brown & Solomon (1989) defined configural information processing as "cognition in which the pattern of stimuli is important to the subsequent judgment/decision" (p. 2). Judgment decision studies aim to investigate the impact of key cues and treatments on judgment decision making and whether they contribute to the findings of the study independently (main effects) or in combination with other effects (configuraly). Ganzach (1997) pointed out that in configural relationships, "the impact (weight) of a given attribute depends on the level of other attributes" (p. 954).

In a linear model of judgment decision making, the relationship between cues and the dependent variables is consistent (monotonic relationship) regardless of the level of other cues; hence, the linear model delivers a good fit for some purposes of decision making (Einhorn, 1970; Libby, 1981). Libby (1981) has also pointed out that a positive relationship occurs when

the increase in one variable leads to an increase in the size effect of another variable and vice versa.

Configural cue processing is influenced by a number of factors, as evidenced by Ganzach (1997), Libby, Artman, & Willingham (1985), Payne (1982), Maletta & Kida (1993) and McGhee et al (1978). Firstly, experienced individuals are more configural in their judgments than less experienced individuals are. Secondly, judgment tasks are associated with less dimensional processing than choice tasks. Thirdly, individuals tend to shift from sophisticated tasks to simplified heuristic methods. Fourthly, environmental risk leads to the use of systematic and sophisticated configural decision processing.

#### 4.2.3.3 Self-Insight & Decision Confidence

Self-insight refers to "an individual's ability to express the relative emphasis he or she places on information cues when generating judgments" (Mear & Firth, 1987, p. 176). Self-insight is also defined as the extent of insight that individuals have in their judgment process (Trotman, 1996). Typically, the degree of self-insight has "been assessed as the correlation, over all cues, between a subject's cue usage and the importance rating subjectively attached to each cue (e.g. by allocating 100 points among the cues)" (Solomon & Shields, 1995, p. 151). In this study, self-insight is investigated by correlating and matching the objective outcomes obtained from cue usage (the 16 scenarios of the factorial experimental questionnaire) against the subjective weightings which have also been gathered from the participants via the same instrument. It is important to perceive the level of self-insight because that helps improve understanding of the learning process (Libby, 1981) and improve the accuracy of judgment as noted by Hooper & Trotman (1996).

Ganzach (1997) pointed out that experienced individuals do not, necessarily, demonstrate more accurate decisions than inexperienced individuals, although they are expected to demonstrate

a higher degree of self-insight (Feldman & Arnold, 1978). Large amounts of information do not guarantee greater accuracy (Slovic, 1972; Goldberg, 1968). The greatest increase in accuracy comes from obtaining data from two additional judges; any further increases in the number of judges contribute relatively little to accuracy (Libby & Blashfield, 1978). Hooper & Trotman (1996) showed that consensus between auditors who processed information configurally was higher than among auditors who did not process configurally. Consensus is often accepted as a reasonable and equitable replacement for accuracy (Libby, 1981; Ashton, 1985) which leads to establish an association between configural processing and improved performance since configural processing of cues decreases inconsistencies in decision making (Hooper & Trotman, 1996).

#### 4.2.4 Experimental Method and Analysis

This study adopted the mixed method approach whereby an experimental design was employed in the quantitative part of this research. Kerlinger (1986) defined the experiment in research as: "a scientific investigation in which an investigator manipulates and controls one or more independent variables and observes the dependent variable or variables for variation concomitant to the manipulation of the independent variables" (p. 293). Systematic experimental design was employed in this thesis In order to present judgment tasks and to infer cause-effect relationships (Libby, 1981; Keppel & Wickens, 2004). "Experimental studies allow causal relationships to be identified. The aim is to manipulate the independent variable (e.g. a reduction in price) in order to observe the effect on the dependent variable (e.g. sales levels of Sunday newspaper)" (Wilson, 2010, p. 110). Analysis of variance (ANOVA) is utilized in the current study. Slovic (1969) pointed out that ANOVA delivers improved quantitative understanding of configural effects (of interactively combined factors) and non-configural effects (independent main effects of the factors) in the use of information in judgments.

A brief historical review of the use of ANOVA techniques shows that it has been applied since the 1960s, initially in the medical arena (Hoffman, Slovic, & Rorer 1968). Its application in finance was introduced by Slovic (1969) who pointed out "The ANOVA technique has considerable promise as a device for describing and furthering the understanding of complex judgment processes. It is likely that this technique can provide even the expert with new insight into his inferential processes" (p. 262). Slovic (1969) concluded his study saying "It is now clear that substantial configural processing of information does occur and can readily be detected by the ANOVA technique" (p. 263). Since then, ANOVA became common in the field of investigating auditors' judgment decisions (e.g. Schneider, 2011; Teoh & Lim, 1996; Hooper & Trotman, 1996; Daniels & Booker, 2011; Trotman, 2005). Trotman (2005) argued that ANOVA has become the most commonly adopted technique in most prestigious research journals.

In this thesis, analysis of variance was used to measure the impact of the selected JCGC factors on the reliability of the audit report and the level of the audit expectation gap. ANOVA is an appropriate statistical technique for experimental studies that aim to investigate both main effects and interactions between the study factors. "ANOVA tests are based on the partitioning of the variability in the dependent variable. Components of variability associated with the main effects of the factors and the joint effects of combinations of factors are obtained and compared with the component that reflects error variability" (Myers & Well, 1995, p. 6). The use of ANOVA has some advantages such as the elimination of multicollinearity which makes the interpretation of the main and interactive effect clear without ambiguity, furthermore, the demonstration of the relative weights of the significance of the cues is also considered one of the important advantages of ANOVA (Libby, 1981; Trotman, 1996).

Slovic et al (1972) stated "the ANOVA technique is important because, by isolating the effects of interactions from those of main effects, it makes the empirical description of configural judgments feasible" (p. 290). They also added that two conditions have to be satisfied to conduct ANOVA analysis: (1) The levels of the variables should be categorical, i.e. not continuous; this condition can be met by designing a suitable survey instrument; (2) The
variables must be orthogonal; this can be satisfied by using a complete factorial design in which all possible combinations of the cues and variable levels are presented (Trotman, 1996).

In this study, the reliability of the audit report and the level of the audit expectation gap were investigated as dependent variables, and an initial Multivariate analysis of variance (MANOVA) test was used to test the significance of the study independent variables as a set. Using MANOVA helps reduce type one statistical errors when individual tests of ANOVA are measured with the same significance level (Coolican, 2009). Coolican (2009) also added "If MANOVA is significant it is legitimate to investigate further and take as significant any of the individual (univariate) ANOVA results which the MANOVA procedure has shown to be significant" (p. 494).

Within-subject design, also called "repeated measures design" (Myers & Well, 2002, p. 210; Mitchell & Jolley, 2012, p. 560; Coolican, 2009, p. 69) is the technique employed for the experiment of this study. Coolican (2009) also pointed out that the technique investigates the differences within each subject or participant. Within-subject designs have been described as a "true experiment" (Heppner, Wampold, & Kivlighan 2008, p. 172). "The true experiment is the only research method that allows the researcher to confidently conclude that A caused B" (Trotman, 1996, p. 7) due to the random allocation of treatments combined with the manipulation of the study independent variables. This technique is widely employed for judgment decision-making studies and in analyzing problems associated with the use of cues (Libby et al, 1985; Trotman, 1996).

Repeated measures design does not just make comparisons between the subjects; it compares each subject's score and assessment using a specific condition with the same subject's assessment using another (different) condition. The advantage of the repeated measures design is that the subjects involved in the study are used as their own controls (Gay & Diehl, 1992; Cox & Reid, 2000; Ryan, 2011; Weathington et al, 2010) which helps in controlling confounding factors (Coolican, 2009; Trotman, 1996) and, consequently, detecting any systematic differences between the treatment conditions, if they appear, therefore increasing the power of the design (Keppel & Wickens, 2004; Heppner et al, 2008). However, allocating

the treatments randomly also allows for the control of systematic differences between subjects.

Like all research designs, the within-subject design has some disadvantages to go with its advantages. The major drawbacks related to the application of repeated measures design are the treatment carryover effects and practice effects (Keppel & Wickens, 2004; Mitchell & Jolly, 1992; Trotman, 1996). Treatment carryover effects have been described as "the effects or side effects of an earlier treatment on responses in later trials" (Mitchell & Jolly, 1992, p. 301). The practice effects refer to the subjects' capability to adjust their perceptions of specific scenarios when they feel that they are being presented, and restricted to, a series of similar scenarios. However, both of these disadvantages are addressed in this study by utilizing a fully crossed, 2 level, factorial experimental design, based on the repeated measures technique, to test the impact of the selected four factors of the JCGC. One of four different patterns of the factorial questionnaire was presented to each subject, these patterns differing only in the order of cases, the order being selected randomly.

"Experimental design is an area of enquiry wholly devoted to the removal of the irrelevant sources of variability for the increase of precision and, therefore, for the increase of the statistical power of tests of null hypotheses" (Cohen, 1988, p. 8). In repeated measures design, all possible levels of treatments are presented to each participant, thus leading Boniface (1995) to say it was "in general a more powerful design than the independent group design" (P. 9).

Keppel (1982, p. 58) stated "A quantitative index of the sensitivity of an experiment is its power". Coolican (2009, p. 342) said "The power of any particular statistical test is the probability of detecting the effect, if it is there, and of not making type II error. It is, therefore, the value  $1 - \beta$ ". Coolican (2009, p. 340) explained "Type II error occurs when the alternative hypothesis is true but we actually get a result that would be quite likely under the null hypothesis".

Levels of power are acceptable if they are higher than 0.8, according to Cohen (1988) and Coolican (2009). They have also classified a level of power higher than 0.8 as a gold standard for power.

"Sample size is indeed important-virtually any desired level of power can be attained in any design simply by making the samples large enough. . . For the range in which most treatment effect sizes fall, the sample size needed to attain high power of level is often much larger than what is customary, or perhaps even possible, under most circumstances of research" (Lipsey, 1989, p. 137). Sample size might be increased to reach a desired high level of power, but doing so is "generally costly and time-consuming to the researcher" (Leong & Austin, 2005, p. 307). Coolican (2009) pointed out that in experimental studies, a sample size of 25 - 30 participants is a "good idea" (p. 393). Burns (2000), Weathington et al (2010) and Trotman (1996) considered the requirement of fewer subjects in within-subjects experimental studies as one of its practical advantages<sup>38</sup>.

Within subject designs rely on the assumption of sphericity. Sphericity refers to the "condition where there is homogeneity of variance among treatment variables and the variances of their differences are also similar" (Coolican, 2009, p. 539). Violating the sphericity assumption raises the likelihood of making type II error and "missing real effects" (Coolican, 2009, p. 523).

Statistically, power is measured by the effect of size. Leong & Austin (2005) described the effect of size as "a standardized indication of the impact of the treatment effects" (p. 307). Size effects could be estimated by the use of partial eta squared ( $\eta^2$ ) as suggested by Coolican (2009, p. 530) and one of the most common method for estimating effect size, where:

<sup>&</sup>lt;sup>38</sup> Referring to 2 experimental studies that have utilized the repeated measures techniques revealed that: out of 72 mailed questionnaires, 42 were returned and analysed (Wood, 2006), and out of 54 mailed questionnaires, 25 were returned and analysed. In this thesis, out of 60 mailed questionnaires, 47 were returned and analysed.



Coolican (2009) categorized size effects with values between 0.01 and less than 0.06 as "small", size effects between 0.06 and less than 0.14 as "medium", and size effects of 0.14 or higher as "large"; this study uses these category definitions to categorize effect sizes.

This thesis also investigated the degree of self-insight among the subjects by "allocating 100 points among the cues" (Solomon & Shields, 1995, p. 151). The scale of 100 points is considered the most common technique to gather subjective data (Cook & Stewart, 1975) due to its simplicity and understandability by the subjects (Wright, 1977), and it has been employed in studies about judgment decision making in finance (Hopkins, 2009; Wood & Ross, 2006; Mear & Firth, 1987). This method generates subjective measures by requesting the participants to allocate 100 points onto the four variables based on their relative importance on their decision-making process. These subjective judgments are matched with the objective judgments obtained from the factorial experimental analysis in order to assess the level of self-insight among the subjects.

In conclusion, the use of experimental design delivered evidence for the study hypothesis H1, and evidence on the impact of relative weights of the selected variables of the JCGC (external auditors' independence, internal auditors' effectiveness, disclosure & transparency, and corporate accountability), as well as the variables' interactions, on the auditing profession in Jordan in respect to the reliability of the audit report and the level of the audit expectation gap

(which addresses research questions 1 & 2) and on the financial analysts' self-insight (addresses research questions 3 & 4).

### 4.2.5 Use of the Factorial Experiment

"A factorial design is a strong experimental design in which two or more independent variables, at least one of which is manipulated, are simultaneously studied to determine their independent and interactive effects on the dependent variable" (Johnson & Christensen, 2010, p. 306). As stated in Chapter One, the factorial experimentation technique was utilized as the most appropriate approach to answer research questions 1 & 2; these research questions aimed to test the relative importance of the study's independent variables on both the perceived reliability of the audit report and the level of the audit expectation gap in Jordan. This technique also allows the researcher to determine both the interactive and independent effects of the study variables. In the fully-crossed factorial experiment, where the researcher manipulates the factors, all possible combinations of the study variables are tested.

Potential interactive effects are considered significant in this thesis. Interactive effects may suggest that the selected factors of the JCGC may work more effectively and appropriately in certain combination rather than independently.

#### 4.2.6 Interview Method and Analysis

Qualitative methods can generate a great depth of knowledge and cover a wide range of perspectives in judgment decision making (Creswell, 2009; Nykiel, 2007). The qualitative method was described by Bryman (1988) as "an approach to the study of the social world which seeks to describe and analyze the culture and behavior of humans and their groups from the point of view of those being studied" (p. 46). One of the main methods of getting qualitative

data is qualitative interviews, also called in-depth interviews (Johnson & Christensen, 2010). Easterby-Smith, Thorpe, Jackson, & Lowe (2008) pointed out that valuable information could be achieved by conducting in-depth interviews since "this approach aims to discover the views, perceptions and opinions" (p. 142) of participants. Johnson & Christensen (2010) also added that this approach aims to "obtain in-depth information about participants' thoughts, belief, knowledge, reasoning, motivations, and feelings about a topic" (p. 202).

In this study, semi-structured interviews have been conducted to gain a depth of understanding of motives and reasoning from the financial analysts' own perspective, and to help disclose potential data that might not be gained by structured interviews. This approach also allows the participants to elaborate and emphasize specific issues and raise particular themes, and gives the researcher greater flexibility to ask follow-up questions if any interesting points arose from participants' answers (Johnson & Christensen, 2010; Bryman, 1988 & 2006; Fontana & Frey, 2003; Wilson, 2010; Collis & Hussey, 2003 & 2009; Burns, 2000).

The qualitative interviews of this thesis were conducted to obtain a depth of understanding of how financial analysts perceive and assess the influence of elements of the JCGC on the reliability of the audit report and the level of the audit expectation gap. Furthermore, interviewees were requested to provide justification and reasons for their conclusions<sup>39</sup>.

Typical interviews "last anywhere from 30 minutes to more than one hour" (Johnson & Christensen, 2010, p. 202). Effective in-depth interviews require up to ten participants (Creswel, 1998). In this study, 10 interviews were conducted, their durations ranging from 45 minutes to 67 minutes.

Creswell (1998) pointed out "Undoubtedly, no consensus exists for the analysis of the forms of qualitative data" (p. 140) while Wilson (2010) argued "there is no definitive series of steps applicable to qualitative data analysis" (p. 254). However, Wilson (2010) suggested four steps to overcome any confusion that may arise when dealing with qualitative data analysis; these steps

<sup>&</sup>lt;sup>39</sup> See Appendix 3: The Interview Protocol

start by transcribing the data, followed by a comprehensive reading to code the data and generate themes, followed by interpretation of the outcomes and, finally, writing the report.

This study's interviews were conducted in Jordan. They were audio recorded and transcribed in Arabic. However, key terms were described in both languages (Arabic and its corresponding English term). Prior to carrying out the qualitative analysis, these transcripts were fully read to get a general view of the respondents' opinions and perspectives (Warren & Karner, 2009), then the transcripts were carefully studied, focusing on themes derived from the respondents' views and feedback. These themes were categorized in order to facilitate associating them with the main variables of the study. Additionally, cross-case synthesis was employed, developing "word tables" to display the data gathered from the participants, with focus on each study variable (Yin, 2009, p. 156). Finally, the tables were analyzed to draw conclusions on the differences and similarities in the cross-case patterns in order to probe the extent of variables' effects on the financial analysts' decision making (Yin, 2009 & 2011).

### 4.2.7 Integrating Experimental and Qualitative Interview Methods

The rationale for the mixed method has been discussed in section 4.2.2. The current section extends this discussion by presenting the benefits of integrating the experimental results with qualitative interviews.

"Experimental studies allow causal relationship to be identified" (Wilson, 2010, p. 303). However, Shadish, Cook & Campbell (2002) go a step further by stating that "Causal inference, even in experiments, is fundamentally qualitative" (p. 6). Hence, integrating experimental technique with qualitative approach enhances the validity of the study.

This thesis investigates a contemporary phenomenon that has financial and economic dimensions. This thesis analyzed the institutional investors' perceptions of the JCGC's impact on

the perceived reliability of the audit report and the level of the audit expectation gap. The JCGC became operational at the beginning of 2009 and, therefore, is still a relatively new factor in the Jordanian environment. The qualitative interview approach adopted in this study focuses on 'how' and 'why' questions to deliver rich insights into the dimensions of this phenomenon and to obtain richer conclusions and enhance the understanding of the experimental study (Yin, 2009). Yin (2009) explained why 'how' and 'why' questions added value, saying "This is because such questions deal with operational links needing to be traced over time rather than mere frequencies or incidence" (p. 9). Furthermore, combining the experiment with the qualitative interview in a single study facilitates the analysis and the interpretation process of the experimental findings since the results of each component complements the other, and the qualitative interview results suggest potential explanations and reasons behind financial analysts' objective judgments (Yin, 2009).

Integrating experimental and qualitative studies increases both internal and external validity. Internal validity concerns satisfying the question: "Do the experimental treatments, in fact, make a difference in the specific experiments under scrutiny or can the differences be ascribed to other factors?" (Burns, 2000, p. 357). In other words, it refers to the validity of whether the influence observed on the study dependent variable(s) was the result of the manipulation and the variation of the study independent variable(s) rather than being caused by other factors (Shadish et al, 2002; Trotman, 1996). Repeated measure designs are associated with strong internal validity (Mitchell & Jolley, 2012; Collis & Hussey, 2003 & 2009; Coolican, 2009).

External validity, generally referred to as generalizability, is concerned with satisfying the question "Given these demonstrable effects, to what populations or settings can they be generalized?" (Burns, 2000, p. 357). Although within-subject designs are described as the most powerful experimental design (Mitchell & Jolley, 2012), experimental studies usually suffer from poor external validity due to the restrictive conditions and controls placed to confirm causal relationships (Shadish et al, 2002). However, integrating the qualitative approach with the experiment approach contributes to enhancing external validity by delivering comprehensive interpretation of the identified relationships (Yin, 2009). Consequently, this

combination improves external validity by lending statistical generalizability, through the experiment, and analytical generalizability, through the qualitative interview (Yin, 2009; Shadish et al, 2002). Moreover, Yin (2009) pointed out that triangulation improves the construct validity of the study. He defined it as "establishing correct operational measures for the concept being studied" (p. 34).

Finally, conducting an interview alone might open a door for criticisms of possible bias given that there are sometimes discrepancies between what people do and what they say, whether the differences are deliberate or unintentional. (Milne & Chan, 1999; Bouwman et al, 1987). That potential criticism could be addressed by reinforcing the qualitative findings with the objective and subjective findings of a well-designed factorial experiment. The experimental survey of this thesis consists of three main parts: Part (A) provides the objective analysis of weightings from the factorial experiment, Part (B) provides a subjective self-insight analysis<sup>40</sup>, and Part (C) is designed to gain some demographic information.

# 4.3 The Unit of Analysis - Institutional Investors

Investors are one of the main annual report user groups. Investors are considered one of the main parties involved in the audit expectation gap. Most studies conducted regarding the 'messages' carried in audit reports and the audit expectation gap have taken into consideration investors' perceptions and opinions. The in-depth understandings generated by these studies help address the question of how to enhance the message conveyed by the audit report and how to bridge the audit expectation gap (Porter, 1993; Kelly & Mohrweis, 1989; O'Reilly, 2009; Humphrey et al, 1992; Citron et al, 2008;; Al-Khadash & Al-Sartawi, 2010; Zureigat, 2010; Hajir, 2001; Al-Khadash et al, 2012; Omari, 2003)<sup>41</sup>

<sup>&</sup>lt;sup>40</sup> See Appendix 4: The Factorial Experiment Questionnaire

<sup>&</sup>lt;sup>41</sup> The latter five studies have been conducted in Jordan

The unit of analysis of this study is the financial analysts whose main job involves reviewing financial statements and accounting reports as part of making investment decisions and recommendations. The study sample was derived from financial analysts who work in investment companies in Jordan. This study surveyed financial analysts from the following types of institutional investors: banks (the investment portfolio department), insurance companies, brokerage houses, investment companies and funds.

# 4.4 Institutional and individual investors in the context of Jordan.

Two main reasons are behind the choice to use institutional investors, rather than individual investors, as the unit of analysis for this study. The first reason is based on the review of related literature on Jordan, and the second reason is to satisfy the experimental method adopted in this study.

The Jordanian literature reveals the necessity of surveying institutional investors rather than individual investors. Furthermore, Al-Khadash & Al-Sartawi (2010) surveyed institutional investors because of the difficulties of accessing individual investors in Jordan and because institutional investors are qualified, sophisticated and knowledgeable in comparison to individual investors. They argued, furthermore, that institutional investors are highly dependent on financial reports in making investment decisions. Their study revealed that the managers of investing and portfolio departments perceive the important role of the external auditors and rely on auditors' independent opinions. Consistent with that opinion, Abu-Nassar & Al-Thnaibat (2005) surveyed institutional investors on the audited financial statements and, due to their financial knowledge and experience, more capable of answering the survey questions. These viewpoints are also consistent with viewpoints of other Jordanian studies (Omari, 2003; Obaidat, 2007).

Another reason for choosing institutional investors, rather than individual investors or any other group, is that not only can they be considered expert judges, but they also influence or control a substantial share of investment decisions and are, therefore, highly influential in Jordanian financial markets (Al-Khadash & Al-Sartawi, 2010). Institutional investors have, for several decades, been playing a significant role in fostering corporate governance best practices as a means of enhancing management performance, long-term value for stakeholders and better return on their investments (Abu-Tapanjeh, 2006). In studying the relationship between ownership structure and audit quality in Jordan, Zureigat (2011b) found that institutional investors in Jordan tend to nominate "high quality auditors" (p. 38). However, the study identified high quality auditors on the basis of the size of the audit firm. That study explained the tendency of institutional investors to hire high quality auditors, typically one of the 'Big 4' audit firms, saying that it was in order to increase the credibility of the audited financial statements for the purpose of decision making.

Regarding the second reason for utilizing institutional investors in this study, respondents who understand what is being asked of them in the experiment can give informed responses. Professional groups and practitioners, such as professional accountants, auditors and financial analysts, have been recommended for the conduct of experimental studies (see, for instance, Teoh & Lim, 1996; O'Reilly, 2009). Teoh & Lim (1996) chose professional accountants in their study because they "represent a group of professionally qualified and trained individuals, capable of making informed judgments" about the effect of the independent variables on the external auditors' independence (p. 239). O'Reilly (2009) argued that expert subjects demonstrate a better understanding of experimental surveys. Financial analysts who work in institutional investment companies were chosen for that study's experiment due to the sophisticated knowledge and experience they demonstrate as prudent investors, which allows them to answer and analyze the treatments and the different scenarios provided in the factorial experimentation questionnaire quickly and more accurately (O'Reilly, 2009). Therefore, surveying institutional investors obtains informed judgment decisions regarding the impact of the adoption of the Jordanian corporate governance code on their perceptions of the reliability of the audit report and size of the expectations gap.

# 4.5 Research Implementation Strategy

The decision of adopting a relevant methodology for any study depends on the study objective and the research questions (Denzin & Lincoln, 2000; Miller & Crabtree, 1999). The mixed method approach is employed in this study to achieve the study goals. Creswell (2008) pointed out that the researchers "could collect quantitative and qualitative data separately in two phases so that data from one source could enhance or complement data from the other source" (p. 554). There are many types of mixed method research design; based on time orientation, they could be classified as concurrent or sequential (Johnson et al, 2007). The simultaneous mixed method technique was adopted in this thesis. This type of simultaneous triangulation method is described as "concurrent or parallel" (Creswell, 2008, p. 557). This implies that the data collection and analysis for both the qualitative and experimental quantitative methods were fulfilled in parallel, and the findings of the two methods integrated at the interpretation stage.

The experimental instrument, interview protocol, information letter to participants and the consent forms were all translated to Arabic and then translated back (to English). These documents have been reviewed by academicians familiar with the area of study.

Prior to conducting the fieldwork for this study, the researcher obtained the ethical approval letter from the Human Ethics Committee at The Australian Catholic University<sup>42</sup>. The data collection process was implemented based on the guidelines of the ethical approval, with formal prior consent from the subjects. Purposes, aims, and procedures of the field work have been outlined in the Information Letter to the Participant and were circulated by the Certified Financial Analysts Society in Jordan (CFA-Jordan). The figure below explains the implementation strategy.

<sup>&</sup>lt;sup>42</sup> See Appendix 5: Ethical Approval Letter from the Australian Catholic University



### Figure 6: Research Implementation Strategy

### 4.5.1 Experimental Treatments

For the factorial experimental treatments, data was collected by distributing a selfadministered survey instrument. This was conducted via postal mail. The postal questionnaire survey method is a popular method for data collection, and it has many advantages relevant to this study, such as respondents arguably provide a "more honest reply" (Wilson, 2010, p. 157) to the scenarios and questions provided due to the absence of the interviewer. Another advantage of self-administered mailed questionnaires, according to Burns (2000), is that "The respondents are free to answer in their own time and their own pace" (p. 581) Burns (2000) also added that this survey technique guarantees confidentiality. This characteristic satisfies part of the human ethics standards' requirements as the respondents feel free to either disclose or conceal their identities.

A well-established questionnaire is capable of producing accurate and reliable data, especially if the instrument is presented in a logical and clear manner (Wilson, 2010). The Arabic version of the questionnaire was distributed. The design of the front cover of the distributed questionnaire satisfied the five elements suggested by Creswell (2008) which are (1) the importance of the respondents to the study; explaining why this particular category of participants has been chosen, stating that this category is the most relevant and capable to provide effective and accurate data, or by emphasizing the perceived added value of the respondents' contribution so as to encourage them to fill in the questionnaire, (2) the purpose of the study, (3) the confidentiality, (4) the status of the sponsorship, such as the institution that supervises the research and using the university insignia and (5) the expected time needed to complete the questionnaire. In addition to the above five points, the cover letter included the instructions for the questionnaire. The 16 experimental scenarios of the study are presented in the internal pages of the questionnaire while the last two pages were designed to gather the subjective weights of the four dimensions and to gain some demographic information, with space left for any additional comments by the respondents.

The questionnaires were designed in four different variants, each differing in only one way: the random order of scenarios, the purpose of the randomization being to mitigate the carryover effects and practice effects (Keppel & Wickens, 2004; Mitchell & Jolly, 1992; Trotman, 1996). The survey instrument contained 16 different scenarios testing the influence of the selected elements of the JCGC on both the reliability of the audit report and the level of the audit expectation gap. Each scenario is made up of four dichotomous factors. The sixteen scenarios represent the fully crossed composition of the four independent variables at two levels, namely, higher and lower. A full factorial design for the four dichotomous factors requires a

total of sixteen scenarios, which is 2\*2\*2\*2. The scenarios were presented in terms of (numbered cases) case one to case 16.

The respondents were requested to consider each scenario, each being a different combination of the study's independent variables, independently of other scenarios, and requested to provide their assessment of the influence of each scenario to the perceived level of the reliability of the audit report and the level of the audit expectation gap. Participants were requested to indicate the perceived level of influence on a scale with seven levels, ranging from 'Substantially Worse' to 'Substantially Better'. The questionnaire included a filled in (completed) example scenario to illustrate the appropriate use of the instrument as shown in Figure 7.



# Figure 7: Example of the Explanatory Scenario

### 4.5.1.1 Independent Variable Definitions

Undoubtedly, the design of any survey instruments should maintain simplicity (Easterby-Smith et al, 2008; Burns, 2000; Cooper & Schindler, 2010). Slovic et al (1972, p. 286) pointed out that "the levels of the factors must be categorical" when designing an experimental survey. Only two levels for the independent variables are used in this study, the reason being to have a manageable number of treatments.

Kida, Cohen & Paquette (1990) investigated whether categorical descriptions attached to cues affect decision making and found that cue importance was significantly influenced by the selection of categories. They also pointed out that categorical descriptions assigned to cues affect the importance of those cues and make it more motivating and stimulating than static descriptions. The design of cues should also maintain consistency and unambiguity whether individually and in combinations (Slovic, 1966).

Each of the four corporate governance dimensions (the four independent variables) is assigned either a 'higher' or 'lower' rating. Higher refers to the level achieved by the top 1/3 of companies on that factor and Lower refers to a level of performance achieved by the 1/3 least successful companies on that factor. The definitions of the independent variables were provided in the front cover of the instrument and were as follows:

- External Auditors' Independence: "Freedom from those pressures and other factors that compromise, or can reasonably be expected to compromise, an auditor's ability to make unbiased audit decisions" (ISB, 2000, para. 4).
- Internal Audit Effectiveness refers to the extent to which the designated internal audit objectives and functions are achieved properly, are unbiased, and are free from management pressure that may compromise the internal auditor's performance. Examples of those designated internal audit functions are safeguarding assets against loss and theft, providing reasonable assurances that the financial and operating information are accurate and reliable, and ensuring the entity's compliance with laws and regulations.

- Disclosure and Transparency: refers to the quality and reliability of information, whether financial or non-financial, provided by the company to the shareholders and the users of the financial reporting.
- Corporate Accountability: refers to the ability of the shareholders and the stakeholders to hold the governing body of the company such as the executive management, the board of directors and the external auditors accountable and answerable to the laws, the regulations and the company's bylaws based on their power and responsibilities.

# 4.5.1.2 Dependent Variable Definitions

The two dependent variables of the study have been assessed separately based on different scenarios of the manipulated independent variables. The respondents were requested to consider each scenario in isolation. Figure 8 shows an example of one treatment taken from one of the survey's four variants whilst figure 9 shows an example of one treatment taken from one of the Arabic versions of the factorial questionnaire.

Case 1									
Lower	External Auditor's Independence	e	Highe	er Disc	losure and	l Transp	arency		
Lower Internal Auditor Effectiveness			Lowe	Lower Corporate Accountability					
Your Assessment on the Following (circle)									
	Substantially Worse Substantially Better								
Reliabi	Reliability of the Audit Report -3 -2 -1 Same +1 +2 +3							+3	
Audit Expectation Gap -3			-2	-1	Same	+1	+2	+3	

# Figure 8: Example of Treatment- English Version



### Figure 9: Example of Treatment- Arabic Version

16 cases of different combinations of independent variables were presented in Part A of the questionnaire, and the respondents were asked to provide their assessment, in each scenario, on (1) the reliability of an audit report (in comparison with the reliability of a "typical" audit report, in the participant's experience in Jordan) on a discrete scale with common intervals and (2) the size of the audit gap (in comparison with the "typical" gap, in the participant's experience in Jordan) on a discrete scale with common intervals.

The scale on which respondents have to indicate their assessments consisted of seven points, ranging from '-3' through 'same' to '+3', where '-3' refers to substantially worse, '+3' refers to substantially better and 'same' was positioned in the middle to allow respondents to demonstrate neutral assessment. The design of the treatments was simple, as suggested by Dillman (2006). For statistical analysis, using SPSS software version 19, these 7 levels were coded from 1 (Substantially Worse) to 7 (Substantially Better).

Part B of the questionnaire requested the respondents to provide their subjective assessment of the weighted influence of the four independent variables on (1) the reliability of the audit report and (2) the size of the audit gap, allowing comparison with the findings of their objective judgments.

The definitions of the dependent variables were provided in the front cover of the instrument and were as follows:

- The audit expectation gap: The audit expectation gap refers to the different perceptions between the auditors' actual performance and to what report users believe the auditors should be performing and achieving as auditors.
- Reliability of the audit report: refers to the extent to which the institutional investor can rely on the audit report to clearly convey significant instances of poor internal controls; accounting errors, biases, or omissions; and concerns regarding the application of the going concern assumption.

# 4.5.2 In-depth Interviews

It has been argued that quantitative studies do not answer 'how' and 'why' questions regarding the phenomenon under study while qualitative studies do (Silverman, 2009; Symon & Cassel, 1998; Denzin & Lincoln, 2000). 'How' and 'why' questions deliver in-depth understanding of the topic and provide exploratory tools that help to gain a clear and rich picture of the process (Symon & Cassel, 1998). "'What' questions, 'who' and 'where' questions ... are likely to favor survey methods .... In contrast 'how' and 'why' questions are more explanatory and likely to lead to the use of case studies, histories and experiments as a preferred research method" (Yin, 2009, p. 9).

The qualitative interview of this study was conducted according to the interview protocol<sup>43</sup> attached in Appendix 7. The design of the interview protocol has been developed based on the related literature review and the main theory employed in this study which is the agency theory. The interview protocol concentrated on 'how' and 'why' questions aiming at investigating how institutional investors' assessment of the selected elements of the JCGC affects their perception of the reliability of the audit report and the level of the audit expectation gap. Subjects were requested to justify their conclusions by 'why' questions.

### 4.5.3 Pretesting: Survey Instrument & Interview Protocol

The design of any survey instrument needs to be pretested in order to ensure that ambiguities, vagueness, and grammatical errors have been avoided, and ensure that the language is understood and the layout of the experiment presented smoothly to the potential participants (Dillman, 2006). Collis & Hussey (2003) also say that pretesting experimental instruments is important.

The experimental instrument of this study was pretested with the help of six financial analyst practitioners. Pretesting resulted in two amendments to the survey instrument. The first amendment was to split the last requirement of part C (which requests subjects to list any other related factors, other than the Jordanian Corporate Governance Codes, that they think would be relevant to their assessment) into two sections, one for each of the dependent variables, i.e. to state if there were any factors that would affect their perception of the reliability of the audit report and then any other JCGC factors that would affect their perception of a brief definition of the Jordanian corporate governance code.

<sup>&</sup>lt;sup>43</sup> Appendix 7 shows the interview protocol of this study. The protocol starts by briefing the contribution of the study. However, beside the request of unprompted impression of what does corporate governance means to the participants and the main 8 questions, there were some questions (follow up & prompt) raised during the interviews. Chapter 5 shows some of them.

Similarly, three preliminary interviews were conducted with financial analyst practitioners. The results of these preliminary meetings contributed to enhancing the definitions of one of the independent variables (internal auditor independence)<sup>44</sup>. Moreover, the participants suggested defining the key terms of the study in English as well as Arabic, a suggestion later implemented in the study.

### 4.5.4 Data Collection process

Prior to conducting the field work for this study, an Information letter to Participants<sup>45</sup> was prepared and distributed to the potential participants, as per the Human Ethics Committee's regulations at the Australian Catholic University (ACU). The information letter to Participants has a particular format designed by ACU, and it aims to communicate to the potential participants some basic information about the study, the research's confidentiality and participants' rights.

The Information letters to Participants were advertised and circulated to the members of the Professional Association of Financial Analysts in Jordan (CFA-Jordan). The interviews were arranged with nine persons who telephoned to volunteer, and a tenth person was suggested by one of the interviewees. Interview times and locations were discussed during those Initial phone calls. All interviews were face-to-face and have been conducted in Amman (the capital city of Jordan). In Jordan, most large companies have their headquarters in Amman. Interviews were audio recorded and conducted at the participants' offices, as per their requests.

<sup>&</sup>lt;sup>44</sup> Interestingly, one participant (in pretesting period) pointed out that the term 'management pressure' may confuse participants: it could mean pressure being put on the internal auditors to make them work harder and to their maximum capacity, or it could mean pressure that aims to jeopardize internal auditors' effectiveness by affecting their reports. This led the researcher to reword the definition of internal auditors effectiveness to make it clear and consistent with the objectives of the study.

<sup>&</sup>lt;sup>45</sup> The form of the Information letters to participants has been downloaded from the Website of the Australian Catholic University. See appendix 1

Addresses and Information about Institutional Investors, as for all types of companies in Jordan, are available for access through the website of the Companies Control Department (CCD) and the Amman Stock Exchange (ASE). The CCD website provides data bases and statistical information about all the Jordanian companies, and the website allows individuals to make queries and search for any type of company using various search criteria such as company name, purposes, capital, etc, (CCD: Companies Inquiry, 2012; ASE: Broker Members, 2012).

As mentioned earlier, experimental studies that use repeated measure techniques require few participants and deliver robust results (Coolican, 1994; Burns, 2000; Weathington et al, 2010; Trotman, 1996). Institutional investors in Jordan have been identified with the help of the CFA-Jordan, CCD and ASE. 60 institutional investing companies were randomly drawn from the list of commercial banks<sup>46</sup>, insurance companies, brokerage houses, and investment funds and companies. Data collection packages containing the survey instrument, the information letter to participants, and a reply envelope with a stamp attached. The packages were distributed to the target sample. 47 usable survey instruments were received and included in the analysis. The response rate was thus 78%.

# 4.6 Methodological Limitations

There are inherent limitations accompanying the methodology adopted in this study due to the fact that the survey instrument were self-administered and were filled out in an uncontrolled environment. This means that instruments might not have been completed by the potential subjects, might not been completed by the subjects independently, or, in the absence of the researcher and his unavailability to answer questions, may not have been completely understood by the participants (Trotman, 1996). In regards to the latter problem (absence of the researcher), this study's research implementation strategy included pretesting of the survey

<sup>&</sup>lt;sup>46</sup> Non-commercial banks were excluded from the study population and left as suggestions for future studies such as (Cities and Villages Development Bank, Industrial Development Bank, Islamic Banks.)

instrument, an approach that helped mitigate the potential for confusion, although the researcher's and the supervisors' contacts (including e-mails and telephone) were presented in the last part of the survey instrument for any clarification that subjects may need. While the first issue (surveys completed by someone other than the selected participant) is unlikely to happen, it is, like in all studies that use self-administered surveys, a possible threat to the validity of this study. Finally,

Another methodological issue is the time and cost restrictions. According to Johnson & Onwuegbuzie (2004) conducting a mixed method research is costly and requires more time than what a single method needs. This is true especially when taking into consideration that the fieldwork study of this thesis was conducted in Jordan (which is another country where the researcher studies in Australia). However, the employment of the concurrent (simultaneous) mixed method strategy mitigated these difficulties.

# 4.7 Chapter Summary

The choice of the appropriate methodology that best addresses the study objective and the research questions is a very crucial decision when researchers plan their studies. This chapter provided an overview of methodology adopted in this thesis, which is the mixed method approach, and also provided the rationale and a justification for the selected method. The experimental survey design and the qualitative interview protocol were discussed in this chapter. The next two chapters discuss and analyze the data collected from both the experiment and the interview in order to develop the study's conclusions and recommendations.

# **Chapter 5: Results & Analysis**

### **5.1 Introduction**

This chapter describes the respondents, and presents separately the findings and analysis of the data obtained after conducting the fieldwork for both the experimental survey and the personal interviews. The quantitative and qualitative data are then integrated in section 5.4.

### **5.2 Experiment Results**

This section presents the findings of the quantitative stage of the study. The results of the experimental survey were analyzed using the SPSS package version 19. The quantitative findings relate to the three main parts of the factorial questionnaire which are considered part A (the objective judgments of the participants), part B (the subjective judgments and the check on the importance of the experimental variables) and part C (particulars on the relevant backgrounds of participants).

### 5.2.1 Description of Response Rates and Respondents

As mentioned in the previous chapter, 47 survey instruments were received and they all were usable giving a 78% response rate. In reviewing recent Jordanian studies that investigated or

surveyed institutional investors such as Al-Khadash and Al-Sartawi (2010); Al-Khadash et al, 2012; Ishhadat, & Abed Al-Jaleel (2012) the response rates of these studies were 63%, 68%, 94% respectively, which suggest that response rate of this study is reasonable in relation to this sector.

The following tables (2, 3 and 4) provide descriptive statistics about the institutional investors responding to the survey regarding their sectors, positions and experiences. The tables reveal that the four major types of institutional investors were adequately covered, and the positions of the financial analysts vary from the manager to the head of the section to the financial analyst. Two respondents chose the other category and indicated that they were an investment consultant and a credit officer respectively. The latter has been accepted for the study analysis since the main job of credit officers in Jordan is to evaluate the financial position and the audit reports of their clients when considering credit facility decisions (Al-Zebdieh & Al-Thuneibat, 2012; Lutfi, 2001; Maswada, 1992). Finally, with 12 years of experience on an average and 11 years being the median, and no one having less than 5 year experience, the subjects of this study could be considered as experienced.

Sector	Frequency
Bank	7
Insurance Company	8
Brokerage House	9
Investment fund/Company	23
Total	47

### Table 2: Respondents' Distribution by Sectors

Position	Frequency
Manager - of the investment portfolio department	16
Head of the investment portfolio section	12
Financial analyst/officer	16
Credit Officer *	1
Investment Consultant *	2
Total	47

# Table 3: Respondents' Distribution by Positions

\* This position was not listed in the survey instrument and has been indicated by the subject as they were allowed to specify another job position if not mentioned.

# Table 4: Respondents' Distribution by Experience

Respondents' Experience	Years
Mean	12
Median	11
Std. deviation	5.4
Minimum	5
Maximum	27

### 5.2.2 The Validity of the Survey Experiment

Part B of the survey instrument aimed to test the validity of using the chosen JCGC factors as well to investigate whether they are considered as the major effective significant factors institutional investors' perceptions of the reliability of the audit report , and the size of the audit expectation gap separately, or if there are some other factors that need to be included in the survey. Participants were requested to indicate on a 1 to 7 Liker scale their level of confidence that the four factors examined in this thesis capture the major determinants of the reliability of the audit report and the size of the audit expectation gap. Figure 10 below shows the questions asked, while the results of their answers are shown in Table 5. A subsequent question requested the respondents to list any other possible related variables from the JCGC they consider relevant to their assessment of the dependent variables separately.

# Figure 10: Level of Confidence that the Selected Variables of the JCGC Represent the Major Variables in terms of their Impact on the Dependent Variables

Please indicate by circling a number on the scale below how confident you feel that the four factors together represent the major determinant of your assessment of the:

(1) Reliability of the audit report	Lower Confidence	1	2	3	4	5	6	7	Higher confidence
(2) Audit expectation gap	Lower Confidence	1	2	3	4	5	6	7	Higher confidence

Dependent variable	Mean	Median	Mode
Reliability of the audit report	5.4	6	6
Audit expectation gap	5.6	6	5

# Table 5: Confidence Level of the Factors

The mean of the confidence level indicated by the 47 subjects' assessments of the reliability of the audit report was 5.4, and 5.6 for the audit expectation gap, where the median for both of the dependent variables was 6. This level of confidence as indicated by the respondents suggests that the model of the study is reasonably strong in capturing the influence of the selected factors of the Jordanian corporate governance code on both the reliability of the audit report and the level of the audit expectation gap.

Table 6 shows the additional variables from the JCGC suggested by the participants and the frequency for each dependent variable.

# Table 6: Frequency of Mention of other Factors Seen as Having an ImportantImpact on the Dependent Variables

Factors	Reliability of the audit report	Audit expectation gap		
The role of the Jordanian Association of Certified Public Accountants	2	3		
Educating the users of the audit report		3		
Audit fees	4	1		
International standard on auditing (ISA)	1			
Audit rotation	3	1		

The above table reveals that the factor that generated the highest number of mentions is the audit fees affecting the reliability of the audit report. It is acknowledged that the factors generating multiple citations for a dependent variable generally do contribute to narrowing the audit expectation gap and in increasing the reliability of the audit report as revealed in the prior literature (Ojo, 2009; Siddiqui et al, 2009; Dickins and Higgs, 2009; Rehana, 2010; Sikka et al, 1992; Sidani and Olayan, 2007; Haniffa & Hudaib, 2007; Fadzly & Ahmad, 2004; Pierce & Kilcommins, 1997; Wolf et al, 1999). Therefore, since they are not derived from the JCGC, they were mentioned in the limitations of the study as other factors that may affect institutional investors' perception of the reliability of the audit report as well as the level of the audit expectation gap.

With regards to the other mentioned factors above, it is worth mentioning that they have been reviewed in Chapter 3 as constituents of external auditor's independence; the audit partner rotation issue was discussed in Section 3.7.4, while audit, fees and remuneration were discussed in Section 3.7.1<sup>47</sup>.

Finally, the assumption of sphericity was checked using the 'Muachy's test'. Field (2009) pointed out that the Muachly's test, "tests the hypothesis that the variances of the differences between conditions are equal" (p. 460), while Coolican (2009) pointed out that the lack of sphericity results in increasing the likelihood of making a type II error. Muachly's test statistic was not significant suggesting the condition of the sphericity was met (Field, 2009).

<sup>&</sup>lt;sup>47</sup> A possible explanation of mentioning audit fees by the subjects could be attributed to the fact that the JCGC did not discuss clearly the role of the audit committee in setting external audit fees, it has generally stated the role of the audit committee of discussing matters related to the nomination of the external auditors. In regard to the audit rotation, it noteworthy to mention that audit rotation in Jordan is a Guideline Code (not imperative) according to the JCGC; listed companies must comply or provide explanations in case of inability to comply.

### **5.2.3 Factor Weightings**

An initial MANOVA test was conducted, and proved significant, thus the two dependent variables of the study were analyzed by using individual ANOVA tests (Meyers, Gamst, & Guarino, 2005; Coolican, 2009; Field, 2009).

# 5.2.3.1 Factor Weightings on the Reliability of the Audit Report

The analysis of the gathered data from the experiment provided evidence in addressing Research Question 1.

# RQ. 1

What are the relative main and interactive weights of:

- External auditor independence;
- Internal auditor effectiveness;
- Disclosure and transparency, and
- Corporate Accountability

in the perceived reliability of the audit report.

In answering this question, subjective and objective measures used to obtain evidence are:

1- Self-reported weights, (the subjective measure) where each financial analyst was requested to allocate 100 points among the selected key dimensions of the JCGC.

2- The effect size (the objective measure) was measured by calculating each of the selected key dimensions of the JCGC measure's main effect and those interactions were found to be

significant. Effect of size is measured through using partial eta-squared which in its turn shows that the proportion of variance explained by each of the four independent variables.

The relative weights of the Jordanian corporate governance code's related factors in contributing to the reliability of the audit report are presented in Table 7.

	External Auditor's Independence	Internal Auditor Effectiveness	Disclosure & Transparency	Corporate Accountability	
Self-Reported Weights (%)					
Mean	40.14%	28.36%	15.05%	16.45%	100%
SD	8.619	9.185	7.159	7.435	
Range	20%-60%	10%-45%	5%-30%	5%-30%	
n=47					
Rank	1	2	4	3	
Effect Size ‡					
Main Effects	22.71%	18.59%	15.70%	9.66%	66.66%
Interactions ‡‡	7.26%	9.19%	8.70%	8.19%	33.34%
Combined Effects	29.97%	27.78%	24.40%	17.85%	100.00 %
n=47					
Rank	1	2	3	4	

# Table 7: Factor Weightings for the Reliability of the Audit Report

**‡** Effect of size is measured by using partial eta-squared

**‡‡** Interaction effects are allocated to each factor involved in the interaction equally.

The figure below shows the Comparisons of the subjective and objective (experimental scenarios) weights in relation to the reliability of the audit report.



Figure 11: Weights of Factors' Influence on the Reliability of the Audit Report

The self-reported weights show a wider distribution between the four independent variables on their effects on the reliability of the audit report; the external auditor's independence factor showed the highest effect followed by the internal auditor's effectiveness, then the corporate accountability which slightly exceeded the disclosure and transparency factor.

The objective measure (the effect size) has a tighter range with regard to the ranked order of importance of factors. The Effect Size suggests that the external auditor's independence gained the highest rank among the other selected factors from the JCGC on the effects on the perceived reliability of the audit report followed by internal auditor's effectiveness, while the corporate accountability had the least contribution. The results obtained from the two measures show that the major influential factors on the perceived reliability of the audit report i.e. external auditor's independence and internal auditor's effectiveness gained the same

ranking order suggesting that Jordanian financial analysts demonstrated a relatively high degree of self-insight.

The detailed Effect Size analysis (Table 8) shows large and statistically significant (at the 1% level) main effects for each of the four variables selected from the JCGC on their individual effect on the reliability of the audit report. The external auditor's independence is the highest followed by the internal auditor's effectiveness, disclosure and transparency, and corporate accountability respectively. The statistical power for all of the four variables is very high and classified as 'gold' standard for power, suggesting there is little likelihood of Type II errors (Cohen, 1988; Coolican, 2009).

With regard to the interactions between the posited factors, the interactions between (External auditors independence \* Internal auditors effectiveness) and (External auditors independence \* Internal auditors effectiveness \* Disclosure & Transparency) and between (Internal auditors effectiveness \* Disclosure & Transparency \* Corporate accountability) reveal large and statistically significant (at the 1% level) main effects and high statistical power. However, the interactions of (External auditors independence \* Corporate accountability) and (External auditors independence \* Corporate accountability) and (External auditors independence \* Internal auditors effectiveness \* Corporate accountability) and (External auditors independence \* Internal auditors effectiveness \* Corporate accountability) and between (External auditors independence \* Internal auditors effectiveness \* Corporate accountability) and between (External auditors independence \* Internal auditors effectiveness \* Corporate accountability) and between (External auditors independence \* Internal auditors effectiveness \* Disclosure & Transparency \* Corporate accountability) show significance at the 5% level and have a strong power level, but failed to attain a magnitude sufficient to be satisfactorily classified as large, therefore classified as moderate according to Coolican (2009).

The finding of the large, statistically significant main effect sizes for all of the four variables plus the findings of moderate and large, statistically significant interactions between the variablesmentioned above - under these controlled conditions supports and further confirms the validity of the study model and the presence of a causality relationship between the posited elements of the JCGC variables and the reliability of the audit report.

Factor	Effect Size ‡	P Value	Power
Main Effects			
External Auditor independence (Ext)	0.933**	0.00*	1.00***
Internal Auditor Effectiveness (Int)	0.764**	0.00*	1.00***
Disclosure & Transparency ( <b>D &amp; T</b> )	0.645**	0.00*	1.00***
Corporate Accountability (Acc)	0.397**	0.00*	1.00***
Total Main Effects	2.739		
Interactions			
Ext * Int	0 .178**	0.003*	0.87***
Ext * D & T	0.075	0 .060	0.47
Ext * Acc	0.106	0.024~	0.62
Int * D & T	0.031	0.235	0.22
Int * Acc	0.011	0.481	0.11
D & T * Acc	0.129	0.012~	0.73
Ext * Int * D & T	0.170**	0.004*	0.85***
Ext * Int * Acc	0.091	0.038~	0.55
Ext * D & T * Acc	0.008	0.549	0.09
Int * D & T * Acc	0.457**	0.00*	1.00***
Ext * Int * D & T * Acc	0.113	0.020~	0.66
Total Significant Interactions at alpha 0.05	1.319		
Total Interactions	1.369	]	
Sum of the Effect Sizes	4.108	]	

 Table 8: Effect Sizes for the Reliability of the Audit Report

\* Significant at alpha = 0.01

~ Significant at alpha = 0.05

\*\* An effect size equals or exceeds 0.14 can be considered large according to (Coolican, 2009)
‡ Effect size is measured by partial eta squared

\*\*\* exceeds the 0.8 considered the 'gold' standard for power (Coolican, 2009; Cohen, 1988)

The analysis of table 8 also shows that one-third of the overall effect size in relation to the perceived reliability of the audit report are attributed to interaction terms between the variables, this result also further confirms and supports Hypothesis 1.

Hypothesis 1 is further supported by the existence of the statistically large and significant interaction effect between (External auditors independence \* Internal auditors effectiveness) and (External auditors independence \* Internal auditors effectiveness \* Disclosure & Transparency) and between (Internal auditors effectiveness \* Disclosure & Transparency) and between (Internal auditors effectiveness \* Disclosure & Transparency).

In addition to evidencing configurable cue processing, these findings are particularly important because they indicated that the external auditor's independence, when considered interactively with internal auditor's effectiveness is a relevant factor in affecting the reliability of the audit report. The findings also revealed that external auditor's independence when considered interactive with disclosure and transparency as well as corporate accountability, are also considered relevant factors in influencing the reliability of the audit report.

Interestingly, the largest interaction showed that even internal auditor's effectiveness when considered interactively with both disclosure and transparency and corporate accountability are also relevant factors in influencing the reliability of the audit report. This finding is particularly noteworthy due to the fact that in this interaction, in which corporate accountability was involved, accounted for 0.457 of the effect size on the reliability of the audit reports is higher than corporate accountability in isolation (as noted at 0.394). However, this interaction implies that the Jordanian financial analysts perceive that internal auditor's effectiveness, the level of the disclosure, and transparency and the nature corporate accountability system that applied on all governing bodies (including external auditors) establish a combined cue that might be termed as 'internal procedures within companies' since

this combination contributes significantly to the reliability of the audit reports. This could be further explained and possibly attributed, based on the interviews and the literature, to the reliance of the external auditors on the work performance and objectivity of internal auditors who were considered the main defense line against fraud, errors and malpractices. Moreover, the subjects pointed out that giving internal auditors direct access to report to an independent and well-established audit committee<sup>48</sup> reinforces their effectiveness and, accordingly, increases the accuracy and the reliability of the financial reporting system (Pickett, 2010; Goodwin, 2003; Qin, 2007; Goodwin & Yeo, 2001; Thnaibat and Shunnaq, 2010). Furthermore, the subjects, based on the interviews, added that a strong corporate accountability system makes all governing bodies (including external and internal auditors) work in the best interest of the company and produce reliable reports that enjoy high levels of disclosure and transparency. Another possible interpretation is that the consistent adoption of positive attributes of governance conveys genuine interest in good corporate governance.

As a summary, the interactive effects may possibly indicate that the financial analysts look for clear messages namely unambiguous signs. The largest main effect they are looking for is to place more confidence in the audit report is a strong independent auditor. The effectiveness of that external audit can primarily be supported by a quality internal audit section (hence the Ext \* Int interaction being the second largest significant term), where both external, and internal audit may indicate that the high assurance of reliable, and adequate disclosure are provided to the outside investors (hence the Ext \* Int \* D & T being the second largest significant term). The other issue they might be considering would be the importance of the management places on governance internally. This would be more likely if the quality of the internal audit, disclosure level, and corporate accountability appear to be uniformly strong and that would be picked up by the three way interaction of Int \* D & T \* Acc. This of course was the highest significant interaction term.

<sup>&</sup>lt;sup>48</sup> Strong audit committee is designed to achieve the appointment of an independent auditor and to increase the likelihood of the internal audit being effective. The current study restricted itself to external audit independence and internal audit effectiveness to achieve clarity of relationships. However given the success of the current experiment the relationship between audit committee independence and external and internal audit quality and performance could be a follow up study.
# 5.2.3.2 Factor Weightings on the Level of the Audit Expectation Gap

The analysis of the gathered data from the experiment also provided evidence in addressing Research Question 2.

RQ. 2	
	are the relative main and interactive weights of: External auditor independence;
*	Internal auditor effectiveness;
*	Disclosure and transparency, and
*	Corporate Accountability
	on the level of the audit expectation gap.

Likewise, the impact on the reliability of the audit report, as shown in the previous section, both subjective (Self-reported weights) and objective (effect size) measures were used to address research question 2.

The relative weights of the Jordanian corporate governance code's related factors in contributing to narrowing the audit expectation gap are presented in Table 9.

# Table 9: Factor Weightings for the Audit Expectation Gap

	External Auditor's Independence	Internal Auditor Effectiveness	Disclosure & Transparency	Corporate Accountability	
Self-Reported Weights (%)					
Mean	21.77%	13.78%	28.76%	35.69%	100%
SD	8.089	6.763	8.491	8.067	
Range	5%-40%	3%-30%	15%-53%	20%-55%	
n=47					
Rank	3	4	2	1	
Effect Size ‡					
Main Effects	16.03%	14.32%	22.94%	23.61%	76.90%
Interactions ‡‡	4.24%	4.70%	7.12%	7.04%	23.10%
Combined Effects	20.27%	19.02%	30.06%	30.65%	100.00%
n=47					
Rank	3	4	2	1	

**‡** Effect of size is measured by using partial eta-squared

**‡‡** Interaction effects are allocated to each factor involved in the interaction equally.

The figure below shows the Comparisons of the subjective and objective (experimental scenarios) weights in relation to the audit expectation gap.



Figure 12: Weights of Factors' Influence on the Level of the Audit Expectation

Gap

The self-reported weights show a wider distribution between the four independent variables on their effect on the level of the audit expectation gap as perceived by financial analysts. Both the corporate accountability and the disclosure, and transparency factors gained the highest influence on reducing the audit expectation gap followed by external auditor's independence and the internal auditor's effectiveness respectively. On the other hand, the objective measure (the effect size) showed a tighter distribution of the effects of the selected JCGC factors on reducing the audit expectation gap. The Effect Size suggests that the corporate accountability gained the highest rank among the other factors which slightly exceeded the disclosure and transparency factor which are then followed by external auditor's independence and the internal auditor's effectiveness respectively. The two sets of the measures are consistent in their ranked order of importance of selected key JCGC factors on the level of the audit expectation gap suggesting that Jordanian financial analysts demonstrated a high degree of self-insight. The detailed Effect Size analysis (Table 10) shows large and statistically significant (at the 1% level) main effects for all of the four variables selected from the JCGC individually on their effect on the size of the audit expectation gap, where corporate accountability is the top followed by disclosure and transparency, external auditor's independence and the internal auditor's effectiveness respectively. The statistical power for all of the four variables is very high and classified as 'gold' standard of power, suggesting there is little likelihood of Type II errors (Cohen, 1988; Coolican, 2009).

In regard to the interactions between the posited factors, interactions between (Corporate accountability \* Disclosure & Transparency) and (External auditors independence \* Internal auditors effectiveness) and between (External auditors independence \* Disclosure & Transparency \* Corporate accountability) reveal large and statistically significant (at the 1% level) main effects and high statistical power. However, the interactions of (Internal auditors effectiveness \* Corporate accountability) and between (Internal auditors effectiveness \* Disclosure & Transparency \* Corporate accountability) show significance at the 5% level and have a strong power level but failed to attain a magnitude sufficient to be satisfactorily classified as large and therefore classified as moderate according to Coolican (2009).

The finding of large, statistically significant main effect sizes for all of the four variables plus the findings of moderate and large statistically significant interactions as indicated between the variables, under these controlled conditions supports and further confirms the validity of the study model and the presence of a causality relationship between the posited JCGC variables and the level of the audit expectation gap.

The analysis of Table 10 shows that approximately one-quarter of the overall effect size in relation to the level of the audit expectation gap are attributed to interaction terms between the variables, this result also further confirms and supports Hypothesis 1.

Factor	Effect Size ‡	P Value	Power
Main Effects			
External Auditor independence (Ext)	0.608**	0.00*	1.00***
Internal Auditor Effectiveness (Int)	0.543**	0.00*	1.00***
Disclosure & Transparency (D & T)	0.870**	0.00*	1.00***
Corporate Accountability (Acc)	0.895**	0.00*	1.00***
Total Main Effects	2.916		
Interactions			
Ext * Int	0 .143**	0.008*	0.77
Ext * D & T	0.074	0.061	0.47
Ext * Acc	0.001	0.839	0.06
Int * D & T	0.040	0.173	0.27
Int * Acc	0.105	0.025~	0.62
D & T * Acc	0.262**	0.000*	0.97***
Ext * Int * D & T	0.000	0.958	0.05
Ext * Int * Acc	0.000	0.972	0.05
Ext * D & T * Acc	0.143**	0.008*	0.77
Int * D & T * Acc	0.091	0.037~	0.56
Ext * Int * D & T * Acc	0.016	0.395	0.13
Total Significant Interactions at alpha 0.05	0.744		
Total Interactions	0.875		
Sum of the Effect Sizes	3.791		

# Table 10: Effect Sizes for the Audit Expectation Gap

\* Significant at alpha = 0.01

~ Significant at alpha = 0.05

\*\* An effect size equals or exceeds 0.14 can be considered large according to (Coolican, 2009)‡ Effect size is measured by partial eta squared

\*\*\* exceeds the 0.8 considered the 'gold' standard for power (Coolican, 2009; Cohen, 1988)

Hypothesis 1 of the study is further supported by the existence of the statistically large and significant interaction effect between (Corporate accountability \* Disclosure & Transparency) and (External auditors independence \* Internal auditors effectiveness) and between (External auditors independence \* Disclosure & Transparency \* Corporate accountability). In addition to evidencing configurable cue processing, these findings are particularly important because they indicate that external auditor's independence, when considered interactively with internal auditor's effectiveness, is a relevant factor in affecting the level of the audit expectation gap accounted for 0.143. Similarly and at an effect size of 0.143, the Jordanian financial analysts are combining external auditor's independence, disclosure and transparency and corporate accountability to form a combined cue that is relevant to their perception on the effect on the level of the audit expectation gap.

The largest interaction as measured by effect size was noted between corporate accountability and the level of disclosure and transparency, accounting for 0.262 of effect size on the audit expectation gap. The Jordanian financial analysts perceived that these two factors form a combined cue which might be termed 'accountability for disclosing adequate information'. The analysis of the interviews suggests the importance of this decision cue. According to the interviews, lack of accountability encourages top executive management to act in their own personal interest, at the expense of the company's overall interest. It may also lead corrupt staff to steal company assets and, consequently, may lead to company insolvency and collapse. The interviews revealed that the level of disclosure and transparency has an inverse relationship with the size of the audit expectation gap. Interviews suggested that more control procedures should be applied by the Jordanian securities commission to ensure that Jordanian listed companies implement the disclosure instructions in the best way possible. In summary, if both corporate accountability and the level of disclosure and transparency are high then as a result it gives a greater confidence that the company management and the board of director accept their responsibilities to outside investors by disclosing transparent and adequate financial data which accordingly reduces the audit expectation gap. Additionally, in the absence of the management commitment to corporate governance there is a reliance on a strongly independent auditor to lower the audit expectation gap hence the common factor in the second and third statistically large and significant term was external auditor's independence whether with the need for the external auditor to be supported by an effective internal audit or by ensuring the provision of adequate disclosure and transparency and the implementation of strong corporate accountability system.

#### **5.3 Interview Data**

This section presents the qualitative interview data obtained from the subjects. The section starts with describing the ten participants' background and their unprompted impression of how they perceive corporate governance, and then that is followed by a cross case comparison section in which the main themes of the outcomes of the interview discussion are presented. Additionally, direct quotations<sup>49</sup> by subjects are also presented to highlight the importance of some issues that where relevant.

<sup>&</sup>lt;sup>49</sup> According to Harvard APA bibliographic guide "When quoting (or paraphrasing) from a publication written in a language other than English, the quotation should be faithfully translated into the English language" See (ENQUIRE GUIDE TO HARVARD APA STYLE BIBLIOGRAPHIC REFERENCING. P. 5) accessed on October, 2012 from: <u>http://www.nottingham.ac.uk/shared/shared\_enquire/PDFs/ENQUIREreferencingguide.pdf</u>. To ensure that these quotations are presented faithfully, the quotations were translated to Arabic and then a back translation step (to English) also had been implemented by academics familiar with the area of the study.

### **5.3.1 Interview Summaries**

Interviews were conducted with the respondents based on the interview protocol. Prior to conducting the interviews, subjects were requested to indicate their experience, job title or position and the type of company in which they work in, similar to the demographic information of the survey instrument. This section provides a brief description of the respondents' financial industry experience and presents their perceptions of what corporate governance means to them.

#### Subject A

Subject A was a financial analyst at an investment fund specializing in the Jordanian capital market and had been in this industry for around 18 years. Subject A started the discussion by referring to the emergence of corporate governance in the nineties of the last century and the several translations of this term to Arabic<sup>50</sup>. Corporate governance as a phenomenon became the focus of attention by financial markets worldwide due to the dramatic collapses and scandals that occurred in the world since the Asian financial crisis in the middle of the nineties of last century and the scandals of the big European, and American corporations. Most of these scandals occurred due to manipulation of the accounts, corruptions committed by insiders and due to the lack of adequate and fair disclosure and transparency. Therefore, there was a need for a tool that works to reduce those financial problems and guarantees that all parties involved in corporate governance work in the best interest of the company. Subject A affirmed that the most important benefits that effective corporate governance should deliver are adequate disclosure and transparency along with ensuring the applicability of corporate accountability. Subject A added that these two elements are essential to ensure equitable treatments amongst shareholders.

<sup>&</sup>lt;sup>50</sup> All of the different translations of corporate governance to Arabic that were mentioned by Subject A were discussed in chapter 2 except for the translation to "Rational management". However, when asked about the better or preferable translation that best describe the term, Subject A replied "Hakemeya or Tahakkom".

Corporate governance should distinguish the tasks, responsibilities and duties of the company management and the board of directors, and this identification and separation of roles is what needs to be activated in Jordan. "*The role of the executive management is to administer the company while the role of the board of directors is to control and to ensure that executives are held accountable*". Accountability was described by Subject A as the shareholders and the stakeholders' right of holding the company management accountable for their actions and decisions through the board of directors.

#### Subject B

Subject B's position was a manager of an investment portfolio department. Almost 25 years of experience were in the investment field in different sectors where the last seven years were in an insurance company. Subject B pointed out that corporate governance gained growing interest recently in developed and developing countries alike. The presence of corporate governance best practices becomes an essential for any company to get facilities and to compete in the contemporary free market.

Subject B believes that, "theoretically" the Jordanian corporate governance code is advanced and close to those codes adopted in developed countries, but the most crucial drawback of these codes is on its weak implementation by the shareholding companies. Subject B provided two examples of the weaknesses of the compliance with the JCGC, the first one is the multimembership by some of the listed companies' board of directors, which should not exceed five memberships in different listed companies for each person. Moreover, there is weak compliance to one of the JCGC policies of the disclosure and transparency, that each listed company has to use its website to provide relevant, on-time and adequate disclosures.

#### Subject C

Subject C was head of a portfolio section in a Jordanian Bank, and worked in this area for 18 years. Subject C perceived that corporate governance aims to ensure companies are directed in

the interest of its owners. "There are many studies associated with corporate governance best practices adopted in certain companies with the quality of the financial reporting for those companies since it becomes the main guidance for decision making".

Subject C pointed out that financial statements are prepared by the company management according to the International Financial Reporting Standards, although these standards aim to secure the financial reporting, ensure that the financial statements are free of errors and biases and that they faithfully presented, *"these standards give the management some flexibility to choose between some alternative accounting methods"* which might be used by executives to achieve personal interests at the expense of the shareholder and stakeholders interests, which contributes to further corporate collapses, and that's why the adoption of effective corporate governance is always important for any company. Subject C also affirmed the vital role that audit committees play to enhance the soundness of the company's financial reporting, and asked to *"strengthen corporate governance by reinforcing and empowering audit committees' effectiveness"* and to focus on its role as a supervisory committee. Moreover, Subject C added that independent and experienced audit committees are more capable of protecting companies' financial accounts and to detect errors, frauds, and distorted reports.

#### Subject D

Subject D was also head of a portfolio section in a Jordanian company specializing in securities investment, and worked in that position for 14 years. Subject D added that the main company activity is to invest in securities in Jordan and real estates in Jordan, also the company's investment in securities outside Jordan are mainly in the Arabian Gulf area market. Subject D highlighted the importance of corporate governance in enhancing disclosure and transparency and accountability in shareholding companies, since corporate governance aims to lend integrity and fairness into the financial reporting system and to maintain transparent policies and to establish bases to protect investors and stakeholders.

Subject D affirmed that lack of effective corporate governance code may lead to corporate failure, as the absence of these effective codes means weak disclosure, transparency policies, internal control systems and weak accountability which may encourage corruption and lead to corporate collapses such as the failures of WorldCom, Enron and Tyco in the U.S.A. Then, and in response to these corporate collapses many legislations and acts have been endorsed globally to limit and mitigate the effects of these corporate scandals.

### Subject E

Subject E was a senior financial analyst in one of the Jordanian investment funds, he worked in this position for nine years and before that time he was a financial analyst in an insurance company for eight years. Subject E described corporate governance from a financial perspective as *"the mechanism that provides shareholders (owners and funders) guarantees that they will receive returns on their investment"*. Subject E also added that corporate governance provides an ideal approach that allows stakeholders to control and supervise their interests in the company, and this could be achieved by establishing an effective bylaw that encourages releasing transparent information to be disclosed to the shareholders equally and to ensure that the executive directors are controlled, and worked under a strong and effective accountability system, subsequently this leads to protection of the company assets.

Subject E considered the definition of the OECD<sup>51</sup> as the most comprehensive definition of corporate governance because it highlighted the supervisory and controlling task for the organizations on the one hand, and the definition identified, and distributed the tasks, roles and responsibilities of all parties and bodies involved with the organization, such as the board of directors, the company management, shareholders and stakeholders.

Subject E pointed out that corporate governance best practices help organizing and coordinating activities between the company management, the board of directors and the audit committees, and it facilitates external auditors' procedures; this leads to a decrease in the

<sup>&</sup>lt;sup>51</sup> The OECD definition was discussed in chapter one.

level of the potential risks of the company which contribute, accordingly, to increase the value of company shares in the capital market and to decrease financing costs (borrowing costs).

#### Subject F

Subject F was a manager and partner in a Jordanian brokerage company, and deals with evaluating and assessing securities, and also works as a financial consultant to the company's clients. 20 years of experience in many Jordanian banks where the last five years were in the brokerage company.

Subject F affirmed that the global attention to establish and adopt best practices of corporate governance emerged along with the financial crisis over the world such as the Asian crisis in 1997 and international corporations especially the big American companies that manipulated their accounts to hide and to cover the losses and to embezzle owners' equities whether those of shareholders or those of creditors. Corporate governance needs tools and mechanisms to ensure the best implementation, which includes the collaboration of the board of directors, the executive managers, audit committees, internal auditors and external auditors with the company's bylaw and regulations as explained by Subject F.

"The most important element of corporate governance is the audit committee and is considered an essential part to lend confidence to the financial reports". Subject F described the audit committee as a sub-committee from the board of directors with its main functions to supervise the whole financial reporting system of the company, to follow up and to evaluate the effectiveness of the internal control system and to follow up with the external auditors of the company. Therefore, *"audit committee members should be independent, and possess relevant and adequate experience that enables them to do their duties effectively"*, subject F also pointed out that many studies revealed that the existence of these characteristics with audit committee members enhances the soundness of the financial reporting system. Subject F pointed out that the Jordanian disclosure regulations required all listed companies to form audit committees, but complained that not all of them required them to be independent members.

Subject F also affirmed that the second important element of corporate governance is the board of directors, since the audit committees are formed by and authorized from the board of directors. Subject F criticized the current corporate law and the related regulations that there is no requirement to separate the positions of the CEO and chairperson of the board of directors, and while the JCGC stated that requirement, it was ineffective because the JCGC provisions are implemented on the basis of comply or explain. Subject F also added that the JCGC prohibited the external auditor from performing any additional services (non-audit services) which is inconsistent with many opinions, and comments of Jordanian professionals.

#### Subject G

Subject G was a manager of one of the Jordanian investment funds and has 27 years of experience in the finance and stock market arenas. Before being the manager of an investment fund subject G was working as a financial analyst then senior financial analyst in the same investment fund for about 8 years. Subject G holds a bachelor degree in accounting and a master's degree in accounting and finance.

Subject G provided two main reasons for the increased demand for adopting corporate governance; starting with the agency theory and the subsequent possibility of generating a conflict of interests between the company management from one side and the company shareholders and stakeholders from the other side, which increased the necessity of establishing effective guides and rules to organize the relations between all parties involved in corporate governance. The other main reason is the increasing occurrence of corporate scandals, insolvency and failures globally, the financial crisis that left some crucial effects on the capital markets led governments, institutions and experts to analyze the reasons behind managerial, and financial corruption.

Subject G perceived corporate governance as a set of legislative, administrative ,and economical requirements that govern organizations in a systematic way and would also interact with the internal control team, external auditors, the board of directors, the company executive management, the audit committees, professional associations and the governmental regulatory bodies to produce outputs that contribute to achieving the company goals which include maintaining the rights of the shareholders and the stakeholders alike, adequate and fair disclosure and transparency and produce reliable financial reports.

Subject G pointed out that it was originally believed that there are deficiencies surrounding the International Financial Reporting Standards (IFRS) which contributed to most of corporate scandals and companies' failures, but later it was discovered that the deficiency was not with the IFRS, it was related to who implement the IFRS and their behaviors. Subject G explained the issue of Enron, and their auditors (Arthur Anderson) who had not followed the standards as an example. Therefore, there must be a mechanism to ensure the compliance with law and the corporate governance code. Subject G affirmed that the board of directors has the responsibility to ensure the implementation of effective corporate governance, and to make the company's bylaw compatible with the corporate governance code.

#### Subject H

Subject H worked as a manager of one the Jordanian investing companies with the major investments in securities, and real-estate spread over the Jordanian market, the Arabian gulf area, and internationally with concentrations in the U.S.A and U.K. Subject H has a total working experience of 22 years most of them in the investment and stock market area, where the first five years were in a Jordanian auditing firm.

Subject H perceives corporate governance as the method that aims to achieve the maximum usage of the company resources and to protect shareholders and stakeholders rights due to the separation of the jobs between the company's owners and the company management. Subject H added that it has gained a great deal of publicity after the unprecedented corporate failures,

and that has also led international organizations to help countries in establishing and implementing corporate governance best practices, and even academicians wrote and presented many articles about how to improve corporate governance.

Subject H pointed out that corporate governance aims at increasing a company's performance in a directed manner at enhancing the soundness, and the reliability of the financial reporting system. With regard to the definition of corporate governance, subject H pointed that there are many definitions of corporate governance because they involve many parties whether internally or externally such as the government, securities commission, professional associations, external auditors and even the society. However, most of the definitions concentrating on protecting shareholder's rights by enhancing disclosure, and transparency, accountability and ensuring the reliability of the released financial reports and statements. Finally Subject H pointed out that corporate governance provides a mechanism against corruption, and malpractices especially the implementation of corporate governance supervised by the top management and the board of directors.

#### Subject I

Subject I was manager of a Jordanian investment company, and has a total of around 20 years of experience in the same field, he also held different positions in several investment companies. Subject I pointed out that corporate governance gained great interest recently as a result of a number of administrative malpractices and financial failures that occurred with many big companies all over the world. *"An investigation of reasons that lead to administrative and financial failure revealed that the absence of effective corporate governance enabled those charged with the company's governance to work for their own interests at the expense of shareholders and creditors interests".* 

Subject I pointed out that nowadays with the globalization and the free movements of the capital across the world have opened the doors toward investors and businessmen, and they seek to invest their wealth in institutions that adopted good corporate governance structure

that allows them to control and supervise their investment fairly and equally with other investors, and provide them with fair disclosure of information and the procedures of those institutions should enjoy high standards of transparency without misleading financial accounts and reports.

"Corporate governance is a term based on a respect basis"; that is respecting the rights of the owners, creditors, employees, government and any related body. Corporate governance contribute in reducing agency theory problems as explained by Subject I, that is by mitigating the conflict of interests and reducing the risks associated with the separation between company's ownership and the company's management. Subject H added that good corporate governance practices attract investors and help in curbing financial corruption and increasing institution's competitiveness.

Subject I pointed out that corporate governance facilitates the job of the external auditors and helps external auditors in communicating with management and the internal auditors of the company. Furthermore, Subject I raised the point and the argument in the United States about the effectiveness of external auditors and the reliability of their reports especially after the subsequent corporate collapses in the beginning of the 21<sup>st</sup> century. On that, Subject I pointed that all blame was placed on the external auditors, beside the company directors, which jeopardize the public confidence in the auditing profession, and in the wake of these scandals, the U.S legislators passed new laws and regulations to retrieve the public confidence in the auditing profession by tightening controls on the executive financial management and the external auditors.

### Subject J

Subject J was a senior financial analyst at one of the Jordanian investment funds, 15 years of experience in different investment companies, and funds, and also possess CMA and CFM professional certifications. Subject J pointed out that undoubtedly the recent highlight of the importance of corporate governance concept has shaped the foundation of corporate

management system in the world. Subject J pointed out that the recent corporate scandals, especially in the American market resulted in questioning of the roles of the board of directors, the roles of top management, external auditors' performance and internal auditors' functions. Subject J added that in order to meet these criticisms, strict laws and systems were legislated in the U.S followed by similar actions in many countries including Jordan where corporate governance codes were established by the Central Bank of Jordan, the Insurance commission and the Jordanian Securities Commission.

"Corporate governance simply refers to institutionalization governance i.e. organization are not to be governed or administered by one person or one party allied with each other and works for their interests as their main priority". Subject J also pointed that in order to achieve this goal organizations must be managed under a certain framework based on considering both the company's interest and the various parties' interests. In the same context, Subject J added that effective corporate governance requires increasing the concept of independent members, whether the composition of the board of directors or the audit committees, furthermore, good corporate governance does not have any added value without full implementation by listed companies, therefore, organizations should establish strict bylaws to ensure the applicability of corporate governance and follow up its implementation and to ensure that non-compliance by individuals makes them liable before the law.

### 5.3.2 Cross Case Comparison

This section views the respondents' viewpoints of how they assess the JCGC elements, and the manner and the reason why their assessments had affected their perceptions of the auditing profession in Jordan that is with regard to their perceptions of the reliability of the audit report and the level of the audit expectation gap. This section also shows the main core themes derived from the ten interviews.

Since four dimensions of the JCGC were investigated in this study, namely, external auditor's independence, internal auditor's effectiveness, disclosure and transparency, and corporate accountability, this section presents core themes derived from institutional investors' perception on the effects of each of these dimensions on: firstly the reliability of the audit report, and secondly the level of the audit expectation gap separately. Further discussions of the respondents' comments are reserved for Chapter 6.

### 5.3.2.1 The Reliability of the Audit Report

#### --- External auditor's independence

The main themes and key remarks that have been derived from the financial analysts of the influence of how and why the external auditor's independence affected their perceptions of the reliability of the audit report are that there is a consensus among respondents that external auditor's independence was perceived as the cornerstone of the auditing profession and that it enhances the integrity and the reliability of the audit report and consequently the audited financial reports and statements. The participants also viewed that the objectivity and neutrality lead external auditors to produce unbiased and reliable reports. Another issue raised by the subjects was the negative impact of management pressure placed on external auditor. Moreover, the subjects stressed the vital role of an independent audit committee in nominating, communicating and overseeing the external auditor's work.

"It is considered the cornerstone of the audited financial reports"

Subject A

"Undoubtedly, external auditor's independence plays a vital role in increasing the reliability of the audit report"

"External auditor's independence is the backbone of the auditing profession, without this independence, there will be neither value nor meanings of the professional auditor's report "

Subject C

"Users of the of the financial statements seek the feeling of confidence in the soundness of the financial reports and they seek the reliability and confidence by appointing independent external auditors"

### Subject G

"Independence is the ability to work objectively without bias . . . . and is an indispensable necessity that would increase the credibility of the of the audited financial information"

Subject E

"The emergence of the need for an independent external auditor stems mainly from the existence of a conflict of interest and a gap between shareholders and the company management"

### Subject H

External auditor's independence is the cornerstone of the auditing profession . . . . and furthermore, it has established a suitable climate to create the profession"

Subject I

"Undoubtedly, external auditor's independence leads to the objectivity and reliability of the audit report"

Respondents also provided several viewpoints of their assessment of the external auditor's independence; independence in appearance, independence in fact, the roles of the audit committees, the roles of the professional associations and ethics code of conduct, auditors' rotation, non-audit services, and management pressure were raised and discussed by subjects.

Subject A pointed out that the reliability of the audit report is enhanced when external auditors enjoy high standards of independence requirements whether those of the JACPA, or those required by the ISA.

### Subject A

"External auditor cannot express an independent opinion unless being neutral and objective . . . and must be strong enough to avoid any possible management pressure"

Interviewer: "could you please explain why independence is important and how to achieve that?"

Subject: "Again I want to repeat that the priority of corporate governance is to generate confidence amongst investors and users of the audit report". Subject A explained that in regard to "'why', that is because the absence of objectivity and neutrality increases the likelihood of investor's doubt in the validity of the financial statements"; in regard to 'how' subject A answered that "by adopting effective corporate governance that describes and distributes responsibilities, tasks and the rights of each party including external auditors. Additionally, compliance with the ethics code of conduct reinforcing objectivity and independence".

Subject B referred to the importance of satisfying independence conditions of the international standards, however, the subject affirmed the role of the professional association in maintaining auditor's performance.

### Subject B

"The output of auditing profession which is the independent audit report depends on the external auditor's independence and performance. Independence also reinforces their attitudes against executive management pressure."

Interviewer: "How could the independence of the external auditors be protected beside the compliance with ISAs?"

Subject: "It is the role of the auditors' professional associations to promote the profession and it should solve the audit fees dilemma, to help auditors not to compromise their duties as to what happened with the case of Enron and Arthur Anderson. Another important point is the nomination process of the external auditors must be carried out with full transparency and clarity, and it should be in the hand of the audit committee".

Subject C also commented on the necessity of enjoying the highest levels of independence and its role of detecting frauds and errors. Subject C explained these levels of independency by referring to the international standards by the IFA and local regulations whether these are set by the JACPA or by companies' law, in addition to independence in fact and independence in appearance, and finally the subject added *"auditors are human beings, where some of them have a strong personality and others have a weak personality while for some others the religious morals play a significant role in their character"*.

"External auditor's Independence and neutrality criteria raised many arguments between auditors and interested parties because it is associated with the attitude and mind of auditors" and that is why it is difficult to find an accurate definition of the independence as explained by subject E. However when asked how to reinforce external auditors independence, the subject replied, the auditors should be neutral, honest, and fair with all parties and provide opinions that are not biased or in favor of any specific body. Subject E argued that the most important issue or concern that might impair an auditor's independence is management pressure. Subject E explained in more details how that pressure could be practiced on auditors:

### Subject E

"The company management represents the greatest source of pressure on the auditor which is at the same time a source of non-confidence of users of financial statements"

Interviewer: "What do you mean by management pressure on auditors and how do they practice it?"

Subject: "management pressure and intervention might start with the beginning of auditor's job in the company even by setting their fees and by making them concentrate on particular accounts, hindering and delaying in providing them with requested audit evidence and hinder the auditors from disclosing frauds and errors and even of some financial facts that are of interest to other parties. The management may also intervene in setting audit fees which negatively affect their independence. Finally, in order to ensure auditors' independence they should not intervene in day-to-day operational and administrative procedures which are the duties of the executives".

Subject I is also consistent with Subject E in regard to management intervention and pressure by stating that *"the company management has many reasons and motives to put pressure on auditors in order to appear to the public as one of best director teams, even if that leads the management to distribute unrealized profits"*. Subject I added that the company management might intervene in selecting particular auditors who are willing to achieve their goals.

"The more the independence of the external auditor, the more the confidence of the shareholders and other users of the financial statements and in the work of the external auditors and their audited financial statements"

Subject H

While subject J affirmed that in order to maintain the vital role of the auditing profession, the complete independence concept must be realized in its two sides; those are independence in mind and independence in appearance and they must not be separated but be in parallel with each other as the absence of one of them may weaken or eliminate the effect of the other one. Subject J added that – which is consistent with the point of views of I & E – bring independent in preparing an audit report and in gathering audit information and evidence, contribute greatly in lending much confidence to the audit report and consequently increasing the reliability of the audited financial statements.

The subjects have also discussed many factors affecting external auditor's independence and those factors which lead to increases in the credibility of the audit reports such as the roles of audit committees, audit fees and remuneration, auditor's selection, auditor's rotation, and the provision of non-audit services.

There was consensus on the role of the audit committee in nominating external auditors, except one subject who preferred to reserve this role to shareholders in the general assembly, justifying that by:

"The issue of selecting and dismissing external auditors should not be left in the hands of the body that manages and prepares the financial accounts, rather, it should be in the hands of the owners of the company"

#### Subject J

However, this point of view does not represent a matter of discrepancy, since the selection of the auditors is still in the hands of the shareholders at the general assembly meeting, but the JCGC stated the role of the audit committees in nominating the external auditors.

Subject I pointed out that the issue of audit fees might be used by the company management to affect the work of the auditors. However, despite the fact that the JCGC stated that one of the audit committee's duties is to discuss all matters related to the selection of the external

auditors and to follow up their compliance with independence criteria as stipulated in legislations in force, Subject C noted that it is common in some of Jordanian listed companies that executive managers dominate and control the board of directors and "*it seems to you that they are like one team*" regardless of the supervisory duty of the board of directors.

Subject I stated that "the company management may put direct and indirect pressure by using their influence of the nomination, re-nomination, dismissing and setting their fees . . . . this caused developed countries to place this decision in the hand of an independent and experienced body which is the audit committee"

Subject F also stated that "*external auditor's independence is affected by audit fees, nomination and dismissing*", this comment is also consistent with the viewpoints of Subject G and subject E who added that "to overcome this problem, the authority of appointing, remunerating and dismissing the auditors must be authorized by the audit committees" to avoid any possible factor that could be used by the company management.

Subject D affirmed the vital roles of audit committees in reinforcing external auditor's independence especially if it is composed of independent members: "in my opinion, the most important body that can ensure the neutrality and the independence of the external auditors is the audit committee . . . . therefore the Jordanian Securities Commission should oversee the nomination process of the audit committee members . . . . to sum up summary, truly independent audit committee may result in selecting effective auditors that the users of the their reports can depend on for decision making".

Auditor rotation was perceived to enhance external auditor's independence as the Jordanian literature revealed<sup>52</sup>. However this factor received little comments by the participants<sup>53</sup>.

<sup>&</sup>lt;sup>52</sup> See section 3.7.4

<sup>&</sup>lt;sup>53</sup> The researcher has not mentioned or reminded the subjects of these factors, because doing so might lead the subjects.

Subject C pointed out that there have been many debates about the impact of providing nonaudit services to author's client where subject C stands for prohibiting these services. Subject C added that *"the JACPA should play crucial roles in overseeing non-audit services"*. While Subject G commented that *"in some developed countries, non-audit services were prohibited to reinforce external auditor's independence"*. Subject E pointed out that *"providing consultation and managerial services to the same company client considered one of the important factors that affect the independence standard and surrounds the audited financial statements with suspicion regarding its credibility"*.

Subject E concluded that "despite the different opinions of the impact of providing non-audit services I would prefer prohibiting it, however I appreciate that the JCGC limited the provision of non-audit services".

Audit partner's rotation was described by Subject G as one of the factors that aims to reinforce external auditor's independence *"it decreases the financial interests that may arise between the external auditors and their clients and consequently increasing auditor's independence"*.

### --- Internal Auditor Effectiveness

The participants appreciated the vital role of the internal auditors for the credibility of the audit report due to two main themes derived from their remarks. The first theme is that the integration between external and internal auditors and that they complement each other as both of them aims to ensure the soundness of the financial reporting. The second theme highlighted the importance of the role of the audit committee in reinforcing the internal auditor's effectiveness, not just in facilitating communications with external auditors, but also in setting their budget and giving them authority to report to the audit committee.

Subject J affirmed the crucial role that audit committees should play to ensure effectiveness of internal auditors: *"the audit committee facilitates the communications between external and internal auditors"* which reflected positively on the extent of the reliability of the audit report

and the audited financial statements since internal auditor's performance increase when they report to the independent audit committee. "*The search for an integrated framework to inspire confidence in the financial information provided to users depending on the degree of integration between the internal and external audit and the degree of harmony with the Audit Committee*"

#### Subject J

"Internal audit is to be seen as one of the constituent and important mechanism of corporate governance . . . . Both external and internal audit complements each other. Internal auditor's independence, albeit hard to be perceived, but it is possible to be achieved taking into consideration that the JCGC authorized the audit committee to nominate and oversee the internal audit function, however, giving internal auditors the authority to report any discovered frauds and errors to the audit committee increases the credibility of the financial reporting system of that company".

#### Subject D

"Definitely internal audit effectiveness affects the reliability of the audit report" as stated by subject B, who added that "As we know, external auditors in the early stages evaluate the effectiveness and the soundness of the internal audit function in order to determine the scope of the audit . . . . . therefore, based on their efficiency of detecting errors and frauds, external auditors design their audit plan"

Subject I referred to generally accepted standards that *"internal auditors should be independent of the company activity"* and that independence could be achieved by providing the internal audit a special position in the organizational structure that allows and facilitates internal auditors to fulfill their job effectively, and by giving internal auditors sufficient power to investigate the whole financial accounts.

"The internal audit function must be under the subordination of the board of directors or audit committee as well as the communication should be at this level"

Subject I

"The importance of the impact of effective internal audit on the reliability of the auditor's report derives from the close cooperation between the two functions" according to Subject C. However, subject C explained many ways that internal and external auditor may cooperate in, such as following up external auditors' notes of their evaluation of the weaknesses of the internal control procedures, and correcting any deficiencies discovered by the external auditors.

Subject C added that "in order to get financial statements with a high level of disclosure and transparency and credibility, the internal audit function should be activated properly . . . . and this could be achieved by the existence of an effective and professional organization that ensures the best cooperation between the board of directors, internal auditors, and external auditors".

Subject E pointed out that "Effective internal control procedures are the biggest support for auditing; hence, external auditors evaluate the efficiency of the internal audit effectiveness to determine the audit scope and the extent of audit evidences". However, Subject E's viewpoint of assessing internal audit effectiveness was based on many criteria such as being independent of the company management, and to be associated with the audit committee that is composed of the board of directors, being competent and experienced which enables internal audit staff to investigate and analyze evidence and work objectively. "Therefore, obviously the internal audit function provides the external auditor great and valuable services which helps to increase the credibility of the report of the external auditor".

"The relationship between internal and external audit is not a new phenomenon, but it got a great deal of interest with the increasing instances of corporate failures and the calls for adopting corporate governance best practices worldwide".

Subject G

Subject G also added that internal audit effectiveness affect the scope, time, procedures and efforts of external auditors when assessing the risk of the company and when seeking audit evidence and when conducting their tests *"that is external auditors pay much attention to understand internal control procedures of their clients"*.

"Indeed, there is a significant effect of the internal audit effectiveness and external auditors on the extent of the reliability of the audit report" as stated by subject A. However consistent with subject G who explained external auditor's purposes of evaluating the effectiveness of internal audit function, Subject A added that "but the key point is that internal auditors are subject to the subordination of the top management and that they depend economically on the (salaries) provided by the company, furthermore, there is a possibility that the top management hire relatives or friends for the internal audit department".

"Adopting effective internal audit becomes a focal mechanism of strengthening corporate governance, therefore, reinforcing internal auditor's effectiveness and strengthening the relationships between corporate governance elements is essential to ensure corporate viability and their ability to achieve their goals".

Subject F

Subject F added that the integration between internal and external auditors conveys benefits to both of them; it increases the experience and performance of the internal auditor as a result of the continuous contact with external bodies, especially when investigating new items and accounts that needs special techniques. On the other side it increases the confidence and the credibility of the audit report and the audited financial statements as that integration guarantees comprehensive auditing processes by external and internal auditors. Subject F also referred to the vital role of the audit committees in following up the efficiency of the internal audit system by stating that "audit committees consult with external and internal auditors about the efficiency of internal audit and then it follows up their notes with the company management"

Subject H highlighted the importance of effective and independent internal auditors. Subject C defined the independence of internal audit as *"It does not mean that it is independent from the company, it means the ability of internal auditors to express an objective and neutral opinion of internal control systems, management risks, and corporate governance without bias or any influence from anybody in the company, this independence will not be realized if there are limitations on the work of internal auditors".* 

Subject H also referred to the importance of internal audit standards issued by the IIA, especially linking internal audit activity with the audit committee of the company. Subject H criticized the reality of the situation and level of the cooperation between audit committees and internal audit and requested the board of directors to pay more attention to reinforce internal auditor's independence and performance. Subject H concluded that the harmonization between internal auditors, audit committees, and the executive management contribute to reinforcing corporate governance pillars and consequently increasing the quality of the financial reporting system.

#### --- Disclosure and transparency

Subjects perceived corporate governance as a tool to ensure that fair disclosure of transparent information has been released to the users of the audit report and audited financial statements. However, subjects were aware of the different types of disclosures such as full, adequate and fair disclosure and they also raised the importance of on-time disclosure that investors rely on for making decisions. Additional issue raised by the subjects is disclosing the effectiveness of the internal audit function in the external auditor's report.

"The level of effective disclosure of financial reports depends on defining the target users of these reports" and then the next step is to identify the nature of the required quantity and quality of disclosure as explained by Subject F.

"Accounting disclosure create a climate of confidence amongst the users of the financial statements on the credibility of these statements"

Subject H

"Adequate disclosure of the financial statement is the role of the company management, where the role of the external auditors is to express their opinion indicating the extent that adequate disclosure assumption was followed and to tell the users of the audit reports that there is no significant information that has been hidden".

Subject J

Subject G stated that "disclosure, and transparency generally mean 'expose information', while in accounting field it means that all of the financial statements should appear with full disclosure of all necessary information that helps the users in their decision making". Subject I concentrated on explaining the characteristics of sufficient disclosure as that they should be ontime, transparent, understandable and unbiased.

In response to explaining the levels of disclosures and transparency, subject B stated that "weak level of disclosure and transparency means that there is some information that was hidden and not disclosed to the users of the financial statements, and vice versa". Subject B added that companies that enjoy high levels of disclosure and transparency increase the credibility of the financial reports and helps its investors in taking right decisions.

"Disclosure and transparency are considered one of the most important principles of corporate governance"

### Subject A

"Accounting reports must disclose all the information necessary in order to be described as nonmisleading"

### Subject E

Subject C pointed that corporate disclosure is a result of the separation of corporate ownership, and corporate management because this separation created a gap between shareholders (company owners) and the financial reporting that are prepared and released by the company management, therefore it was necessary to report the real financial position of the company for these shareholders by the independent external auditors. Subject C also added that a special report about the extent of the internal auditor's effectiveness should be prepared and disclosed by the external auditors because having that report affects shareholders and investors' perceptions in relation to the reliability of the audit report. This viewpoint is also in line with comments by subjects J, H and G.

Subject D pointed out that because of the growing importance of accounting disclosure, international professional associations were careful about issuing special standards of disclosure principles. Subject D referred to the International Accounting Standards Committee (IASC) rules and guidance on information that must be disclosed in the financial statements. Subject D considered the external auditor as *"the guarantor person of the integrity of the financial statements since financial statements are prepared according to the IFRS where full disclosure is one of its components"*.

#### --- Corporate Accountability

Corporate accountability was perceived by the subjects as the mechanism that holds all governing parties accountable for their decisions and makes them work under their roles, duties and responsibilities. There was a consensus that effective corporate accountability ensures the soundness of the financial reporting system. However, they raised the issue of ensuring the best implementation of the rules, regulations and legislations in force referring to the potential role that the JSC should play in this regard.

"Corporate accountability is a vital element of corporate governance therefore every company should establish a strong corporate accountability system. Unfortunately, an accountability system could be weakened by the existence of corrupted staff and directors, and it also can be weakened when there is no specific identification of the roles, duties, powers and authorities for all parties working inside the company . . . . again, the accountability affects the reliability of the audit report since the auditors when they feel that they work in a company under appropriately-organized systems and structures that ensures there are no interactions of powers and duties, and when they feel they are liable to answerability and accountability, then the auditors will practice professional due care and work under the auditing standard and local laws without deviations".

### Subject D

Subject C pointed out that the concept of accountability is a major dimension of corporate governance where companies should establish a strong organizational structure that includes defining and documenting the roles, duties and responsibilities of all staff regardless of their level in the organization. The subject added that the existence of such systems in the company makes it easier for the auditor tracking weaknesses and shortcomings and gaps that corrupted staff and directors might go through, thus, adopting an effective corporate accountability system contribute to an increase in the reliability of the auditor's report and provide more assurance that the presented financial statements are not prepared in favour of the company management and are not misleading. Interestingly, the subject called for more governmental

control and roles in fighting corporate corruption, and attributed the increasing number of the companies that have been sent to the court to the Arab uprising where 'anti-corruption' becomes a common slogan in most of the demonstrations.

Subject C:

"Governmental regulatory bodies in the capital market should not be lenient or tolerant with corrupted directors in Jordanian companies"

Interviewer: "do you call for more governmental control of the market?"

Subject: "In fact, in the past Jordanian governmental bodies were lenient in dealing with corrupted companies, but for about approximately one year we used to hear that many of the Jordanian corporations were sent to the court after discovering malpractices by the company management and staff. If you go back one year and view the newspaper you will realize the dramatically increasing number of companies that have been sent to the court due to committing corruption"

Interviewer: "What is the reason for that from your point of view?"

Subject: "I think the reason behind that could be attributed to the Arab Spring uprising, where anti-corruption, implementing accountability and ensuring transparent procedures become the slogan of many people in the demonstrations, and the reforms are required in the financial sectors and it is not just a political issue".

Similar to Subject C, Subjects B, H, and I also called on the government for more control over listed companies by the JSC, ASE, and CCD.

Subject A pointed out that a clear definition of the functions, powers, and responsibilities of the company management, audit committees, internal auditors and external auditors helps in tracing those responsible for the errors and frauds and send them to the court. This opinion is close to that of subject G who added that *"it increases the credibility of their reports especially when these parties feel that they are liable to an effective accountability system"*.

"The robustness of accountability regulations depend on firstly, the support it got from top management to ensure that its implementation to all ranks and staff categories. Secondly, the existence of clear organizational structure in which there are no overlapping in duties and powers between company staff. These points – from my viewpoint - identify the levels of accountability whether strong, moderate or weak. Therefore external auditors who work in a company that has a clear organizational structure and all staff comply with its regulations will increase the reliability of their reports".

"It is the professional association's responsibility to set strict guidelines to hold all auditors accountable . . . . . in order to produce reasonable assurance that the audited financial statements are free of error and fraud"

#### Subject A

"Accountability establishes for retribution, so auditors would fulfill their tasks honestly as they would not forget that they are liable for accountability. This explains why some auditors choose words carefully in their reports because they fear responsibility and consequences".

#### Subject F

Subject I explained the three types of auditor's responsibilities that govern auditors scope which include civil, criminal and disciplinary responsibility, the latter one belongs to the professional association. Subject I added that it is a crucial point to ensure accountability over the company management and the board of directors. *"External auditors should not be left working alone as the only guarantor of the accuracy of financial accounts"*, the securities commission should play its role on overseeing strict accountability over the companies' management.

Therefore, when management and internal auditors work altogether with external auditors under a clear division between the managerial roles and the control roles, that in turn, should help external auditors working effectively.

Reinforcing corporate accountability could be enhanced by an extra assurance provided by the external auditors about the entity's compliance with the corporate governance code, albeit it increases an auditor's tasks which may consequently imply a parallel demand for more audit fees. Subject I affirmed that the benefits of that assurance exceeds its costs, and interestingly, the subject pointed out that many of the audit tasks and duties already existed in the JCGC.

### Subject I

"Undoubtedly corporate accountability supports corporate governance and both top management and the board of directors should not just ensure the existence of the accountability system, they should also ensure the practical and actual implementation of the corporate accountability and corporate governance code".

Interviewer: "How do you increase corporate compliance with the corporate governance code"?

Subject: "I believe that external auditors should check and follow up listed companies' compliance with the corporate governance code i.e. it should be added to the auditors' duties and included in their reports".

Interviewer: "What about the potential burdens whether costs or extra tasks that might occur, as you know auditors in Jordan consider low audit fees compromise their performance?

Subject: "There will be consequential costs, but we also should look at the potential benefits. Reporting about the entity's compliance with corporate governance increases external auditors' responsibilities and awareness of issuing a factual audit report. In regard to the potential additional cost, I would like to mention here that it is the management's duty to report about the company's compliance with the corporate governance code which means the auditors' role will be checking and auditing this report rather than preparing it. However, if we review corporate governance code we will find that many of its items are already essential parts of auditors' responsibilities whether those stated in the companies' law or the regulations of JACPA or even the circulations and the regulations of the Jordanian Securities Commission".

## 5.3.2.2 The Audit Expectation Gap

### --- External auditor's independence

The external auditors were perceived as one side of the audit expectation gap phenomenon, and the other side consists of the users of the financial reporting system whether as preparers of the financial statements, investors, lenders, and the respondents also pointed out that it could be expanded to the whole of society. The key themes derived from the participants' viewpoints is the corporate collapses and scandals are the major contributors to the widening of the audit expectation gap, and they also affirmed the crucial role of the signals that the audit report should convey with regard to the entity's ability to continue as a going concern.

### Subject B

"External auditors are main part of the phenomena of the expectation gap, which reflects the differences in perceptions between the users of the audit report and the auditors themselves . . . . I believe that external auditor's job should not be just to express their opinions on the financial statements, it should exceed that to provide evidence of the viability of companies by evaluating their ability to continue as a going concern, since the phenomenon of expectation gap emerges with the beginning of corporate collapses"

Interviewer: "How can the auditors ensure the best evaluation of viability of companies 'going concern assumption'?"

Subject: "this might not be achieved by only detecting frauds and errors, the auditors should work in the interest of the owners of the company who appointed them for this purpose. Auditors should report to the anti-corruption department, the Jordanian Securities Commission and the company control department when they discover serious violations and breaches committed by the company management before the effects of these breaches increase and subsequently leads to corporate failure and a widening of the audit expectation gap"

"The audit expectation gap occurs when the users of the audit report feel that their expectation of the message of the audit report was not realized . . . . . and usually this phenomenon appears
along with the occurrence of corporate collapses especially after the issuance of an unqualified audit opinion. Therefore external auditors must be independent . . . . the existence of qualified auditors does not mean anything unless they are being independent"

## Subject J

"If we seek to reduce the audit expectation gap, we should consider external auditor's independence, otherwise, external auditors may work for their personal interest rather than shareholders, and they might cover errors and fraudulent which if discovered later, it will widen the audit expectation gap from the shareholders' perception who believed that auditors were working for their benefit".

## Subject E

Subject A pointed out that the users of the audit report may not be aware of the nature of the audit process and called for educating them:

## Subject A

"The gap results because of the difference of external auditor's performance and the expectations of the users of the audit report. Auditors may not audit all transactions and may just audit samples of transactions, external auditors who fulfil their job independently and without management intervention helps in narrowing the gap because this enables the auditors to convey a more factual audit report that reflects the real financial position of the company".

Interviewer: "Why do you think users of audit report are not aware of the audit job, such as 'sampling in auditing' – as you mentioned-, and how to make them more awareness of auditor's duties and tasks?"

Subject: "Most of investors and users of the audit report are not aware of the practical job of the external auditors. Therefore they should be educated by conducting seminars and workshops about external auditor's functions and issuing booklets that explains external auditor's fieldwork procedures and mechanisms, this helps in bridging those different views". Subject G pointed out that external auditor's independence affects the level of the gap whether increasing or decreasing it, as the users of the audit report expect and consider external auditors to be neutral and honest. The subject added that if auditors are not independent they might be involved with management in committing frauds and the problem is that this collusion is hard to discover in the early stages, but when discovered later it creates the gap.

Subject D highlighted the importance of the audit committees in reinforcing external auditor's independence:

## Subject D

"External auditor's independence is an important pillar in the auditing profession, and the Jordanian legislator should limit any thing that compromises their independence, because non-independent auditors result in widening the audit expectation gap"

Interviewer: "Could you please explain how does external auditor's independence affect or contribute in widening the audit expectation gap?"

Subject: "Because the audit expectation gap emerges with the occurrence of corporate scandals and collapses, and as we know external auditors are considered as the agent to protect the wealth of the shareholders who appointed them because they expect they have appointed independent and strong auditors who are able to send them a message about the viability of the company, while weak and non-independent auditors may succumb to the demands of the management of not disclosing frauds and errors if detected."

Interviewer: "How can you reinforce external auditor's capability to confront management pressure?"

Subject: "In my opinion the body that should help external auditors in confronting management pressure is the audit committee. An independent and strong audit committee is an essential element of reinforcing external auditor's independence and performance against management pressure"

Subject H stated that "In my opinion the relationship between external auditor's independence and the audit expectation gap depends on the auditor's efficiency of reporting about the entity's ability to continue as a going concern . . . . when auditors express a qualified opinion, investors will re-consider their assessment of that company, and if that company collapsed, auditors would not be blamed and even would not be responsible for widening the audit expectation gap.". In line with Subject's H opinion, Subjects G and J have similar viewpoints.

Subject F was not optimistic of the capability of the external auditor alone in bridging the gap, although he agreed with the importance of their independence in reducing the gap. *"External auditors may not be able alone to audit and ensure financial control over company's accounts, so, efforts of internal auditors and audit committees are needed to work with each other and with external auditors, because for a long time the professional associations of auditors have always tried to reduce the audit expectation gap, but the gap emerges from time to time, therefore I believe that leaving the matter of reducing the audit expectation gap in the hands of auditors alone is a prejudice against the auditors and everybody who deals with the capital market"* 

"External auditor's independence whether in fact or in appearance are fundamental for their performance . . . . and increases investor's confidence of the real financial position of the company in order to make rightful decisions which accordingly helps in reducing the audit expectation gap . . . . the reason behind selecting external auditors is that shareholders need an independent and neutral body that could not be influenced by the company management".

Subject I

#### ---- Internal Auditor Effectiveness

The subjects viewed internal auditors as a primary protector of the company assets. The key theme is that the internal audit function is fundamentally established in order to detect frauds, errors and malpractices which ensure the final reports, are transparent and unbiased leading to an increased likelihood of a company's viability being discernible and thus reducing the audit expectations gap. Moreover, the subjects pointed out the important role of the audit committees in reinforcing internal auditor's performance and working as a liaison between internal auditors.

"The most important quality that characterized the job of the internal auditors is their involvement in day-to-day operations, which enables them to control the company activity . . . . Internal auditors dedicate the majority of their time to check the accuracy of the accounts of the company, more than the time spent by the external auditors . . . . in regard to the expectation gap, although this term is associated with the external auditors, we should not forget the correlation and the cooperation between them as they integrate, and complement each other. Hence, internal auditor's effectiveness helps and facilitates external auditors in detecting and preventing frauds, errors and breaches and it helps them to the best evaluation of the company's ability to continue as a going concern."

#### Subject I

Subject F stressed the importance of the role of an independent audit committee in nominating internal auditors:

## Subject F

"The aim of an internal audit function is to ensure the accuracy of the accounts and that they are free of errors and frauds, therefore, companies must pay attention to who they recruit in the internal audit department, they should be competent and possess a high level of related experience. . . . . Nowadays it is common to outsource internal auditors, notwithstanding, and from my experience, that staffs still work according to what the company management tells them to do, because they worry about their salary and about re-appointment. However, the JCGC gave the audit committees the right to nominate the internal auditor, and my opinion if this condition is implemented properly it will increase internal audit effectiveness in detecting frauds"

Interviewer: "From your opinion how could we reach the best implementation of internal auditors' appointment?"

Subject: "The essential requirement is the audit committee members must be fully independent and have sufficient experience in accounting and auditing"

Subject J pointed out that the success of internal audit depends on the support they get from the board of directors and from the audit committee in particular, as they are responsible for providing them with adequate powers, independence from the company management, and with any necessary requirements and provisions. Subject J suggested that internal auditors should be under the supervision of the audit committee as recommended by most of the initiatives in developed countries, *"internal auditors draw their strength and power from the audit committee because the audit committee makes them more independent as a result of distancing them from the company management authority*". Subject J stressed that audit committee members should be non-executives and independent members in order to overcome executive management's pressure on auditors. The audit committee should also providing direct lines of communication between internal auditors on the one hand and with external auditors and the board of directors on the other hand. Subject J concluded that audit committees oversee the company management and, therefore *"this integration helps internal and external auditors in hindering frauds, errors and corporate failures and consequently reducing the audit expectation gap"*.

Similar to this viewpoint, Subject B and D have the same opinion as to the vital role of the audit committees:

"It is very important to maintain the effective internal audit function; the company should appoint qualified internal auditors and this appointment must be in the hand of the audit committees as well as the appointment of external auditors and working as a liaison between them"

Subject B

Subject D criticized the independence condition requirements of the JCGC.

## Subject D

"An internal audit function aims to detect frauds, errors and malpractices, therefore, they should be qualified to fulfil their duties and must be appointed by the independent audit committee as stipulated in the Jordanian corporate governance"

Interviewer: "To what extent do you believe that the independence of the audit committee is important, taking into consideration that the JCGC stipulated that the independent audit committee members should not be less than two?"

Subject: "In my opinion all of them must be independent; although there are many committees that could be formed by the board of directors, I believe that the audit committee is different and I consider the audit committee as a vital artery to ensure the soundness of the financial reporting system . . . . they also should be qualified with experience in accounting and auditing"

Subjects G, H and A have the same viewpoints in regard to the internal auditor's role of detecting frauds and errors and maintaining the viability of corporations and also to the importance of the integration with the external auditor. Subject G described that integration as "these two functions are considered as the faithful custodian of the company's accounts". However, Subjects H, C and I perceived that the audit expectation gap as a phenomenon primarily relates to the external auditor's responsibility to reduce the audit expectation gap since this gap represents the gap between the auditors and the users of the audit report. Internal audit, albeit its significant roles, can be seen by external auditors as an assistant function... and an effective external auditors may cover the deficiencies of internal auditors"

#### --- Disclosure and transparency

The participants stressed the importance of fair and on-time disclosure in reducing the audit expectation gap, they have also mentioned several sorts of financial facts the company management sometimes prefer to hide. The key theme is that the participants perceive the fair and on-time disclosure as the medium that tells the investors, shareholders, and stakeholders about the true position of the company and pointed out that providing these parties with transparent information makes them informed about the real financial position of the company, accordingly, making correct decisions.

Subject B argued that "disclosure and transparency levels are much associated with the audit expectation gap whether weak, medium or strong. Disclosure and transparency have an inversely relationship with the audit expectation gap, that is the more the level of disclosure and transparency, the lower the level of the audit expectation gap, because based on that level of disclosure and transparency the users of the audited financial statements will be aware of what is going on inside the company". Subject B added that, regardless of the nature of the disclosure and transparency whether they are in favor of the company management or not (such as bad

news about unexpected losses and so on), the disclosure of these information will contribute in bridging the audit expectation gap, because informing investors and the users of the audit report about the facts surrounding the company will make them unsurprised when financial problem faces the company and even if these financial problems may cause the company to be insolvent or leads to corporate failure.

Subject B provided an example of the importance of the disclosure and transparency in Jordan which is how the Jordanian Central Bank (JCB) dealt with the problem of 'one Jordanian bank' about 3 years ago [in the middle of 2008]: "The governor of the JCB himself immediately appeared in the media and TVs to justified the reasons behind dissolving the board of directors of that bank and explained that clearly and in transparent detail, from the dissolving step, appointing another board of directors instead of the previous one until the election of a new board. The bank now is working like any other bank. Just imagine if the procedures of the JCB were vague without transparency, how would the situation of that bank be and imagine how that widens the audit expectation gap."

"There are many ways of providing adequate disclosure to the investors which enables them to make their informed decisions and accordingly helps in reducing the audit expectation such as: disclosure within the context of the financial statements, using clear and understandable terms, explanations that might appear between brackets, footnotes and the accompanying appendices of notes, attaching explanatory tables and reports, the board of director's report, and finally the external auditor's report as well."

#### Subject G

"Fair disclosure is the guarantor of reducing the audit expectation gap. Fair disclosure is considered the most influential element on the audit expectation gap because it aims to provide information about the company activities and put them at the disposal of shareholders; thus this reduces the audit expectation gap" Subject J pointed out that the accounting and auditing profession play a significant role in affecting decision making and that depends on the level of disclosure and transparency released to the public because of the fundamental goal of the disclosure and transparency concept is to tell the investors and shareholder the truth about the financial position of the companies.

Subject J stressed that the provided information must be adequate, objective, on time, and away from speculations and rumors. Subject J explained how the company executives and directors manipulate the disclosed information. The subject stated that companies may not comply with disclosure requirements whether intentionally, unintentionally or even due to the insufficient knowledge about the disclosure requirements set by the Jordanian Securities Commission, this may include hiding information about transactions with the related parties and insiders, inadequate disclosure in the annual reports and the board of director's report, and it could be by manipulating the business results in newspapers before the official publication of the financial statements.

Subjects E, F and A have similar viewpoints in regard to the association of corporate disclosure and transparency on the level of the audit expectation gap and the importance of making investors and shareholders informed of the real financial position of the company. However, Subjects I, D, and C criticized the role of JSC in maintaining and overseeing Jordanian listed companies' compliance with the disclosure requirements:

"When a company's shares increase dramatically, the JSC uses it power by asking the company to justify that sudden increase of its shares, especially when there are no published reasons or causes for that increase such as published statements of a meeting with strategic partners. However the normal answers from those companies were: we do not have specific justification for that increase and it may relate to the market mechanism and the rules of supply and demand. However, we always hear that routine answer but we, as financial analysts, have

doubts that the insiders<sup>54</sup> at the company know the real reasons behind that increase . . . . such as this example about poor level of disclosure and transparency leads to widen the audit expectation gap"

## Subject C

Subject I pointed out that strong disclosure and transparency measured by its capability of providing shareholders clear clues about the entity's ability to continue as a going concern. The subject added that investors suppose that auditors and the JSC are doing their job sufficiently, otherwise, the JSC and ASE participate in widening the expectation gap. Interestingly, Subject I added other bodies that may participate in widening the audit expectation gap. Subject I also explained how the Jordanian ASE and JSC can play their roles in reducing the audit expectation gap by stating that: "[their role] is not just by demanding listed companies to comply with disclosure requirements, they also need to follow up the compliance with the disclosure instructions and ensure that the companies do not disclose vague information and to ensure that they are provided on the right time, furthermore, the JSC must give shareholders the right to obtain information from listed companies in a transparent manner, if they want to satisfy the corporate governance code of ensuring equitable treatments of shareholders".

## --- Corporate Accountability

The participants perceived that the existence of an effective corporate accountability system increases the performance of the external auditors and reinforces the effectiveness of internal auditors in protecting the assets of the company and increases the likelihood of its ability to continue as a going concern. They also pointed out that increasing the awareness of the

<sup>&</sup>lt;sup>54</sup> The JCGC defined insider at the company as "A person who has access to internal information by virtue of his position or function within the company, including the chairman and members of the board of directors and the company's general manager, financial manager and internal auditor, the representative of the legal person, and relatives" (Corporate Governance Codes in Jordan 2009, Chapter One, Definition section, p. 6)

directors that they work under a strict accountability system and are liable to be accountable reduces instances of management fraud. Another noteworthy issue raised by the participants is the large interest of accountability and anti-corruption in both private and public sectors as a result of the Arab Spring.

"The matter of effective corporate accountability is a vital element and indicator of companies' strength. Therefore, all staff and employees working in the company should recognize that they are liable to be held accountable and that the company is not as we say in slang language 'free money that anyone can steal'. However, I believe that the key thing is to ensure the implementation of the regulations that were essentially founded to maintain the company assets out of theft and loss".

#### Subject E

Subject C stressed the importance of corporate accountability and pointed out that it plays the significant role of limiting the audit expectation gap, because "the company that enjoys strict and high level of accountability applicable on all levels of staff from the small position to the top management and the board of directors are directed properly according to regulations, bylaws and legislations in force". Subject C also explained 'why' he reached that conclusion by pointing out that both external and internal auditors will report any serious breaches and malpractices committed by the executive management to the securities commission and even to the company shareholders to clear their responsibilities. However, Subject C affirmed again the importance of implementing strong corporate accountability over Jordanian listed companies and their directors, the subject added that now, there are many calls for anti-corruption in companies demanded by the public as we can see in the newspapers. Subject C highlighted the importance of the accountability and repeated again the unprecedented number of the Jordanian listed companies which have been sent to the court and attributed that to the unprecedented number of demonstrations demanding economic reforms beside the political reforms.

Similar to the opinions subjects E and C, subject F added that "effective corporate accountability must be documented", while Subject B argued that the absence of effective corporate accountability system makes all staff who works with the company believe that the company is 'like what we say in slang language a house with low fences' which leads to an increase in malpractices instances and therefore to corporate failure and accordingly, increases the audit expectation gap. Subject A pointed out that a strong accountability system makes staff hesitant and reluctant to commit frauds, accordingly, makes the users of the audit reports more convinced about the soundness of the outcomes of the financial reporting, and on the assessment of the ability of the entity to continue as a going concern. Subject D's opinion is consistent with Subject F and stated that "Corporate accountability stands like a bulwark to preserve corporate governance" and called for more governmental intervention in the capital market.

Subject J pointed out that the term audit expectation gap appears whenever corporate collapse occurs and due to the insufficient assurance provided by the company and by the auditors about the ability of the companies to continue as a going concern therefore, all those responsible for manipulating company's accounts must be held accountable and sent to the court. Subject J argued that many of the causes of audit expectation gap relate to 'business failure' rather 'auditing failure' and added that the gap also emerges because the users of the audit report perceive the auditors as the guarantor of the integrity and the accuracy of the financial reporting system.

Subject I stressed that there is misconception that when auditors express a clean opinion it means the financial statements and the financial position of the company are 100% free of errors and frauds, while the generally accepted auditing standards required auditors to practice the maximum levels of *"due professional care"*. The subject added that another aspect of widening the audit expectation gap is the auditor's responsibility of reporting about the company's ability to continue as a going concern and also pointed out that there are many cases in which the external auditors were sued because of negligence in fulfilling their job from a shareholder's perception. However, in their opinion the company management should be

accountable and liable for strict accountability regulations if they are corrupted or even if they do not cooperate with the external auditors. The subject also pointed out that audit committees help in narrowing the audit expectation gap since they are independent and non-executive committees and because they aim to settle disputes and to reinforce external and internal audit performance. However, the subject concluded that the audit committees are the strongest party that is capable to ensure the right implementation of *"the financial accountability"* over the company.

In regard to suing and sending corrupt staff, auditors, directors and managers to the court, Subject H affirmed the vital role that government bodies should play: "the Jordanian Securities Commission should help auditors to ensure the highest level of corporate accountability over the companies by adding more strict items to the current JCGC, and it should send all directors involved in corruption to the court". While subject G pointed out that strict corporate accountability limits committing corruption, and added that "these days we are used to hearing about companies being sent to court and to the anti-corruption department which is a good indicator of implementing a high level of accountability, the accountability becomes a public demand these days as we can see the demonstrations". Consistent with that, subject J added "large instances of sending corrupted directors to the court these days as we see from time to time in the media is an example of good implementation of the accountability".

## 5.4 Integration of the Mixed Methods Data

The mixed method approach helps the researcher to gather quantitative and qualitative data from different sources. This allows the researcher to analyze these data and then integrate them to deliver a comprehensive understanding of the phenomenon, as noted earlier in the previous chapter; adopting quantitative and qualitative methods complement each other and facilitate interpreting the results obtained from the other. This section shows the findings of the study to provide a comprehensive analysis of the impact of the JCGC on the value of the audit opinions as basically measured by the reliability of the audit report and the level of the audit expectation gap.

#### 5.4.1 Context

It is clear that institutional investors in Jordan do perceive the importance of the constituents of the Jordanian corporate governance code and its impact on the reliability of the audit report and the level of the audit expectation gap even if that evidence was often identified with the impact of failures to conform. The effect of the four variables whether individually or in combination (configurally) was obtained from both data sets, those were, the results of the experimental survey and the additional comments to the survey instrument, and the direct comments by interviewees. The findings of the survey method suggest the integrity of the study model and provided a degree of quantification and robustness. The qualitative interviews also supported the study model and provided in-depth understanding of how and why the model factors affect the value of the audit opinions as basically measured by the reliability of the audit report and the level of the audit expectation gap from the institutional investors' perceptions.

#### 5.4.2 Self insight

Both data sets suggest a relatively strong self-insight on the part of institutional investor's assessment of the evaluation of the constituents of the JCGC on both the perceived reliability of the audit report and the level of the audit expectation gap. The experimental survey instrument revealed relatively consistent rankings between the objective (effect size) and subjective data for the factor's influence on both the reliability of the audit report and the level of the audit expectation gap. On the other hand, the interview findings also showed relatively strong self-insight of institutional investor's assessment of the evaluation of the constituents of the JCGC on both the reliability of the audit expectation gap. However,

one subject referred to the role of the board of directors and the shareholders in nominating external auditors and one subject referred to the role of the company management to appoint internal auditors 'ignoring the role of the audit committee as stipulated in the JCGC'.

These dispositions of some subjects' comments, however, they could be attributed to the recent implementation of the JCGC, and that the premises of the role of the board of directors and the shareholders in nominating external auditors still in their mind, as well as the role of the company management to appoint the internal auditors. Additionally, their comments possibly could also reflect a political reality that management or the dominant shareholder/directors still dominate. It is perhaps a reflection that cultures in the Jordanian companies are resistant to change.

### 5.4.3 External Auditor's Independence

External auditor's independence was shown to be a significant, positive factor in influencing the auditing profession in regard to increasing the perceived reliability of the audit report and in reducing the audit expectation gap as well as being the highest subjectively weighted factor in considering the reliability of the audit report. The experiment suggested that external auditor's independence gained the highest rank of importance of its effect on the reliability of the audit report. This could be explained, based on the interviewees' opinions, as there were consensuses in perceiving external auditor's independence as the cornerstone of the profession and it leads to objectivity, credibility and then to the reliability of the audit report. However, external auditor's independence gained the third rank of the influence in the bridging the audit expectation gap.

## 5.4.4 Internal Auditor's Effectiveness

Internal auditor's effectiveness was shown to be a significant, positive factor in influencing the auditing profession in regard to increasing the perceived reliability of the audit report and in reducing the audit expectation gap as the factor satisfied the standard statistical tests for that suggestion. The experiment suggested that internal auditor's effectiveness gained the second rank of importance of its effect on the reliability of the audit report. This could be attributed, based on the interviewees' opinions, to the crucial role of effective internal auditors in detecting and preventing frauds, errors, and malpractices because of the internal auditor's wide knowledge of the detailed activities of the company, moreover, that also might be attributed to the subjects' perceptions of the efficient integration of both internal and external functions in maintaining the soundness of the financial reporting system. However, Internal auditor's effectiveness gained the last rank of the influence in the bridging the audit expectation gap. This perhaps could be explained by the fact that in viewing the expectation gap the institutional investors were judging on the integrity of management and the quality of the audit to determine whether the required standard has been reached. Failure to convince management to be honest and forthright could raise issues as to the effectiveness of the internal audit. In addition to that, two subjects have primarily associated the concept of the audit expectation gap with the external auditors<sup>55</sup>.

#### **5.4.5 Disclosure and Transparency**

Disclosure and transparency were shown to be significant, positive factors in influencing the auditing profession in regard to increasing the perceived reliability of the audit report and in reducing the audit expectation gap as the factor satisfied the standard statistical tests for that suggestion. The experiment suggested that disclosure and transparency was a major factor in

<sup>&</sup>lt;sup>55</sup> Traditionally the audit expectation gap has been attributed to the external auditors failing to show sufficient independence from management as indicated in the common reference of the auditors to management as their clients rather than seeing the shareholders and creditors as the clients.

regard to the effect on the level of the audit expectation gap since the results showed that the disclosure and transparency gained the second rank of effect, slightly less than the corporate accountability factor, as the influential factor on the level of the audit expectation gap. This could be explained, based on the interviewees' opinions, as that the more the level of disclosure and transparency, the lower the level of the audit expectation gap, because based on that level of disclosure and transparency the users of the audited financial statements will be aware of what is going inside the company. However, disclosure and transparency gained the third rank of the influence on the reliability of the audit report.

#### 5.4.6 Corporate Accountability

Corporate accountability was shown to be a significant, positive factor in influencing the auditing profession with regard to increasing the perceived reliability of the audit report, and in reducing the audit expectation gap as the factor satisfied the standard statistical tests for that suggestion. The results of the experiment suggested that corporate accountability gained the first rank of (effect size) of importance on the level of the audit expectation gap. This could be explained, based on the interviewees' opinions, as that when all parties involved in corporate governance such as the board of directors, the executive management, external auditors and internal auditors work with a company that has an effective accountability system and good organizational structures in which all responsibilities, powers, duties and roles are stipulated clearly, such points make those parties liable to the strict accountability system and make them work in the best interests of company rather than their personal interests, therefore, increasing the capability of the company continuing as a going concern.

## 5.4.7 Significant Interactions

The objective measure revealed three large and statistically significant (at the 0.01% level) interactions and four moderate interactions related to influence of the perceived reliability of the audit report (the first dependent variable), the sum of these interactions accounted for approximately one-third of the total effect size. On the other hand, the objective measure has also revealed three large, and statistically significant (at the 0.01% level) interactions and two moderate interactions related to influence of the level of the audit expectation gap (the second dependent variable), and the total of the effect of these interactions accounted for approximately one guarter of the total effect size. These interactions have been discussed in the early section of this chapter. However, the analysis of data derived from the qualitative interviews sheds considerable light, and provided depth understanding on this phenomenon. The existence of these significant interactions supported hypothesis 1 which states [Institutional investors process decision making information configurally when considering the influence of the selected components of the Jordanian corporate governance code]. These indicated interaction terms along with the statistically significance (at the 0.01% level) main effects for all of the four variables individually, supported the model of this study, and the causality relationship between the four posited independent variables, and the dependent variables. Where these interactions have been discussed in the early sections of this chapter, the implications of these interactions are discussed in section 6.6.2.

## **Chapter 6: Discussion and Conclusions**

## **6.1 Introduction**

This chapter consists of eight main sections, not including this brief introduction. Section 6.2 reviews the independent variables of this study. Sections 6.3 and 6.4 bring together the results of the quantitative and qualitative elements of this study and relate them back to the relevant literature. Theoretical and practical implications of the study are summarized in sections 6.5 and 6.7, respectively. Section 6.6 presents the findings of the study in relation the original research questions. This chapter ends with a discussion of the limitations of the study (section 6.8) and suggests a number of areas for further research (section 6.9).

## 6.2 Discussion of the Independent Variables

The audit expectation gap has plagued the auditing profession almost from inception. The profession struggles to mitigate this phenomenon (Epstein & Geiger, 1994) while confidence in audit reports and audited financial statements are undermined by dramatic instances of corporate failure arising without warning from external auditors. Examples of such failures include Enron, WorldCom and the recent global financial crisis.

The literature reveals that the audit expectation gap exists in Jordan, and is partially affected by (1) the external auditor's performance and independence, according to some authorities (Al-

Thuneibat, 2003; Asfoor, 2003 Omari, 2003; Al-Khadash & Al-Sartawi, 2010) and (2) the level of disclosure and transparency (Hajir, 2001). The literature also shows that inadequate evaluation of the going concern assumption (Al-Awaqleh, 2008) and management pressure on auditors (Omari, 2003; Bani-Ahmad, 2000; Matar, 1995; Al-Adeli, 1993; Malhas, 1992) reduces the reliance placed on the audit report.

This section discusses the results of the study in respect of each of the four independent variables.

### 6.2.1 External Auditors' Independence

The value of the assurance provided by the external auditors to shareholders and stakeholders relies on the auditor's perceived ability to detect breaches, irregularities and errors in the financial reporting system, in addition to the auditor's perceived ability to resist management pressure to hide some or all of those breaches (Pott et al, 2009). Based on the interviews conducted in this study, external auditors' independence is seen as the cornerstone of the auditing profession and should lead to reliable and credible financial statements. The intent behind independent auditor opinion is to provide confidence, supposedly being a genuine and expert opinion about the likelihood of absence of material errors and fraud.

This study adopted the definition of external auditor independence provided by the Independence Standard Board (ISB) as noted in Chapter One, section 1.7. The definition focuses on the strength of auditors' attitudes and their performance in the face of any possible pressure that may compromise their effectiveness. The ISB also states "Quality audits improve the reliability and enhance the credibility of the financial reporting process, thereby contributing to its usefulness and to the efficient functioning of the capital markets, which serves the public interest" (ISB, 2000, para. 3). Both this statement and the previously mentioned definition emphasize the key role of external auditors' independence in the

perceived reliability of the audit report and the level of the audit expectation gap. The figure below shows two aspects of auditor's independence, i.e. independence in fact and appearance, as well as the extent to which they are essential parts of maintaining the objectivity of an auditor's performance (Strohm, 2006). This finding is in conformity with Fearnley & Beattie (2004) and Pott et al (2009) who argued that failure of the external auditor's independence is enough to undermine the audit process and leads to a loss of trust in the financial reports, thus destabilizing the capital market. Such failures in independence also raise questions about the role and objectivity of external auditors and contribute to a widening of the audit expectation gap.



Figure 13: External auditor Independence and Reliability of the Audit Report

This figure is quoted from Strohm (2006, p. 19)

According to Duska, Duska, & Ragatz (2011), there is no "absolute [degree of] independence" or "total independence" (p. 127). They argued there will always be some interest and bias among auditors. While Mcgrath, Siegel, Dunfee, Glazer and Jaenicke (2001) argued that the ISB concept of auditor's independence does not require external auditors to be totally free of all influences on their ability to express an unbiased audit report; it requires them to be free from influences significant enough to compromise that ability. The ISB provides guidance that enables external auditors to identify whether undue bias exists in specific circumstances and also suggests several safeguards for external auditor independence.

The results of the objective measures (as captured by the experimental instrument) and the subjective measures (based on self-reporting by participant) revealed that external auditor independence is the most influential factor in the perceived reliability of the audit report, according to Jordanian financial analysts. This result is consistent with results in the existing literature highlighting the association between external auditor independence and the perceived reliability of audited financial statements (Gupta, 2004; Teoh, 1992). This perceived influence could be explained by the interviewees' consensus that external auditors' independence is the cornerstone of the profession and that it leads to objectivity, credibility and thus to the perceived reliability of the audit report.

In regard to the audit expectation gap, the results of this study are not consistent with prior Jordanian literature, those earlier studies having revealed external auditors' independence and performance to be major factors affecting the gap (Al-Thuneibat, 2003; Asfoor, 2003; Omari, 2003; Al-Khadash & Al-Sartawi, 2010). The experiment and the subjective weights in this study indicated independence to be an important factor but not as significant as disclosure and transparency, and accountability. Based on the interviews of financial analysts in this study, this could possibly be attributed to the increasing Jordanian interest in accountability over the past two years<sup>56</sup>. Furthermore, the objective measure showed that the most significant interaction affecting the size of the audit expectation gap was the interaction between 'corporate accountability' and 'disclosure and transparency', implying that the institutional investors

<sup>&</sup>lt;sup>56</sup> For more details, see section 6.2.4

consider the ability of shareholders and stakeholders to hold corporate governance bodies accountable for their level of disclosure to be a key factor affecting the level of the audit expectation gap in Jordan.

## **6.2.2 Internal Auditors' Effectiveness**

The results of this study showed that the internal auditor's effectiveness was a significant positive factor in both the perceived reliability of the audit report and changes in the level of the audit expectation gap. The internal auditor's effectiveness was the second most significant influence on the reliability of the audit report and accounted for 18.59% of the total effect sizes, and had the least, but nevertheless significant, influence on lowering the audit expectation gap as it accounted for 14.32% of the total effect sizes. These two different rankings of significance could also be clearly seen in the financial analysts' subjective weighting of the independent variables. They weighted internal auditor's effectiveness as the least significant factor affecting the size of the audit expectation gap and the second factor affecting the reliability of the audit report. This difference in ranking could be partially attributed, based on the interviews of Subjects H, C and I, to the perception that the audit expectation gap phenomena primarily relates to the external audit profession rather than the internal audit function, although the subjects were aware of the importance of the integration between external and internal audits. The difference in ranking may also be partially attributed to financial analysts' perception of increasing interest in the effect of corporate accountability on the reduction of the audit expectation gap along with the importance of disclosure and transparency<sup>57</sup>.

Based on the interviews, the audit committee was seen as a great supporter of the internal auditor's effectiveness. The interviewees pointed out that an effective audit committee reinforces internal auditors' effectiveness in detecting frauds and errors. They also added that

 $<sup>^{57}</sup>$  See the next two sections: 6.2.3 & 6.2.4

the internal auditors should have the authority to report directly to the audit committee in order to enhance their independence and performance. This is, in turn, increases the accuracy of the financial reporting system. Accordingly, due to the reliance of the external auditors on the work performance and objectivity of internal auditors, and due to the vital role of the audit committee in facilitating the communications between both external and internal auditors, this should increase the reliability of the audit report. These viewpoints of the subject are also consistent with the literature that affirmed the substantial roles of the audit committees in maintaining the effectiveness of the internal auditors (Zureigat, 2011a; Goodwin & Seow, 2000; Goodwin, 2003; Braiotta, 2004; Qin, 2007; Krishnan & Visvanathan, 2007; Christopher et al, 2009).

#### 6.2.3 Disclosure & Transparency

The results of the study showed that 'disclosure and transparency' was a significant positive influence on both the perceived reliability of the audit report and the level of the audit expectation gap. The results showed that the main effect of disclosure and transparency on the change in the size of the audit expectation gap was higher than its effect on the reliability of the audit report (23% and 16% respectively).

A possible explanation, derived from the interviews, is that disclosure and transparency are seen as having an inverse relationship with the level of the audit expectation gap; that is, the higher the level of disclosure and transparency, the lower the level of the audit expectation gap. This is because when information is disclosed appropriately, the users of the audited financial statements become aware of what is going on inside the company. Another possible explanation, again derived from the interviews, might be that there is considerable skepticism relating to the actions of management, particularly the obstruction of auditors' (internal and external) efforts to increase transparency and accountability. With such obstruction, auditors may be perceived by investors as having insufficient access to information. It could be argued that the results of this study, in regard to the impact of disclosure and transparency on the level of the audit expectation gap, are consistent with Hajir (2001). Hajir found that the level of the disclosure and transparency was one the main factors of influence on the size of the audit expectation gap between external auditors and institutional investors in Jordan. Similarly, the results of the experiment in this study showed that the level of disclosure and transparency was the second most significant factor reducing the level of the audit expectation gap, only slightly less significant than the first-ranked factor. Hajir (2001) explained this by stating "investors perceive that the level of disclosure is not adequate and that they then look for higher levels of disclosure from the auditors" (p. 51). Hajir also recommended that the Jordanian regulatory bodies and the JACPA provide audit firms with a mandate, going beyond the expression of opinions about going concern assumptions, to disclose significant facts and financial information, and all uncertainties surrounding the company as such actions would help reduce the audit expectation gap.

## 6.2.4 Corporate Accountability

The results of the study showed that corporate accountability is a significant positive influence on both the perceived reliability of the audit report and changes in the level of the audit expectation gap. However, the experiment findings showed that corporate accountability was the main influence on the changes in the size of the audit expectation gap, but the least significant factor influencing the perceived reliability of the audit report.

It is noteworthy to recall that most related literature found the external auditor's independence to be the most significant factor influencing the level of the audit expectation gap<sup>58</sup>. However, interestingly, this thesis evidenced that corporate accountability was the most influential factor on changes in the level of the audit expectation gap between the external auditors and financial analysts. The interviews suggest that this noteworthy result could be attributed to the

<sup>&</sup>lt;sup>58</sup> For more details, refer to section 3.5.4 of Chapter 3

increasing interest in the role of accountability in fighting corruption in Jordan. Such interest arises from the implication that weak corporate accountability suggests that the top management are not accountable and, if they are also major shareholders, they may consider it appropriate to view the company as theirs to do whatever they wished with. This implication has negatively affected trading in the shares of some companies. Similarly, poor corporate accountability has led to a perception by analysts that external auditors in Jordan are failing to accomplish their role as agents of shareholders<sup>59</sup>.

In the past two years, the issue of corporate accountability has received more attention than normal in Jordan, in the light of a series of corporate scandals. "Corruption exceeded expectations, especially in public shareholding companies," said Samih Bino, the president of the Jordanian Anti-Corruption Commission (JACC) (Bino, 2013)<sup>60</sup>. The interviewees in the current study pointed out that subsequent to the Arab Spring, Jordan witnessed the largest number of demonstrations demanding both economic reforms and political reforms. Tunisia, the Arab country that arguably triggered the Arab Spring, established an accountability act, seeking to reinforce anti-corruption efforts and establish a national accountability board (Seghaier, 2013).

In an effort to reinforce the principles of transparency and accountability, the Jordanian Transparency Center (JTC) was established in March 2011. The JTC aims to increase public awareness of the importance of accountability and is also trying to increase public awareness of the efforts being made to stamp out corruption in both the private and public sectors. In the same context, Jordanian Partners Center launched a series of educational films aiming at spreading awareness within Jordanian society of their roles of advocating accountability and transparency practices (Ammon news, 2013). Calls for accountability and anti-corruption became a slogan raised in most of demonstrations in Jordan. According to some interviewees, Jordan witnessed numerous cases of companies and well-known directors and executives who

<sup>&</sup>lt;sup>59</sup> For more details, see also Qadamani (2007) and Khaza'alah (2001).

<sup>&</sup>lt;sup>60</sup> From a speech during a meeting with the Financial and Economic Committee of the Jordanian House of Representatives

have been sent to the Jordanian Anti-Corruption Commission and to the courts in the past two years. The interviewees pointed out that a review of Jordanian media in the past two years would reveal a dramatic number of companies being sent to the court. A review of the media at that time confirms the interviewees' comments. For instance, in one headline, "48 listed companies were sent to the court", according to information revealed by the general observer of the Jordanian Companies Control Department (CCD) (Al-Talhouni, 2012). "The Arab Spring should be a winter for corruption", said the news reporter Akl (2013). Akl pointed out that public frustration with the dramatic instances of corruption was one of the major catalysts behind the Arab uprisings in the Middle East.

## 6.3 Discussion of Analysts' Decision-Making

This study provided evidence on the decision-making processes of Jordanian financial analysts and the degree of their understanding of these processes. The two sections below discuss the configurality and the level of self-insight of Jordanian financial analysts in comparison with previous research.

#### **6.3.1 Configurality**

The research revealed that Jordanian financial analysts process cues configurally (in combination) when assessing the elements of the JCGC, namely external auditor's independence, internal auditor's effectiveness, disclosure & transparency, and corporate accountability. This finding is consistent with some studies that supported configural cue processing among financial analysts in other countries (e.g. Hopkins, 2009; Ebert & Kruse, 1978; Mear & Firth, 1987; Nguyen & Ross, 2006; Slovic, 1972; Wood, 2002; Teoh, 1996).

## 6.3.2 Self Insight

This study is the first to establish the degree of self-insight among Jordanian financial analysts and it revealed relatively high degrees of self-insight into their decision-making processes. This conclusion is based on both quantitative and qualitative data.

This degree of self-insight demonstrated by Jordanian financial analysts is not consistent with other prior studies of financial analyst decision-making, such as a study in which financial analysts in Australia demonstrated limited and poor self-insight. Those studies revealed that financial analysts tend to overestimate the importance of minor cues and underestimate the importance of major cues (Hopkins, 2009; Mear & Firth, 1987; Libby, 1981). On the other hand, studies using external auditors, accountants (Solomon & Shields, 1995; Savich, 1977) and professional managers (Wood, 2002; Gibbins & Swieringa, 1995) as subjects have shown relatively high degrees of self-insight. The high degree of self-insight among auditors was attributed to the implementation of professional standards on auditing and to regular training performed by the professional associations (Libby, 1981; Pike, Sharp & Kantor 1988). In contrast, the high degree of self-insight among professional managers was attributed to their professional managers of self-insight among professional managers was attributed to their professional associations (Libby, 1981; Pike, Sharp & Kantor 1988). In

It has been evidenced that the financial analysts with more work experience demonstrate moderate degrees of self-insight in the U.S and New Zealand (Feldman & Arnold, 1978; Mear & Firth, 1987; Slovic et al, 1972). In this thesis, the experience of the survey respondents who completed the experiment ranged between 5 to 27 years, with a mean of 12 years<sup>61</sup>. Those who were interviewed for the qualitative study had experiences which ranged between 14 and 27 years. This high level of subjects' experience, along with their professionalism, might be a possible explanation for that relatively high degree of self-insight among the Jordanian financial analysts.

<sup>&</sup>lt;sup>61</sup> See Table 4: Respondents' distribution by experience.

## 6.4 Perceptions of the Elements of the Corporate Governance Code in Jordan

This section discusses the interviewee's viewpoints in regard to the Jordanian corporate governance code. Prior to being presented with the eight core questions, as specified in the attached interview protocol<sup>62</sup>, the participants were requested to give unprompted impressions of what corporate governance means to them, thus revealing the extent of their understanding of the elements of the Jordanian corporate governance code.

The financial analysts demonstrated a high level of awareness of the elements of the Jordanian corporate governance code and their importance. The prominent consensus among the participants was that they attributed the emergence, development and adoption of codes of best practice to the increasing number of large-scale financial scandals that have taken place worldwide. Most of the participants commenced their meeting by referring to these scandals, particularly the Asian financial crisis and the collapse of some large European and American corporations such as Enron and WorldCom as common examples. The participants pointed out that these recent corporate collapses resulted in a questioning of the roles of the board of directors and the external auditors alike. They added that, in response to these financial scandals, investors, stakeholders and all those interested in dealing with the capital markets demanded the adoption of corporate governance best practices in order to ensure better protection of their interests.

Jordanian financial analysts perceived corporate governance, in general, as being important and pointed out that such systems ensure that corporations work in the best interests of their shareholders, employees, and society in general. They added that these corporate governance systems aim to reduce agency theory problems by limiting any potential conflict of interest between the company management, shareholders and other stakeholders. Poor accountability encourages directors to achieve personal interests at the expense of the owners' interests. Such corruption can lead to corporate collapses. The participants added that well-governed

<sup>&</sup>lt;sup>62</sup> See Appendix 7: Data Collection – Interview

companies are those companies that have strong corporate accountability systems in which all governing bodies and staff comply with the company's bylaws and regulations, and ensures that equitable rights of the shareholders, especially their rights to timely and adequate disclosure and transparent financial information, are provided for.

The major concern of the Jordanian financial analysts was whether listed companies embraced the JCGC. Some analysts pointed out that the JCGC is similar to those codes adopted in developed countries, but the problem lies in compliance with the code and not with the code itself. The analysts called for a more effective role for government enforcement to achieve compliance.

## **6.5 Theoretical Implications**

As mentioned earlier, several theoretical frameworks emerged to explain corporate governance. These theories differed slightly since they viewed corporate governance from different angles (Solomon, 2010). In regard to auditing, the information hypothesis, insurance hypothesis, and agency theory evolved to justify the use of auditing (Leung, Paul, & Cooper 2011).

Agency theory is commonly employed in corporate governance (Saunders et al, 2012; Dalton et al, 2003). The research model for the current study is primarily grounded in the agency theory. The associated conflict of interests between corporate governance mosaics (in particular, the board of directors, the executive managements, the external auditors and the owners of the company) that arose because of the separation of the entities and control of those entities could be mitigated by adopting effective corporate governance mechanisms. Since poor corporate governance contributes to financial crisis and corporate collapses, it is associated with a widening of the audit expectation gap. This gap partially exists due to the lack of perceived external auditor independence, weak internal auditor effectiveness, insufficient

disclosure and transparency, and poor corporate accountability systems, according to both the interviews and the literature review in chapter 3.

Separation of ownership and control of entities means that outside (i.e. non-manager) owners need an independent body to provide assurance. The auditing profession was created in an effort to help resolve the agency theory problem, and is being further developed to ensure that shareholders and investors can rely, with confidence, on audit reports and audited financial statements. However, perceived reliance and confidence was compromised due to the instances of corporate failures that occurred without any warning signals in audit reports.

In the Jordanian context, lack of perceived reliance on the audit report and the expansion of the audit expectation gap were partially attributed to the problems associated with the agency theory and conflict of interests that may arise because of the separation between different owners, executive directors and the appointed external auditors. One aspect of agency problems (i.e. compromised auditor independence) was attributed to management pressure on the auditors, weak corporate accountability, inadequate audit fees determined by company management rather than by an independent body (e.g. the audit committee), social and personal relationships between the auditors and the board of directors and the management, which has affected the integrity of external auditor selection. Moreover, the soundness of financial reporting has been negatively impacted by the lack of well-established audit committees, with insufficient roles and authorities, as well as a lack of communication with both external and internal auditors (Swaiti, 2006; Malkawi, 2008; Abdullatif, 2006; Abu-Tapanjeh, 2006; Dahmash, 1989; Al-Basheer, 2003; Matar, 1995; Al-Saudi, 2007; Matar, 2000; Momany, 1994; Al-Awaqleh, 2008).

The interviews also revealed that the conflict of interest can be seen when the external auditors appointed by the company owners (shareholders) depend on the company management for their fees. It is common in Jordan for executive management to, effectively, appoint the external auditors and, at the same time, set their fees. Those perceptions, however, are consistent with the Jordanian literature (Malkai, 2008; Matar, 1994). The subjects considered

company management intervention on the audit fees as a tool of management pressure on auditors. The subjects also stressed the necessity of handing the duties of selecting, dismissing and remunerating the external auditors to a body independent from the company management, which should be the audit committee.

In summary, the research model appears theoretically robust. The four posited independent variables that are derived from the JCGC contributed to an increase in the perceived reliability of the audit report and in bridging the audit expectation gap. The posited four independent variables were found to be statistically significant in their influence. In addition to that, several statistically significant interactions were identified between the posited four independent variables in their influence on both the reliability of the audit report and on the reduction of the audit expectation gap.

## 6.6 Conclusions from the Analysis and Discussion of Findings

This study program allowed for in-depth understanding of the key selected elements of the JCGC, in so far they relate to auditing, and their relative main and interactive influence on the perceived reliability of the audit report and the level of the audit expectation gap, in the Jordanian context. The study program and the adopted mixed method technique addressed the research questions of the study. Next, the findings of the study will be summarized.

#### 6.6.1 The relative main and interactive weights of the JCGC Selected Dimensions

#### 6.6.1.1 The reliability of the audit report

The research technique adopted in this study succeeded in identifying and assessing the relative main and interactive weights of the selected elements of the JCGC and their influence

on the auditing profession, as measured by the perceived reliability of the audit report and changes in the level of the audit expectation gap. The relative main and interactive weights of the four hypothesized independent factors on the perceived reliability of the audit report (the first research question) were measured using objective and subjective techniques. For the objective technique, the 'Effect Size' (Coolican, 2009), was utilized.

The results of the objective measure revealed that perceived external auditor's independence had the greatest effect on the perceived reliability of the audit report among the selected factors of the JCGC. It accounted for 30% of effect size, when interaction terms are allocated back to their parent factors. The second-most influential factor was the internal auditor's effectiveness, which accounted for 27.80% of influence on the perceived reliability of the audit report. Disclosure & transparency was another statistically significant influence, measured at an effect size of 24.40%. Of the JCGC elements selected for investigation in this study, corporate accountability had the least influence on the perceived reliability of the audit report, with an effect size of 17.9%. Table 11 shows the percentage of influence for statistically significant Effect Sizes of the four selected dimensions of the Jordanian corporate governance code in relation to the reliability of the audit report:

# Table 11: Effect Sizes as a Percentage of Total Effect Sizes- Reliability of theAudit Report

Main Effects	Percentage <sup>63</sup>
External Auditor Independence (Ext)	22.71%
Internal Auditor Effectiveness (Int)	18.59%
Disclosure & Transparency (D & T)	15.70%
Corporate Accountability (Acc)	9.66%

## **Interactions**

Significant (large*) effects and interactions at alpha 0.01	19.6%	
Other significant interactions at alpha 0.05	10.70%	
Sum of non-significant effects and interactions	<u>3.04%</u>	
Sum of effects of interaction		<u>33.34%</u>
		100%

\* According to Coolican (2009), an effect size equal to or greater than 0.14 is large.

Table 11 shows that external auditor independence is notably the main effective factor on perceived reliability of the audit report, in the perception of institutional investors.

This result is consistent with several studies (Strohm, 2006; Fearnley & Beattie, 2004; Pott, Mock, & Watrin, 2009) that pointed out that compromising the external auditor's independence results in a loss of confidence in financial reporting and destabilizes the capital market. "Quality audits improve the reliability and enhance the credibility of the financial

<sup>&</sup>lt;sup>63</sup> The percentage of the total variability explained by both main and interactive effects.

reporting process, thereby contributing to its usefulness and to the efficient functioning of the capital markets, which serves the public interest" (ISB, 2000, para. 3). This result was supported by the comments of interview subjects in the current study as there was a consensus that the concept of external auditor's independence is the cornerstone of auditing and the main guarantor of the auditor expressing an independent audit report.

### 6.6.1.2 The audit expectation gap

The model of this study was also successful in identifying and assessing the relative main and interactive weights of the selected elements of the JCGC and their influence on the size of the audit expectation gap whether reducing or widening the gap (the second research question). The relative main and interactive weights of the four hypothesized independent factors, as regards reducing the audit expectation gap, were measured using objective and subjective techniques.

The results of the objective measure revealed that corporate accountability had the greatest effect on reducing the size of the audit expectation gap among the selected factors of the JCGC. Corporate accountability accounted for 30.65% of the effect size, when the interaction terms are allocated back to their parent factors. The second-most influential factor was disclosure & transparency, coming slightly behind corporate accountability in terms of overall influence. External auditor's independence factor was another statistically significant underlying dimension, measured at 20.27 under the objective measure. The least influential factor affecting the level of the audit expectation gap was the internal auditor's effectiveness, which accounted for 19.20% of effect size. Table 12 shows the percentage of influence for statistically significant Effect Sizes of the four selected dimensions of the Jordanian corporate governance code in relation to the change in the size of the audit expectation gap:

# Table 12: Effect Sizes as a Percentage of Total Effect Sizes - Audit ExpectationGap

Main Effects	Percentage <sup>64</sup>
External Auditor Independence (Ext)	16.03%
Internal Auditor Effectiveness (Int)	14.32%
Disclosure & Transparency (D & T)	22.94%
Corporate Accountability (Acc)	23.61%

## **Interactions**

Significant (large*) effects and interactions at alpha 0.01	14.46%	
Other significant interactions at alpha 0.05	5.20%	
Sum of non-significant effects and interactions	<u>3.44%</u>	
Sum of effects of interaction		<u>23.10%</u>
		100%

\* According to Coolican (2009), an effect size equal to or greater than 0.14 is large.

Interestingly, the results of the study suggested that corporate accountability was the most influential factor on the size of the audit expectation gap. Much of the literature related to the audit expectations gap (Abdel-Qader, 2002; Hajir, 2001; Al-Thuneibat, 2003; Omari, 2003; Al-Khadash & Al-Sartawi, 2010; Jarbou, 2004; Jarbou, 2007; Sidani & Olayan, 2007; Al-Qarni, 2004; Elmeligy, 2006; Al-Husaini, 2000; Humphrey et al, 1993; Koh & Woo, 1998; Dewing & Russell, 2002; Salehi & Rostami, 2009; Gloeck & Jager, 1993; Gray & Manson, 2008; Lin & Chen, 2004)

<sup>&</sup>lt;sup>64</sup> The percentage of the total variability explained by both main and interactive effects
have referred to the external auditor's perceived lack of independence as the main factor in widening the audit expectation gap.

The interviews suggest that the perceived importance of cooperate accountability is related to the increasing public interest within Jordan regarding accountability and corruption. Accountability, political and economic reforms and anti-corruption actions in all Jordanian sectors have attracted a great deal of publicity in the past two years. The call for reform in Jordan was not an accident nor divorced from what was happening in neighboring countries. Indeed, the Arab uprising affected the situation in Jordan and accelerated the pace of reform. Some of the interviewees affirmed that these demonstrations succeeded in bringing about action against corruption in Jordan, as demonstrated by the dramatic number of directors and companies that have been sent to the Jordanian Anti-Corruption Commission and to the courts in the past two years.

## 6.6.2 The Constituents of the JCGC and Cue Processing

The hypothesis of the study suggests that the Jordanian financial analysts process information configurally when considering the selected dimensions of the Jordanian corporate governance code. Several significant interactions between the study variables have been identified on both the perceived reliability of the audit report and the level of the audit expectations gap. These interactions accounted for approximately one-third of the effect on the reliability of the audit report and approximately one-quarter of the effect on the level of the audit expectations gap. Furthermore, the existence of 'large' statistically significant interactions between the study variables further confirms and supports the hypothesis of the study.

The existence of these significant interactions in the impact of the selected elements of the JCGC, based on the perceptions of Jordanian financial analysts, sends a clear message to the policy makers and Jordanian regulatory bodies, especially the JSC which is responsible for

253

issuing, revising and monitoring listed companies' compliance with the JCGC. The message is that Jordanian financial analysts, during their judgment decisions, take into consideration the joint effects of these key factors, not just their individual effects. Therefore, the JSC must not just focus on the most decisive factor in their enforcement program, but must realize that the complete environment, in the form of the complementary factors, influence analyst decisions. The JSC must consider how these factors interact.

### 6.6.3 A Note on Self-Insight

Research questions three and four were aimed at investigating the degree of self-insight that institutional investors in Jordan demonstrate in their assessment of the impact of the selected elements of the Jordanian corporate governance code on both the perceived reliability of the audit report and changes in the size of the audit expectation gap. After completing the 16 hypothetical scenarios (the objective measure), the financial analysts were asked to complete a self-reported weighting (the subjective measure) in which they allocate 100 points among the four selected elements of the JCGC. In conclusion, the comparison of these two measures plus the analysis of the qualitative in-depth interviews delivered further evidence of relatively high degrees of self-insight among Jordanian financial analysts into their decision making processes.

## 6.6.4 In-depth interviews

This section addresses this study's fifth research question "How and why do institutional investors' assessments of the elements of the JCGC affect their perception of the reliability of the audit report and changes in the level of the audit expectation gap?" Jordanian institutional investors were asked to explain why the selected elements of the JCGC influence their assessments. The analysis of the participants' viewpoints regarding each of the independent variables, focusing on identifying recurring themes among the various viewpoints, contributed

to an in-depth understanding of the participants' perceptions of the influence of the JCGC, facilitated the interpretation of relative main and interactive weights of the study factors, and contributed to ascertaining the degree of self-insight among the Jordanian financial analysts.

The financial analysts demonstrated high levels of awareness of the importance of and the nature of the elements of the JCGC. The participants pointed out that recent corporate collapses resulted in a questioning of the roles of the directors, and the external and internal auditors, and enhanced the importance placed by financial analysts on good corporate governance, with particular focus on ensuring adequate disclosure and transparency and effective corporate accountability.

# **6.7 Practical Implications**

The insights gained could be used as follows:

- a) Better targeting of enforcement actions by regulators of companies and auditors so as to have a positive impact on the stock market.
- b) The evidence would provide an important message for communicating to directors the impact that their governance decisions will have on institutional investors' willingness to invest in their shares.
- c) To reorient academic research in this area to include interactional dimensions.
- d) To encourage a reorientation of governance research in other developing countries to include cause and effect relationships.

# 6.8 The limitations of the Study

As with all research, there are actual and potential limitations to the current study. This study's limitations can be categorized into three types: limitations of design, limitations of scope, and limitations of application.

# 6.8.1 Limitations of Design

"Limitations of design are principally related to the limited contact between the researcher and respondents, which does not really allow the researcher to gain clarification of responses or additional information on points of interest" (Wood, 2002, p. 257).

This thesis aims to identify the impact of the selected elements of the Jordanian corporate governance code in so far as they relate to the impact on the perceived reliability of the audit report and changes in the size of the audit expectations gap, according to institutional investors in Jordan. This thesis focuses on the external auditor's independence, the internal auditor's effectiveness, disclosure & transparency, and corporate accountability. Through this investigation, this thesis endeavors to contribute to narrowing the expectation gap. Other studies (Ojo, 2009; Siddiqui et al, 2009; Dickins & Higgs, 2009; Rehana, 2010; Sikka et al, 1992; Sidani & Olayan, 2007; Haniffa & Hudaib, 2007; Fadzly & Ahmad, 2004; Pierce & Kilcommins, 1997; Wolf et al, 1999) have investigated and pointed out the importance of other factors such as the role of accounting professional associations, the code of professional ethical conduct in relation to auditing, and the role of audit education in narrowing the audit expectations gap. Some studies also highlighted the importance of auditors' experience and integrity, alongside auditors' independence, in regards to maximizing auditors' objectivity and ultimately increasing the reliability of the audit report (Taylor et al, 2003). Other studies (Teoh et al, 1996; Gul, 1991) posited the impact of disclosing the extent of non-audit services, and/or restrictions on the provision of non-audit services to audit customers, on the perceived reliability of the audit report. The repeated measures design, applied to four independent factors, yielded 16 scenarios. Adding more variables, such as some of those outlined in other studies, would have increased the number of the scenarios to such an extent that they might have rendered the questionnaire impractical (i.e. participants could find it harder to complete), which consequently, could jeopardize the response rate.

The questionnaire contained 16 hypothetical scenarios, any of which could have been difficult to conceptualize for some respondents. This potential difficulty contributed to the decision to confine the unit of analysis to institutional investors only. Individual investors were excluded in order to enhance the experimental design adopted in this study. Professional financial analysts are the better choice when conducting experimental studies because they demonstrate a better understanding of experimental surveys and have more skills with which to analyze the treatments provided in the questionnaire. Furthermore, they are more capable of making informed judgments about the causal relationship of the factors in the study (O'Reilly, 2009; Teoh & Lim, 1996).

Another inherent limitation of the factorial experiment design employed in this study is the decision to limit scenario variations to two levels: lower and higher. Including additional levels would result in a dramatic increase in the number of scenarios and would have rendered the questionnaire unfeasible (for instance, the use of three levels instead of two, under the four posited independent variables, results in increasing the number of scenarios from 16 to 81). The use of two levels is common in research and has enabled various studies to achieve their objectives (Hopkins, 2009; Wood & Ross, 2006; Teoh & Lim, 1996; Nguyen & Ross, 2006).

257

### 6.8.2 Limitations of Scope

There are three sets of corporate governance codes in Jordan; one for the banking sector, another for the insurance sector and a third one for publicly traded companies at the Amman Stock Exchange, the latter one being the code under investigation in this thesis.

The audit expectation gap phenomenon emerges because of the difference in perceptions between auditors and non-auditors regarding the responsibilities of the auditors. Many parties rely on the message conveyed by the auditor's report including the company managers, members of the board of directors, creditors and the credit staff at banks, and the preparers of the financial statements such as the company financial managers and accountants. These parties, in addition to the investors, are the principal parties impacted by the quality of the audit report, and they constitute the other side of the audit expectation gap (see for instance, Abdullatif, 2003; Al-Khadash et al, 2012; Jarbou, 2007; Al-Khadash & Al-Sartawi, 2010; Hajir, 2001; Omari, 2003). However, the scope of this thesis is confined to a unit of analysis made up of institutional investors in Jordan, enabling this study to achieve precise and focused results. In regards to suggestions for future studies, research covering those groups of financial statement users excluded from this study may generate further insights and contribute to a more comprehensive understanding of how corporate governance codes impact the independence of auditors and the audit expectations gap.

This study's field work in Jordan was conducted during the 'Arab Spring'. Based on the interviews, the participants highlighted the importance of accountability in both private and public sectors and asked for more governmental intervention in controlling and regulating the capital market. Slogans demanding strict implementation of accountability and the fighting of corruption became common in Jordanian demonstrations. This situation probably influenced the viewpoints of the interviewees, therefore, the findings of the study must be considered in this context, and caution should be exercised in applying the findings outside this context.

258

## 6.8.3 Limitations in Applications

This thesis was limited to testing the corporate governance code in so far as it relates to the auditing profession in Jordan. The general concepts and elements of corporate governance are likely to be comparable in developing and developed countries. However, some differences are likely because of specific legal and social contexts. Examples of such differences include the composition and the authority of the audit committees, and the regulations regarding auditors providing non-audit services. Furthermore, business culture may vary from country to country, with, for example, varying degrees of dominant shareholdings, crony capitalism, and the influence of religious values, which may limit the ability to generalize the study's findings. Therefore, cautions must be taken in generalizing the findings of the study to other countries.

# 6.9 Suggestions for Future Research

The findings of this thesis provide useful insights into how the external auditor's independence, the internal auditor's effectiveness, disclosure & transparency, and corporate accountability affect institutional investors' perception of the reliability of the audit report and the level of the audit expectation gap. The results also established a starting point for future research.

While effective corporate governance was found to have a positive influence on the Jordanian auditing profession in regard to increasing the perceived reliability of the audit report and on narrowing the audit expectation gap, more empirical evidence is needed since the JCGC was implemented in 2009 and is thus relatively recent. Further research is suggested to investigate the potential roles of the JCGC on other issues such as the practice of earnings management and the characteristics of effective audit committee as a key pillar of corporate governance. Such studies might add a valuable contribution to understanding the importance of maintaining effective corporate governance systems. Moreover, the other (excluded) users of the financial statements, mentioned in the limitation of scope section, and the other factors mentioned in

the limitation of design section, including those elements of the JCGC not selected for this study, present opportunities for further studies to be undertaken in Jordan in order to identify a comprehensive approach to be adopted to minimize the audit expectation gap.

Financial analysts, as expert judges, are expected to possess high degrees of self-insight since that reflects the extent of the accuracy, consistency and confidence in their judgement decision-making process (Libby, 1981; Hooper & Trotman, 1996). Libby (1981) and Hooper & Trotman, (1996) also pointed out that understanding the contributors to self-insight is an crucial issue in the learning process, in training and education, and for maintaining consistency and accuracy in judgment decision making. Therefore, since both qualitative and quantitative methods suggested that Jordanian financial analysts possess a relatively high degree of selfinsight, more research is needed to investigate the causes and contributors to that degree of self-insight, and to develop tools to maximize and maintain a high degree of self-insight.

# References

- Abbott, L. J., Park, Y., & Parker, S. (2000). The effects of audit committee activity and independence on corporate fraud. *Managerial Finance*, *26*(11), 55–68. doi:10.1108/03074350010766990
- Abdel-Al, T. (2007). Corporate governance: Public, Private and Banking sectors (2nd edition). Egypt-Cairo: Aldar Aljame'ya.
- Abdel-Qader, W., (2002). An evaluation of the International Auditing Standards and their application to the audit of listed corporations in Jordan, Unpublished PhD thesis, The University of Western Sydney, Australia.
- Abdel-khalik, A. R. (2002). Reforming corporate governance post Enron: Shareholders' Board of Trustees and the auditor. *Journal of Accounting and Public Policy*, *21*(2), 97–103. doi:10.1016/S0278-4254(02)00040-6
- Abdullah, H. and Valentine, B. (2009). Fundamental and Ethics Theories of Corporate Governance. *Middle Eastern Finance and Economics*, 4, 88-96.
- Abdullah, K.A. (1982). The Audit Profession in Jordan and Kuwait: A Comparative Analytical Study. *Dirasat*, 9(2), 131-151.
- Abdullah, K.A. (2007). The Science of Auditing: Theoretical & Scientific Approach (4th ed.). Dar Wael Publishing- Amman- Jordan.
- Abdullatif, M. (2003). The Role of Auditing in Jordan: An Empirical Study of Responsibilities and Expectations. Unpublished PhD thesis, The University of Manchester, UK.
- Abdullatif, M. (2006). The Effectiveness of Audit Committees in Jordanian Public Shareholding Companies and Potential Characteristics Affecting It: Perceptions from Auditors in Jordan, *Dirasat: Administrative Sciences*, 33(2), 450-467.
- Abdullatif, M. (2007). Audit Committees' Responsibilities and Their Relevance to the Corporate Governance Process: Perceptions from Auditors in Jordan. University of Jordan *Dirasat: Administrative Sciences*, 34(2). 363-384.
- Abed, S., Al-Attar, A., & Suwaidan, M. (2012). Corporate Governance and Earnings Management: Jordanian Evidence. International Business Research, 5(1). doi:10.5539/ibr.v5n1p216

- Abu-Nassar, M., Al-Thnaibat, A., (2005). Importance of Disclosure Requirements Issued by Securities Exchange Commission, and their Adequacy for Serving Users of Financial Statements. University of Jordan. *Dirasat: Administrative Sciences. 32*(1), 115-141
- Abu-Tapanjeh, A. M. (2006). Good corporate governance Mechanism and Firms' operating and financial performance: insight from the perspective of Jordanian industrial companies. *Journal of King Saud University. Administrative Sciences.* 19(2), 101-121
- Abu-Tapanjeh, A., M. (2009). Corporate governance from the Islamic perspective: A comparative analysis with OECD principles. *Critical Perspectives on Accounting* 20(2009). 556–567.
- Adams, M. B., (1994). Agency Theory and the Internal Audit. *Managerial Auditing Journal, 9*(8), 8-12
- Adelopo I. (2010). The impact of corporate governance on auditor independence: A study of audit committees in the UK listed companies, Unpublished PhD thesis, De Montfort University, Leicester, UK.
- Afify, H. A. E. (2009). Determinants of audit report lag: Does implementing corporate governance have any impact? Empirical evidence from Egypt. *Journal of Applied Accounting Research*, 10(1), 56–86. doi:10.1108/09675420910963397
- Ahmad, A. M., (2003). Experimental study of the effects of corporate governance on the quality of financial reporting and the effectiveness of the audit process in Jordan, *Journal of* financial and commercial studies, Cairo University, 1, 276-534
- Ahmed, S. M. R. R. (2011). The Role of Audit Committee as a Supportive Tool of Corporate Governance on Enhancing the Quality of Financial Reports An Applied Study on Egyptian Pharmaceutical Corporations. *Jordan Journal of Business Administration*, 7(1). Retrieved from http://journals.ju.edu.jo/JJBA/article/view/2217
- Akl, N. (2013). The Arab Spring should be a winter for corruption. (2013). The Daily Star Newspaper - Lebanon. Retrieved February 20, 2013, from http://www.dailystar.com.lb/Opinion/Commentary/2013/Jan-15/202265-the-arabspring-should-be-a-winter-for-corruption.ashx
- Al-Adeli, S. (1993). The conflict of interest between the management and the external auditors, and its impact on the external auditor's independence. Unpublished Master thesis. The University of Jordan. Amman.
- Al-Aqdeh, S., Juda, A., Saadah, Y., (2011). Future of Corporate Governance in Jordan, retrieved on March 2012, available at: eco.asu.edu.jo/ecofaculty/wpcontent/uploads/2011/04/42.doc
- Al-Awaqleh, Q. A., (2008). Going Concern Evaluation in Jordan. Unpublished PhD thesis, University Utara: Malaysia. UTARA - MALAYSIA
- Al-Barghouthi, S. (2009). The role of internal audit in accountability to enhance the corporate governance in the Arab bank. Unpublished Master Thesis. Applied Science University. Amman- Jordan.

- Al-Basheer, M. (2003). Corporate governance and the role of auditors. Working paper presented at the 5th scientific conference of Jordanian Association of Certified Public Accountants held in Jordan. Amman during 24 - 25 September 2003. Retrieved from http://aldaralarabia.net/2009-06-01-15-52-11/15-2009-07-27-07-07-50/94-2009-09-09-09-49-49.html
- Al-Duneibat, A. A. (1998). Towards An Understanding of Auditing Change in the UK: The Case of the Development of the Unqualified Audit Report During the Period 1940-1993, Unpublished Doctoral Dissertation, University of Essex, Essex. UK.
- Al-Farah, A.M.S. (2001). The Effectiveness of the Audit Committees in Jordanian Public Shareholding Companies, An Analytical Field Study. Unpublished Master Thesis. The University of Jordan. Amman- Jordan.
- Al-Hanini, A., Dahmash N. (2008). Developing a System for Corporate Governance in Jordanian Stock Corporations to Reinforce the Independence of the Auditor. *An-Najah University Journal for Research*, 22(5), 1481-1518.
- Al-Hasani, S. (1999). The Independence of the Auditor: Analytical and Comparative Study in Light of International Standards and Regulations Controlling Auditing. *Dirasat: Administrative Sciences, 26*(1), 52-77.
- Al-Husaini, W. (2000). The Kuwait Investment Office (KIO) Scandal: A Study of Auditing and Audit Expectations in an International Context. Unpublished PhD thesis, Sheffield University. UK
- Al-Jazy, O. M. H., (2005). Private Sector Initiatives Advancing Corporate Governance. Working paper presented at the Regional Corporate Governance Forum: Private Sector Consultative Meeting, held on 25th January, 2005. Retrieved on April, 2011 from: http://www.aljazylaw.com/pdf/Regional-Corporat-Governance.pdf
- Al-khabash, A. A., & Al-Thuneibat, A. A. (2009). Earnings management practices from the perspective of external and internal auditors: Evidence from Jordan. *Managerial Auditing Journal, 24*, 58–80.
- Al-Khadash, H., & Al-Sartawi, M. A. (2010). The Capability of Sarbanes-Oxley Act in Enhancing the Independence of the Jordanian Certified Public Accountant and its Impact on Reducing the Audit Expectation Gap - An Empirical Investigation from the Perspectives of Auditors and Institutional Investors. *Jordan Journal of Business Administration*, 6(3), 294-316
- Al-Khadash, H., Al-Zoy, I., & Nassar, M. (2012). The Effect of ISA 700 (Revised) "The Independent Auditor's Report on Narrowing the Audit Expectations Gap": An Empirical Investigation from the Perspectives of Auditors, Preparers of Financial Statements, Investors and Academicians. Jordan Journal of Business Administration, 7(4). Retrieved from http://journals.ju.edu.jo/JJBA/article/view/2726
- Al-Khafaji, A., Aly, I. M., Gheyara, K. F., & Metawae, S. (1993). Behavioral Judgment and Cue Utilization: The Case of Corporate Takeovers. *Managerial Finance*, 19(5), 43–54. doi:10.1108/eb013723

- Al-Khouri, R. (2005). Corporate governance and firms value in emerging market: The case of Jordan. *Advances in Financial Economics, 11*, 31–50. doi:10.1016/S1569-3732(04)11002-5
- Al-Lehaidan, I. (2006). *Audit committee effectiveness: Australia and Saudi Arabia*. Unpublished PhD thesis, Victoria University, Retrieved from http://eprints.vu.edu.au/1438/
- Alleyne, P., Howard, M., & Greenidge, D. (2006). The role of audit committees in Barbados. *Corporate Governance*, 6(5), 567–581. doi:10.1108/14720700610706063
- Allison, D.L. (1994). Internal auditors and audit committees. Internal Auditor, 51(4), 50-55
- Al-Matarneh G. F. (2009) *Contemporary Auditing*. (2nd ed.). Dar Al-Maseera for publishing, distributing and printing. Amman- Jordan
- Al-Manzou, Z. (2009). The effect of corporate governance on the dividend policy of the big Jordanian industrial companies listed in Amman stock exchange, Unpublished Master thesis, Al-Albeit University, Jordan.
- Al-Moataz, E. (2003). The Effectiveness of Audit Committees within Saudi Corporations: An Empirical Investigation, PhD Thesis, Loughborough University:UK.
- Al-Najjar, D. M. (2007). *Direct and indirect determinants of transparency, corporate governance and firm value*. Unpublished Master Thesis. Yarmouk University.
- Al-Qam, M. A. (1997). Factors affecting the external auditor's change in the Jordanian shareholding companies. Unpublished Master Thesis. The University of Jordan. Amman-Jordan.
- Al-Qarni, A. A. (2004). The audit expectations gap in Saudi Arabia: perceptions of auditors, preparers and financial statement users. Unpublished PhD thesis, The University of Dundee. UK. Retrieved on January 2012 from: http://discovery.dundee.ac.uk/handle/10588/680
- Alrai (2011). 60 Million dinars the new capital for Jordan Magnesia. *Alrai Newspaper*. Retrieved in August 2012, from <u>http://www.alrai.com/article/519622.html</u>
- Al-Rheily, A. (2008). Audit Committees as one of the corporate governance pillars. *journal of economic and management, University of king Abdulaziz.* 22(1), 179-218.
- Al-Saudi, S. (2007). Factors that Affect Decisions on Selecting and Changing Public Accountants in the Jordanian Public Companies from the Perspective of the Audit Committees, Unpublished Master thesis, Amman Arab University – Jordan
- Al-Talhouni, b. (2012). 48 listed companies were sent to the court. *Rum online news*. Retrieved on September, 2012 from: http://rumonline.net/more-73515-0-
- Al-Tamimi, H. (2006). Introduction to the Auditing: (3th ed.). Dar Wael Publishing- Amman-Jordan.
- Al-Thuneibat, A. (2003). Structure and causes of the audit expectation gap: evidence from Jordan. *Derasat-Managerial Sciences*, 30(1), 108-127

- Al-Thuneibat, A. A., Khamees, B. A., & Al-Fayoumi, N. A. (2007). The effect of qualified auditors' opinions on share prices: evidence from Jordan. *Managerial Auditing Journal*, 23(1), 84– 101. doi:10.1108/02686900810838182
- Al-Waked, M. (2007). Rational corporate governance and financial firms. Al-Mudaqeq. *The periodic of the Jordanian Association of Certified Public Accountants, June*(17, 72), 22.
- Al-Wardat, K.A. (2003). Audit Committees between Practice and Ambition. Working paper presented at the 5th scientific conference of Jordanian Association of Certified Public Accountants held in Jordan. Amman during 24 25 September 2003.
- Al-Zebdieh, R., & Al-Thuneibat, A. (2012). The Effect of External Auditor's Report on the Credit Decision for Jordanian Commercial Banks. *Jordan Journal of Business Administration*, 8(3). Retrieved from http://journals.ju.edu.jo/JJBA/article/view/3322
- Ameera, M. S. (2004). Determining and measuring external auditor selection in the Jordanian public shareholding companies. Unpublished Master Thesis, Amman Arab University Jordan
- Ammon news, (2013). Educational movies to reinforce awareness about transparency and the fight against corruption. *Ammon news*. Retrieved on February, 12, 2013 from: http://www.ammonnews.net/article.aspx?articleno=142264
- Andersen, J. V. (2008). Models of Investment Decision Making in Finance. In Encyclopaedia of Complexity and Systems Science. Retrieved on June 2012 from: http://www.inln.cnrs.fr/IMG/pdf/models\_of\_decision\_making\_in\_finance\_v1.pdf
- Andrew, S., & Halcomb, E. J. (2009). *Mixed Methods Research for Nursing and the Health Sciences*. John Wiley & Sons.
- Antle, R. (1982). The Auditor As an Economic Agent. *Journal of Accounting Research*, 20(2), 503–527. doi:10.2307/2490884
- Applied Corporate Governance. (2012). Definition of Corporate Governance. Retrieved on February 2012 from: http://www.applied-corporate-governance.com/definition-ofcorporate-governance.html
- Archambeault, D., DeZoort, F. T. and Holt T. P. (2008). Governance transparency and the need for an internal audit report to external stakeholders. *Accounting Horizons*, 22(4), 375-388.
- Arens A., Best P., Shailer G., Fiedler B., Elder R., and Beasley M. (2010). *Auditing, Assurance services and ethics in Australia: An integrated approach*, (8th Ed), Pearson Australia, French Forest NSW 2086.
- Ariño, A. (2002). Measures of strategic alliance performance: an analysis of construct validity. *Journal of International Business Studies, 34*(1), 66–79. doi:10.1057/palgrave.jibs.8400005
- Arnold, V., Bedard, J. C., Phillips, J. R., & Sutton, S. G. (2011). Do section 404 disclosures affect investors' perceptions of information systems reliability and stock price predictions?

International Journal of Accounting Information Systems, 12(4), 243–258. doi:10.1016/j.accinf.2011.05.001

- ASE (2011). Amman Stock Exchange "ASE", Privatization in Jordan: Why is Privatization a need in Jordan? Retrieved at September 2011 from: http://www.ase.com.jo/en/privatization-jordan
- ASE: Broker Members (2012). Amman Stock Exchange (ASE). Last visit on January 2012. Broker Members webpage. Available at: http://www.ase.com.jo/en/contact-information
- Asfoor, M. (2003). The effects of the social and economic factors on the work of the external auditor under the domination of the establishments. Working paper presented at the 5th scientific conference of Jordanian Association of Certified Public Accountants held in Jordan. Amman during 24 25 September 2003.
- Ashmawi, M. A., (2005). Accounting framework proposal for the role of corporate governance in activating the financial market. Analytical approach for activating knowledge economy. The 5th International Scientific conference. Alzaytoonah University: Amman Jordan.
- Ashton, A. H. (1985). Does consensus imply accuracy in accounting studies of decision making? *The Accounting Review, 60*(2), 173–185.
- Ashton, R. H. (1974). Cue utilization and expert judgments: A comparison of independent auditors with other judges. *Journal of Applied Psychology*, *59*(4), 437–444. doi:10.1037/h0037314
- Ashton, R. H., & Ashton, A. H. (1995). Perspectives on judgment and decision-making research in accounting and auditing. In *Judgment and Decision-making Research in Accounting and Auditing*. Cambridge University Press. Retrieved from http://dx.doi.org/10.1017/CBO9780511720420.003
- Awartani, H. (2011). Jordan Magnesia. Losses of 41 million Dinars and a syndicated loan led by "Arab Bank" with 30 million. *Ammon news*. Retrieved in October, 2011 from <u>http://www.ammonnews.net/article.aspx?articleNO=92999</u>
- Baker, K., & Nofsinger, J. (2002). Psychological Biases of Investors. Financial Services Review,11(2),97–117.Availableonlinehttp://www2.stetson.edu/fsr/abstracts/vol\_11\_num2\_p97.pdf
- Ballesta, J. P. S., & García-Meca, E. (2005). Audit qualifications and corporate governance in Spanish listed firms. *Managerial Auditing Journal, 20*(7), 725–738. doi:10.1108/02686900510611258
- Bani-Ahmad, A. (2000). Factors Affecting the Selection of Audit Samples in the Jordanian Commercial Banks. Unpublished Master thesis, University of Aal Al-Bayt. Mafeaq -Jordan.
- Bashtawi, H., .and Suleiman, S. (2003). *The Impact of the Economic and Social Factors on the External Auditor's Performance under the Organizational Control*. Working paper presented at the 5th scientific conference of Jordanian Association of Certified Public Accountants held in Jordan. Amman during 24 25 September 2003.

- Beasley, M. S., & Salterio, S. E. (2001). The Relationship between Board Characteristics and Voluntary Improvements in Audit Committee Composition and Experience\*. *Contemporary Accounting Research*, 18(4), 539–570. doi:10.1506/RM1J-A0YM-3VMV-TAMV
- Ben Zur, H., & Breznitz, S. J. (1981). The effect of time pressure on risky choice behaviour. *Acta Psychologica*, *47*(2), 89–104.
- Beyer, A., & Sridhar, S. S. (2006). Effects of Multiple Clients on the Reliability of Audit Reports. Journal of Accounting Research, 44(1), 29–51. doi:10.1111/j.1475-679X.2006.00197.x
- Bino, S. (2013). Corruption exceeded expectations, especially in public shareholding companies. *Alrai Newspaper*. Retrieved March 16, 2013, from http://alrai.com/article/573293.html
- Black, F. (1986). Noise. *The Journal of Finance*, *41*(3), 529–543. doi:10.1111/j.1540-6261.1986.tb04513.x
- Block, D. J., & Epstein, M. A. (1995). *The corporate counsellor's deskbook*. Aspen Publishers Online.
- Boniface, D. R. (1995). *Experiment Design and Statistical Methods: For Behavioural and Social Research.* Chapman & Hall.
- Borkan, J. M. (2004). Mixed Methods Studies: A Foundation for Primary Care Research. *Annals* of Family Medicine, 2(1), 4–6. doi:10.1370/afm.111
- Bouwman, M. J., Frishkoff, P. A., & Frishkoff, P. (1987). How do financial analysts make decisions? A process model of the investment screening decision. *Accounting, Organizations and Society*, *12*(1), 1–29. doi:10.1016/0361-3682(87)90013-4
- Boutros-Ghali Y.(2001) Corporate Governance Practices in Egypt: The Road Ahead, Paper presented at a Conference in Cairo sponsored by the Centre for International Private Enterprise (CIPE) and the Egyptian Capital Market Association. Retrieved on September 2010 from: <www.cipe.org/publications/fs/articles/CGinEgypt.pdf>,
- Braiotta, L. (2004). *The audit committee handbook*. (4th ed.) Wiley and Sons.
- Brannen, J. (1992). Combining Qualitative and Quantitative Approaches: An overview (pp. 3– 37). *Mixing methods: qualitative and quantitative research*. Aldershot: Avebury.
- Brown, C. E., & Solomon, I. (1989). Auditor configural information processing in control risk appraisal. FACULTY WORKING PAPER NO. 89-1585. College of Commerce and Business Administration. University of Illinois at Urbana-Champaign, August 1989. Retrieved on June 2012, from: http://ia600808.us.archive.org/3/items/auditorconfigura1585brow/auditorconfigura15 85brow.pdf
- Brown, C. E., & Solomon, I. (1990). Auditor configural information processing in control risk assessment. *Auditing: A journal of practice and theory 9*(3), 17–38.
- Brown, C. E., & Solomon, I. (1991). Configural Information Processing in Auditing: The Role of Domain-Specific Knowledge. *The Accounting Review*, *66*(1), 100–119.

- Brown, P. (1983). Independent Auditor Judgement in the Evaluation of Internal Audit Functions. Journal of Accounting Research. Autumn, 21(2). p.444-455.
- Bryman, A. (1988). *Quantity and Quality in Social Research*. Psychology Press.
- Bryman, A. (1992). Quantitative and qualitative research: Further reflections on their integration. In J. Brannen (Ed.), *Mixing methods: Qualitative and quantitative research* (pp. 57–78). Aldershot: Avebury. Ashgate Publishing.
- Bryman, A. (2006). *Mixed Methods.* 1. Sage Publications: London- UK http://www.sagepub.com/home.nav
- Bryman, A., & Bell, E. (2007). Business Research Methods. Oxford University Press.
- Bryman, A. (2008). Why do researchers integrate/combine/mesh/blend/mix/merge/fuse quantitative and qualitative research? In M. Bergman (Ed.) *Advances in mixed methods research*. Thousand Oaks, CA: Sage.
- Buglear, J. (2012). Stats Means Business. CRC Press.
- Burns, R., (2000). Introduction to Research Methods (4th ed.). London: Sage Publication.
- Bushman R. M., Piotroski J.D., & Smith A.J. (2004). What determines corporate transparency. *Journal of Accounting research.* 42(2), 207-252
- Bylaw of Practicing the Auditing Profession. (2006). Issued by the virtue of articles No. 29 and<br/>45 from the Law of Regulating the auditing profession No. 73 for the year of 2003.<br/>Retrieved on August 2011 from:<br/>http://www.lob.gov.jo/ui/bylaws/search\_no.jsp?no=7&year=2006
- Cadbury, A. (2002). *Corporate Governance and Chairmanship: A Personal View*. Oxford University Press, USA.
- Cadbury Report, (1992). Report of the committee on the financial aspects of corporate governance: The code of best practice, Gee professional publishing. London UK. Retrieved on August 2011 from <a href="http://www.ecgi.org/codes/documents/cadbury.pdf">http://www.ecgi.org/codes/documents/cadbury.pdf</a>
- Cameron, R. (2011). Mixed Methods Research: The Five Ps Framework. *Electronic Journal Of Business Research Methods, 9*(2), 96-108.
- Carcello, J. V., and Neal, T. L. (2003). Audit Committee Characteristics and Auditor Dismissals Following "New" Going-Concern Reports. *The Accounting Review, 78*(1), 95-117
- Carcello, J. V., and Neal, T. L. (2000). Audit Committee Composition and Auditor Reporting. *The* Accounting Review, 75(4) 453-467
- Carlon, S., Mladenovic-McAlpine, R., Loftus, J., Palm, C., Kimmel, P. D., Kieso, D. E., and Weygandt J. J. (2008). *Accounting: Building Business Skills* (3rd ed.). John Wiley & Sons Australia.
- CCD: Companies Inquiry (2012). Companies Control Department Companies Inquiry web page. Last visit on January 2012. available at: http://www.ccd.gov.jo:7777/ccd\_gov\_jo\_2/CompanyInquiry/CompanyInquiry\_en.jsp

- Chan, K., Wong, J., Wong, N. (2002). Non audit services and auditor independence, Chartered Accountants. *Journal of New Zealand*, *81*(5), 27-32.
- Cheng. P., & Abdel-Qader, W. (2010). *Are Auditors Guilty for Surprise Corporate Failures*? Working Paper presented at the 8th Annual International Conference on Finance. Athens Institute for Education and Research. 5-8 July 2010, Athens, Greece.
- Chewning Jr, E. G., & Harrell, A. M. (1990). The effect of information load on decision makers' cue utilization levels and decision quality in a financial distress decision task. *Accounting, Organizations and Society, 15*(6), 527–542. doi:10.1016/0361-3682(90)90033-Q
- Ching, K. W., Tan, J. S., & Ching, C. R. G. (2006). *Corporate Governance in East Asia: The Road Ahead : Analysis and Case Studies in China, Hong Kong, Taiwan, Korea, and Japan*. Prentice Hall/Pearson Education South Asia Pte Limited.
- Christopher, J., Sarens, G., & Leung, P. (2009). A critical analysis of the independence of the internal audit function: evidence from Australia. *Accounting, Auditing & Accountability Journal, 22*(2), 200–220. doi:10.1108/09513570910933942
- CIPE, (2011). Advancing Corporate Governance in the Middle East and North Africa: Stories and Solutions, Center For International Private Enterprise, Global Corporate Governance Forum. Retrieved on August 2011 from: http://www.cipe.org/publications/bookstore/pdf/advancingCGinTheMENA.pdf
- Citron, D., Taffler, R., and Jinn-Yang, U. (2008). Delays in reporting price-sensitive information: the case of going concern, *Journal of Accounting & Public Policy*. 27, 19-37
- Clark, V. L. P., & Creswell, J. W. (2007). The mixed methods reader. Sage Publications.
- Cohen, J. (1988). Statistical Power Analysis for the Behavioral Sciences. Routledge.
- Cohen, I,. & Manion L. (2007). *Research Methods in Education*. Routledge.
- Cohen, S. S., & Boyd, G. (2000). *Corporate governance and globalization: long range planning issues*. Edward Elgar Publishing.
- Colbert, J. L. (2002). Corporate governance: communications from internal and external auditors. *Managerial Auditing Journal, 17*(3), 147-152.
- Collins, K. M. T., Onwuegbuzie, A. J., & Sutton, I. L. (2006). A model incorporating the rationale and purpose for conducting mixed methods research in special education and beyond. Learning Disabilities: *A Contemporary Journal*, *4*, 67-100.
- Collis, J., & Hussey, R. (2003). Business Research. Palgrave Macmillan.
- Collis, J., & Hussey, R. (2009). Business Research: A Practical Guide for Undergraduate and Postgraduate Students. Palgrave Macmillan.
- Companies Law (1997). The Companies Law No. 22 of 1997 and its amendments. Published in the Official Gazette No. 4204 dated 15/5/1997. Retrieved on August 2011 from: http://www.mit.gov.jo/portals/0/tabid/502/Companies%20Law.aspx

- Company Law Draft (2011). Draft for the new upcoming Companies' Law in Jordan. Retrieved on August 2011 from: http://www.ccd.gov.jo/english/inside.php?src=ml&id=15
- Cook, R. L., & Stewart, T. R. (1975). A comparison of seven methods for obtaining subjective descriptions of judgmental policy. *Organizational Behavior & Human Performance*, *13*(1), 31–45. doi:10.1016/0030-5073(75)90004-5
- Coolican, H. (1994). *Research methods and statistics in psychology* (2nd ed.). London: Hodder & Stoughton
- Coolican, H. (2009). Research Methods and Statistics in Psychology, (5th ed.). Hachette UK.
- Cooper B.J., & Grose, R. (2010). Trust Me: I'm An Auditor. *International Review of Business Research Papers 6*(5), 160 – 166
- Cooper, D., & Schindler, P. (2010). *Business Research Methods*. McGraw-Hill Companies, Incorporated.
- Council of the European Communities (2006). Eight Council Directive 2006/43/EC of the European Parliament and of the Council of 17 May 2006 on statutory audits of annual accounts and consolidated accounts. Brussels. Retrieved on May 2012 from http://ec.europa.eu/internal\_market/auditing/directives/index\_en.htm
- Corporate Governance Code for Banks in Jordan (2007). Issued by the Central Bank of Jordan. Accessed on December 2009. Available at: http://www.sdcjordan.com/english/index.php?option=com\_content&task=view&id=3 26&Itemid=391
- Corporate Governance Code in Jordan, (2009). The Jordanian Corporate Governance Codes that are issued for the publicly trading company on the Amman Stock exchange by the board of the commissioners of the Jordanian Securities Commission (JSC) on the 29th of July 2008. Retrieved on September 2009 from the Jordanian Securities Commission: http://www.jsc.gov.jo
- Cox, D. R., & Reid, N. (2000). *The Theory of the Design of Experiments*. CRC Press.
- Crane, A., & Matten, D. (2010). Business Ethics: Managing Corporate Citizenship and Sustainability in the Age of Globalization (3rd ed.). Oxford University Press, USA.
- Creswell, J. W. (1998). *Qualitative Inquiry And Research Design: Choosing Among Five Traditions*. Thousand Oaks: Sage Publications.
- Creswell, J. W. (2003). *Research Design: Qualitative, Quantitative, and Mixed Methods Approaches*. SAGE.
- Creswell, J. W. (2008). Educational Research: Planning, Conducting, and Evaluating Quantitative and Qualitative Research. Pearson/Merrill Prentice Hall.
- Creswell, J. W. (2009). Research design: qualitative, quantitative and mixed methods approaches, (3rd Ed.) Thousand Oaks, CA: Sage.

- Creswell, J. W. (2010). Mapping the developing landscape of mixed methods research. In A. Tashakkori & C. Teddlie (Eds.), *Handbook of mixed methods in social & behavioral research* (2nd ed.; pp. 45-68). Thousand Oaks, CA: Sage.
- Dahmash, N., (1989). Auditing Profession in Jordan. *The journal of Arabic certified accountants,* (44), 21-25.
- Dahmash, N, Hajjier, I., & Al-Farah, A. (2003). Audit Committees; Their Establishment, Duties, Responsibilities, and Role in Increasing Disclosure. Paper presented at Jordanian Association of Certified Public Accountants' 5th conference. Retrieved from http://infotechaccountants.com/forums/showthread.php/5832
- Dahmash, N and Abu Zerr, A. (2003) Corporate governance and its relationship with accounting and auditing profession. Working paper presented at the 5th scientific conference of Jordanian Association of Certified Public Accountants held in Jordan. Amman during 24 - 25 September 2003.
- Dalton D, Daily C, Certo S, Roengpitya R. (2003). Meta-analysis of financial performance and equity: fusion or confusion. *Academy of Management Journal.* 46(1), 269–90.
- Daniels, B. W., & Booker, Q. (2011). The effects of audit firm rotation on perceived auditor independence and audit quality. *Research in Accounting Regulation, 23*(1), 78–82. doi:10.1016/j.racreg.2011.03.008
- Datta, L. (1994). Paradigm wars: A basis for peaceful coexistence and beyond. In A. Bryman (Ed.), *Mixed Methods*. 1, 31–52. London: Sage Publications.
- Davies, M. (2009). Effective working relationships between audit committees and internal audit—the cornerstone of corporate governance in local authorities, a Welsh perspective. *Journal of Management & Governance, 13*(1-2), 41–73. doi:10.1007/s10997-008-9070-9
- DeCuir–Gunby, J. T. (2008). 8 Mixed Methods Research in the Social Sciences. Best Practices in Quantitative Methods (pp. 125–136). Thousand Oaks California . United States of America: SAGE Publications, Inc. Retrieved from http://srmo.sagepub.com.ezproxy2.acu.edu.au/view/best-practices-in-quantitativemethods/d11.xml
- Demsetz, H., and Lehn, K. (1985). The Structure of Corporate Ownership: Causes and Consequences. *The Journal of Political Economy*, *93*, 1155-1177.
- Demirag, I.S. and Solomon, J.F. (2003). Developments in International Corporate Governance and the Impact of Recent Events. *Corporate Governance: An International Review*, 11(1), 1-7.
- Denzin, N. K., & Lincoln, Y. S., (2000). Introduction. In N. K. Denzin, & Y. S. Lincoln (Eds.), Handbook of qualitative research, (2nd. Ed). Sage
- Dewing, I., and Russell, P. (2002). UK fund managers, audit regulation and the new Accountancy Foundation: towards a narrowing of the audit expectations gap. Journal: *Managerial Auditing Journal, 17* (9), 537-545

- Dickins, D., & Higgs, J. L. (2009). Bridging the expectations gap. *Journal of Corporate Accounting* & *Finance*, 21(1), 51–62. doi:10.1002/jcaf.20547
- Dillman, D. A. (2006). *Mail and Internet Surveys: The Tailored Design Method -- 2007 Update with New Internet, Visual, and Mixed-Mode Guide*. Wiley.
- Directives for Listing Securities on the Amman Stock Exchange (2004). Issued by virtue of the provisions of Article 72 of the Securities Law No. 76 of 2002 by the decision of the board of commissioners of Jordan Securities Commission No (149/2004). Retrieved from http://www.ase.com.jo/en/listing-securities-directives
- Duska, R., Duska, B. S., & Ragatz, J. A. (2011). Accounting Ethics. John Wiley & Sons.
- Easterby-Smith, M., Thorpe, R., Jackson, P., & Lowe, A. (2008). Management Research. SAGE.
- Ebert, R. J., & Kruse, T. E. (1978). Bootstrapping the security analyst. *Journal of Applied Psychology*, *63*(1), 110–119. doi:10.1037/0021-9010.63.1.110
- Einhorn, H. J. (1970). The use of nonlinear, noncompensatory models in decision making. *Psychological bulletin*, *73*(3), 221–230.
- Einhorn, H. J., & Hogarth, R. M. (1981). Behavioral decision theory: *Processes of judgment and choice. Annual Review of Pschology*, *32*, 53–88.
- Elmeligy, I. (2006). The impact of the corporate governance mechanisms on the audit expectation gap in Egypt. *Journal of Faculty of Commerce for Scientific Research. Alexandria University.* 43(1) retrieved on January 2011 from: http://faculty.ksu.edu.sa/72669/Documents/%D9%85%D9%84%D9%8A%D8%AC%D9 %8A%2011.doc
- ERKKILÄ, T. (2007). GOVERNANCE AND ACCOUNTABILITY A SHIFT IN CONCEPTUALISATION. *Public Administration Quarterly*, *31*(1/2), 1–38. doi:10.2307/41288281
- Epstein, M. J., & Geiger M.A. (1994). Investor Views of Audit Assurance: Recent Evidence of the Expectation Gap. *Journal of Accountancy*, 177(January, 1994), 60-66.
- Fadzly, M. N., & Ahmad, Z. (2004). Audit expectation gap: The case of Malaysia. *Managerial Auditing Journal, 19*(7), 897–915. doi:10.1108/02686900410549420
- Fama, E. F. (1998). Market efficiency, long-term returns, and behavioural finance. *Journal of Financial Economics*, 49 (3), 283–306.
- Farrar, J.H. (2003). Corporate Governance and the Judges. *Bond Law Review: 15*(1), 63-101, Article 5. Available at: http://epublications.bond.edu.au/blr/vol15/iss1/5
- Farrar J, H., (2005). *Corporate Governance: Theories, Principles, and Practice* (2nd ed.) Melbourne: Oxford University Press.
- Fearnley, S., & Beattie, V. (2004). The Reform of the UK's Auditor Independence Framework after the Enron Collapse: An Example of Evidence-based Policy Making. *International Journal of Auditing*, 8(2), 117–138. doi:10.1111/j.1099-1123.2004.00085.x

- Felo, A. J., Solieri, S. A. (2009). Are all Audit Committee Financial Experts Created Equally?. International Journal of Disclosure and Governance, 6(2), 150-166.
- Felton R. F., Hudnut, A ., & Heeckren, V. (1996). Putting a Value on Board Governance. 4 McKinsey Quarterly, 170-175. Retrieved on September 2011, from https://www.mckinseyquarterly.com/PDFDownload.aspx?ar=191.
- Feldman, D. C., & Arnold, H. J. (1978). Position choice: Comparing the importance of organizational and job factors. *Journal of Applied Psychology*, 63(6), 706–710.
- Field, A. (2009). Discovering Statistics Using SPSS. SAGE Publications Ltd.
- Firth, M. (1980). A note on the impact of audit qualifications on lending and credit decisions. *Journal of Banking and Finance.* 4(3), 257-67.
- Fontana, A., & Frey, J. H. (2003). The Interview: From Structured Questions to Negotiated Text. In N. K. Denzin, & Y. S. Lincoln (Eds.), *Collecting and Interpreting Qualitative Materials*, (2nd ed., pp. 61-105). London, UK: Sage Publications Inc.
- Ganzach, Y. (1994). Theory and configurality in expert and layperson judgment. *Journal of Applied Psychology, 79*, 439-448.
- Ganzach, Y. (1997). Theory and configurality in clinical judgments of expert and novice psychologists. *Journal of Applied Psychology*, *82*(6), 954–960.
- García Benau, M.A., Humphrey, C., Moizer, P. and Turley, S. (1993). Auditing Expectations and Performance in Spain and Britain: A Comparative Analysis, *The International Journal of Accounting*, 28, 281-307.
- Gay, G., & Simnett, R. (2010). Auditing and Assurance Services in Australia. McGraw-Hill Australia.
- Gay, L. R., & Diehl, P. L. (1992). *Research methods for business and management*. Macmillan Pub. Co.
- Gibbins, M., & Swieringa, R. J. (1995). Twenty years of judgment research in accounting and auditing. In Judgment and Decision-making Research in Accounting and Auditing.
  Cambridge University Press. Retrieved from http://dx.doi.org/10.1017/CB09780511720420.011
- Gendron Y., and Bedard J., (2006). On the constitution of audit committee effectiveness. Accounting, Organizations and Society 31 (April) 211-239.
- Gill, J., & Johnson, P. (2010). Research Methods for Managers. SAGE.
- Gloeck, JD., and Jager, H de., (1993). The focus point of the audit expectation gap in the Republic of South Africa, *Meditari*, 1-42. Retrieved from: http://www.up.ac.za/dspace/bitstream/2263/11197/1/Gloeck\_Focus%281993%29.pd f
- Goldberg, L. R. (1968). Simple Models or Simple Processes? Some Research on Clinical Judgments. *American Psychologist*, 23(7), 483–496.

- Gómez-Guillamón, A. D., (2003). The usefulness of the audit report in investment and financing decisions. *Managerial Auditing Journal, 18* (6/7), 549 559. DOI:10.1108/02686900310482687
- Gompers, P. A., Ishii, J. L., & Metrick, A. (2001). *Corporate Governance and Equity Prices*. National Bureau of Economic Research Working Paper Series, No. 8449. Retrieved from http://www.nber.org/papers/w8449
- Goodwin, J. (2003). The Relationship Between the Audit Committee and the Internal Audit Function: Evidence from Australia and New Zealand. *International Journal of Auditing*, 7(3), 263–278. doi:10.1046/j.1099-1123.2003.00074.x
- Goodwin, J., and Seow, J. L (2000). Corporate governance in Singapore: perceptions of investors, directors and auditors. *Accounting and Business Review* 7(1), 39-68.
- Gramling A. A., Maleta M.J., Schmeider A., and Church B. K. (2004). The role of internal audit function in corporate governance: a synthesis of the extant internal auditing literature and directions for future research. *Journal of accounting literature, 2004*(23), 194-244.
- Gray, I. and Manson, S. (2008) *The Audit Process: Principles, Practice and Cases*. (4<sup>th</sup> edition), Thomson Learning, London : UK.
- Green, D. L. (2006). Auditor Independence in Canada: A Historical Perspective From Shareholder Auditors to Modern-Day Audit Committees. *Canadian Accounting Perspectives*, 5(1), 37–65. doi:10.1506/CGMP-AV06-YWG9-RJ40
- Greene, J. C., Caracelli, V. J., & Graham, W. F. (1989). Toward a Conceptual Framework for Mixed-Method Evaluation Designs. *Educational Evaluation and Policy Analysis*, 11(3), 255–274.
- Gregary H. (2000). The Globalisation of Corporate Governance. Retrieved on March, 2012 from: http://rru.worldbank.org/documents/paperslinks/globalisation\_of\_corporate\_govern ance.pdfl
- Gregory H. J., & Simms M. E. (1999). Corporate governance: what it is and why it matters. Working paper presented at the 9th International Anti-Corruption Conference, 10-15 October 1999, Durban, South Africa. Retrieved on November 2011, from http://9iacc.org/papers/day2/ws3/dnld/d2ws3\_hjgregorymesimms.pdf
- Guillamon, G. (2003). The usefulness of the audit report in investment and financing decisions. *Managerial Auditing Journal, 18*(6/7), 549-59.
- Gul, F.A. (1990). Qualified audit reports, field dependence, cognitive style, and their effects on decision making. Accounting and Finance. 30(2), 15-27.
- Gul, F. A. (1991). Size of Audit Fees and Perceptions of Auditors'Ability to Resist Management Pressure in Audit Conflict Situations. *Abacus, 27*(2), 162–172. doi:10.1111/j.1467-6281.1991.tb00264.x
- Gupta. (2004). *Contemporary Auditing*, (6<sup>th</sup> ed). Tata McGraw-Hill Education.

- Hajir, I., (2001). *Expectations gap and the responsibilities of auditors: Jordanian Investors' perspective*, Unpublished Master thesis, Jordanian university, Amman-Jordan.
- Hakim, S. R., (2002). Islamic Banking, Challenges and Corporate Governance, LARIBA. Retrieved on February 2011 from: http://www.kantakji.com/fiqh/Files/Manage/90086.pdf
- Hamdan, M. (1996). The Degree of Auditors' Implementation of necessary Procedures for Assessing Corporate Failure and Financial Problems, Unpublished Master thesis, The University of Jordan. Amman - Jordan www.ju.edu.jo
- Haniffa R., and Hudaib M., (2007) Locating audit expectations gap within a cultural context: The case of Saudi Arabia, Journal of the International Accounting. *Auditing and Taxation*, *16* (2007) 179–206
- Harris, L. & Spannier, R. (1976). Accountability, answerability and liability. *Journal for the Theory* of Social Behaviour, 6(3), 253-260.
- Hassan, T., (2004). Chairman, Securities and Exchange Commission of Pakistan, Speech delivered at the ICAP conference in Lahore Pakistan on 16 April 2004, retrieved on February 2011 from: http://www.secp.gov.pk/CS/ChairmanSpeeches/PDF/160404 ICAP Lhr.pdf
- Hatherly, D. & Skuse, P. (1991), Audit Reports, in Sherer, M. and Turley, S. *Current Issues in Auditing*, (2nd ed.). London. Paul Chapman Chapman Publishing Ltd, 115-130.
- Healy, J. (2003). *Corporate governance & wealth creation in New Zealand*. Palmerston North, N.Z.: Dunmore Press.
- Heberlein, T. A., & Baumgartner, R. (1978). Factors Affecting Response Rates to Mailed Questionnaires: A Quantitative Analysis of the Published Literature. American Sociological Review, 43(4), 447–462. doi:10.2307/2094771
- Heppner, P. P., Wampold, B. E., & Kivlighan, D. M. (2008). *Research Design In Counseling*. Cengage Learning.
- Herbohn, K., Ragunathan, V., & Garsden, R. (2007). The horse has bolted: revisiting the market reaction to going concern modifications of audit reports. Accounting & Finance, 47(3), 473–493. doi:10.1111/j.1467-629X.2007.00216.x
- Higgs Report, (2003). *Review of the role and effectiveness of non-executive directors.* Department of Trade and Industry. London. UK
- Hilbert, M. (2012). Toward a synthesis of cognitive biases: How noisy information processing can bias human decision making. *Psychological Bulletin*, *138*(2), 211–237. doi:10.1037/a0025940
- Hoffman, P. J., Slovic, P., & Rorer, L. G. (1968). An analysis-of-variance model for the assessment of configural cue utilization in clinical judgment. *Psychological bulletin, 69*(5), 338–349. doi:10.1037/h0025665
- Hofstedt, T. R., & Hughes, G. D. (1977). An Experimental Study of the Judgment Element in Disclosure Decisions. *The Accounting Review*, *52*(2), 379–395.

- Hoitach, R., Hoitach, U. (2009). The role of audit committees in managing relationships with external auditors after SOX, Evidence from the USA, *Managerial Auditing Journal*, 24(4), 368-397.
- Hooper, C., & Trotman, K. T. (1996). Configural Information Processing in Auditing: Further Evidence. *Accounting & Business Research*, *26*(2), 125–136.
- Hopkins, M. (2007). Corporate social responsibility and international development: is business the solution? Earthscan.
- Hopkins, M. (2009). *Determinants of private equity firm certification at initial public offering*. Unpublished PhD thesis. Macquarie Graduate School of Management. Macquarie University. Australia.
- Howe, K. R. (1988). Against the Quantitative-Qualitative Incompatibility Thesis or Dogmas Die Hard. *Educational Researcher*, *17*(8), 10–16. doi:10.3102/0013189X017008010
- Humphrey, C. (1997). Debating Audit Expectations, In M. Sherer and, S. Turley (3rd ed.), *Current Issues in Auditing*. London: Paul Chapman Publishing Ltd.
- Humphrey, C., Moizer, P., & Turley, S. (1992). The Audit Expectations Gap in the United Kingdom. Institute of Chartered Accountants in England and Wales, Research Board.
- Humphrey, C., Moizer P., and Turley, S (1993). The Audit Expectation Gap in Britain: An empirical investigation. *Accounting and Business Research*, 23(91A), 395-411
- Hunter, J. E., & Coggin, T. D. (1988). Analyst judgment: The efficient market hypothesis versus a psychological theory of human judgment. *Organizational Behavior and Human Decision Processes*, 42 (3), 284–302.
- Hurmerinta-Peltomäki, L., & Nummela, N. (2006). Mixed Methods in International Business Research: A Value-added Perspective. *MIR: Management International Review, 46*(4), 439–459.
- IFAC, (2010a). International Standard on Auditing 260 "Communications of audit matters with those charged with governance" International Federation of Accountants, New York http://web.ifac.org/download/a014-2010-iaasb-handbook-isa-260.pdf
- IFAC, (2010b). International Standard on Auditing ISA. "Glossary of Terms". International Federation of Accountants, Retrieved on December 2011 from: http://www.ifac.org/publications-resources/2010-handbook-international-qualitycontrol-auditing-review-other-assurance-a
- IIA (2009), Practice advisories, Institute of Internal Auditors (IIA), Retrieved from: www.theiia.org/ on December 2009.
- IIA (2011). Glossary of International Standards of January 2011. The Chartered Institute of Internal Auditors (IIA). Retrieved on November 2011 from: http://www.iia.org.uk/en/Knowledge\_Centre/global\_professional\_guidance/internati onal-standards/2011Standards\_glossary.cfm

- IIA (2012). What are the principles of good corporate governance? The Chartered Institute of Internal Auditors (IIA). Retrieved on January 2012 from http://www.iia.org.uk/en/Knowledge\_Centre/Resource\_Library/corporategovernance.cfm
- Imhoff, E. (2003). Accounting Quality, Auditing and Corporate Governance. Accounting Horizon: special issue in accounting quality. Supplement 2003, 117-128. (SSRN Scholarly Paper No. ID 374380). Rochester, NY: Social Science Research Network. Retrieved from http://papers.ssrn.com/abstract=374380
- Instructions for Issuing Companies Disclosure, Accounting and Auditing Standards (2004). Jordan Securities Commission, The Hashemite Kingdom of Jordan. Retrieved on May 2011 from http://www.jsc.gov.jo/Public/English.aspx?Site\_ID=1&Page\_ID=717
- International Chamber of Commerce. (2011). International Chamber of Commerce The world Business Organization "Disclosure and Transparency". Retrieved on March 2012, from http://www.iccwbo.org/corporate-governance/id3081/index.html
- ISB (2000). Statement of Independence Concepts: A Conceptual Framework for Auditor Independence, Exposure Draft ED 00-2, Independence Standards Board .New York: American Institute of Certified Public Accountants (AICPA). Retrieved on October 2011 from http://business.illinois.edu/doogar/303/Articles/ED002.pdf
- Ishhadat, M., & Abed Al-Jaleel, T. (2012). The Effect of Corporate Governance Principles Adoption in The Institutional Investor's Decisions in Amman Bourse. *Jordan Journal of Business Administration, 8*(1). Retrieved from http://journals.ju.edu.jo/JJBA/article/view/2802
- Jarbou, Y. (2004). The Expectation Gap between the Financial Community and the External Auditors and the Methods of How To Narrow This Gap. *Islamic University Journal, series of humanitarian studies, 12*(2), 367-389.
- Jarbou, Y. (2007). The Expectation Gap between the Financial Community and the External Auditors and How to Narrow This Gap - Analytical Study of: External Auditors, Banks Managers, Income Tax Employees' Point of View In Gaza Province-Palestine, retrieved On 12 January 2011 from: http://site.iugaza.edu.ps/yjarbou/files/2010/02/The Expectation Gap.pdf
- Jensen, M. C., & Meckling, W. H. (1976). Theory of the firm: Managerial behavior, agency costs and ownership structure. *Journal of Financial Economics*, 3(4), 305–360.
- Jesover F & Kirkpatrick G, (2005). *The Revised OECD Principles of Corporate Governance and their relevance to Non-OECD Countries* (2004). Working paper - Corporate Governance: an International Review in January 2005. Retrieved on August 2011, from http://www.oecd.org/dataoecd/41/38/33977036.pdf
- Johnson, B., & Christensen, L. B. (2010). *Educational Research: Quantitative, Qualitative, and Mixed Approaches* (4th ed.). SAGE.

- Johnson, R. B., & Onwuegbuzie, A. J. (2004). Mixed methods research: A research paradigm whose time has come. *Educational Researcher*, 33(7), 14-26. doi:10.1177/1558689806298224
- Johnson, R. B., Onwuegbuzie, A., & Turner, L. (2007). Toward a definition of mixed methods research. *Journal of Mixed Methods Research*, 1(2), 112-133.
- Johnstone, P. L. (2004). Mixed methods, mixed methodology health services research in practice. *Qualitative health research*, 14(2), 259–271. doi:10.1177/1049732303260610
- Jordanian Forum for Economic Development. (2004). *The Policy Debate on Economic Issue*, special issue for corporate governance in Jordan. (5, Summer), 1-12. Amman- Jordan. Retrieved on May 2010, from: http://www.ujrc-jordan.net/pdf/economic\_policy/05-2004,Arb.pdf
- Jones, F.L. (1996). The information content of the auditor's going concern evaluation. *Journal of Accounting & Public Policy. 15*(1), 1-27.
- Jordan's Membership WTO. (2011). Jordan's Membership to the World Trade Organization. Retrieved on August 2011, from: http://www.mit.gov.jo/portrals/0/tabid/514/Jordan%27s%20Membership%20to%20 WTO.aspx
- Jordan Securities Commission, (1998). Instructions on Disclosure and Accounting Standards and Auditing Standards and Conditions to be Met by Auditor of Parties Under the Supervision of the Jordan Securities Commission (No. 1 - 1998). The Hashemite Kingdom of Jordan.
- Jubb, C. Rittenberg, L. Johnstone, K. Gramling, A. (2012). *Auditing and assurance: a business risk approach*. South Melbourne, Vic.: Cengage Learning.
- Jumaa A. (1999). Toward effective and sufficient audit committees to support the strategic control of the Jordanian shareholding companies. Faculty of Commerce. Al-Mansoura University. *Egyptian journal for commercial studies 23*(1), 137-191.
- Kahneman, D. (2011). *Thinking, Fast and Slow*. Farrar, Straus and Giroux.
- Kahneman, D., & Tversky, A. (1972). Subjective probability: A judgment of representativeness. *Cognitive Psychology*, 3(3), 430–454.
- Keats, B. W. (1988). The Vertical Construct Validity of Business Economic Performance Measures. The Journal of Applied Behavioral Science, 24(2), 151–160. doi:10.1177/0021886388242002
- Kelly, A. S., and Mohrweis, L. C., (1989). Bankers and investors perceptions of the auditors' role in financial statement reporting: *The impact of SAS No. 58, Journal of practice and theory. 9*(1), 87-79
- Keppel, G. (1982). *Design and Analysis: A Researcher's Handbook* (2nd ed.). Englewood Cliffs, NJ: Prentice Hall.

- Keppel, G., & Wickens, T. D. (2004). *Design and Analysis: A Researcher's Handbook* (4th ed.). Pearson.
- Kerlinger, F. N. (1986). *Foundations of behavioral research* (3rd ed.). New York: Holt, Rinehart & Winston
- Kevin, J. (2003). The Effects of Internal Audit Structure on Perceived Financial Statement Fraud Prevention. *Accounting Horizons.* 17 (4), 1-17
- Khaza'alah (2001). The directorate of companies control moves towards activating its role by studying the situation of 200 publicly traded Companies. *Al-Rai Daily newspaper* 11/4/2001, p 26.
- Khouri, N.S. (1994). The Evolution of the Audit Profession in Jordan. Al-Iqtisadi Al-Urduni The Jordanian Economic. Special issue of January/February, 82-83.
- Khsharmh, H., (2003). Factors Affecting Auditors' Selection in Jordanian Public Shareholding Companies – An Empirical Study. *Dirasat Journal, Administrative sciences 30*(1), 1-22
- Kida, T., Cohen, J., & Paquette, L. (1990). The Effect of Cue Categorization and Modeling Technique on the Assessment of Cue Importance. *Decision Sciences*, 21(2), 357–372. doi:10.1111/j.1540-5915.1990.tb01690.x
- Kieso, D. E., Waygandt, J. J., & Warfield, T. D. (2010). *Intermediate Accounting*. (13 ed). Wiley Publisher. USA.
- Klein, A. (2006). Audit Committee, Board of Director Characteristics, and Earnings Management. Law and Economic Research Paper Series. Working Paper No. 06-42. SSRN eLibrary. Retrieved from http://ssrn.com/paper=246674
- Knapp, M. (1985). Audit Conflict: An Empirical Study of the Perceived Ability of Auditors to Resist Management Pressure. *The Accounting Review*, *9*(2), 202-211.
- Koh, H.C., and Woo, E., (1998). The expectation gap in auditing. *Managerial Auditing Journal* 13(3), 147–154
- Knutson, P.H. (1994). In the public interest is it enough?, CPA Journal, 64(January), 32-4.
- Kirkpatrick G. (2005). Improving Corporate Governance Standards: The Work of the OECD and the Principles. Retrieved on August 2011 from: www.oecd.org/dataoecd/45/24/33655111.pdf
- Krishnamoorthy, G. (2002). A Multistage Approach to External Auditors' Evaluation of the Internal Audit Function. *Auditing Journal of Practice & Theory, 21*(1). Retrieved from http://ssrn.com/paper=285359
- Krishnan, G. V., & Visvanathan, G. (2007). Reporting Internal Control Deficiencies in the Post-Sarbanes-Oxley Era: The Role of Auditors and Corporate Governance. International Journal of Auditing, 11(2), 73–90. doi:10.1111/j.1099-1123.2007.00358.x
- Law of Regulating the auditing profession No. 73, (2003). Released on the Jordanian official gazette No. 4604 on 16/June/2003 page 3293. Retrieved on August 2011 from: http://www.mit.gov.jo/Default.aspx?tabid=163

- Leong, F. T. L., & Austin, J. T. (2005). *The Psychology Research Handbook: A Guide for Graduate Students and Research Assistants*. SAGE.
- Leung, P., & Cooper, B. J. (2005). Accountants, Ethical Issues and the Corporate Governance Context. Australian Accounting Review, 15(35), 79–88. doi:10.1111/j.1835-2561.2005.tb00255.x
- Leung, P., Cooper, B. J., & Perera, L. (2011). Accountability structures and management relationships of internal audit: An Australian study. *Managerial Auditing Journal*, 26(9), 794–816. doi:10.1108/02686901111171457
- Leung, P., Paul, C. P., Cooper B. J., and Richardson. (2011). *Modern auditing & assurance services*, (4th ed.) Australia: John Wiley and Sons.
- Levitt, A., (1998). The Numbers Game, Speech by Chairman Arthur Levitt, U.S. Securities and Exchange Commission, at New York University's Center for Law and Business on SEPTEMBER 28, 1998. New York, NY: SEC. Retrieved from: http://www.sec.gov/news/speech/speecharchive/1998/spch220.txt
- Levitt, A. (1999). An Essential Next Step in the Evolution of Corporate Governance. Speech to the Audit Committee Symposium on June 29, 1999. NewYork. Retrieved on April 2011 from http://www.sec.gov/news/speech/speecharchive/1999/spch289.htm
- Libby, R. (1981). Accounting and human information processing: Theory and applications. Englewood Cliffs: Prentice Hall.
- Libby, R., Artman, J. T., & Willingham, J. J. (1985). Process susceptibility, control risk and audit planning. *Accounting Review*, 60(2), 212–230.
- Libby, R., & Blashfield, R. K. (1978). Performance of a composite as a function of the number of judges. *Organizational Behavior and Human Performance*, 21(2), 121–129.
- Libby, R., & Lewis, B. L. (1982). Human information processing research in accounting: The state of the art in 1982. Accounting, *Organizations and Society*, 7(3), 231–285. doi:10.1016/0361-3682(82)90004-6
- Libby, R., Libby, P. A., & Shor, D. G. (2001). *Financial Accounting* (3rd. ed). McGraw-Hill Companies, Incorporated
- Liggio, C.D. (1974). The Expectation Gap: *The Accountant's Legal Waterloo Journal of Contemporary Business.* 3(3), 27-44.
- Lin, Z. J., & Chen, F. (2004). An Empirical Study of Audit "Expectation Gap" in The People's Republic of China. *International Journal of Auditing*, 8(2), 93-115. doi:10.1111/j.1099-1123.2004.00084.x
- Lindlof T. R. and Taylor, B. C. (2002). Qualitative Communication Research Methods (second Ed.), Sage Publications, Thousand Oaks, CA (2002).
- Lipsey, M. W. (1989). Design Sensitivity: Statistical Power for Experimental Research. SAGE.
- Low, C. K. (2002). Corporate governance: an Asia-Pacific critique. Sweet & Maxwell Asia.

- Lutfi, M, M,. (2001). Impression of audit report on credit officers. *Dirasat: Administrative Sciences*, 28(2), 395-406
- Luyt, R. (2012). A Framework for Mixing Methods in Quantitative Measurement Development, Validation, and Revision A Case Study. *Journal of Mixed Methods Research*, 6(4), 294– 316. doi:10.1177/1558689811427912
- Malaysian Institute of Accountants. (1990). A case for the establishment of audit committees (A memorandum submitted to the Registrar of Companies, the Capital Issues Committee and the Kuala Lumpur Stock Exchange, reproduced in summary form in) National Accountant (Malaysia), October, 25-27.
- Maletta, M. J., & Kida, T. (1993). The Effect of Risk Factors on Auditors' Configural Information Processing. *The Accounting Review*, *68*(3), 681–691. doi:10.2307/248209
- Malhas, M. (1992). Some Worries of Auditors. Al-Mudaqeq. *The periodic of the Jordanian* Association of Certified Public Accountants (17) 14-15
- Malkawi, B, (2008). Building of a Corporate Governance System in Jordan: A Critique of the Current Framework. *The journal of business law (UK). 6*, 488-507.
- Martinov-Bennie, N., Cohen, J., & Simnett, R. (2011). Impact of the CFO's affiliation on auditor independence. *Managerial Auditing Journal, 26*(8), 656–671. doi:10.1108/02686901111161322
- Mason, J. (1996) Qualitative Research, Thousand Oaks: Sage Publications.
- Maswada, S, N. (1992). The extent to which financial companies (Banks and financial institutions) rely on the audited financial statements issued by the Jordanian listed companies for investment and credit decisions making. Unpublished Master Thesis. Jordanian University- Amman, Jordan.
- Matar, M. (1994). Means of reinforcing external auditor independence in Jordan: An Empirical Analytical Study of Views of Auditors and Users of Financial Information and the Profession' Supervisory and Control Authorities. *Dirasat: Humanitarian Studies 21*(3), 321-373.
- Matar M., (1995). The Accounting Profession in Jordan. *The Banks in Jordan*, (June, No. 5), 20-27, Amman-Jordan.
- Matar, M. (2000). The Extent of the Auditor's Responsibility for Assessing Entities' Ability to Continue as a Going Concern. *Albasaer* 4(2) September, 143-198.
- Matar, M. (2003). The role of accounting disclosure in reinforcing corporate governance. Working paper presented at the 5th scientific conference of Jordanian Association of Certified Public Accountants held in Jordan. Amman during 24 - 25 September 2003.
- Matar, M., & Nour, A. (2007). The Compliance of Jordanian Shareholding Companies with the Principles of Corporate Governance: An Analytical Comparative Study between the Banking and the Industrial Sectors. *Jordan Journal of Business Administration*, 3(1). Retrieved from http://journals.ju.edu.jo/JJBA/article/view/1350

- Mautz, R.K. and Neary, R.D. (1979) Corporate Audit Committee- Quo Vadis? *Journal of Accountancy* 148(4) 83-88.
- Mautz, R. K. (1964). Fundamentals of auditing. (2nd ed.) John Wiley.
- McEnroe, J. & Martens, S. (2001). Auditors' and Investors' Perceptions of the Expectation Gap. Accounting Horizons. 15(4), 345-358.
- McGhee, W., Shields, M. D., & Birnberg, J. G. (1978). The effects of personality on a subject's information processing. *The Accounting Review*, *53*(3), 681–697.
- Mcgrath, S., Siegel, A. Dunfee, T. W., Glazer, A. S., and Jaenicke, H. R. (2001): A framework for auditor's independence. *Journal of Accountancy Online; AICPA*, January. Retrieved on October 2012 from: http://www.journalofaccountancy.com/Issues/2001/Jan/AFrameworkForAuditorInde pendence.htm
- McKinsey & company, (2000). McKinsey & Company Investor Opinion Survey on Corporate Governance. Retrieved on September 2011, from: http://www.oecd.org/dataoecd/56/7/1922101.pdf
- McMullen, D. A. (1996). Audit committee performance: An investigation of the consequences associated with audit committees. *Auditing: A Journal of Practice & Theory 15* (Spring), 87-103.
- Mear, R. W. T., & Firth, M. A. (1987). Mear, R., & Firth, M. (1987). Cue Usage and Self-Insight of Financial Analysts. *The Accounting Review*, *62*(1), 176–182. doi:10.2307/248053
- MENA Corporate Governance Workshop, (2003). Middle East and North Africa Regional Corporate Governance Workshop. Retrieved on September 2011 from: www.gcgf.org/ifcext/cgf.nsf/AttachmentsByTitle/MENA\_Sep\_03\_CG\_in\_MENA\_count ries/\$FILE/MENA\_Sep\_03\_MENA\_CG\_Report.pdf
- MENA-OECD investing program, (2005). Advancing the corporate governance agenda in the middle east and north Africa: A survey study of legal and institutional frameworks, Working Group 5. Retrieved on October 2011 from: <u>http://www.oecd.org/dataoecd/43/59/38186933.pdf</u>
- Megat, N. K. (2000). Board committees important mechanisms for listed companies. *Business Times*. Retrieved from http://search.proquest.com/docview/266857548?accountid=8194
- Merkle, E. C. (2009). The disutility of the hard-easy effect in choice confidence. *Psychonomic Bulletin & Review*, 16(1), 204–213.
- Meyers, L. S., Gamst, G., & Guarino, A. J. (2005). *Applied Multivariate Research: Design and Interpretation.* SAGE.
- Miller, W. L., & Crabtree, B. F. (1999). *Doing Qualitative Research*. Thousand Oaks, CA: Sage Publications.
- Millstein I. (1998). The Evolution of Corporate Governance in the United States. Remarks to the World Economic Forum, that was held in Davos Switzerland on February 2, 1998.

- Milne, M. J., & Chan, C. C. (1999). Narrative corporate social disclosures: How much of a difference do they make to investment decision-making? *The British Accounting Review*, 31 (4), 439–457. doi:10.1006/bare.1999.0108
- Mitchell, M. L., & Jolley, J. M. (1992). *Research Design Explained*. Harcourt Brace Jovanovitch College Publisher. Forth Worth
- Mitchell, M. L., & Jolley, J. M. (2012). *Research Design Explained*. Cengage Learning.
- Mohamed, E., & Hussain, M. M. (2005). The Role of Audit Committees in Enhancing a Transparent Corporate Reporting. *Humanomics*, 21(1), 30–47. doi:10.1108/eb018899
- Momany, H., (1994). *The impact of non-audit services on the external auditor's independence in Jordan*. Unpublished Master Thesis, Yarmouk University, Irbid – Jordan.
- Monks, R. A. G., & Minow, N. (2011). Corporate Governance. (5th ed.). John Wiley & Sons.
- Morck, R. K. (Ed.). (2006). A History of Corporate Governance around the World: Family Business Groups to Professional Managers. University Of Chicago Press.
- Morgan, D. (2007). Paradigms Lost and Pragmatism Regained: Methodological Implications of Combining Qualitative and Qualitative Methods, *Journal of Mixed Methods Research* 1(1), 48-76.
- Muneer, A. M., (2010). *Improving communication: An approach to narrow the audit expectations gap; An Empirical Study*. Unpublished Master thesis. Assiut University Egypt.
- Myers, J. L., & Well, A. D. (1995). *Research Design and Statistical Analysis*. Lawrence Erlbaum Associate Publisher, UK.
- Myers, J. L., & Well, A. D. (2002). *Research Design and Statistical Analysis*. Psychology Press.
- Nagy, A. L. (2005). Mandatory Audit Firm Turnover, Financial Reporting Quality, and Client Bargaining Power: The Case of Arthur Andersen. *Accounting Horizons*, 19(2), 51–68. doi:10.2308/acch.2005.19.2.51
- Naser, K., & Nuseibeh, R. (2007). Determinants of audit fees: empirical evidence from an emerging economy. *International Journal of Commerce and Management*, 17(3), 239-254. doi:10.1108/10569210710833635
- Ng, T. B., & Tan, H. (2003). Effects of Authoritative Guidance Availability and Audit Committee Effectiveness on Auditors' Judgments in an Auditor-Client Negotiation Context. *The Accounting Review, 78*(3), 801–818. doi:10.2308/accr.2003.78.3.801
- Nguyen, H. H., & Ross, D. G. (2006). Project finance asset risk pricing decisions: Australian evidence. Macquarie University Research Online, Project finance asset risk pricing decisions: Australian evidence. Retrieved from http://www.researchonline.mq.edu.au/vital/access/manager/Repository/mq:10546
- Nykiel, R. (2007). Handbook of Marketing Research Methodologies for Hospitality and Tourism. Routledge.

- NYSE and NASD, (1999). New York Stock Exchange and National Association of Securities Dealers. Report and Recommendations of the Blue Ribbon Committee on Improving the Effectiveness of Corporate Audit Committees, New York: New York Stock Exchange and National Association of Securities Dealers.
- Obaidat, A., (2007). Accounting Information Qualitative Characteristics Gap: Evidence from Jordan, *International Management Review*. 3(2), 26-32
- O'Brien, P. C., & Bhushan, R. (1990). Analyst Following and Institutional Ownership. Journal of Accounting Research, 28, 55–76.
- OECD, (2004) Organization for Economic Co-operation and Development. The OECD Principles of Corporate Governance. Retrieved from: http://www.oecd.org/dataoecd/32/18/31557724.pdf
- OECD Glossary, (2007). The OECD Glossary of Statistical Terms. Retrieved on March 2010 from: http://stats.oecd.org/glossary/glossaryPDF.zip
- OECD, (2009) website. Organization for Economic Co-operation and Development. Retrieved on October 2009 from: www.oecd.org
- Ojo, M. (2009). Eliminating the Audit Expectations Gap: Reality or Myth? *Journal of Forensic Accounting*, 8(1, 2). (SSRN Scholarly Paper No. ID 1407199). Rochester, NY: Social Science Research Network. Retrieved from http://papers.ssrn.com/abstract=1407199
- Omari, B. (2003). *The expectation gap between investors and auditors in Jordan*. Unpublished Master thesis, Yarmouk university, Irbid-Jordan.
- Onwuegbuzie, A. J., & Leech, N. L. (2005). On becoming a pragmatist researcher: The importance of combining quantitative and qualitative research methodologies. International Journal of Social Research Methodology: *Theory & Practice, 8*, 375-387.
- O'Reilly, D. M. (2009). Do investors perceive the going-concern opinion as useful for pricing stocks? *Managerial Auditing Journal, 25*(1), 4-16. doi:10.1108/02686901011007270
- Pankaj, S., (2003). The relationship between corporate governance and management fraud: An empirical analysis. *Academy of accounting and financial studies*. 7(3), 21-37
- Parker, S. (2000). The association between audit committee characteristics and the conservatism of financial reporting. Working paper, Santa Clara University.
- Parkinson, J. E. (1995). *Corporate Power and Responsibility*: Issues in the Theory of Company Law. Clarendon Press.
- Payne, J. W. (1982). Contingent decision behavior. *Psychological Bulletin*, 92(2), 382–402.
- Pickett, K. H. S. (2010). The Internal Auditing Handbook (3rd ed.). John Wiley and Sons.
- Pierce, B., Kilcommins, M. (1997). The audit expectations gap: role of auditing education. *The Irish Accounting Review*. 7(2), 39-60.
- Pike, R., & Neale, B. (2006). Corporate Finance and Investment: Decisions & Strategies. Financial Times. (5th ed.). Prentice Hall.

- Pike, R., Sharp, J., & Kantor, J. (1988). The Role of Accounting Information in Valuing Unlisted Shares. *Accounting and Business Research, 18*(71), 249–255. doi:10.1080/00014788.1988.9729371
- Plessis, J. du, McConvill, J., & Bagaric, M. (2005). *Principles of Contemporary Corporate Governance*. Cambridge University Press.
- Porter, B. A. (1991). Narrowing the Audit Expectation-Performance Gap: A Contemporary Approach *Pacific Accounting Review.* 33(1), 1-36.
- Porter, B. A. (1993). An Empirical study of the Audit Expectation-Performance Gap. Accounting and *Business Research.* 24(93), 49-68.
- Porter, B. A. (2009). The audit trinity: the key to securing corporate accountability. *Managerial Auditing Journal*, 24(2), 156–182. doi:10.1108/02686900910924563
- Pott, C., Mock, T. J., & Watrin, C. (2009). Review of empirical research on rotation and nonaudit services: auditor independence in fact vs. appearance. *Journal für Betriebswirtschaft*, *58*(4), 209–239. doi:10.1007/s11301-009-0043-0
- Prentice, D. & Holland, P. (1993). *Contemporary Issues in Corporate Governance*, Oxford University Press, Oxford.
- Pronin, E. (2007). Perception and misperception of bias in human judgment. *Trends in Cognitive Sciences, 11*(1), 37–43. doi:10.1016/j.tics.2006.11.001
- Qin, Bo (2007), The Influence of Audit Committee Financial Expertise on Earnings Quality: U.S. Evidence. *The Icfai Journal of Audit Practice*, 4(3) 8-28, Available at SSRN: http://ssrn.com/abstract=799645
- Qadamani, I., (2007). White Revolution in Banks. *Ammon News*, retrieved from: http://www.ammonnews.net/article.aspx?articleno=6747
- Raghunandan, K., Rama, D. V., & Read, W. J. (2001). Audit Committee Composition, "Gray Directors," and Interaction with Internal Auditing. *Accounting Horizons*, 15(2), 105–118. doi:10.2308/acch.2001.15.2.105
- Rehana, F. (2010). An Empirical Study on Audit Expectation Gap: Role of Auditing Education in Bangladesh (*MPRA Paper No. 22708*). University Library of Munich, Germany. Retrieved from http://ideas.repec.org/p/pra/mprapa/22708.html
- Reichardt, C. S., & Cook, T. D. (1979). Beyond qualitative versus quantitative methods. In A. Bryman (Ed.), *Mixed Methods 1*, 213–237. Sage.
- Reuters (2009). Amman bourse resumes trade in Capital Bank shares, Retrieved on November 2009 from: http://www.reuters.com/article/idUSLJ58335420090719
- Rittenberg, L. E., & Schwieger, B. J. (2007). Auditing: Concepts for Changing Environment. (6th ed.) *Cengage Learning.*
- Ross, D. (1991). *The Trade Credit Offer for Expert Sales: Canadian Evidence and Experience.* Unpublished PhD thesis. University of Bradford, UK.

- Ryan, T. P. (2011). Design of Experiments. Statistical Methods for Quality Improvement (pp. 435–511). John Wiley & Sons, Inc.
- Saadah, Y.M. (1993). Legal Barriers facing the Roles of the Auditing Profession in Serving Development in Jordan. *Dirasat: Humanitarian Sciences, 20*(4), 144-171.
- Saaydah, M. (1998). The impact of the qualified external auditor's report on the on the share price and return. Abhath Al-Yarmouk Humanities and Social Sciences. 14(1), 163-191.
- Saeed H. (2003). Corpoarte Governance from a banking perspective. *Egyptian Banking Institute.* Cairo. Egypt, 2003. (3). Retrieved on October 2011 from http://www.ebi.gov.eg/downloads/EBI%20Lnk\_Basel%20II\_ARB\_V.Final\_104.08.pdf
- Salehi, M., Rostami, V. (2009). Audit Expectation Gap: International Evidence. *International Journal of Academic Research* 1(1), 141-147
- Sale, J. Lohfeld, L. & Brazil, K. (2002). Revisiting the quantitative-qualitative debate: Implications for mixed-methods research. In A. Bryman (Ed.), *Mixed Methods* (Vol. 1, pp. 133–144). London: Sage Publications.
- Schipper, K. (1991). Analysts' Forecasts. Accounting Horizons, pp. 105–121.
- Schneider, A. (1984). Modeling External Auditors' Evaluations of Internal Auditing. *Journal of Accounting Research (Autumn), 22* (2), 657–678.
- Schneider, A. (2011). Is investment decision-making influenced by perceptions relating to auditors' client dependence and amount of audit fees? *Advances in Accounting*, 27(1), 75–80. doi:10.1016/j.adiac.2010.10.006
- Schultz Jr., J. J., Bierstaker, J. L., & O'Donnell, E. (2010). Integrating business risk into auditor judgment about the risk of material misstatement: The influence of a strategicsystems-audit approach. Accounting, Organizations and Society, 35(2), 238–251. doi:10.1016/j.aos.2009.07.006
- Securities law. (2002). The Securities Law No. (76) for the Year 2002. Retrieved on August 2011 from: http://www.jsc.gov.jo/Public/English.aspx?Site\_ID=1&Page\_ID=1698
- Seghaier, Roua. (2013). Accountability Pact Seeks to Boost Anti-Corruption Efforts Tunisia Live,<br/>Retrieved March 16, 2013, from http://www.tunisia-<br/>live.net/2013/01/16/accountability-pact-seeks-to-boost-anti-corruption-efforts/
- Shadish, W. R., Cook, T. D., & Campbell, D. T. (2002). *Experimental and Quasi- Experimental Designs for Generalized Causal Inference*. Boston: Houghton Mifflin.
- Shanikat, M., & Abbadi, S. (2011). Assessment of Corporate Governance in Jordan: An Empirical Study. *Australasian Accounting Business and Finance Journal*, *5*(3), 93–106.
- Sharaby, N. (2010). The role of audit committees in narrowing the audit expectations gap in the commercial banks of Yemen An Empirical Study. Unpublished Master thesis, Menoufiya University Egypt.

- Sharar, Z. (2006) A comparative analysis of the corporate governance legislative frameworks in Australia and Jordan measured against the OECD Principles of Corporate Governance 2004 as an international benchmark, SJD, ePublications@bond, Faculty of Law
- Shehadeh, S., & Barghouthy, S. (2009). *Pillars of corporate governance and its role in controlling earnings management in the banking environment in the light of the global financial crisis.* Working paper presented at the international conference about 'the global financial crisis and corporate governance. University of Farahat Abbas, Algeria. Available at: <u>http://www.univ-ecosetif.com/seminars/financialcrisis/50.pdf</u>
- Sherer, M., & Turley, S. (1997). Current Issues in Auditing. SAGE.
- Sheridan, T., & Kendall, N. (1992). Corporate governance: an action plan for profitability and business success. Pitman Publishing. London.
- Shleifer, A., & Vishny R. (1997a). A Survey of Corporate Governance. *Journal of Finance 52*(2), 37–83.
- Shleifer, A., & Vishny, R. (1997b). Large Shareholders and Corporate Control. *Journal of Political Economy*, 94(3), 461-488. Retrieved from: http://dx.doi.org/10.1086/261385
- Sidani, Y. M., Olayan S. (2007). The audit expectation gap: evidence from Lebanon. *Managerial Auditing Journal.* 22(3), 288-302
- Siddiqui, J., Nasreen, T., and Lema, A. (2009). The audit expectations gap and the role of audit education: the case of an emerging economy. *Managerial Auditing Journal, 24*(6), 564 - 583
- Sikka, P., Puxty, T., Willmott, H. and Cooper, C. (1992). Eliminating the Expectations Gap? *Certified Research Report No. 28* (Chartered Association of Certified Accountants). June 1992.
- Silverman, D. (2009). Doing Qualitative Research. SAGE.
- Slovic, P. (1966). Cue-Consistency and Cue-Utilization in Judgment. *The American Journal of Psychology*, *79*(3), 427–434. doi:10.2307/1420883
- Slovic, P. (1969). Analyzing the expert judge: A descriptive study of a stockbroker's decision processes. *Journal of Applied Psychology*, 53(4), 255–263.
- Slovic, P. (1972). Psychological Study of Human Judgment: Implications for Investment Decision Making. *The Journal of Finance*, 27(4), 779–799.
- Slovic, P., Fleissner, D., & Bauman, W. S. (1972). Analyzing the Use of Information in Investment Decision Making: A Methodological Proposal. *The Journal of Business*, *45*(2), 283–301.
- Slovic, P., & MacPhillamy, D. (1974). Dimensional commensurability and cue utilization in comparative judgement. Organizational Behavior and Human Performance, 11(2), 172–194.
- Smith, M., (2003). Research Methods in Accounting, Thousand Oaks: Sage Publications.

- Snowball, D. (1986). Accounting laboratory experiments on human judgement: Some characteristics and influences. *Accounting, Organizations & Society, 11*(1), 47–69.
- Solomon, J., (2010). *Corporate Governance and Accountability*, (3rd Edition) West Sussex, UK: John Wiley & Sons Ltd.
- Solomon, I., & Shields, M. D. (1995). Judgement and decision making research in auditing. In R.
  H. Ashton & A. H. Ashton (Eds.), Judgment and decision making research in accounting and auditing. Cambridge M A: Cambridge University Press. Available at: http://dx.doi.org/10.1017/CBO9780511720420
- Solomon, I., & Trotman, K. (2003). Experimental judgment and decision research in auditing: The first 25 years of AOS. Accounting, *Organizations and Society*, *28*(4), 395–412
- Statement of Accounting Concepts SAC 2.5 (2012). "Objective of General Purpose Financial Reporting". Australian Accounting Standards Board. Retrieved on March 2012, from http://www.aasb.gov.au/admin/file/content102/c3/SAC2\_8-90\_2001V.pdf
- Statman, M. (2005). Normal investors, then and now. *Financial Analysts Journal*, 61(2), 31–37.
- Steginga, S. k., Occhipinti, S., Gardiner, R. a., Yaxley, J., & Heathcote, P. (2002). Making decisions about treatment for localized prostate cancer. *BJU International, 89*(3), 255–260. doi:10.1046/j.1464-4096.2001.01741.x
- Stocks, M. H., & Harrell, A. (1995). The impact of an increase in accounting information level on the judgment quality of individuals and groups. *Accounting, Organizations and Society,* 20(7–8), 685–700. doi:10.1016/0361-3682(95)00012-X
- Strawser, J. R., & Strawser, R. H. (2001). *Auditing: Theory and Practice*. Dame/Thomson Learning
- Strohm, C. (2006). United States and European Union Auditor Independence Regulation: Implications for Regulators and Auditing Practice. Springer.
- Sudsomboon, S., Ussahawanitchakit, P. (2009). Professional Audit Competencies: Effects on Thai CPAs' Audit Quality Reputation and Success. *Review of Business Research*, 9(3), 66-85
- Suleiman, S. (2011). Audit committees and their role in the control, Working paper available at The Arab Society of Certified Accountants (ASCA.). Retrieved on November 2011, from http://ascajordan.org/?lang=en
- Saunders, M. N. K., Lewis, P., & Thornhill, A. (2012). *Research Methods for Business Students*. Financial Times Prent.Int.
- Swaiti, M. (2006). A Proposed Model for the Role of Audit Committees in the Jordanian Public Shareholding Companies, and their Influence on the Effectiveness and Independence of External Audit. Unpublished Ph.D Thesis, Amman Arab University – Jordan
- Symon, G., & Cassell, C. (Eds.). (1998). *Qualitative Methods and Analysis in Organizational Research: A Practical Guide*. Sage Publications Ltd.
- Tafara, E. (2006). A race to the top: International regulatory reform post-Sarbanes Oxley. *International Financial Law Review.* 25(September), Available at http://www.sec.gov/news/speech/2006/spch091106et.htm
- Tashakkori, A., & Teddlie, C. (Eds.). (2003). *Handbook of mixed methods in social and behavioral research*. Thousand Oaks, CA: SAGE.
- Taylor, R. N. (1975). Age and experience as determinants of managerial information processing and decision making performance. *Academy of Management Journal, 18*(1), 74–81.
- Taylor, M. H., DeZoort, F. T., Munn, E., & Thomas, M. W. (2003). A Proposed Framework Emphasizing Auditor Reliability over Auditor Independence. Accounting Horizons, 17(3), 257–266.
- Teddlie, C., & Tashakkori, A. (2009). *Foundations of mixed methods research*. Thousand Oaks, CA: Sage.
- Teed, D. G. (2010). *The Changing Role and Responsibilities of Audit Committees in the United States.* Dissertation. University of North Texas. Retrieved January 16, 2012, from http://digital.library.unt.edu/ark:/67531/metadc30519/citation/
- Teoh, H. Y., & Lim, C. C. (1996). An empirical study of the effects of audit committees, disclosure of nonaudit fees, and other issues on audit independence: Malaysian evidence. *Journal of International Accounting, Auditing and Taxation, 5*(2), 231–248. doi:10.1016/S1061-9518(96)90007-5
- The Economist. (2003). The fall of a kingmaker. *The Economist*. Retrieved on 11/March/2012, from http://216.35.68.215/node/1930467
- The UK Corporate Governance Codes, (2010). Financial Reporting Council in the UK. Retrieved on January 2012 from http://www.frc.org.uk/documents/pagemanager/Corporate\_Governance/UK%20Corp %20Gov%20Code%20June%202010.pdf
- Thnaibat, A. A., & Shunnaq, B. K. (2010). External Auditor's Assessment of the Internal Audit Function in the Light of Implementation of the International Standard on Auditing No. (610): Analytical Study from the Point of View of the External and Internal Auditors in Jordan. Jordan Journal of Business Administration, 2(2), 187-212. Retrieved from http://journals.ju.edu.jo/JJBA/article/view/1313
- Tomar, S., & Bino, A. (2012). Corporate Governance and Bank Performance: Evidence from Jordanian Banking Industry. *Jordan Journal of Business Administration*, 8(2). Retrieved from http://journals.ju.edu.jo/JJBA/article/view/3092
- Torrance, H. (2012). Triangulation, Respondent Validation, and Democratic Participation in Mixed Methods Research. Journal of Mixed Methods Research, 6(2), 111–123. doi:10.1177/1558689812437185
- Tricker, R. (1984). Corporate Governance: Practices, Procedures and Powers in British Companies and Their Boards of Directors. Gower Press, Aldershot, UK.

- Tricker, R. (2009). The Ongoing Financial Crisis: Where Were the Auditors?. Oxford University Press blog. Retrieved on January 2012 from: http://blog.oup.com/2009/03/corporate governance/
- Trotman, K. T. (1996). *Research Methods for Judgement and Decision Making Studies in Auditing.* Coopers & Lybrand and the AAANZ.
- Trotman, K. T. (2005). DISCUSSION OF Judgment and Decision Making Research in Auditing: A Task, Person, and Interpersonal Interaction Perspective. *AUDITING: A Journal of Practice & Theory, 24*(s-1), 73–87. doi:10.2308/aud.2005.24.s-1.73
- TUAC, (2004). Trade Union Advisory Committee to the OECD (2004), The OECD Principles of Corporate Governance: An Evaluation of the 2004 Review by the TUAC Secretariat (2004), Retrieved on at 02 November 2011: <u>http://old.tuac.org/statemen/communig/TUACassesmentreview1004p.pdf</u>
- Tversky, A., & Koehler, D. J. (1994). Support theory: A nonextensional representation of subjective probability. *Psychological Review*, 101(4), 547–567.
- UNCTAD: Guidance on good practices in corporate governance disclosure (2006), ISBN 92-1-112704-1, page 25. Available at http://www.unctad.org/en/docs/iteteb20063\_en.pdf
- Veal, A. J. (2005). *Business Research Methods: A Managerial Approach*. Addison-Wesley Longman, Incorporated.
- Vinten, G., & Lee, C. (1993). Audit Committees and Corporate Control. *Managerial Auditing Journal*, 8(3). doi:10.1108/02686909310036232
- Wang, K. J., & Tuttle, B. M. (2009). The impact of auditor rotation on auditor–client negotiation. *Accounting, Organizations and Society, 34*(2), 222–243. doi:10.1016/j.aos.2008.06.003
- Warren, C. A. B., & Karner, T. X. (2009). Discovering Qualitative Methods: Field Research, Interviews, and Analysis (2nd ed.). Oxford University Press, USA.
- Weathington, B. L., Cunningham, C. J. L., & Pittenger, D. J. (2010). *Research Methods for the Behavioral and Social Sciences.* John Wiley & Sons.
- Wenger, G. C. (1999). Advantages Gained By Combining Qualitative and Quantitative Data in a Longitudinal Study. *Journal of Aging Studies*, 13(4), 369–376.
- Whittington, O.R. and Pany, K (2012). *Principles of Auditing and Other Assurance Services*. (18th ed.) McGraw-Hill.
- Wild, J. J. (1994). Managerial Accountability To Shareholders: Audit Committees And The Explanatory Power Of Earnings For Returns. *The British Accounting Review*, 26(4), 353– 374. doi:10.1006/bare.1994.1024
- Wilson, J., (2010). Essential of Business Research (1st Edition), Sage publications Ltd, London UK.
- Winer, B. J., Brown, D. R., & Michels, K. M. (1991). *Statistical principles in experimental design*. McGraw-Hill.

- Wolfensohn. (1999). Cited by the Encyclopedias of Corporate Governance in the Article on "What is Corporate Governance". *Financial Times*, 21 June.
- Wolf, F.M., Tackett, J.A. & Claypool, G.A. (1999). Audit Disaster Futures: Antidotes for the Expectation Gap?. *Managerial Auditing Journal*. 14(9), 468-478.
- Wood, D. (2002). The influence of environmental social controls on the capital investment decision-making of the firm: Australian Evidence. Unpublished PhD thesis. The University of Western Sydney, Australia.
- Wood, D. & Ross, D. G. (2006). Environmental social controls and capital investments: Australian evidence. *Accounting & Finance*, *46*(4), 677–695.
- World Bank: Accounting and Auditing, (2004). Report on the Observance of Standards and Codes (ROSC): Hashemite Kingdom of Jordan: Accounting and Auditing, available at: http://www.worldbank.org/ifa/rosc\_aa\_jor.pdf
- World Bank: Corporate Governance Assessment (2004). Report on the Observance of Standards and Codes (ROSC): Hashemite Kingdom of Jordan: Corporate Governance Country Assessment. JORDAN. June 2004 available at: http://www.worldbank.org/ifa/jor\_rosc\_cg.pdf
- World Bank ROSC experience in MENA and the World, (2005). Jordan Corporate Governance Assessment, Policy Practice, Corporate Governance Department - The World Bank, February 14, 2005, available at: http://www.oecd.org/daf/corporateaffairs/corporategovernanceprinciples/34918132.p df
- Wright, D. (1996). *Evidence on the relation between Corporate Governance characteristics and the quality of financial reporting*. working paper, University of Michigan, 1996.
- Wright, P. (1974). The harassed decision maker: Time pressures, distractions and the use of evidence. *Journal of Applied Psychology*, *59*(5), 555–561.
- Wright, W. F. (1977). Self-insight into the cognitive processing of financial information. *Accounting, Organizations and Society, 2*(4), 323–331.
- Yauch, C. & Steudel, H. (2003). Complementary Use of Qualitative and Quantitative Cultural Assessment *Methods, Organizational Research Methods, 6*(4), 465-481
- Yin, R. K. (2009). *Case study research: Design and methods* (4rd ed.). Applied Social Research Methods Series, 5, Sage Publications
- Yin, R. K. (2011). Applications of Case Study Research. SAGE.
- Zachary, M. A., McKenny, A., Short, J. C., & Payne, G. T. (2011). Family Business and Market Orientation Construct Validation and Comparative Analysis. *Family Business Review*, 24(3), 233–251. doi:10.1177/0894486510396871
- Zayadeen, K. (2006). A mechanism to compensate the labours of Jordan Magnesia. Retrieved on July 2011 from <u>http://alrai2.dot.jo/article/177992.html</u>

- Zingales, L. (1994). The Value of the Voting Right: A Study of the Milan Stock. Exchange Experience. *Review of Financial Studies 7* (Spring), 125–48.
- Zureigat, Q. M. (2006). The Effect of Modified Auditors Opinions on Shares Prices Evidence from Amman Stock Exchange. Unpublished PhD thesis. AABFS - Arab Academy for Banking and Financial Sciences. Amman - Jordan
- Zureigat, Q. M. (2010). The Effect of Modified Auditors Opinions on Shares Prices Evidence from Amman Stock Exchange. *Jordan Journal of Business Administration, 6*(2). Retrieved from http://journals.ju.edu.jo/JJBA/article/view/744
- Zureigat, Q. M. (2011a). The Effect of Corporate Governance Guide Issued by the Central Bank of Jordan in Improving Internal Audit Quality in Jordanian Banks. *Jordan Journal of Business Administration*, 7(2). Retrieved from <u>http://journals.ju.edu.jo/JJBA/article/view/2348</u>
- Zureigat, Q. M. (2011b). The Effect of Ownership Structure on Audit Quality: Evidence from Jordan, *International Journal of Business and Social Science*. 2(10) 38-46.

## Appendices

## Appendix 1: Ethics Approval

Human Research Ethics Committee

Committee Approval Form

Principal Investigator: Waleed Abdel-Qader

Student Researcher: Mohammad Shbeilat

Ethics approval has been granted for the following project:

The Jordanian Corporate Governance Code: A study of Institutional Investors Perception of the Reliability on the Audit Report and the Audit Expectation

for the period: 02/07/2012 - 30/06/2013

Human Research Ethics Committee (HREC) Register Number: 2012 116N

#### Special Condition/s of Approval

*Prior to commencement of your research,* the following permissions are required to be submitted to the ACU HREC:

N/A

# The following <u>standard</u> conditions as stipulated in the *National Statement on Ethical Conduct in Research Involving Humans* (2007) apply:

- (i) that Principal Investigators / Supervisors provide, on the form supplied by the Human Research Ethics Committee, annual reports on matters such as:
  - security of records
  - compliance with approved consent procedures and documentation
  - compliance with special conditions, and

- (ii) that researchers report to the HREC immediately any matter that might affect the ethical acceptability of the protocol, such as:
  - proposed changes to the protocol
  - unforeseen circumstances or events
  - adverse effects on participants

The HREC will conduct an audit each year of all projects deemed to be of more than low risk. There will also be random audits of a sample of projects considered to be of negligible risk and low risk on all campuses each year.

Within one month of the conclusion of the project, researchers are required to complete a *Final Report Form* and submit it to the local Research Services Officer.

If the project continues for more than one year, researchers are required to complete an *Annual Progress Report Form* and submit it to the local Research Services Officer within one month of the anniversary date of the ethics approval.

Jo Mushi Signed:

Date: 06/12/2012

(Research Services Officer, Melbourne Campus)



North Sydney Campus (Mackillop) Level 10, 8-20 Napier Street, Tenison House North Sydney NSW 2060 Australia PO Box 968 North Sydney NSW 2059 Telephone 612 9739 2361 Facsimile 612 9739 2088 www.acu.edu.au

#### INFORMATION LETTER TO PARTICIPANTS

TITLE OF PROJECT:	Corporate Governance and the Auditing Profession
PRINCIPAL SUPERVISOR:	Dr Waleed Abdel-Qader
STUDENT RESEARCHER:	Mohammad Shbeilat
PROGRAMME ENROLLED	Doctor of Philosophy

#### **Dear Participant**

You are invited to participate in a study about The Jordanian Corporate Governance Code. This survey is part of a PhD study being undertaken by Mohammad Shbeilat of the faculty of business at the Australian Catholic University. You are invited due to your experience as a financial analyst. The primary objective of this thesis is to improve understanding of how the implementation of the newly adopted Jordanian corporate governance code has affected the perceived reliability of an independent audit opinion and the extent to which the code might help in narrowing the audit expectation gap. The audit expectation gap refers to the different perceptions between the auditors' actual performance and to what report users believe the auditors should be performing and achieving as auditors. If you participate in this study, you will be invited to take part in a one-on-one interview with the researcher, in addition to the survey questionnaire.

As professional investors, the participants will share equally with other Jordanian investors the social benefits of improved knowledge of the influence of the Jordanian corporate governance code. We do not foresee any significant risk in participating in this study. The interview will not ask after details of your personal life, other private matters, or any organizational specific questions, but it will take on the form of a conversation about your experience of the Jordanian corporate governance and the auditing profession.

The duration of the interview will be between 45-60 minutes, where the time needed to fill in the questionnaire is approximately 20 minutes. No preparation on your part is necessary. The interview will be held at your office, alternatively, you can request for the interview to be held in a different location and whether to be conducted during work hours or during the break. The interviews will be recorded using a mobile recording device.

The purpose of this study is to fill the gap in knowledge surrounding institutional investors' perception in evaluating the impact of the newly adopted Jordanian Corporate Governance Code on the auditing profession and to Increase knowledge of how the elements of the Jordanian corporate governance codes affects the reliability of the audit report. It is expected that the results and the potential recommendations will also be of the benefit of the Jordanian capital market regulatory bodies.

Participation is voluntary and you are free to refuse consent or withdraw from project at any time without any penalty and without giving a reason.

This study will ensure the confidentiality of your Participation. No identifying factors such as name or contact details will be disclosed to anyone but the researcher. When the study is published, your name and the name of your workplace (or any other identifying information) will be replaced with pseudonyms, and may appear like this:

Katrina (pseudonym) worked at Organization A (pseudonym),,,,

Interview transcripts and recordings will be stored in the student laptop which has a security password while conducting the study in Jordan, and then in secure facilities at ACU, with no identifying information attached.

296

Please return the completed questionnaire in the envelope provided to: P.O.Box 410969 Jabal Al-Taj 11141 Amman – Jordan. Please indicate if you would like to receive a copy of a summary conclusions report on the published results at the end of the meeting or by answering question 4 of Part C of the questionnaire. If you would like further information about this study, please contact the student researcher or the Principal Supervisor at any time.

Principal Supervisor

Dr Waleed Abdel-Qader

Waleed.Abdel-Qader@acu.edu.au School of Business

<u>Australian Catholic University</u> T: +61 2 9739 2307 F: +61 2 9739 2088 Tenison House Level 10, 8-20 Napier Street, North Sydney NSW 2060

<u>Student Researcher</u> Mohammad Shbeilat <u>Mohammad.Shbeilat@acu.edu.au</u> School of Business/ Australian Catholic University T: +61 2 9739 2113 F: +61 2 9739 2088 Level 10, 8-20 Tenison House. North Sydney NSW 2060

Please be advised that this study has been approved by the Human Research Ethics Committee at Australian Catholic University. In the event you have any complaint or concern or if you have any query that the supervisor and Student Researcher have not been able to satisfy, you may write to the Chair of the Human Research Ethics Committee on the following address:

NSW and ACT: Chair, HREC

C/- Research Services

Australian Catholic University North Sydney Campus PO Box 968 NORTH SYDNEY NSW 2059 Tel: 02 9739 2105 Fax: 02 9739 2870 res.ethics@acu.edu.au

Any complaint or concern will be treated in confidence and fully investigated. The participant will be informed of the outcome.

If you agree to participate in this project, you should sign both copies of the Consent Form, retain one copy for your records and return the other copy to the Principal Supervisor or Student Researcher.

Dr Waleed Abdel-Qader

Mohammad Shbeilat

Principal Investigator

Student Researcher

## **Appendix 3: Letter to the Professional Association of Financial Analysts**



North Sydney Campus (Mackillop) Level 10, 8-20 Napier Street, Tenison House North Sydney NSW 2060 Australia PO Box 968 North Sydney NSW 2059 Telephone 612 9739 2361 Facsimile 612 9739 2088 www.acu.edu.au

Letter to the Professional Association of Financial Analysts

TITLE OF PROJECT: Corporate Governance and the Auditing Profession

PRINCIPAL SUPERVISOR): Dr Waleed Abdel-Qader

STUDENT RESEARCHER: Mohammad Shbeilat

PROGRAMME ENROLLED: Doctor of Philosophy

Dear Professional Association of the Financial Analysts in Jordan

Mohammad Shbeilat is a student researcher at the faculty of business at the Australian Catholic University and conducting a study about The Jordanian Corporate Governance Code and its impact on the auditing profession in Jordan. Mohammad is seeking permission to advertise the attached file (Information Letter to Participants) amongst the members of Professional Association of Financial Analysts in Jordan. Please refer to the attached Information Letter to Participants for more details about the study aims, location, and duration.

Dr Waleed Abdel-Qader

Mohammad Shbeilat

**Principal Investigator** 

Student Researcher

## **Appendix 4: Reply From CFA-Jordan**

Dear Mohammad,

"After consulting with our board we can help in disseminating the information letter to participants and the questionnaire to the contact list that we have, with any replies going back directly to you.

You would then have to manage any further correspondence on your own (interviews etc) directly based on the replies you get".

All the best,

President of CFA Society Jordan



North Sydney Campus (Mackillop) Level 10, 8-20 Napier Street Tenison House North Sydney NSW 2060 Australia PO Box 968 North Sydney NSW 2059 Telephone 612 9739 2361 Facsimile 612 9739 2088 www.acu.edu.au

#### CONSENT FORM

TITLE OF PROJECT: The Jordanian Corporate Governance Code: A study of Institutional Investors' Perception of the Reliability on the Audit Report and the Audit Expectation Gap

PRINCIPAL SUPERVISOR: Waleed Abdel-Qader

STUDENT RESEARCHER: Mohammad Shbeilat

I ...... have read and understood the information provided in the Letter to Participants. Any questions I have asked have been answered to my satisfaction. I agree to participate in this approximately one hour audiotaped interview or in the questionnaire that takes about 20 minutes to complete, realising that I can withdraw my consent at any time without adverse consequences. I agree that research data collected for the study may be published or may be provided to other researchers in a form that does not identify me in any way.

NAME OF PARTICIPANT:	
SIGNATURE	DATE:
SIGNATURE OF PRINCIPAL SUPERVISOR: Waleed Abdel-Qader	DATE:
SIGNATURE OF STUDENT RESEARCHER: Mohammad Shbeilat	DATE:

## Appendix 6: Data Collection - Experimental Survey



The Jordanian Corporate Governance Codes

&

The Auditing Profession Study

This questionnaire is part of a PhD study being undertaken by Mohammad Shbeilat of the faculty of business at the Australian Catholic University into the institutional investors' perceptions of the reliability of the audit report and the audit expectations gap

Your views will contribute greatly to the level and quality of information being gathered. Please complete all 3 parts of the questionnaire yourself and without discussion with colleagues.

Your responses and comments are strictly confidential. This questionnaire is anonymous unless you opt to provide contact details to receive a copy of the research report. No responses or comments will be individually attributed in any published report and any comments used will be de-identified. Participation in the study is entirely voluntary. The questionnaire should take less than 20 minutes to complete.

Please return the completed questionnaire in the envelope provided to:

P.O.Box 410969 Jabal Al-Taj 11141 Amman – Jordan

OR You can send a soft copy to mohammad.shbeilat@acu.edu.au

#### **COMPLETION INSTRUCTIONS – PART A**

Please read the following definitions in order to best visualize the scenarios presented.

#### The Jordanian Corporate Governance Codes (JCGC):

Refers to the JCGC that has been approved and issued by board of the commissioners of the Jordanian Securities Commission on the 29th of July 2008 for the publicly traded companies listed on the Amman Stock Exchange and which has come into effect as of the first of January 2009. The JCGC contains five chapters as follows: Definitions, The Board of Directors of the Shareholding Company, General Assembly Meetings, Shareholders Rights, and the Disclosure and Transparency respectively. The latter chapter has also discussed the roles and duties of the audit committees, external auditors and the internal auditors.

**External Auditors' Independence:** Freedom from those pressures and other factors that compromise, or can reasonably be expected to compromise, an auditor's ability to make unbiased audit decisions (ISB, 2000).

**Internal Audit Effectiveness** refers to the extent to which the designated internal audit objectives and functions are achieved properly, are unbiased, and are free from management pressure that may compromise the internal auditor's performance. Examples of those designated internal audit functions are safeguarding assets against loss and theft, providing reasonable assurances that the financial and operating information are accurate and reliable, and ensuring the entity's compliance with laws and regulations.

**Disclosure and Transparency:** Refers to the quality and reliability of information, whether financial or non-financial, provided by the company to the shareholders and the users of the financial reporting.

**Corporate Accountability:** Refers to the ability of the shareholders and the stakeholders to hold the governing body of the company such as the executive management, the board of directors and the external auditors accountable and answerable to the laws, the regulations and the company's bylaw based on their power and responsibilities.

**The audit expectation gap:** The audit expectation gap refers to the different perceptions between the auditors' actual performance and to what report users believe the auditors should be performing and achieving as auditors.

**Reliability of the audit report:** refers to the extent to which the institutional investor can rely on the audit report to clearly convey significant instances of poor internal controls; accounting errors, biases, or omissions; and concerns regarding the application of the going concern assumption.

#### **PART A - The Exercise**

You are presented with 16 hypothetical investment scenarios. Each scenario is slightly different. Each investment scenario requires 2 responses. Please consider each in isolation relative to your Typical Assessment of each scenario on the reliability of an audit report and the typical size of the audit expectation gap and then score all the cases relative to those two typical scenarios. Each of the four corporate governance dimensions is assigned either a high or low rating. High refers to the a level achieved by the top 1/3 of companies on that factor and Low refers to a level of performance achieved by the 1/3 least successful companies on that factor.



Case 1									
Lower External Auditor's Independence Higher Disclosure and Transparency									
Higher Internal Auditor Effectiveness Lower Corporate Accountability									
Your Assessment on the Following (circle)									
	Substantial	ly Worse				Substant	ially Better		
Reliability of the Audit Report	-3	-2	-1	Same	+1	+2	+3		
Audit Expectation Gap	-3	-2	-1	Same	+1	+2	+3		

Case 2									
Lower External Auditor's Independence Lower Disclosure and Transparency									
Higher Internal Auditor Effectiveness Lower Corporate Accountability									
Your Assessment on the Following (circle)									
	Substantial	ly Worse				Substant	tially Better		
Reliability of the Audit Report	-3	-2	-1	Same	+1	+2	+3		
Audit Expectation Gap -3			-1	Same	+1	+2	+3		

Case 3										
Higher	External Auditor's Independence			Higher	Disclosure and Transparency					
Higher	Internal Auditor Effectiveness			Lower	Corporate Accountability					
	Your Assessment on the Following (circle)									
		Substantial	ly Worse	9			Substant	tially Better		
Reliabilit	y of the Audit Report	-3	-2	-1	Same	+1	+2	+3		
Audit Ex	Audit Expectation Gap -3 -2			-1	Same	+1	+2	+3		

Case 4										
Higher	External Auditor's Indepe	Lower	Disclosure and Transparency							
Higher	Internal Auditor Effectiveness				Corporate Accountability					
	Your Assessment on the Following (circle)									
		Substantial	ly Worse	9			Substant	tially Better		
Reliabilit	y of the Audit Report	-3	-2	-1	Same	+1	+2	+3		
Audit Ex	Audit Expectation Gap -3 -2			-1	Same	+1	+2	+3		

Case 5										
Higher	Higher External Auditor's Independence Lower Disclosure a						Fransparer	ю		
Lower	Lower Internal Auditor Effectiveness Higher Co					e Accou	ntability			
	Your Assessment on the Following (circle)									
		Substantiall	y Worse				Substan	tially Better		
Reliabilit	Reliability of the Audit Report -3 -			-1	Same	+1	+2	+3		
Audit Ex	Audit Expectation Gap			-1	Same	+1	+2	+3		

Case 6										
Lower	External Auditor's Indepe		Higher Disclosure and Transparency							
Lower	Internal Auditor Effective	F	Higher Corporate Accountability							
	Your Assessment on the Following (circle)									
	Substantially Worse Substantially Better									
Reliabili	Reliability of the Audit Report -3 -2				Same	+1	+2	+3		
Audit Ex	Audit Expectation Gap -3			-1	Same	+1	+2	+3		

Case 7									
Higher External Audi	Higher External Auditor's Independence			Higher	Disclosure and Transparency				
Higher Internal Auditor Effectiveness Higher Corporate Accountability									
Your Assessment on the Following (circle)									
		Substantiall	y Worse				Substant	ially Better	
Reliability of the Audit I	Report	-3	-2	-1	Same	+1	+2	+3	
Audit Expectation Gap	Audit Expectation Gap -3 -2			-1	Same	+1	+2	+3	

Case 8										
Lower	External Auditor's Independence			Higher Disclosure and Transparency						
Lower	Lower Internal Auditor Effectiveness Lower Corporate Accountability									
	Your Assessment on the Following (circle)									
	Substantially Worse Substantially Better									
Reliabili	ty of the Audit Report	-3	-2	-1	Same	+1	+2	+3		
Audit Ex	Audit Expectation Gap -3 -2			-1	Same	+1	+2	+3		

Case 9										
Higher	External Auditor's Indepe		Lower Disclosure and Transparency							
Higher	Internal Auditor Effective	I	ower Corporate Accountability							
	Your Assessment on the Following (circle)									
		Substantiall	y Worse				Substant	tially Better		
Reliabilit	y of the Audit Report	-3	-2	-1	Same	+1	+2	+3		
Audit Ex	Audit Expectation Gap -3			-1	Same	+1	+2	+3		

Case 10	Case 10									
Lower External Auditor's Independence Lower Disclosure and T							су			
Lower Internal Auditor Effectiven	Lower Internal Auditor Effectiveness									
Your Assessment on the Following (circle)										
	Substantial	ly Worse				Substant	tially Better			
Reliability of the Audit Report	-3	-2	-1	Same	+1	+2	+3			
Audit Expectation Gap	Audit Expectation Gap -3			Same	+1	+2	+3			

Case 11									
Higher	Higher External Auditor's Independence Higher Disclosure and Transparency								
Lower	Lower Internal Auditor Effectiveness Lower Corporate Accountability								
	Your Assessment on the Following (circle)								
	Substantially Worse Substantially Better								
Reliabilit	y of the Audit Report	-3	-2	-1	Same	+1	+2	+3	
Audit Exp	Audit Expectation Gap -3 -2				Same	+1	+2	+3	

Case 12								
Lower E	External Auditor's Indeper	ndence		Lower	Disclosur	e and T	ransparen	су
Lower I	Lower Internal Auditor Effectiveness Higher Corporate Accountability							
	Your Assessm	nent on the Fe	ollowing	; (circle)				
		Substantial	y Worse	!			Substan	tially Better
Reliability	of the Audit Report	-3	-2	-1	Same	+1	+2	+3
Audit Expe	ectation Gap	-3	-2	-1	Same	+1	+2	+3

Case 13							
Higher External Auditor's Indepen	idence		Lower	Disclosu	re and <sup>-</sup>	Transparen	су
Lower Internal Auditor Effectivene	Lower Internal Auditor Effectiveness Lower Corporate Accountability						
Your Assessme	Your Assessment on the Following (circle)						
	Substantial	ly Worse				Substant	ially Better
Reliability of the Audit Report	-3	-2	-1	Same	+1	+2	+3
Audit Expectation Gap	-3	-2	-1	Same	+1	+2	+3

Case 14								
Higher External Auditor's Independence Higher Disclosure and Transparency								
Lower Internal Auditor Effectivene	Lower Internal Auditor Effectiveness Higher Corporate Accountability							
Your Assessme	Your Assessment on the Following (circle)							
	Substantial	y Worse				Substant	ially Better	
Reliability of the Audit Report	-3	-2	-1	Same	+1	+2	+3	
Audit Expectation Gap	-3	-2	-1	Same	+1	+2	+3	

Case 15							
Lower External Auditor's Indepen	dence		Lower	Disclosur	e and T	Fransparen	су
Higher Internal Auditor Effectiven	Higher Internal Auditor Effectiveness Higher Corporate Accountability						
Your Assessm	ent on the F	ollowing	g (circle)				
	Substantial	ly Worse	9			Substan	tially Better
Reliability of the Audit Report	-3	-2	-1	Same	+1	+2	+3
Audit Expectation Gap	-3	-2	-1	Same	+1	+2	+3

Case 16									
Lower External Auditor's Indeper	idence		Higher	Disclosu	re and <sup>-</sup>	Fransparen	ю		
Higher Internal Auditor Effectiven	Higher Internal Auditor Effectiveness				Higher Corporate Accountability				
Your Assessm	Your Assessment on the Following (circle)								
	Substantial	ly Worse				Substant	tially Better		
Reliability of the Audit Report	-3	-2	-1	Same	+1	+2	+3		
Audit Expectation Gap	-3	-2	-1	Same	+1	+2	+3		

#### PART B

Please indicate the relative importance each of the four variables had on your judgments by allocating 100 points between them for each of the outcome measures (i.e. each column should total 100):

	Reliability of the Audit Report	Audit Expectation Gap
External Auditor's Independence		
Internal Auditor Effectiveness		
Disclosure and Transparency		
Corporate Accountability		

Please indicate by circling a number on the scale below how confident you feel that the four factors together represent the major determinant of your assessment of the:

(1) Reliability of the audit report	Lower Confidence	1	2	3	4	5	6	7	Higher confidence
(2) Audit expectation gap	Lower Confidence	1	2	3	4	5	6	7	Higher confidence

Please list other related factors from the Jordanian Corporate Governance Codes that you think would be relevant to your assessment of your perceptions of:

A greater reliability of the audit report

A narrowing of the expectation gap

PART C
1- Please indicate the type of the company you are currently work in:
a- Bank b- Insurance Company c- Brokerage company d- Investment fund
e- Other, please specify
2- How many years have you worked as a financial analyst, Investment analyst/officer or investment manager?
3- Please indicate your position:
a. Manager (of the investment department) b. Head of the investment section
c. Investment analyst/officer d. If others, please specify
<ul> <li>4- Please indicate if you would like to receive a copy of a summary conclusions report. Yes / No</li> <li>If you have answered Yes to the above, please provide your contact details:</li> </ul>
Name Position
Company
Telephone email

If you have any questions regarding this questionnaire please contact:

Mohammad Shbeilat	OR	Dr. Waleed Abdel-Qader	OR	Professor Donald Ross	OR	Professor Garry Tibbits
Tel: +962		Tel: +61 (02) 9739 2307		Tel: +61 (02) 9739 2356		
Mohammad.shbeilat@acu.edu	i.au	Waleed.Abdel-Qader@acu.edu.	<u>au</u>	Donald.Ross@acu.edu.au		g.tibbits@uws.edu.au

## **Appendix 7: Data Collection – Interview**

#### **Interview Protocol**

- Thank
- Explain the terms used in the study and deliver a brief introduction to study contribution
- Hardcopy of the corporate governance code was available during the meetings
- Request unprompted impressions of what does the corporate governance means to them.
- 1- How does your assessment of the external auditor's independence affect your perception of the reliability of the audit report?

Can you please tell me why?

- 2- How does your assessment of the internal auditor's effectiveness affect your perception of the reliability of the audit report?
  - Can you please tell me why?
- 3- How does your assessment of the level (poor, medium, strong) of the disclosure and transparency affect your perception of the reliability of the audit report?
  - Can you please tell me why?
- 4- How does your assessment of the level (poor, medium, strong) of corporate accountability affect your perception of the reliability of the audit report?
  - Can you please tell me why?
- 5- How does your assessment of the external auditor's independence affect your perception on the level of the audit expectation gap?
  - Can you please tell me why?

- 6- How does your assessment of the internal auditor's effectiveness affect your perception on the level of the audit expectation gap?
  - Can you please tell me why?
- 7- How does your assessment of the level (poor, medium, strong) of the disclosure and transparency affect your perception on the level of the audit expectation gap?
  - Can you please tell me why?
- 8- How does your assessment of the level (poor, medium, strong) of corporate accountability affect your perception on the level of the audit expectation gap?
  - Can you please tell me why?

Thanks Interviewee

## Appendix 8: List of Abbreviations

ACU	Australian Catholic University
AICPA	American Institute of Certified Public Accountants
ASCA	Arab Society of Certified Accountants
ASE	Amman Stock Exchange
ASXCGC BP	THE Australian Stock Exchange Corporate Governance Council
	Best Practices
BoD	Board of Directors
CBJ	Central Bank of Jordan
CCD	Companies Control Department in Jordan
CEO	Chief Executive Officer
CFA - Jordan	Professional Association of Financial Analysts in Jordan
CIPE	Centre for International Private Enterprise
CFM	Certified Financial Management
СМА	Certified Managerial Accounting
СРА	Certified Public Accountant
FRC	The Financial Reporting Council in the UK
FASB	Financial Accounting Standard Board
GAAS	Generally accepted auditing standard
GATT	General Agreement on Tariffs and Trade
IASC	International Accounting Standards Committee
IFAC	International Federation of Accountants
IFRS	International Financial Reporting Standards
IIA	Institute of Internal Auditors
IMF	International Monetary Fund
IRC	Insurance Commission: in Jordan
ISA	International Standards on auditing
JACC	Jordanian Anti-Corruption Commission
JACG	The Jordanian Association of Corporate Governance
JACPA	Jordanian Association for Certified Public Accountants
	Jordanian Corporate Governance Code for the publicly-traded
JCGC	companies which have been approved by the JSC board of the
	commissioners on the 29th of July 2008 and came into effect as
	of the first of January 2009
JSC	The Jordanian Securities Commission
JTC	Jordanian Transparency Center
MENA	Middle East and North Africa
NYSE	New York Stock Exchange
NASDAQ	National Association of Securities Dealers Automated
	Quotations
NAS	Non-audit services
OECD	The Organization for Economic Co-operation and Development
ROSC	Report on the Observance of Standards and Codes
SDC	Securities Depository Center – for Jordan

SEC	Securities and Exchange Commission in American
SOX	Sarbanes-Oxley Act of 2002
SPSS	Statistical Package for Social Sciences
WTO	World Trade Organization

## **Appendix 9: Proofreading Certificate**



#### To Whom It May Concern:

Thesis Title: The Jordanian Corporate Governance Code: a Study of Institutional Investors' Perception of the Reliability of the Audit Report and the Audit Expectation Gap

University: Australian Catholic University

Student Name: Mohammad K. Shbeilat

This is to certify that Clarity Editors have edited the above titled Ph.D. thesis by Mohammad Shbeilat, providing feedback on the language, completeness and consistency of the document. The services rendered were limited to copyediting and proofreading, as covered in Standards D and E of the Australian Standards for Editing Practice. Clarity Editors did not modify or provide feedback on the substance and structure of the thesis beyond drawing the author's attention to potential problems.

Sincerely,

21 May 2013

Hisham Alhubaishi Chief Editor Clarity Editors

Clarity Editors, P.O. Box 954355, Amman Mall 11954, Amman, Jordan Website: <u>www.clarityeditors.com</u> email: <u>info@clarityeditors.com</u>